

# PhD-FSTM-2024-038 The Faculty of Science, Technology and Medicine

## **DISSERTATION**

Defence held on 30/05/2024 in Esch-sur-Alzette

to obtain the degree of

# DOCTEUR DE L'UNIVERSITÉ DU LUXEMBOURG EN MATHÉMATIQUES

by

## Leonardo MAINI

Born on 8 March 1996 in Borgo San Lorenzo (Italy)

# ON SOME ASYMPTOTIC RESULTS ON FUNCTIONALS OF WEAKLY STATIONARY RANDOM FIELDS

## Dissertation defence committee

Dr Ivan Nourdin, dissertation supervisor *Professor, Université du Luxembourg* 

Dr Giovanni Peccati, Chairman Professor, Université du Luxembourg

Dr Guillaume Poly *Professor, Université de Rennes* 

Dr Maurizia Rossi Professor, Università degli Studi di Milano-Bicocca

Dr Ciprian Tudor Professor, Université de Lille

# Acknowledgements

My first big thank you goes to my supervisor, Ivan Nourdin, who has always guided and supported me. Thank you, Ivan, for the many insightful discussions, for teaching me how to approach problems and present mathematical results, while still giving me the freedom to develop my own style and ideas. As the kind of oscillatory covariance functions we studied together, this PhD has been full of ups and downs. This thesis would not have been written if you hadn't believed in me during the downs!

I would like to thank Giovanni Peccati, Guillaume Poly, Maurizia Rossi and Ciprian Tudor for agreeing to be members of my jury. My papers are based on and strongly inspired by your seminal contributions, and it is indeed a great honour for me to present them in front of you! Special thanks go to Guillaume Poly and Maurizia Rossi, for generously accepting to review my thesis.

I would like to thank my co-authors Anna Paola, Francesca, Francesco, Ivan and Nikolai, for the insightful discussions that led to the results of this thesis. I have truly enjoyed working with you and I hope that those in this thesis are only the first of many future collaborations.

To all the wonderful people I met at the University of Luxembourg, some of whom I consider true friends: thank you for the discussions (mathematical and otherwise), for the ping-pong games, for the coffee breaks, for the jokes, and in general for the good time spent together. A special thank you to Arianna, Francesca, Krzysztof and Michele, who always had kind words and gestures for me, especially in times of need.

I would like to thank Emilie, Katharina and Marie from the math office for their commendable professionalism.

Thanks to my lifelong friends, whose presence, support and affection I feel strongly even from afar. Spending time with you and seeing each other after a long time is always nice, and every time I feel like I see you every day.

My last immense thank you goes to the people I love and to whom this thesis is dedicated. I would like to thank my beautiful family. You always make me feel your affection and support.

Thank you to my father Nicola, whose unwavering spirit of sacrifice, often taken for granted, always inspires me to give more.

Thank you to my sister Elena, the best person I know. Thank you for being who you are, and without realizing it, for being an example to us all.

Thank you to my mother, Rossella, for your fighting spirit and for never giving up. What you went through has changed the way I look at life, problems, sadness and happiness. Thank you for fighting, for winning and for giving us the best Christmas present ever.

Finally, thank you, Manu. Thank you for always being there for me, for being a point of reference and a source of joy, serenity and inspiration. I have shared these three years with you, between wonderful moments, difficult phases, distance, closeness, sorrow, immense joy and sadness. What we have gone through and overcome together, the bond I feel with you, is something that cannot be explained.

I gratefully acknowledge support from the Luxembourg National Research Fund PRIDE17/1224660/GPS, in the framework of which this thesis was elaborated.

#### **Abstract**

Functionals of random fields have always been a central topic in probability theory, since its inception as a subject of study. The latter include, among others, partial sums of random variables and geometric quantities associated to random functions on manifolds. In this thesis, we investigate the asymptotic probabilistic behaviour of integral functionals of weakly stationary random fields on expanding Euclidean domains, with a special focus on additive (or nonlinear) functionals of stationary Gaussian fields.

In Chapter 1 we first introduce the main mathematical objects and tools encountered in this work, concluding with an overview of the state of the art and our new contributions related to the main research questions of this thesis. The two main questions are the following: first, as the integration domain expands, does a central limit theorem hold? Second, given two expanding integration domains, what is the asymptotic covariance between their integral functionals?

Chapter 2 contains the paper "Spectral central limit theorem for additive functionals of isotropic and stationary Gaussian fields", written in collaboration with Ivan Nourdin. In this chapter, we prove that a large class of additive functionals of stationary, isotropic Gaussian fields satisfies a central limit theorem if an easily verifiable condition on the spectral measure holds. This result brings to light a new class of "strongly correlated" Gaussian fields whose additive functionals satisfy a central limit theorem. This fact contradicts the intuition forged in the last four decades, starting from the seminal works by Breuer, Dobrushin, Major, Rosenblatt and Taqqu.

Chapter 3 contains the paper "Fluctuations of polyspectra in spherical and Euclidean random wave models", written in collaboration with Francesco Grotto and Anna Paola Todino. Our main result provides the variance rate of any additive functional of Euclidean (Berry's random wave model) and spherical random waves, a problem that was left as a conjecture ten years ago. To do this, we exploit a relation between random waves and Pearson's random walks.

Chapter 4 contains the paper "Asymptotic covariances for functionals of weakly stationary random fields". Here we compute the asymptotic covariances of integral functionals of weakly stationary random fields on expanding domains under assumptions that encompass the ones in the literature, deriving an explicit formula that involves the directional derivative of the cross covariogram of two domains.

Chapter 5 contains the preprint "Limit theorems for p-domain functionals of stationary Gaussian fields", written in collaboration with Nikolai Leonenko, Ivan Nourdin and Francesca Pistolato. In this chapter we consider more general families of additive functionals, which we call p-domain functionals, including as a special case spatio-temporal functionals and 1-domain functionals considered in the previous chapters. In this setting, we are able (under suitable assumptions) to reduce the study of p-domain functionals to that of some 1-domain functionals, explaining some recent findings in the literature in a new light.

# **Contents**

| 1                                  | Intr  | Introduction                 |  |     |  |  |
|------------------------------------|-------|------------------------------|--|-----|--|--|
|                                    | 1.1   | 1 Mathematical objects       |  |     |  |  |
| 1.1.1 Functionals of random fields |       | Functionals of random fields | 1  |     |  |  |
|                                    |       | 1.1.2                        | Invariant random fields  | 4   |  |  |
|                                    |       | 1.1.3                        | Gaussian fields  | 5   |  |  |
|                                    | 1.2   | Mathe                        | ematical tools   | 7   |  |  |
|                                    |       | 1.2.1                        | Spectral representations   | 7   |  |  |
|                                    |       | 1.2.2                        | Chaotic decompositions and Fourth Moment theorems                | 10  |  |  |
|                                    |       | 1.2.3                        | Cross covariograms   | 20  |  |  |
|                                    | 1.3   | Resear                       | rch questions and existing results                               | 21  |  |  |
|                                    |       | 1.3.1                        | Functionals of random fields with short memory                   | 22  |  |  |
|                                    |       | 1.3.2                        | Functionals of random fields with long memory                    | 25  |  |  |
|                                    |       | 1.3.3                        | p-domain functionals of random fields                            | 26  |  |  |
|                                    | 1.4   | New r                        | esults of the thesis   | 27  |  |  |
|                                    |       | 1.4.1                        | Chapter 2: Spectral central limit theorem for additive function- |     |  |  |
|                                    |       |                              | als of isotropic and stationary Gaussian fields                  | 28  |  |  |
|                                    |       | 1.4.2                        | Chapter 3: Fluctuations of Polyspectra in Spherical and Eu-      |     |  |  |
|                                    |       |                              | clidean Random Wave Models                                       | 30  |  |  |
|                                    |       | 1.4.3                        | Chapter 4: Asymptotic covariances for functionals of weakly      |     |  |  |
|                                    |       |                              | stationary random fields   | 32  |  |  |
|                                    |       | 1.4.4                        | Chapter 5: Limit theorems for p-domain functionals of station-   | 0.4 |  |  |
|                                    |       |                              | ary Gaussian fields  | 34  |  |  |
| 2                                  | Spe   | ctral ce                     | entral limit theorem for additive functionals of isotropic and   | l   |  |  |
|                                    | stati | ionary                       | Gaussian fields  | 37  |  |  |
| 2.1 Introduction                   |       |                              | uction   | 37  |  |  |
|                                    |       | 2.1.1                        | Previous results   | 39  |  |  |
|                                    |       | 2.1.2                        | Motivating examples  | 40  |  |  |
|                                    |       | 2.1.3                        | Main result  | 41  |  |  |
|                                    |       |                              |  |     |  |  |

iv Contents

|   |      | 2.1.4   | Possible natural extensions of Theorem 2.1.2                        | 44   |
|---|------|---------|---|------|
|   |      | 2.1.5   | Plan of the paper   | 45   |
|   | 2.2  | A few   | preliminaries for the proof of Theorem 2.1.2                        | 45   |
|   |      | 2.2.1   | Well-posedness of $Y_t$   | 45   |
|   |      | 2.2.2   | Bessel functions and Fourier transform of the indicator of the      |      |
|   |      |         | unit ball   | 46   |
|   |      | 2.2.3   | Reduction to the $R$ th chaos                                       | 47   |
|   |      | 2.2.4   | Elements of Malliavin calculus and the Fourth Moment theorem        | 1 49 |
|   | 2.3  | Proof   | of Theorem 2.1.2 when $R \geq 4$ even $\ldots \ldots \ldots \ldots$ | 52   |
|   | 2.4  |         | of Theorem 2.1.2 when $R=2$   | 54   |
|   | 2.5  | Proof   | of theorem 2.1.2 when $R = 1 \dots \dots \dots \dots$               | 62   |
|   | 2.6  | An exa  | ample of application of Theorem 2.1.2                               | 64   |
| 3 | Fluc | tuation | ns of polyspectra in spherical and Euclidean random wave            | 2    |
|   | mod  |         |   | 67   |
|   | 3.1  | Introd  | uction and Main Results   | 67   |
|   |      | 3.1.1   | Random Wave Models  | 68   |
|   |      | 3.1.2   | Polyspectra   | 69   |
|   |      | 3.1.3   | Variance Asymptotics  | 70   |
|   | 3.2  | Pearso  | on's Random Walk  | 72   |
|   |      | 3.2.1   | Random Walks and Polyspectra  | 72   |
|   |      | 3.2.2   | Density of short random walks                                       | 73   |
|   | 3.3  | Fluctu  | ation Asymptotics and Integration Domains                           | 75   |
|   |      | 3.3.1   | Variance of Euclidean Polyspectra                                   | 75   |
|   |      | 3.3.2   | Variance of Hyperspherical Polyspectra                              | 76   |
| 4 | Asyr | nptotic | covariances for functionals of weakly stationary random fields      | 81   |
|   | 4.1  | Introd  | uction  | 81   |
|   |      | 4.1.1   | Classical assumptions, existing literature, motivating examples     | 83   |
|   |      | 4.1.2   | Main result   | 85   |
|   |      | 4.1.3   | Plan of the paper   | 88   |
|   | 4.2  |         | ninaries for the proof of Theorem 4.1.1                             | 88   |
|   |      | 4.2.1   | Well-posedness of (4.2) and proof of (4.4)                          | 88   |
|   |      | 4.2.2   | Cross covariograms  | 90   |
|   |      | 4.2.3   | Regularly varying functions   | 91   |
|   | 4.3  | Proof   | of Theorem 4.1.1  | 93   |
|   | 4.4  | Non-li  | near functionals of stationary Gaussian fields                      | 95   |
|   |      | 4.4.1   | The Breuer-Major case   | 96   |
|   |      | 4.4.2   | The long memory case  | 98   |
|   |      | 4.4.3   | The Berry's case  | 99   |
|   | 4.5  | Non-ra  | adial covariance functions  | 106  |
| 5 | Limi |         | •   | 111  |
|   | 5.1  | Introd  | uction  | 111  |
|   |      | 5.1.1   | Separable case  | 113  |
|   |      | 5.1.2   | Non separable case  | 116  |
|   |      | 5.1.3   | Plan of the paper   | 116  |

<u>Contents</u> <u>v</u>

| 5.2  | Preliminaries    |  |     |  |  |
|------|------------------|--|-----|--|--|
|      | 5.2.1            | Elements of Malliavin-Stein method                   | 116 |  |  |
|      | 5.2.2            | Classical results for 1-domain functionals           | 118 |  |  |
| 5.3  | Proof            | of Theorems 5.1.1 and 5.1.2                          | 120 |  |  |
|      | 5.3.1            | Proof of Theorem 5.1.1 when $\varphi = H_q$          | 120 |  |  |
|      | 5.3.2            | A Breuer-Major theorem for $p$ -domain functionals   | 124 |  |  |
|      | 5.3.3            | Reduction to $R$ th chaos and proof of Theorem 5.1.1 | 127 |  |  |
|      | 5.3.4            | Proof of Theorem 5.1.2                               | 130 |  |  |
| 5.4  | Examp            | oles   | 133 |  |  |
| 5.5  | Going            | beyond the separability assumption                   | 137 |  |  |
|      | 5.5.1            | Gneiting covariance functions                        | 137 |  |  |
|      | 5.5.2            | Additively separable covariance functions            | 140 |  |  |
| Bibl | Bibliography 149 |  |     |  |  |

# CHAPTER 1

# Introduction

In this chapter, we highlight what are the novel ideas of this thesis. To do this, we proceed as follows. In Section 1.1 we define the main mathematical objects of the thesis. In Section 1.2 we provide a concise introduction to the needed mathematical tools. In Section 1.3 we give an overview of the state of the art related to our main research questions. Finally, in Section 1.4 we present our contributions in a concise form, comparing them with the existing results of the previous section.

## 1.1 Mathematical objects

In this section, we introduce the main mathematical objects of the thesis: functionals of random fields and the classes of random fields whose functionals will be studied. In particular, we deal with functionals of random fields in Subsection 1.1.1, invariant random fields in Subsection 1.1.2 and Gaussian fields in Subsection 1.1.3.

#### 1.1.1 Functionals of random fields

Fix a probability space  $(\Omega, \mathcal{F}, \mathbb{P})$ , a set E and a measurable space  $(S, \mathcal{S})$ . We start with the definition of S-valued random field on E.

**Definition 1.1.1.** A S-valued random field A on E is a function

$$A: \Omega \times E \to S, \qquad (\omega, x) \mapsto A_x(\omega),$$
 (1.1)

such that  $A_x := A_x(\cdot)$  is a S-valued random variable for every  $x \in E$ .

Equivalently, a S-valued random field on E is a collection of S-valued random variables  $A = (A_x)_{x \in E}$ , or a  $S^E$ -valued random variable, where  $S^E$  is the set of

all the functions  $E \mapsto S$  equipped with the product sigma-algebra. Since the three definitions are equivalent, we will use them interchangeably.

**Terminology.** In this manuscript, a random variable is a measurable function  $(\Omega, \mathcal{F}) \to (\mathbb{R}, \mathcal{B}(\mathbb{R}))$ , where  $\mathcal{B}(\mathbb{R})$  denotes the Borel sigma-algebra. Moreover, unless otherwise stated, random *objects* (variables, vectors, fields) are all real-valued. Alternatively, if they take values in a measurable space  $(S, \mathcal{S})$ , we call them S-valued random *objects*. For example, a random field on E is a  $\mathbb{R}^E$ -valued random variable.

Random fields can model virtually every uncertain quantity  $A_x$  that depends on a variable x in a given window of values E. In applications, they appear whenever an exact prediction is impossible. For instance E may represent a time interval, a space domain, a set of space-time configurations, a set of indices associated with individuals in a given population, or every possible inputs of a neural network with random weights.

The second definition of this thesis is that of functional of a random field, which we state here in full generality.

**Definition 1.1.2.** Let A be a random field on E, seen as a  $\mathbb{R}^E$ -valued random variable  $A: \Omega \to \mathbb{R}^E$ , with realizations in  $\mathrm{Im}(A) := \{A(\omega) : \omega \in \Omega\} \subseteq \mathbb{R}^E$ . A **functional** of A is a random variable  $Y = f \circ A$ , where  $f: \mathcal{U} \to \mathbb{R}$ , with  $\mathrm{Im}(A) \subseteq \mathcal{U}$ .

Examples of functionals of random fields abound in the literature, and arise in every kind of application. We focus here on three examples, which can well highlight their omnipresence.

**Example 1.1.1** (Sums of random variables). As a first example, just consider a sequence  $A_1, A_2, \ldots$  of random variables on the same probability space.  $A = (A_i)_{i \in \mathbb{N}}$  is obviously a random field on  $\mathbb{N}$ . An example of functional of A is the sum of the first  $n \in \mathbb{N}$  elements of A, namely

$$Y_n := \sum_{i=1}^n A_i, \quad n \in \mathbb{N}.$$
 (1.2)

In this case  $Y_n = f_n \circ A$ , where  $f_n(a_1, a_2, \dots) = a_1 + \dots + a_n$ , is a functional of A for every  $n \in \mathbb{N}$ , see Definition 1.1.2. The study of  $Y_n$  as  $n \to \infty$  has always been a central topic in probability, since its inception as a subject of study; just think of the law of large numbers, or the classical central limit theorem. For less classical examples, see e.g. [IL71; MPU19; Pel90].

**Example 1.1.2** (Geometric functionals in  $\mathbb{R}^d$ ). Let A be a random field on  $\mathbb{R}^d$ . In this case, numerous functionals have been studied in the literature. Under suitable regularity assumptions on A, we have that  $\operatorname{Im}(A) \subseteq L^1_{loc}(\mathbb{R}^d)$ , so that one can consider for example a continuous version of (1.2)

$$Y_t := \int_{tD} A_x \, dx, \quad t > 0,$$
 (1.3)

where D is a compact domain in  $\mathbb{R}^d$ . Note that integral functionals can also be used to investigate the large domain geometry of random fields in growing domains,

studying  $Y_t$  as  $t \to \infty$ . Indeed, if for example  $A_x = \mathbf{1}_{[a,\infty)}(B_x)$ , where B is a random field on  $\mathbb{R}^d$  and  $a \in \mathbb{R}$ , then  $Y_t$  is the so called **excursion volume** of B at level a in a growing window tD, namely

$$Y_t = \operatorname{Vol}\left(\left\{x \in \mathbb{R}^d : B_x \ge a\right\} \cap tD\right).$$

The latter has been extensively studied in the literature, see e.g. [DER20; KS18; LO14; MN24; Not21]. Moreover, many other classes of geometric functionals have been investigated, some of which do not admit a representation (1.3). Some examples are topological quantities related to the excursion (random) set  $\{x \in \mathbb{R}^d : B_x \ge a\}$ , such as its number of connected components, see e.g. [BMM22; NS16; NS20], or other functionals related to lower dimensional random sets, like the number of zeros, the number of critical points, or the lower dimensional volume of level sets, see e.g. [DEL21; GS23; NPV23; NPR19; PV20; Vid21].

**Example 1.1.3** (Functionals of random waves). Let  $(\mathcal{M}, g)$  be a compact Riemannian manifold without boundary. Fix an orthonormal basis  $\{\psi_{\lambda_j}\}$  of  $L^2(\mathcal{M})$ , composed by eigenfunctions of the Laplace-Beltrami operator  $\Delta_g$ , so that

$$\Delta_g \psi_{\lambda_j} = \lambda_j^2 \psi_{\lambda_j} \quad \langle \psi_j, \psi_k \rangle = \delta_{jk}.$$

The monochromatic random wave  $A^{\lambda}=(A_{x}^{\lambda})_{x\in\mathcal{M}}$  is the random field on  $(\mathcal{M},g)$  defined as the random linear combination

$$A_x^{\lambda} := \sum_{\lambda_j \in [\lambda, \lambda + 1]} \gamma_j \, \psi_{\lambda_j},$$

where  $\gamma_1, \gamma_2, \ldots$  are i.i.d. standard Gaussian random variables. Geometric functionals of random waves have been extensively studied in the literature on different manifolds, see e.g. [Cam19; CMR23; Die+23; GP23; MW14; Tod19; Zel09]. A relevant fact proved by Canzani and Hanin in [CH15; CH18] (see also [Gas23]) is that, under suitable assumptions on  $\mathcal{M}$ , monochromatic random waves are locally approximated by a random field on  $\mathbb{R}^{\dim(\mathcal{M})}$ , the so-called Berry random wave model, which will be formally defined at the end of the section. As a consequence, one may try to reduce the small scale study of Riemannian geometric functionals of monochromatic random waves, as  $\lambda \to \infty$ , to that of Euclidean geometric functionals of Berry's random wave model. This has been done for example in [CH20; Die+23].

Throughout the thesis we will be interested in the following heuristic question: given a random field A on E and a family of functionals  $Y = (Y_t)_{t>0}$  of A, what is the probabilistic behavior of  $Y_t$  as  $t \to \infty$ ?

To be more precise, different features will be analyzed, such as the mean, the variance, or the distribution of  $Y_t$  as  $t \to \infty$ . In Example 1.1.1, for instance, if  $A_1, A_2, ...$  are i.i.d. (independent and identically distributed) with mean  $m \in \mathbb{R}$  and variance  $\sigma^2 \in (0, \infty)$ , then the mean and variance of  $Y_n$  are

$$\mathbb{E}[Y_n] = m n, \quad \operatorname{Var}(Y_n) = n \sigma^2,$$

and the **classical central limit theorem** ensures that, as  $n \to \infty$ ,

$$\frac{Y_n - \mathbb{E}[Y_n]}{\sqrt{\operatorname{Var}(Y_n)}} = \frac{1}{\sqrt{n}} \sum_{i=1}^n \frac{A_i - m}{\sigma} \xrightarrow{d} N(0, 1),$$

where N(0,1) is a standard Gaussian and  $\stackrel{d}{\rightarrow}$  denotes the convergence in distribution. At this point of the discussion, the definition of central limit theorem follows naturally.

**Definition 1.1.3.** Let A be a random field on E. Let  $Y = (Y_t)_{t>0}$  be a family of functionals of A. Assume that for t sufficiently large  $Var(Y_t) \in (0, \infty)$ . Then, Y satisfies a **central limit theorem (CLT)** if

$$\frac{Y_t - \mathbb{E}[Y_t]}{\sqrt{\operatorname{Var}(Y_t)}} \xrightarrow{d} N(0, 1), \qquad \text{ as } t \to \infty.$$

In order to prove central limit theorems (or other asymptotic results) for a family of functionals of A, we need assumptions on its distribution. The rest of this subsection is devoted to introducing the two main classes of random fields whose functionals are considered in this thesis: invariant random fields and Gaussian fields.

#### 1.1.2 Invariant random fields

A very popular class of assumptions for random fields are **invariance properties**, such as stationarity (invariance by translations) and isotropy (invariance by rotations). These properties can all be defined through the action of a certain group  $(\mathbb{G}, +)$  on the index set E of the random field.

In the classical CLT, for example, since  $A_1, A_2, \ldots$  are i.i.d., A is invariant with respect to the action of any group acting on  $\mathbb{N}$ . Let us give the precise definition.

**Definition 1.1.4.** Let A be a random field on E. Let  $(\mathbb{G}, +)$  be a group acting on E (from the left). A is said **G-invariant** if its finite dimensional distributions are invariant under the action of  $\mathbb{G}$  (from the left), that is,  $\forall n \in \mathbb{N}, \forall x_1, \ldots, x_n \in E$ ,

$$(A_{x_1}, \dots, A_{x_n}) \stackrel{d}{=} (A_{g \cdot x_1}, \dots, A_{g \cdot x_n}) \qquad \forall g \in \mathbb{G},$$

where  $\cdot$  denotes the group action and  $\stackrel{d}{=}$  the equality in distribution.

Depending on the group  $\mathbb{G}$  acting on E, one can have many different kinds of invariance, see e.g. [Mal13; MP11]. In particular, the invariance property that stands out above all the others in this thesis is that of stationarity.

**Definition 1.1.5.** Let A be a random field on E. Suppose that E has a group structure (E, +), and consider the standard (left) action on itself, namely  $g \cdot x = g + x$  for every  $g, x \in E$ . A is said **stationary** if A is E-invariant, that is,  $\forall n \in \mathbb{N}, \forall x_1, \dots, x_n \in E$ ,

$$(A_{x_1}, \dots, A_{x_n}) \stackrel{d}{=} (A_{g+x_1}, \dots, A_{g+x_n}) \quad \forall g \in E.$$
 (1.4)

The easier cases to visualize stationarity are  $E \in \{\mathbb{R}^d, \mathbb{Z}^d\}$ , equipped with the sum operation. Indeed, in the latter situations A stationary means that the distribution of

A is invariant by translations<sup>1</sup>. We shall also consider the following weaker version of stationarity.

**Definition 1.1.6.** Let A be a random field on E. Suppose that E has a group structure (E, +), and consider the standard (left) action on itself, namely  $g \cdot x = g + x$  for every  $g, x \in E$ . A is said **weakly stationary** if for some function  $K : E \to \mathbb{R}$  we have

$$\mathbb{E}[A_x] = \mathbb{E}[A_y], \quad \operatorname{Cov}(A_x, A_y) = K(x - y), \quad \forall x, y \in E.$$

If this is the case, K is said the **covariance function** of A.

Another important invariance property for random fields on  $\mathbb{R}^d$  (or  $S^{d-1}$ ) is **isotropy**, i.e. invariance by rotations.

**Definition 1.1.7.** Let A be a a random field on  $E = \mathbb{R}^d$  (or  $E = S^{d-1}$ ). Consider the special orthogonal group  $\mathbb{G} = SO(d)$  of rotations, equipped with the standard matrix multiplication, and acting on E with the left multiplication  $g \cdot x = g x$ , for g matrix in SO(d) and  $x \in E$ . A is said **isotropic** if A is SO(d)-invariant, that is,  $\forall n \in \mathbb{N}, \forall x_1, \ldots, x_n \in E$ ,

$$(A_{x_1}, \dots, A_{x_n}) \stackrel{d}{=} (A_{g x_1}, \dots, A_{g x_n}) \qquad \forall g \in SO(d).$$

For some examples of anisotropic random fields, see e.g. [EF20; Xia09]. In the following, especially to prove central limit theorems for functionals of  $A = (A_x)_{x \in \mathbb{R}^d}$ , we assume that A is stationarity and/or isotropic. When these assumptions are removed, another type of invariance is usually required, for instance self-similarity (i.e. scale invariance, see e.g. [HLX23; Jar+23]).

#### 1.1.3 Gaussian fields

We start with the definition of Gaussian field. From now on, to specify that a random field is Gaussian, we shall denote it by the letter B (instead of A).

**Definition 1.1.8.** A random field B on E is said to be a **Gaussian field** on E if its finite dimensional distributions are Gaussian, that is,  $\forall n \in \mathbb{N}, \forall x_1, \dots, x_n \in E$ ,  $(B_{x_1}, \dots, B_{x_n})$  is a Gaussian (random) vector.

Gaussian fields are popular among random fields for two main reasons. First, they often turn out to be scaling limits of families of random fields, see e.g. [NPR19, p.100] or [EF20, (2.1)], hence their interpretation as good models in many applications. Second, their Gaussian finite dimensional distributions are known explicitly and satisfy very convenient analytical properties, such as being completely determined by the **mean function**  $m: E \to \mathbb{R}$ ,

$$m(x) := \mathbb{E}[B_x],$$

and the **covariance kernel**  $C: E \times E \to \mathbb{R}$ ,

$$C(x,y) := Cov(B_x, B_y).$$

Following this idea, one may also say that a random field A on  $E \in \{\mathbb{R}_+, \mathbb{N}\}$  is stationary if (1.4) holds for every  $x_1, \ldots, x_n, g \in E$ , even if (E, +) is not a group.

Combining this with Kolmogorov existence theorem, we have the following.

**Fact 1.** Let *E* be a set. Then, the following statements are equivalent:

1.  $C: E \times E \to \mathbb{R}$  is a **non-negative definite kernel** on E, that is,  $\forall n \in \mathbb{N}, \forall x_1, \dots, x_n \in E, \forall a_1, \dots, a_n \in \mathbb{R}$ ,

$$\sum_{i,j=1}^{n} a_i a_j \, \mathcal{C}(x_i, x_j) \ge 0.$$

2. There exist a Gaussian field B on E with covariance kernel C.

Since the distribution of a Gaussian field B is completely characterized by its mean function m and its covariance kernel  $\mathcal C$ , also the invariance properties considered in the previous section can be expressed in terms of m and  $\mathcal C$ . In particular, if B is a Gaussian field on E, and (E,+) has a group structure, then B is stationary if and only if it is weakly stationary, with constant mean function  $m:E\to\mathbb R$ , and covariance function  $C:E\to\mathbb R$ ,

$$C(x - y) := \mathcal{C}(x, y), \quad \forall x, y \in E.$$

In the same vein, if B is a Gaussian field on  $\mathbb{R}^d$  (or  $S^{d-1}$ ), then B is isotropic if and only if  $\forall x, y \in E, \forall g \in SO(d)$ 

$$m(x) = m(g \cdot x), \qquad \mathcal{C}(x, y) = \mathcal{C}(g \cdot x, g \cdot y).$$
 (1.5)

Note that if B is on  $S^{d-1}$ , then (1.5) implies that C(x,y) only depends on the (spherical) distance between x and y, which is generally not true (considering the Euclidean distance) for an isotropic field B on  $\mathbb{R}^d$ . However, if B is a stationary Gaussian field on  $\mathbb{R}^d$ , with constant mean function m and covariance function C, then B is isotropic if and only if the covariances depend only on the distances, that is,  $\forall x, y \in \mathbb{R}^d$ ,

$$C(x,y) = C(x-y) =: \rho(|x-y|),$$

where  $|\cdot|$  denotes the standard Euclidean norm. In this case, the function  $\rho: \mathbb{R}_+ \to \mathbb{R}$  is called the **isotropic covariance function** of B.

A Gaussian field that will often recur in the thesis is Berry's random wave model, which is defined as follows.

**Definition 1.1.9.** The (*d*-dimensional) Berry random wave model is the unique (in distribution) stationary, centered (i.e. with  $m \equiv 0$ ), isotropic Gaussian field on  $\mathbb{R}^d$  with isotropic covariance function

$$\rho(r) = b_d(r) := \Gamma\left(\frac{d}{2}\right) \left(\frac{2}{r}\right)^{\frac{d}{2}-1} J_{\frac{d}{2}-1}(r)$$

where  $\Gamma$  denotes the Gamma function, and  $J_{\nu}$  is the Bessel function of the first kind of order  $\nu$ , defined as

$$J_{\nu}(r) := \sum_{k=0}^{\infty} \frac{(-1)^k}{k!\Gamma(k+\nu+1)} \left(\frac{r}{2}\right)^{2k+\nu}.$$

Berry's random wave model can be also characterized as the limit in finite dimensional distributions of a superposition of J independent plane waves with fixed wave number 1, as  $J \to \infty$ ; or the unique isotropic centered Gaussian field B with unit variance satisfying  $\Delta B + B = 0$ , see e.g. [NPR19].

As we anticipated in Example 1.1.3, this Gaussian field, introduced by Berry in [Ber77], arises as the local scaling limit of random waves on a class of manifolds, hence its importance in applications. A crucial fact is that its covariance function,  $x \to b_d(|x|)$ , is the Fourier transform of the uniform distribution on  $S^{d-1}$ , a property that will be exploited several times in the sequel.

### 1.2 Mathematical tools

This section is devoted to the introduction of the main mathematical tools which will be applied in the thesis. In particular: in Subsection 1.2.1 we focus on the spectral representations for the covariance functions of invariant random fields; in Subsection 1.2.2 we focus on chaotic decompositions and Fourth Moment theorems for functionals of Gaussian fields; in Subsection 1.2.3 we define cross covariograms and discuss their properties and importance.

#### 1.2.1 Spectral representations

In this subsection, we introduce one of the main tool of this thesis: spectral representations. We start with the definition of non-negative definite function.

**Definition 1.2.1.** Fix a group (E,+). A symmetric function  $C: E \to \mathbb{R}$  is nonnegative definite on E if  $\forall n \in \mathbb{N}, \forall x_1, \dots, x_n \in E, \forall a_1, \dots, a_n \in \mathbb{R}$ ,

$$\sum_{i,j=1}^{n} a_i a_j C(x_i - x_j) \ge 0.$$

Equivalently, a non-negative definite function C on a group E is the covariance function of a stationary centered Gaussian field B on E (see Fact 1), namely

$$C(z) = \mathbb{E}[B_0 B_z] = \mathbb{E}[B_x B_{x+z}] \qquad \forall x \in E. \tag{1.6}$$

Using this fact, we easily obtain these useful properties.

**Proposition 1.2.1.** Fix a group (E, +). Let  $C, C_1, \ldots, C_n$  be non-negative definite functions on E and  $a_1, \ldots, a_n \geq 0$ . Then:

- 1.  $C(0) \ge 0$  and  $|C(z)| \le C(0)$ .
- 2.  $a_1C_1 + \cdots + a_nC_n$  and  $\prod_{i=1}^n C_i$  are non-negative definite functions on E.
- 3. If E is a metric space, then C is continuous if and only if C is continuous in 0.

*Proof.* 1. follows by (1.6) and Cauchy-Schwarz inequality. 2. follows by the fact that  $a_1C_1+\cdots+a_nC_n$  and  $\prod_{i=1}^n C_i$  are the covariance functions of linear combinations and

products (respectively) of n independent stationary Gaussian fields with covariance functions  $C_i$ . Finally, 3. follows by (1.6), observing that

$$|C(z) - C(z+h)| = |\mathbb{E}[B_0(B_z - B_{z+h})]| \le \sqrt{C(0)}\sqrt{2(C(0) - C(h)}.$$

The most important theorem of the subsection is Bochner's theorem.

**Theorem 1.2.2** (Bochner). Let  $C : \mathbb{R}^d \to \mathbb{R}$  be a non-negative definite function on  $\mathbb{R}^d$ , with C continuous and C(0) = 1. Then C is the Fourier transform of a unique Borel probability measure G on  $\mathbb{R}^d$ , namely

$$C(z) = \int_{\mathbb{R}^d} e^{i\langle z, x \rangle} G(dx). \tag{1.7}$$

Note that Bochner's theorem has been generalized by Yaglom in [Yag60] for non-negative definite functions on every group E satisfying very general assumptions, see also [Mal13, Theorem 2.18]. We only consider the Euclidean case  $E = \mathbb{R}^d$  for simplicity, since we will not focus on other cases.

In probabilistic terms, Bochner's theorem ensures that the covariance function  $C:\mathbb{R}^d\to\mathbb{R}$  of a stationary Gaussian field, if continuous, is the characteristic function of a random vector  $\Lambda$ , where  $\Lambda\sim G$ . Moreover, considering the random norm  $|\Lambda|\sim\mu$  and the random direction  $\widehat{\Lambda}:=\mathbf{1}_{\Lambda\neq 0}\Lambda/|\Lambda|\sim\sigma$ , with  $\mu,\sigma$  probability measures on  $\mathbb{R}_+$ ,  $S^{d-1}$  respectively  $^2$ , then we have

$$C(z) = \mathbb{E}\left[e^{i|\Lambda|\langle z, \widehat{\Lambda}\rangle}\right] = \int_0^\infty \mu(dr) \mathbb{E}\left[e^{ir\langle z, \widehat{\Lambda}\rangle} \middle| |\Lambda| = r\right]. \tag{1.8}$$

In particular, if  $|\Lambda|$ ,  $\widehat{\Lambda}$  are independent, we can write

$$C(z) = \int_0^\infty \mu(dr) \int_{S^{d-1}} \sigma(d\theta) \, e^{ir\langle z,\theta\rangle} \, .$$

**Definition 1.2.2.** Let  $B=(B_x)_{x\in\mathbb{R}^d}$  be a stationary Gaussian field on  $\mathbb{R}^d$  with continuous covariance function  $C:\mathbb{R}^d\to\mathbb{R}$ , C(0)=1. The **spectral measure** of B is the unique probability measure G on  $\mathbb{R}^d$  associated to C through (1.7). The **isotropic spectral measure** of B is the unique probability measure  $\mu$  on  $\mathbb{R}_+$  associated to C through (1.8). The **spherical spectral measure** of B is the unique probability measure  $\sigma$  on  $S^{d-1}$  associated to C through (1.8).

If a given Gaussian field  $B = (B_x)_{x \in \mathbb{R}^d}$  is stationary, then recall that it is isotropic if and only if its covariance function is radial, namely

$$C(z) = \rho(|z|)$$

where the function  $\rho: \mathbb{R}_+ \to \mathbb{R}$  is said the **isotropic covariance function** of B. Moreover, C is radial if and only if C(z) = C(Pz), or equivalently  $P\Lambda \sim \Lambda$ , for every

<sup>&</sup>lt;sup>2</sup>If d = 1, then  $S^0 = \{-1, 1\}$  and  $\sigma = \frac{1}{2}(\delta_1 + \delta_{-1})$ .

 $P \in SO(d)$ . Furthermore, since the latter is equivalent to say that the spherical spectral measure  $\sigma$  is the uniform probability measure on  $S^{d-1}$ , we obtain the following result, the so-called Schoenberg theorem.

**Theorem 1.2.3** (Schoenberg). Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a stationary Gaussian field with continuous covariance function  $C : \mathbb{R}^d \to \mathbb{R}$ , C(0) = 1, spectral measure G, isotropic spectral measure G, and spherical spectral measure G. Let G be the uniform probability measure on  $G^{d-1}$ . Then G is isotropic if and only if G = G. Moreover, by (1.8), if G is isotropic with isotropic covariance function G: G, we have

$$\rho(r) = \int_0^\infty b_d(rs)\mu(ds),\tag{1.9}$$

where  $b_d$  is the characteristic function of  $\nu = \sigma$ . In particular, we have

$$b_d(s) = \int_{S^{d-1}} e^{i\langle z,\theta\rangle} \nu(d\theta) = \Gamma\left(\frac{d}{2}\right) \left(\frac{s}{2}\right)^{1-\frac{d}{2}} J_{\frac{d}{2}-1}(s), \qquad s = |z|,$$

where  $J_{\frac{d}{2}-1}$  is the Bessel function of the first kind and order  $\frac{d}{2}-1$ .

- (1.9) could be seen as the isotropic version of (1.7), obtained fixing the spherical spectral measure  $\sigma$  equal to the uniform distribution on the sphere. The two spectral representations by Bochner, (1.7), and Schoenberg, (1.9), play a crucial role in the sequel, for two main reasons:
  - (1.7) is our gateway to Fourier analysis. Indeed, it allows to rewrite important quantities related to functionals of B in terms of Fourier transforms and integrals with respect to the spectral measure G. The variance of  $\int_{tD} B_x dx$ , for example, which has the form

$$\operatorname{Var}\left(\int_{D} B_{x} dx\right) = \int_{\mathbb{R}^{d}} \left| \int_{D} e^{i\langle x, \lambda \rangle} dx \right|^{2} G(d\lambda),$$

may (and will) be estimated under suitable conditions on G and D.

• (1.9) is used in Chapter 2, to estimate the decay of  $\rho$  under suitable conditions on  $\mu$ , see in particular Lemma 2.3.2. We conjecture that one may deduce similar results for C when B is not isotropic, i.e. when  $\sigma$  is not the uniform distribution  $\nu$ , assuming that the spherical spectral measure  $\sigma$  is "regular" enough. This may be done using estimates for the decay of the Fourier transform of measures supported on surfaces, see e.g. [Ste93, p.348, Theorem 1].

**Remark 1.2.1.** We conjecture that some of the arguments that we are going to use in the sequel, based on (1.7) and (1.9), can be extended from non-negative definite functions  $C: \mathbb{R}^d \to \mathbb{R}$  to **harmonizable** non-negative definite kernels  $\mathcal{C}: \mathbb{R}^d \times \mathbb{R}^d \to \mathbb{R}$  of the form

$$C(x,y) = \int_{\mathbb{R}^d} \int_{\mathbb{R}^d} e^{i\langle \lambda, x \rangle - i\langle \lambda', y \rangle} \mathcal{G}(d\lambda, d\lambda'), \tag{1.10}$$

where  $\mathcal{G}$  is a non-negative definite bimeasure of bounded variation, see e.g. [Swi02]. Such a class of kernels is much more general than the class of translation-invariant

kernels C(x,y) = C(x-y), which corresponds to the just discussed case of non-negative definite functions. In the latter case,  $\mathcal{G}$  is supported on the diagonal  $\lambda = \lambda'$ , and coincides with the spectral measure G of C. In other words, we conjecture that representations of the form (1.10) would allow to extend some of the results presented in this thesis from stationary Gaussian fields to non-stationary Gaussian fields with harmonizable covariance kernel C. Nevertheless, these possible extensions are not immediate and would require novel ideas, which are left for future research.

#### 1.2.2 Chaotic decompositions and Fourth Moment theorems

In this subsection, we introduce some concepts and results which will allow in the sequel to decompose the functional of a Gaussian field as the sum of easier functionals, the so-called chaotic components, and reduce the study of the functional to the one of its chaotic components.

#### **Isonormal Gaussian Processes**

Let us begin by introducing isonormal Gaussian processes, a special class of Gaussian fields on Hilbert spaces. In the following we denote by  $\mathcal H$  a real separable Hilbert space with inner product  $\langle \cdot, \cdot \rangle_{\mathcal H}$  and induced norm  $\| \cdot \|_{\mathcal H}$ .

**Definition 1.2.3.** An **isonormal Gaussian process** on  $\mathcal{H}$  is a Gaussian field  $I = (I(h))_{h \in \mathcal{H}}$  satisfying two properties:

- I is centered, that is  $\mathbb{E}[I(h)] = 0 \ \forall h \in \mathcal{H}$ .
- $I:\mathcal{H}\to L^2(\Omega)$  is an isometry, that is, I has covariance kernel

$$C(h,g) = \langle I(h), I(g) \rangle_{L^2(\Omega)} = \mathbb{E}[I(h)I(g)] = \langle h, g \rangle_{\mathcal{H}}, \quad \forall h, g \in \mathcal{H}.$$

**Remark 1.2.2.** Given a real separable Hilbert space  $\mathcal{H}$ , we can always construct an isonormal Gaussian process on  $\mathcal{H}$ , see e.g. [NP12, Proposition 2.1.1]. However, this construction is generally abstract. The effectiveness of isonormal Gaussian processes as mathematical tools comes from the fact that one can identify a Gaussian field B with a subset of an isonormal Gaussian process  $I = (I(h))_{h \in \mathcal{H}}$ , where the Hilbert space  $\mathcal{H}$  is suitably chosen depending on B and on the problem we are dealing with, see e.g. the construction provided in Proposition 1.2.4 and [NP12, Examples 2.1.3-2.1.7] for other instructive examples. Then, once B is identified as a subset of I, we can use powerful tools on I to study functionals of B.

In this thesis, we only deal with isonormal Gaussian processes as global environments containing stationary Gaussian fields on  $\mathbb{R}^d$ . In particular, a construction based on the spectral measure of a starting Gaussian field on  $\mathbb{R}^d$  is the following.

**Proposition 1.2.4.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a centered, stationary Gaussian field, with continuous covariance function  $C : \mathbb{R}^d \to \mathbb{R}$  and spectral measure G. Let us consider the real separable Hilbert space

$$L^{2}(G) := \left\{ h : \mathbb{R}^{d} \to \mathbb{C} : \int_{\mathbb{R}^{d}} |h(\lambda)|^{2} G(d\lambda) < \infty, \, \overline{h(\lambda)} = h(-\lambda) \right\}, \tag{1.11}$$

where  $|\cdot|$  denotes the complex norm and  $\bar{\cdot}$  the complex conjugate. Consider an isonormal Gaussian process  $I=(I(h))_{h\in L^2(G)}$  (recall Remark 1.2.2). For  $x\in \mathbb{R}^d$ , let  $e_x\in L^2(G)$  be defined as  $e_x(\lambda):=e^{i\langle x,\lambda\rangle}$ . Then B can be identified with a subset of I as follows

$$(B_x)_{x \in \mathbb{R}^d} \stackrel{d}{=} (I(e_x))_{x \in \mathbb{R}^d} \subseteq (I(h))_{h \in L^2(G)}.$$

*Proof.* The fact that the inner product  $\langle h,g\rangle_{L^2(G)}$  is real follows by the symmetry of G and the fact that h,g are even. The fact that B has the same distribution of  $(I(e_x))_{x\in\mathbb{R}^d}$  follows by Bochner's theorem and the definition of I

$$\mathbb{E}[B_x B_y] = C(x - y) = \int_{\mathbb{R}^d} e_x(\lambda) \overline{e_y(\lambda)} G(d\lambda) = \langle e_x, e_y \rangle_{L^2(G)} = \mathbb{E}[I(e_x) I(e_y)].$$

Proposition 1.2.4 allows to identify a stationary Gaussian field B on  $\mathbb{R}^d$  with a subset of an isonormal Gaussian process on  $L^2(G)$ . This construction will be important in the sequel, in particular for proving non-central limit theorems for functionals of B.

Note that, by definition of isonormal Gaussian process,  $I: \mathcal{H} \to L^2(\Omega)$  is **linear** and **closed**, since

$$\mathbb{E}\left[(I(h) - I(g))^2\right] = \|h - g\|_{\mathcal{H}}^2 \qquad \forall h, g \in \mathcal{H}.$$

This property is crucial, and may be considered as the first step to represent functionals of Gaussian fields in terms of operators acting on Hilbert spaces. If we consider linear combinations of a stationary Gaussian field B with continuous covariance function, for example, then by Proposition 1.2.4 and the linearity of I we have

$$a_1B_{x_1} + \dots + a_nB_{x_n} \stackrel{d}{=} I(a_1e_{x_1} + \dots + a_ne_{x_n}), \quad x_i \in \mathbb{R}^d, a_i \in \mathbb{R}.$$

Moreover, since I is closed, if  $\int_{D} |B_x| dx$  is well defined and in  $L^2(\Omega)$ , then

$$\int_{D} B_{x} dx \stackrel{d}{=} \int_{D} I(e_{x}) dx = I\left(\int_{D} e_{x} dx\right).$$

Such functionals of B, as well as all those which can be represented as evaluations of I in an element of  $\mathcal{H}$ , are said to be **linear functionals** of B. Note in particular that every linear functional is Gaussian.

As we are going to see, every square integrable functional of B can actually be written in terms of operators acting on Hilbert spaces.

#### Hermite polynomials, Wiener chaoses and Wiener-Itô integrals

In Definition 1.2.3,  $L^2(\Omega)$  is the space of square integrable random variables on  $(\Omega, \mathcal{F}, \mathbb{P})$ , the underlying probability space on which I is defined as a Gaussian field. From now on, we will implicitly assume that

$$\mathcal{F} = \sigma(I) = \sigma(I(h), h \in \mathcal{H})$$
.

or equivalently that  $L^2(\Omega) = L^2(\Omega, \sigma(I), \mathbb{P})$  is the space of square integrable functionals of I. Under this assumption, every functional of I in  $L^2(\Omega)$  can be decomposed as the sum of orthogonal elements in "easier" Hilbert subspaces, called Wiener chaoses. The qth Wiener chaos is defined as the closed linear subspace of  $L^2(\Omega)$ 

$$\overline{H_q} := \overline{\operatorname{span}\{H_q(I(h)), h \in \mathcal{H}, \|h\|_{\mathcal{H}} = 1\}}^{L^2(\Omega)}$$

where  $H_q$  is the qth Hermite polynomial, defined by the recurrence relation

$$H_0(x) = 1$$
,  $H_1(x) = x$ ,  $H_q(x) = xH_{q-1}(x) - (q-1)H_{q-2}(x)$ . (1.12)

Note that  $\overline{H_0}=\mathbb{R}$  and  $\overline{H_1}=I=(I(h))_{h\in\mathcal{H}}.$ 

As stated in the following proposition, the family of Wiener chaoses  $(\overline{H_q})_{q\in\mathbb{N}}$  plays in  $L^2(\Omega)$  the same role that the orthogonal basis of Hermite polynomials  $(H_q)_{q\in\mathbb{N}}$  plays in  $L^2(\mathbb{R},\gamma(dx))$ , where  $\gamma$  is the standard Gaussian measure

$$\gamma(dx) := \frac{e^{-x^2/2}}{\sqrt{2\pi}} dx.$$

**Terminology.** In the sequel, **integrating by parts** with respect to a standard Gaussian random variable  $N \sim N(0,1) \sim \gamma$  will mean

$$\mathbb{E}[f(N)N] = \int_{\mathbb{R}} f(x) \, x \, \gamma(dx) = \int_{\mathbb{R}} f'(x) \gamma(dx) = \mathbb{E}[f'(N)],$$

for f absolutely continuous with  $f' \in L^1(\mathbb{R}, \gamma(dx))$  (see e.g. [NP12, Lemma 1.1.1]).

**Proposition 1.2.5.** Let  $(H_q)_{q\in\mathbb{N}}\subseteq L^2(\mathbb{R},\gamma(dx))$  be the family of Hermite polynomials and  $(\overline{H_q})_{q\in\mathbb{N}}\subseteq L^2(\Omega)$  be the family of Wiener chaoses. Then:

• For  $N_1, N_2 \sim N(0,1)$  jointly Gaussian, we have the **isometry property** 

$$\mathbb{E}\left[H_q(N_1)H_p(N_2)\right] = \mathbf{1}_{\{q=p\}} \, q! (\mathbb{E}[N_1 N_2])^q. \tag{1.13}$$

•  $(H_q)_{q\in\mathbb{N}}$  is an orthogonal basis in  $L^2(\mathbb{R}, \gamma(dx))$  and  $L^2(\Omega)$  coincides with the direct sum of  $(\overline{H_q})_{q\in\mathbb{N}}$ , namely

$$L^{2}(\Omega) = \bigoplus_{q=0}^{\infty} \overline{H_{q}}.$$
(1.14)

*Proof.* First of all, note that by definition of Hermite polynomials one can easily prove by induction that  $H'_q(x) = qH_{q-1}(x)$  for  $q \ge 1$ . Now let us prove that  $\mathbb{E}[H_q(N)] = 0$ ,  $N \sim N(0,1)$  for  $q \ge 1$ . For q=1 the claim is trivially true. For  $q \ge 2$  we have by definition of Hermite polynomial

$$\mathbb{E}[H_q(N)] = \mathbb{E}[NH_{q-1}(N)] - (q-1)\mathbb{E}[H_{q-2}(N)] = 0, \tag{1.15}$$

where the last equality follows by integration by parts and  $H'_q(x) = qH_{q-1}(x)$ . Now we can prove the isometry property (1.13). Let us assume without loss of generality

that  $q \le p$  and proceed by induction on q. If q = 0, then (1.13) is (1.15). If  $q \ge 1$ , by definition of  $H_q$  and inductive hypothesis we have

$$\mathbb{E}[H_q(N_1)H_p(N_2)] = \mathbb{E}[N_1H_{q-1}(N_1)H_p(N_2)].$$

Then the proof of (1.13) is concluded observing that  $(N_1, N_2)$  has the same distribution of  $(N_1, \rho N_1 + \sqrt{1 - \rho^2} N_1')$ , where  $N_1'$  is an independent copy of  $N_1$  and  $\rho = \mathbb{E}[N_1 N_2]$ . Indeed, integrating by parts with respect to  $N_1$  and recalling that  $H_q'(x) = qH_{q-1}(x)$ , we obtain (here we use the convention  $H_{-1} = 0$  if q = 1)

$$\mathbb{E}[H_q(N_1)H_p(N_2)] = \mathbb{E}\left[N_1H_{q-1}(N_1)H_p\left(\rho N_1 + \sqrt{1-\rho^2}N_1'\right)\right]$$

$$= (q-1)\mathbb{E}[H_{q-2}(N_1)H_p(N_2)] + p\rho\mathbb{E}[H_{q-1}(N_1)H_{p-1}(N_2)]$$

$$= p\rho\mathbb{E}[H_{q-1}(N_1)H_{p-1}(N_2)].$$

To prove the second part of the statement, we need to show that  $(H_q)_{q\geq 1}$  is dense in  $L^2(\mathbb{R},\gamma(dx))$  and  $\operatorname{span}\{H_q(I(h)),\ q\in\mathbb{N},\ h\in\mathcal{H},\ \|h\|_{\mathcal{H}}=1\}$  is dense in  $L^2(\Omega)$ . These facts follow by standard considerations, mainly based on the density of monomials in  $L^2(\mathbb{R},\gamma(dx))$ , the linearity of  $I:\mathcal{H}\to L^2(\Omega)$  and the fact that  $H_q$  is a polynomial of degree q for every  $q\in\mathbb{N}$ , see e.g. [NP12, Proposition 1.1.5, Theorem 2.2.4].

The previous ensures that, just as for a function  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$  we can write

$$\varphi = \sum_{q=0}^{\infty} a_q H_q, \qquad a_q := \frac{1}{q!} \int_{\mathbb{R}} \varphi(x) H_q(x) \gamma(dx), \qquad (1.16)$$

similarly for every functional  $Y \in L^2(\Omega)$  we have the orthogonal decomposition

$$Y = \sum_{q=0}^{\infty} Y[q], \qquad Y[q] := \text{proj}_{|_{\overline{H_q}}}(Y),$$
 (1.17)

where the projection  $Y[q] \in \overline{H_q(B)}$  of Y is said the qth chaotic component of Y and (1.17) is said the Wiener-Itô chaotic decomposition of Y.

Let us now focus on the single chaotic components. Note that  $Y[0] = \mathbb{E}[Y]$  since  $\overline{H_0} = \mathbb{R}$ . Moreover,  $Y[1] = I_1(h_1)$  for some  $h_1 \in \mathcal{H}$  since  $\overline{H_1} = I$ . Thus,  $I: \mathcal{H} \to \overline{H_1}$  is an isomorphism between  $\mathcal{H}$  and the first Wiener chaos. Thanks to this fact, one can study the first chaotic component as an element of  $\mathcal{H}$ . In the same vein, we would like to study the qth chaotic component ( $q \geq 2$ ) as an element of an Hilbert space. To do this, we introduce multiple Wiener-Itô integrals.

Let  $\mathcal{H}^{\otimes q}$  be the Hilbert space obtained as the q-fold tensor product of  $\mathcal{H}$ , generated by the set  $\{h_1 \otimes \cdots \otimes h_q : h_i \in \mathcal{H}, \|h_i\|_{\mathcal{H}} = 1\}$ , endowed with the standard tensor inner product  $\langle \cdot, \cdot \rangle_q$  and norm  $\|\cdot\|_q$ , satisfying

$$\langle h_1 \otimes \cdots \otimes h_q, g_1 \otimes \cdots \otimes g_q \rangle_q := \prod_{i=1}^q \langle h_i, g_i \rangle.$$

Let  $H^{\odot q}$  be the symmetrized q-fold tensor product of  $\mathcal{H}$ , defined as the Hilbert

subspace of  $\mathcal{H}^{\otimes q}$  generated by the set

$$H^{\odot q} = \overline{\operatorname{span}\{\operatorname{Sym}(h_1 \otimes \cdots \otimes h_q) : h_i \in \mathcal{H}, \|h_i\|_{\mathcal{H}} = 1\}}^{\mathcal{H}^{\otimes q}}.$$

Here Sym is the symmetrization operator defined as

$$\operatorname{Sym}(h_1 \otimes \cdots \otimes h_q) = \frac{1}{q!} \sum_{\sigma \in S_q} h_{\sigma(1)} \otimes \cdots \otimes h_{\sigma(q)},$$

where  $S_q$  is the group of all the permutations of  $\{1, \ldots, q\}$ . In order to define Wiener-Itô integrals, we use the following classical polarization formula, whose proof can be found e.g. in [Flo97].

**Lemma 1.2.6** (Polarization formula). For  $q \ge 1$  and  $h_0, h_1, \ldots, h_q \in \mathcal{H}$ , we have

$$\operatorname{Sym}(h_1 \otimes \cdots \otimes h_q) = \frac{1}{2^q q!} \sum_{\epsilon_i = \pm 1} \epsilon_1 \cdots \epsilon_q (h_0 + \epsilon_1 h_1 + \cdots + \epsilon_q h_q)^{\otimes q}.$$

where  $h^{\otimes q} := h \otimes \cdots \otimes h$ , or in other words

$$\operatorname{Sym}(h_1 \otimes \cdots \otimes h_q) = \frac{1}{q!} \mathbb{E} \left[ \varepsilon_1 \dots \varepsilon_q (h_0 + \varepsilon_1 h_1 + \dots + \varepsilon_q h_q)^{\otimes q} \right]$$

where  $\epsilon_1, \ldots, \epsilon_q$  are independent Rademacher random variables<sup>3</sup>. In particular

$$H^{\odot q} = \overline{\operatorname{span}\{h^{\otimes q} : h \in \mathcal{H}, \|h\|_{\mathcal{H}} = 1\}}^{\mathcal{H}^{\otimes q}}.$$
 (1.18)

**Definition 1.2.4.** Let the notations above prevail. Fix  $q \geq 1$ . The qth **Wiener-Itô** integral with respect to the isonormal Gaussian process I is the unique linear closed operator  $I_q: \mathcal{H}^{\odot q} \to \overline{H_q}$  such that for every  $h \in \mathcal{H}$ , ||h|| = 1,

$$I_q(h^{\otimes q}) = H_q(I(h)).$$

In particular,  $I_q/\sqrt{q!}$  is an isomorphism and  $\mathcal{H}^{\odot q}, \overline{H_q}$  are isomorphic Hilbert spaces.

Proof of Definition 1.2.4. Let us prove that the previous definition is well posed and that  $I_q/\sqrt{q!}$  is an isomorphism between the Hilbert spaces  $H^{\odot q}$  and  $\overline{H_q}$ . First, we define the qth Wiener-Itô integral  $I_q$  on q-fold tensors  $h^{\otimes q}$ , ||h||=1, as

$$I_q(h^{\otimes q}) = H_q(I(h)),$$

and extend by linearity the definition to span $\{h^{\otimes q}, \|h\| = 1\}$ . Then, recalling the isometry property of Hermite polynomials we have

$$\mathbb{E}[I_q(v_1)I_q(v_2)] = q! \langle v_1, v_2 \rangle_q, \tag{1.19}$$

 $\forall v_1, v_2 \in \operatorname{span}\{h^{\otimes q}, \|h\| = 1\}$ . Thus, by (1.18) we can extend  $I_q$  to an operator  $I_q : \mathcal{H}^{\odot q} \to \overline{H_q}$ , such that the isometry property (1.19) holds for every  $v_1, v_2 \in H^{\odot q}$ . Moreover, by definition of Wiener chaos  $\overline{H_q}$  and again (1.19), we have that every

<sup>&</sup>lt;sup>3</sup>Actually the result holds more in general for  $\epsilon_1, \ldots, \epsilon_q$  independent with mean 0 and variance 1, see [Flo97].

element in  $\overline{H_q}$  can be written as  $I_q(v)$  for some  $v \in \mathcal{H}^{\odot q}$ . We have proved that  $I_q/\sqrt{q!}$  is a surjective isometry, i.e. an isomorphism .

Combining the Wiener-Itô chaotic decomposition (1.17) and the just proved fact  $\overline{H_q}=(I_q(h))_{h\in\mathcal{H}^{\odot q}}$ , we are finally able to decompose every square integrable functional Y of an isonormal Gaussian process I, i.e.  $Y\in L^2(\Omega)$ , as the infinite sum of Wiener-Itô integrals evaluated in elements of  $H^{\odot q}$ , namely

$$Y = \mathbb{E}[Y] + \sum_{q=1}^{\infty} I_q(h_q)$$
(1.20)

for some unique symmetric tensors  $(h_q)_{q=1}^{\infty}$ ,  $h_q \in \mathcal{H}^{\odot q}$ . This representation plays a crucial role in the study of square integrable functionals of Gaussian fields. Indeed, one can often reduce the study of Y to the one of the single chaotic components  $I_q(h_q)$ , which are usually much easier to deal with.

#### Contractions and product formula

Formula (1.20) is very general, but also abstract. In other words, even if every functional  $Y \in L^2(\Omega)$  is uniquely determined by the tensors  $h_q \in \mathcal{H}^{\odot q}$ , the latter are not always easy to derive. One of the most powerful tools to solve this kind of issues is the **product formula**, which allows to write the product of Wiener-Itô integrals  $I_q(f)I_p(g)$  as the sum of Wiener Itô integrals evaluated in the symmetrized **contractions** of f and g. As a consequence, the product formula also allows to derive explicit expression for the powers and moments of Wiener-Itô integrals. In order to introduce this important formula, we first need the definition of contraction.

**Definition 1.2.5.** Let  $p, q \ge 1$ ,  $1 \le r \le p \land q$ . Let  $(e_i)_{i \in \mathbb{N}}$  be an orthonormal basis of  $\mathcal{H}$ . Let  $f \in \mathcal{H}^{\odot q}$  and  $g \in \mathcal{H}^{\odot p}$ , with orthogonal expansions

$$f = \sum_{i_1, \dots, i_q \in \mathbb{N}} f_{i_1, \dots, i_q} e_{i_1} \otimes \dots \otimes e_{i_q}, \qquad g = \sum_{j_1, \dots, j_p \in \mathbb{N}} g_{j_1, \dots, j_p} e_{j_1} \otimes \dots \otimes e_{j_p},$$

where the coefficients  $f_{i_1,\dots,i_q}$  and  $g_{j_1,\dots,j_p}$  are invariant by permutations of the indices. Let us use the notation

$$\langle f, e_{k_1} \otimes \cdots \otimes e_{k_r} \rangle := \sum_{i_1, \dots, i_q \in \mathbb{N}} f_{i_1, \dots, i_q} e_{i_1} \otimes \cdots \otimes e_{i_{q-r}}.$$

The rth contraction of f and g is the tensor  $f \otimes_r g$  in  $\mathcal{H}^{\otimes p+q-2r}$  defined as

$$f \otimes_r g := \sum_{k_1, \dots, k_r \in \mathbb{N}} \langle f, e_{k_1} \otimes \dots \otimes e_{k_r} \rangle \langle g, e_{k_1} \otimes \dots \otimes e_{k_r} \rangle.$$

**Remark 1.2.3** (Some useful inequalities). For r=0, we formally define  $f\otimes_0 g=f\otimes g$ . Moreover, for the rth contraction of  $f\in\mathcal{H}^{\odot q}$  with itself, we have by Cauchy-Schwarz inequality (see e.g. [NP12, (B.4.3)])

$$||f \otimes_r f||_{2q-2r} \le ||f \otimes_0 f||_{2q} = ||f||_q^2.$$

Combining the previous and the equivalence (see e.g. [NP12, Lemma 6.2.1])

$$||f \otimes_r g||_{q+p-2r}^2 = \langle f \otimes_{q-r} f, g \otimes_{p-r} g \rangle_{2r},$$

which for f=g provides the symmetry  $\|f\otimes_r f\|_{2q-2r}=\|f\otimes_{q-r}\|_{2r}$ , we also obtain

$$||f \otimes_r g||_{q+p-2r} \le ||f \otimes_{q-r} f||_{2r} ||g \otimes_{p-r} g||_{2r} \le ||f||_q^2 ||g||_p^2$$

**Remark 1.2.4.** Note that for a q-fold tensor product  $h_1^{\otimes q}$  and a p-fold tensor product  $h_2^{\otimes p}$ ,  $h_1, h_2 \in \mathcal{H}$ , the rth contraction  $h_1^{\otimes q} \otimes_r h_2^{\otimes p}$  coincides with

$$h_1^{\otimes q} \otimes_r h_2^{\otimes p} = \langle h_1, h_2 \rangle^r h_1^{\otimes q-r} \otimes h_2^{\otimes p-r}. \tag{1.21}$$

Moreover, thanks to Lemma 1.2.6, the rth contraction  $\otimes_r$  may be also defined imposing (1.21) for every  $h_1^{\otimes q}$ ,  $h_2^{\otimes p}$ ,  $h_1, h_2 \in \mathcal{H}$ . Indeed, one can linearly extend the definition (1.21) to  $f \in \operatorname{span}\{h^{\otimes q}, \|h\| = 1\}$ ,  $g \in \operatorname{span}\{h^{\otimes p}, \|h\| = 1\}$ , and then prove the relations in Remark 1.2.3 for every  $f \in \operatorname{span}\{h^{\otimes q}, \|h\| = 1\}$ ,  $g \in \operatorname{span}\{h^{\otimes p}, \|h\| = 1\}$ , to extend the bilinear operator  $f \otimes_r g$  to  $f \in \mathcal{H}^{\odot q}$ ,  $g \in \mathcal{H}^{\odot p}$  by density (see (1.18)).

We are finally ready to state the product formula.

**Theorem 1.2.7** (Product formula). Let  $q, p \ge 1$ . Then, for  $f \in \mathcal{H}^{\odot q}$  and  $g \in \mathcal{H}^{\odot p}$ , we have

$$I_q(f)I_p(g) = \sum_{r=0}^{q \wedge p} r! \binom{q}{r} \binom{p}{r} I_{p+q-2r}(f \tilde{\otimes}_r g),$$

where  $f \tilde{\otimes}_r g$  is a concise notation for  $\operatorname{Sym}(f \otimes_r g)$ . In particular, if  $e_1, \ldots, e_k$  are orthonormal in  $\mathcal{H}$ , the formula implies

$$\prod_{i=1}^k H_{p_i}(I(e_i)) = \prod_{i=1}^k I_{p_i}(e_i^{\otimes p_i}) = I_{p_1 + \dots + p_k}(e_1^{\otimes p_1} \tilde{\otimes} \dots \tilde{\otimes} e_k^{\otimes p_k}).$$

*Proof.* By Lemma 1.2.6, Remark 1.2.4, since  $I_q$ ,  $\operatorname{Sym}$  are linear and  $I_q$  is closed, it is enough to prove the result for  $f=h_1^{\otimes q}$  and  $g=h_2^{\otimes p}$ ,  $h_i\in\mathcal{H}, \|h_i\|=1$ . Here we prove the result only for q=1, see e.g [Nua06, Proposition 1.1.3] for the inductive extension to  $q\geq 1$ . For q=1, we have to prove

$$I(h_1)I_p(h_2^{\otimes p}) = I_{p+1}(h_1\tilde{\otimes}h_2^{\otimes p}) + pI_{p-1}(h_1\tilde{\otimes}_1h_2^{\otimes p})$$

$$= I_{p+1}(h_1\tilde{\otimes}h_2^{\otimes p}) + p\langle h_1, h_2\rangle I_{p-1}(h_2^{\otimes p-1}).$$
(1.22)

First of all, note that if  $h_1=h_2$  the result trivially follows by definition of Hermite polynomial and Wiener-Itô integral. Suppose now that the result holds when  $\langle h_1,h_2\rangle=0$ . Then, by a Gram-Schmidt argument (writing  $h_1=h_1'+\langle h_1,h_2\rangle h_2$ ) and linearity of all the operators that come into play, the result holds true for general  $h_1$ . Thus, without loss of generality we can prove (1.22) for  $\langle h_1,h_2\rangle=0$ . By Lemma 1.2.6 and the definitions of Wiener-Itô integral and Hermite polynomial,  $I_{p+1}(h_1\tilde{\otimes}h_2^{\otimes p})$ 

can be rewritten as (when p = 1 the third addend is 0)

$$\frac{1}{(p+1)} \left( I(h_1) I_p(h_2^{\otimes p}) + p I(h_2) I_p(h_1 \tilde{\otimes} h_2^{\otimes p-1}) - p (p-1) I_{p-1}(h_1 \tilde{\otimes} h_2^{\otimes p-2}) \right).$$

Thus, for p=1 the result is true. For  $p \ge 2$ , assume that (1.22) holds for  $k \le p-1$ . Then the proof is concluded if the following holds

$$I(h_2)I_p(h_1 \tilde{\otimes} h_2^{\otimes p-1}) - (p-1)I_{p-1}(h_1 \tilde{\otimes} h_2^{\otimes p-2}) = I(h_1)I_p(h_2^{\otimes p}).$$

By using the inductive hypothesis on the LHS and the defintion of Hermite polynomial on the RHS, both sides are equal to

$$I(h_1)I(h_2)I_{p-1}(h_2^{\otimes p-1}) - (p-1)I(h_1)I_{p-2}(h_2^{\otimes p-2})$$

and the proof is concluded.

#### Fourth Moment theorems and related results

A consequence of the product formula is that in a fixed chaos q we are able to derive explicit expressions for powers and moments of  $I_q(f)$  in terms of f. Thus, one may use the method of moments to study the distribution of  $I_q(f)$  as  $t \to \infty$ , where  $(f_t)_{t>0} \subseteq H^{\odot q}$ . Fortunately, thanks to Nualart and Peccati, we do not need to study every moment, but only (the second and) the fourth. Indeed, these authors proved in [NP05] a surprising result on the asymptotic behavior of Wiener-Itô integrals in a fixed Wiener chaos: the convergence in distribution to a Gaussian random variable  $N(0,\sigma^2)$  is equivalent to the convergence of only the second moment to  $\sigma^2>0$  and of the fourth moment to  $3\sigma^2$ . From that time to the present, the so-called Fourth Moment theorem of Nualart and Peccati has been extended, adapted and applied in many different contexts, see e.g. [Azm+16; Cam+16; DP17; DP18; HMP24; NP09; Nou+16; NP05; PT05; Zhe19].

**Theorem 1.2.8** (Fourth Moment Theorem). Let I be an isonormal Gaussian process on  $\mathcal{H}$ . Fix  $q \geq 2$  and consider  $(I_q(f_t))_{t>0} \subseteq \overline{H_q}$ , with  $(f_t)_{t>0} \subseteq H^{\odot q}$ . Suppose that  $\operatorname{Var}(I_q(f_t)) = q! \|f_t\|_q^2 \to \sigma^2 \in (0, \infty)$ . Then, the following assertions all are equivalent:

- $I_q(f_t) \to N(0, \sigma^2)$ .
- $\mathbb{E}\left[I_q(f_t)^4\right] \to 3\sigma^2$ .
- $\max_{r=1,\dots,q-1} ||f_t \otimes_r f_t||_{2q-2r} \to 0$ .

There are many proofs of the Fourth Moment Theorem. Here we refer in particular to the proof in [Nou11], which consists in a recursive use of the product formula to compute the moments of  $I_q(f_t)$  and a combinatorial manipulation of the obtained expressions. In particular, one obtains (see e.g. [Nou11][Lemma 4.1])

$$c(q) \max_{r=1,\dots,q-1} \|f_t \otimes_r f_t\|_{2q-2r}^2 \le \mathbb{E}[I_q(f_t)^4] - 3 \le c'(q) \max_{r=1,\dots,q-1} \|f_t \otimes_r f_t\|_{2q-2r}^2,$$
(1.23)

for some constants c'(q) > c(q) > 0, which explains the equivalence between the last two conditions of the Fourth Moment Theorem.

The following multidimensional version of Theorem 1.2.8 was proved by Peccati and Tudor in [PT05]. In particular, it asserts that, for Wiener-Itô integrals, componentwise convergence to Gaussian random variables is equivalent to joint convergence to Gaussian random vectors.

**Theorem 1.2.9** (Multidimensional Fourth Moment Theorem). Let I be an isonormal Gaussian process on  $\mathcal{H}$ . Fix  $n \geq 2$ ,  $q_1, \ldots, q_n \geq 1$  and consider  $(I_{q_1}(f_{1,t}), \ldots, I_{q_n}(f_{n,t}))$ , with  $f_{i,t} \in \mathcal{H}^{\odot q_i}$ , t > 0. Finally, assume that the following limits exist finite for  $i, j \in \{1, \ldots, n\}$ 

$$\sigma_{ij} := \lim_{t \to \infty} \operatorname{Cov}\left(I_{q_i}(f_{i,t}), I_{q_j}(f_{j,t})\right) < \infty.$$
(1.24)

Let  $\underline{N} = (N_1, \dots, N_n)$  be a centered Gaussian vector with covariance matrix (1.24). Then, the following assertions are all equivalent:

- $(I_{q_1}(f_{1,t}), \dots, I_{q_n}(f_{n,t})) \xrightarrow{d} \underline{N}$
- For  $i=1,\ldots,n$ , we have  $\mathbb{E}\left[I_q(f_{i,t})^4\right] o 3\sigma_{ii}$ .

The previous result is not only particularly useful to prove multidimensional central limit theorems for functionals of Gaussian fields, but also, combined with the chaotic decomposition (1.20), to prove central limit theorems for general functionals of Gaussian fields, see e.g. [NP12, Theorem 6.3.1].

Another important generalization of Theorem 1.2.8 are quantitative bounds for the normal approximation. The first quantitative Fourth Moment Theorem was proved by Nourdin and Peccati in [NP09]. In particular, these authors considered the Fortet-Mourier, Wasserstein, Kolmogorov and total variation distances. In the following, for simplicity, we only state the result for the total variation distance between the laws of two random variables  $F_1$  and  $F_2$ , defined (with a slight abuse of notation) as

$$d_{TV}(F_1, F_2) = \sup_{E \in \mathcal{B}(\mathbb{R})} |\mathbb{P}(F_1 \in E) - \mathbb{P}(F_2 \in E)|.$$

For the proof of the next statement, see e.g. [NP12, Theorem 5.2.6].

**Theorem 1.2.10** (Quantitative Fourth Moment Theorem). Let I be an isonormal Gaussian process on  $\mathcal{H}$ . Fix  $q \geq 2$  and let  $I_q(f) \in \overline{H_q}$  be a functional of I in the qth Wiener chaos, with  $f \in H^{\odot q}$ . Suppose that  $\operatorname{Var}(I_q(f)) = q! ||f||_q^2 > 0$ . Then, if  $N \sim N(0,1)$ , we have

$$d_{TV}\left(\frac{I_q(f)}{\sqrt{\operatorname{Var}(I_q(f))}}, N\right) \le 2\sqrt{\frac{q-1}{3q}}\sqrt{\mathbb{E}\left[I_q(f)^4\right] - 3}.$$
 (1.25)

Note that the bound (1.25) may be combined with (1.23), obtaining

$$d_{TV}\left(\frac{I_q(f)}{\sqrt{\text{Var}(I_q(f))}}, N\right) \le \sqrt{c'(q)} \max_{r=1,\dots,q-1} \|f \otimes_r f\|_{2q-2r},$$

an expression which is particularly useful in applications. Regarding the proof of Theorem 1.2.10, we remark that Nourdin and Peccati actually proved in [NP09] a more general version of the previous result, providing a bound for  $d_{TV}(F, N)$  for

every  $F\in L^2(\Omega)$  with square integrable Malliavin derivative. This condition is equivalent to ask that

$$\sum_{q=1}^{\infty} q \, q! \, \|f_q\|_q^2 < \infty \,, \tag{1.26}$$

where the  $f_q$  are the symmetric tensors uniquely associated to F through its chaotic decomposition (1.20). To prove these general bounds under assumptions on the Malliavin derivative of F, they introduced the so-called Malliavin-Stein method, which is nowadays very popular and well-developed, and has been extended and applied in different frameworks, see e.g. [APY21; LPS16; Led12]. In a nutshell, this method consists of two main steps:

• Exploiting Stein's method to bound  $d_{TV}(F, N)$  with an expression that explicitly depends only on F, i.e.

$$d_{TV}(F, N) \le \sup_{f \in \mathcal{F}_{TV}} |\mathbb{E}[f'(F)] - \mathbb{E}[F f(F)]|,$$

where  $\mathcal{F}_{TV}$  is a suitable set of absolutely continuous functions. The dependence on  $N \sim N(0,1)$  is implicit, hidden in the expression of the bound. Indeed, by Stein's lemma (see e.g. [NP12, Lemma 3.1.2]), F is Gaussian if and only if  $E[f'(F)] - \mathbb{E}[F f(F)] = 0$  for every f differentiable with  $f' \in L^1(\mathbb{R}, \gamma(dx))$ .

• Expressing the bound obtained in the previous step in terms of Malliavin operators, see e.g. [NP12][Proposition 5.1.1]. This yields bounds for general classes of Malliavin differentiable random variables. In particular, as an application, one obtains the quantitative Fourth Moment Theorem 1.2.10.

In this thesis, mentioning the Malliavin-Stein method, we intend to refer to results such as Theorem 1.2.10, obtained as a consequence of the two steps above.

#### Application to additive functionals of Gaussian fields

We conclude the section with an important application of the previous results: additive functionals of Gaussian fields.

Let us consider a stationary, continuous Gaussian field  $B=(B_x)_{x\in\mathbb{R}^d}$ , with  $B_x\sim N(0,1)$ , covariance function  $C:\mathbb{R}^d\to\mathbb{R}$  and spectral measure G. The spectral measure G is well defined by Bochner's theorem, because every stationary, continuous Gaussian field has continuous covariance function. By Proposition 1.2.4, we can identify B with a subset of an isonormal Gaussian process I on  $\mathcal{H}=L^2(G)$ , so that

$$(B_x)_{x \in \mathbb{R}^d} \stackrel{d}{=} (I(e_x))_{x \in \mathbb{R}^d}, \qquad e_x(\cdot) := e^{i\langle x, \cdot \rangle}.$$

Moreover, we have:

- $\mathcal{H}^{\otimes q}$  can be identified with the space  $L^2((\mathbb{R}^d)^q, G^{\otimes q})$  of complex-valued even functions that are square integrable with respect to the product measure  $G^{\otimes q}$ ;
- $\mathcal{H}^{\odot q}$  can be identified with the space  $L^2_s((\mathbb{R}^d)^q, G^{\otimes q})$  of symmetric functions in  $L^2((\mathbb{R}^d)^q, G^{\otimes q})$ , where  $f:(\mathbb{R}^d)^q \to \mathbb{C}$  symmetric means

$$f(\lambda_1, \dots, \lambda_q) = f(\lambda_{\sigma(1)}, \dots, \lambda_{\sigma(q)}) \quad \forall \sigma \in S_q.$$

In particular, note that the q-fold tensor product  $e_x^{\otimes q}$  can be seen as the function  $e_x^{\otimes q}:(\mathbb{R}^d)^q\to\mathbb{C}$ , defined as  $e_x^{\otimes q}(\lambda_1,\dots,\lambda_q)=e^{i\langle x,\lambda_1+\dots+\lambda_q\rangle}$ .

Consider now a function  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$  and a compact domain  $D \subseteq \mathbb{R}^d$ , with  $\operatorname{Vol}(D) > 0$ . We define the family of additive functionals  $Y = (Y_t)_{t>0}$  of B as

$$Y_t := \int_{tD} \varphi(B_x) dx, \qquad t > 0.$$

By (1.16) and (1.13), the chaotic decomposition of  $Y_t$  (see (1.17)) is

$$Y_t - \mathbb{E}[Y_t] = \sum_{q=1}^{\infty} a_q \int_{tD} H_q(B_x) dx$$
 (1.27)

where we recall that  $a_q$  and  $H_q$  are (respectively) the qth coefficient of  $\varphi$  and the qth Hermite polynomial appearing in (1.16). Moreover, by Definition 1.2.4 (recall  $I(e_x) \stackrel{d}{=} B_x \sim N(0,1) \implies ||e_x|| = 1$ ) and the continuity of  $I_q$ , we obtain

$$Y_t - \mathbb{E}[Y_t] = \sum_{q=1}^{\infty} I_q(f_{q,t}),$$

with  $f_{q,t} \in H^{\odot q}$  identifiable with the symmetric function  $f_{q,t} : (\mathbb{R}^d)^q \to \mathbb{C}$ 

$$f_{q,t} := \int_{tD} e_x^{\otimes q} dx, \qquad f_{q,t}(\lambda_1, \dots, \lambda_q) := \mathcal{F}[\mathbf{1}_{tD}](\lambda_1 + \dots + \lambda_q),$$

where  $\mathcal{F}$  denotes the Fourier transform. This last identification will be particularly useful to prove non-central limit theorems for Y.

In this setting, one can often reduce the study of  $Y_t$  to that of its chaotic components  $I_q(f_{q,t}) = a_q \int_{tD} H_q(B_x) dx$ . Thus, recalling the aforementioned Fourth Moment theorems, it becomes important to study the contractions  $f_{q,t} \otimes_r f_{q,t}$ ,

$$f_{q,t} \otimes_r f_{q,t} = \int_{tD} dx \int_{tD} dy C(x-y)^r e_x^{\otimes q-r} \otimes e_y^{\otimes q-r},$$

with squared norm

$$||f_{q,t} \otimes_r f_{q,t}||_{2q-2r}^2 = \int_{(tD)^4} C(x-y)^r C(u-v)^r C(x-u)^{q-r} C(y-v)^{q-r} dx dy du dv.$$
(1.28)

This expression can then be combined with the Nualart-Peccati, Peccati-Tudor and Nourdin-Peccati Fourth Moment theorems to prove central limit theorems for  $I_q(f_{q,t})$ . This fact will be crucial for many of the coming discussions.

#### 1.2.3 Cross covariograms

We conclude this section with the last simple (but crucial) tool of this thesis: cross covariograms. For all the missing details and proofs, we refer to [Gal11].

**Definition 1.2.6.** Let  $D, L \subseteq \mathbb{R}^d$  be compact sets. The cross covariogram of D, L is

the function  $g_{D,L}:\mathbb{R}^d \to \mathbb{R}$  defined as

$$g_{D,L}(z) := \operatorname{Vol}(D \cap (L+z)) = \mathbf{1}_D * \mathbf{1}_{-L}(z), \qquad z \in \mathbb{R}^d.$$

The covariogram of D is the function  $g_D := g_{D,D}$ .

Covariograms and cross covariograms are scale-invariant  $(g_{tD,tL}(z) = t^d g_{D,L}(z/t))$  and uniformly continuous (with compact support D-L). Moreover,  $g_D$  is a Lipschitz function if and only if its perimeter is finite, and  $g_{D,L}$  is Lipschitz with Lipschitz constant  $\operatorname{Per}(D) \wedge \operatorname{Per}(L)$  if  $\operatorname{Per}(D) \wedge \operatorname{Per}(L) < \infty$ , where we recall that the definition of (generalized) perimeter is

$$\operatorname{Per}(D) := \sup \left\{ \int_{D} \operatorname{div} \varphi(x) \, dx : \varphi \in C_{c}^{1}(\mathbb{R}^{d}, \mathbb{R}^{d}), \|\varphi\|_{\infty} \le 1 \right\}. \tag{1.29}$$

See Section 4.2.2 for additional details and properties.

The reason why we introduce them is that they play a fundamental role in our analysis. If we consider the qth chaotic component  $\int_{tD} H_q(B_x) dx$  considered in (1.27), for instance, we know that its variance is

$$\operatorname{Var}\left(\int_{tD} H_q(B_x) dx\right) = q! \int_{tD} \int_{tD} C^q(x-y) dx dy = q! \int_{\mathbb{R}^d} C^q(z) g_{tD}(z) dz.$$

In other words, its variance can be written only in terms of C (which depends only on the Gaussian field B) and  $g_{tD}$  (which depends only on the domain). Analogously, one could write the covariance for two domains D, L only in terms of C and  $g_{tD,tL}$ :

$$\operatorname{Cov}\left(\int_{tD} H_q(B_x) dx, \int_{tL} H_q(B_y) dy\right) = q! \int_{\mathbb{R}^d} C^q(z) g_{tD,tL}(z) dz. \tag{1.30}$$

This fact, combined with the properties of cross covariograms, will allow us to derive the exact asymptotics for the (co)variances as  $t \to \infty$ , under simple and general assumptions on C, D, L (that encompass the classical assumptions in the literature).

A less immediate fact is that also the norms of the contractions (1.28), which have to be estimated to apply Fourth Moment theorems to  $\int_{tD} H_q(B_x) dx$ , can be rewritten only in terms of C and cross covariograms, namely

$$\int_{\mathbb{R}^{3d}} C(u)^r C(v)^r C(w)^{q-r} C(u+v+w)^{q-r} g_{D_t(-w),D_t(u)}(v) du \, dv \, dw \,, \tag{1.31}$$

where  $D_t(u) := tD \cap tD + u$ . This expression, combined with Bochner's theorem, yields a reformulation of (1.28) in terms of the spectral measure G of B and Fourier transforms of cross covariograms, an idea which will allow us to prove central limit theorems under general assumptions on G.

## 1.3 Research questions and existing results

In this section, we introduce our main research questions, giving an overview of the state of the art related to them.

From now on, our study concentrates on functionals of the form

$$\int_{tD} A_x dx, \qquad t > 0, \tag{1.32}$$

where  $A = (A_x)_{x \in \mathbb{R}^d}$  is a weakly stationary random field and  $D \subseteq \mathbb{R}^d$  is compact. In this section, we focus on the existing results related to our main research questions:

**Question 1** (CLT for additive functionals of Gaussian fields). If  $A_x = \varphi(B_x)$  for some stationary Gaussian field  $B = (B_x)_{x \in \mathbb{R}^d}$ ,  $B_x \sim N(0,1)$ , and  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$ , the functionals in (1.32) are the additive functionals of B previously introduced,

$$Y_t := \int_{tD} \varphi(B_x) dx, \quad t > 0.$$

In this setting, we ask ourselves: does  $Y = (Y_t)_{t>0}$  satisfy a central limit theorem? We mainly deal with Question 1 in Chapter 2 and Chapter 5.

**Question 2** (Asymptotic covariances). Given a family of compact sets  $\mathcal{D}$  of  $\mathbb{R}^d$  and  $D, L \in \mathcal{D}$ , can we compute the asymptotic covariances of  $(\int_{tD} A_x dx)_{D \in \mathcal{D}}$ ? That is, can we compute the limit

$$\lim_{t \to \infty} \text{Cov}\left(\frac{\int_{tD} A_x dx}{r_t}, \frac{\int_{tL} A_y dy}{r_t}\right), \tag{1.33}$$

for some  $r_t \to \infty$  suitably chosen so that the previous limit is finite and not identically 0 for every  $D, L \in \mathcal{D}$ ? We mainly deal with Question 2 in Chapter 3 and Chapter 4.

Note that the two questions are very related to each other. Indeed, Question 2 with D=L corresponds to the study of the asymptotic variance of  $\int_{tD} A_x dx$ , which is usually the first step to study the fluctuations of the functionals in Question 1. Moreover, multi-dimensional central limit theorems for additive functionals of Gaussian fields can often be obtained (thanks to Theorem 1.2.9) by a simple combination of CLTs as in Question 1 and exact asymptotics of the covariances in Question 2.

We split the exposition in three parts: in Subsections 1.3.1-1.3.2 we focus on the existing results related to Question 1 and Question 2 when A has short memory and long memory, respectively; in Subsection 1.3.3 we focus on Question 1 for more general families of p-domain functionals.

#### 1.3.1 Functionals of random fields with short memory

In this section, we analyze situations where A exhibits **short memory**. Short memory for random fields is defined in many different ways in the literature, see e.g. [Mak+21] and the references therein. Heuristically, the common denominator among all these definitions is the **low global dependence** (or correlation) between the random variables composing the random field, where the "threshold" to not exceed depends on the problem under study.

In order to speak about short memory, we need to measure (or classify) the dependence or the correlation between the random variables of A. A popular way to do this are strong mixing conditions, see e.g. [IL71; Pel90]. When A is weakly

stationary, one may also classify the memory of A in terms of its covariance function  $K: \mathbb{R}^d \to \mathbb{R}$ , see the upcoming Definition 1.3.1. Of course, whatever reasonable definition of short memory we choose, the extreme case of the classical central limit theorem, in which the random variables in A are all independent, should correspond to "zero" memory. Likewise, the opposite extreme case, where all the random variables are equal, should correspond to "infinite" memory.

The concept of short memory is convenient because it allows to give an immediate interpretation of many central limit theorems' statements. To give few examples, just consider the classical central limit theorem, where the independence of the random variables is assumed (i.e. zero memory), or the upcoming Breuer-Major theorem. In all these results, one assumes that the random field satisfies some kind of short memory condition (independence in the classical CLT,  $C \in L^R$  in Theorem 1.3.1).

In our setting, where A is a weakly stationary random field, the most popular definition of short memory is the following, see e.g. [Azo09; Lav06; WC10].

**Definition 1.3.1.** Let  $A = (A_x)_{x \in \mathbb{R}^d}$  be a weakly stationary random field, with measurable covariance function  $K : \mathbb{R}^d \to \mathbb{R}$ . We say that A has **short memory** if

$$\int_{\mathbb{R}^d} |K(z)| dz < \infty. \tag{1.34}$$

**Terminology.** Short memory is sometimes referred to as **short-range dependence** (or weak dependence). These terms are often used interchangeably, see for example [Lav06; WC10]. When we are dealing with Gaussian fields, we also use them interchangeably. Since in our analysis the weakly stationary random field A is not necessarily Gaussian, we use the word "memory" (with a slight abuse of terminology when  $d \geq 2$ , since memory is intuitively associated with time) instead of "dependence", to specify that (1.34) is in general not a condition on the low global dependence of the random variables in A, but rather on their low global correlation. We avoid the term "short-range correlation" because it does not seem to be a common nomenclature in the literature.

Let us start considering Question 1 when A has short memory. If  $A_x = \varphi(B_x)$ , with  $B = (B_x)_{x \in \mathbb{R}^d}$  stationary Gaussian field,  $B_x \sim N(0,1)$ , and  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$ , the functionals in (1.32) are additive functionals of B, namely

$$Y_t := \int_{tD} \varphi(B_x) dx, \qquad t > 0. \tag{1.35}$$

Since  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$ , we have

$$\varphi = \mathbb{E}[\varphi(N)] + \sum_{q=R}^{\infty} a_q H_q, \quad a_R \neq 0,$$

and

$$\operatorname{Var}(\varphi(N)) = \sum_{q=R}^{\infty} q! a_q^2 < \infty, \qquad N \sim N(0, 1). \tag{1.36}$$

 $R := \inf\{q \ge 1 : a_q \ne 0\}$  is said the **Hermite rank** of  $\varphi$ . To avoid trivialities, we assume that  $\varphi$  is not constant, so that R is a well defined finite integer.

If K, C are the covariance functions of A and B (respectively), by (1.13) we have

$$K(z) = \sum_{q=R}^{\infty} q! a_q^2 C^q(z).$$
 (1.37)

In this setting, the most important and celebrated theorem is Breuer-Major theorem, first proved in [BM83] for discrete Gaussian fields and extended in many different directions, see e.g. [BH02; CNN20; NNP21; NPY19; NZ20].

**Theorem 1.3.1** (Breuer-Major). Let B be a stationary Gaussian field with  $B_x \sim N(0,1)$  and covariance function  $C: \mathbb{R}^d \to \mathbb{R}$ ,  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$  with Hermite rank  $R < \infty$  and consider the family  $Y = (Y_t)_{t>0}$  defined in (1.35). If  $C \in L^R(\mathbb{R}^d)$ , then

$$\frac{Y_t - \mathbb{E}[Y_t]}{t^{d/2}} \stackrel{d}{\to} N(0, \sigma^2), \quad \text{as } t \to \infty.$$

Moreover, if  $\sigma^2 := \operatorname{Vol}(D) \int_{\mathbb{R}^d} K(z) dz > 0$ , then  $\operatorname{Var}(Y_t) \sim \sigma^2 t^d$  and Y satisfies a CLT.

 $C\in L^R(\mathbb{R}^d)$  implies<sup>4</sup> that  $K\in L^1(\mathbb{R}^d)$ , i.e. that  $A_x=\varphi(B_x)$  has short memory in the sense of Definition 1.3.1. Moreover, since B is Gaussian,  $C\in L^R$  is also a condition on the low global dependence of the random variables in B, where the "low" depends on the Hermite rank R of  $\varphi$ , i.e. from the considered functional . Thus, Breuer-Major theorem provides **Gaussian fluctuations in short memory situations**.

Let us now deal with Question 2. If A has short memory, by (1.34) we have the following fact: for every  $D, L \subseteq \mathbb{R}^d$  compact sets,

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2}}, \frac{\int_{tL} A_y dy}{t^{d/2}}\right) \to \operatorname{Vol}(D \cap L) \int_{\mathbb{R}^d} K(z) dz, \quad \text{as } t \to \infty. \quad (1.38)$$

This simply follows by dominated convergence theorem and

$$\operatorname{Cov}\left(\int_{tD} A_x dx, \int_{tL} A_y dy\right) = \int_{tD} \int_{tL} K(x - y) dx dy = t^d \int_{\mathbb{R}^d} K(z) g_{D,L}(z/t) dz$$

where we recall that  $g_{D,L}$  denotes the cross covariogram of D and L.

Thus, we have two possible situations:

- If  $\int_{\mathbb{R}^d} K(z) dz = 0$ , then  $r_t = t^{d/2}$  is not a correct choice to compute the asymptotic covariances, since (1.38) is identically 0 for every D, L compact.
- If  $\int_{\mathbb{R}^d} K(z) dz > 0$ , then  $r_t = t^{d/2}$  is a correct choice for computing the asymptotic covariances. In this case, up to a scaling factor, the asymptotic covariance structure is that of a Gaussian noise (see [AT07, p.24]), i.e.  $\operatorname{Vol}(D \cap L)$ .

The above analysis provides an additional possible interpretation of the short memory of A in terms of the asymptotic covariances (when  $\int_{\mathbb{R}^d} K(z) dz > 0^5$ ): if D, L are disjoint, then  $\int_{tD} A_x dx$  and  $\int_{tL} A_y dy$  (once suitably rescaled) are asymptotically uncorrelated.

<sup>&</sup>lt;sup>4</sup>This follows by (1.36) and  $||C||_{\infty} \le 1$ . Note that the converse implication is not true in general.

<sup>&</sup>lt;sup>5</sup>See Example 4.4.1 for a situation where  $\int_{\mathbb{R}^d} K(z) dz = 0$ .

#### 1.3.2 Functionals of random fields with long memory

Following the trend in the literature, if A has not short memory, i.e.  $K \notin L^1(\mathbb{R}^d)$ , we say that A has **long memory**. When this is the case, we usually need more specific assumptions on K in order to answer Question 1 and Question 2. The most classical family of non-integrable<sup>6</sup> covariance functions considered in the literature is that of radial and **regularly varying** functions with index  $-\beta \in (-d, 0)$ ,

$$K(z) = k(|z|) := \ell(|z|)|z|^{-\beta}, \qquad \beta \in (0, d),$$
 (1.39)

where  $\ell: \mathbb{R}_+ \to \mathbb{R}$  is **slowly varying**, i.e.  $\ell$  is definitively positive and  $\ell(rs)/\ell(s) \to 1$  as  $s \to \infty$ ,  $\forall r > 0$ . An explicit example is given by Cauchy covariance functions, of the form

 $K(z) = \frac{1}{(1+|z|^2)^{\beta/2}}, \qquad \beta \in (0,d).$ 

Let us consider Question 1 in the long-memory situations (1.39). Consider again  $A_x = \varphi(B_x)$ , where K is the covariance function of A. The main theorem in this setting is the following Theorem 1.3.2. In the discrete case, a first proof for functionals with Hermite rank R = 1, 2, was given by Taqqu in [Taq75], and then generalized to any R by Dobrushin and Major in [DM79]. The following sticks to the continuous case and a proof can be found e.g. in [LO14].

**Theorem 1.3.2** (Dobrushin-Major-Taqqu). Let B be a stationary Gaussian field with  $B_x \sim N(0,1)$  and covariance function  $C: \mathbb{R}^d \to \mathbb{R}$ . Consider  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$  with Hermite rank  $R < \infty$ , Rth coefficient  $a_R \neq 0$ , see (1.16). Consider the family  $Y = (Y_t)_{t>0}$  defined in (1.35). Suppose that C satisfies the following conditions:

• C is radial and regularly varying with index  $-\beta \in (-d/R, 0)$ , i.e.

$$C(z) = \ell(z)|z|^{-\beta}, \qquad \beta \in (0, d/R).$$
 (1.40)

• The spectral measure G of C is absolutely continuous, i.e.  $G(d\lambda) = g(\lambda)d\lambda$ , and the spectral density g satisfies the following condition for some constant c > 0

$$g(\lambda) \sim c \, \ell\left(|\lambda|^{-1}\right) |\lambda|^{\beta-d}, \qquad \text{as } |\lambda| \to 0.$$
 (1.41)

Then, as  $t \to \infty$ , we have

$$\widetilde{Y}(t) \xrightarrow{d} \frac{\operatorname{sgn}(a_{R})I_{\nu,R}(f_{R,D})}{\sqrt{\operatorname{Var}(I_{\nu,R}(f_{R,D}))}},$$

where  $\nu$  is the measure on  $\mathbb{R}^d$  defined as  $\nu(dx) := |x|^{\beta - d} dx$ ,  $f_{R,D}$  is defined as

$$f_{R,D}(\lambda_1,\ldots,\lambda_R) := \int_D e^{i\langle x,\lambda_1+\cdots+\lambda_R\rangle} dx,$$

<sup>&</sup>lt;sup>6</sup>Note that if  $\beta > d$ , then  $K \in L^1(\mathbb{R}^d)$  and we fall in the previous case of short memory. If  $\beta = d$ , then  $K \notin L^1(\mathbb{R}^d)$ , but one has a CLT, and asymptotic covariances as in the short memory case. The case  $\beta = d$  may be considered a critical case between short and long memory, at least for the class of covariance functions (1.39).

and  $I_{\nu,R}$  is the Rth Wiener-Itô integral acting on  $\mathcal{H}_{\nu}^{\odot R}$ , where  $\mathcal{H}_{\nu}:=L^{2}(\nu)$  is defined as in (1.11). In particular,  $I_{\nu,R}(f_{R,D})$  is not Gaussian as soon as  $R\geq 2$ .

**Remark 1.3.1** (The extra spectral condition (1.41)). Note that the extra condition on the spectral density (1.41) is not needed in the discrete version of Theorem 1.3.2 proved by Taqqu and Dobrushin-Major. To the best of our knowledge, the continuous version of the result, i.e. Theorem 1.3.2, has never been proved without assumption (1.41). For more details on the relation between assumption (1.40) and (1.41) see [LO13]. In many situations, e.g. when C is a Cauchy covariance function, the two conditions both hold (see [LO13, Example 3]).

Combining relation (1.37) and assumption (1.40), we have that (1.39) is satisfied and A has long memory. Thus, Theorem 1.3.2 provides non-Gaussian fluctuations in long-memory situations, as soon as  $R \geq 2$ . This phenomenon is opposite to the one in the previous subsection, where Theorem 1.3.1 gives Gaussian fluctuations in short memory situations.

Let us now consider Question 2 in the long memory situations (1.39). Since K(z) = k(|z|) is radial and regularly varying with index  $-\beta \in (-d,0)$ , for  $D, L \subseteq \mathbb{R}^d$  compact we have

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^d k(t)^{1/2}}, \frac{\int_{tL} A_y dy}{t^d k(t)^{1/2}}\right) = \int_D \int_L \frac{k(|x-y|t)}{k(t)} dx dy \to \int_D \int_L |x-y|^{-\beta} dx dy,$$

where the convergence follows by dominated convergence theorem and Potter's bounds for regularly varying functions (see Theorem 4.2.5). In particular, this means that  $r_t = t^d k(t)^{1/2}$  is a correct choice for computing the asymptotic covariances (1.33), since  $\int_D \int_L |x-y|^{-\beta} dx dy \in (0,\infty)$  for every D,L compact with positive volume. When d=1 and D=[0,s], L=[0,r], then the asymptotic covariance structure is the one of a fractional Brownian motion, see Remark 4.1.2.

This provides an additional possible interpretation of the long memory of A in terms of the asymptotic covariances: if D,L are disjoint and A satisfies (1.39), then  $\int_{tD} A_x dx$  and  $\int_{tL} A_y dy$  are not necessarily asymptotically uncorrelated.

#### 1.3.3 p-domain functionals of random fields

A possible extension of Question 1 is the following: consider  $B = (B_x)_{x \in \mathbb{R}^d}$  and  $\varphi$  as in Question 1, with  $d \geq 2$ , but instead of integrating on a uniformly growing domain tD as in (1.35), let us consider the p-domain functional

$$Y(t_1, \dots, t_p) := \int_{t_1 D_1 \times \dots \times t_p D_p} \varphi(B_x) dx, \qquad t_1, \dots, t_p \to \infty$$

where  $2 \leq p \leq d$ , the compact sets  $D_i \subseteq \mathbb{R}^{d_i}$  have positive Lebesgue measure,  $d_1 + \cdots + d_p = d$ , and  $t_1, \ldots, t_p$  can grow possibly at different rates. The two main reasons to study this extended class of functionals are random field with separable covariance functions and spatio-temporal models.

#### Random fields with separable covariance function

In several applications, the covariance function C of B is assumed to be separable, that is, it can be written as  $C = C_1 \otimes \cdots \otimes C_p$ , where  $C_i$  is a non-negative definite function on  $\mathbb{R}^{d_i}$ . An example which has been studied in deep, see e.g. [PR16; RST12], are the Hermite variations of the rectangular increments of a fractional Brownian sheet, corresponding to  $\varphi = H_q$ , p = 2 and  $C_i$  of the form

$$C_i(x_i) = \frac{1}{2} \left( |x_i + 1|^{2H_i} + |x_i - 1|^{2H_i} - 2|x_i|^{2H_i} \right), \qquad H_i \in (0, 1).$$

In [RST12], the authors observed<sup>7</sup> that  $Y(t_1, t_2)$  satisfies a CLT as  $t_1, t_2 \to \infty$  if and only if  $H_i \le 1 - 1/2q$  for at least one i. In particular, the authors exploited the Malliavin-Stein method introduced by Nourdin and Peccati (see Subsection 5.2.1) to study quantitative limit theorems. This work was followed by several papers, see e.g. [Bre11; PR16]. Other examples of separable models in hydrology and fluid dynamics can be found e.g. in [Chr92, Chapter 5].

#### Spatio-temporal models

In applications, B may model a random quantity of interest depending on several (also non-Euclidean) variables  $x_1, \ldots, x_p$ , see e.g. [AO18; BPP22; Chr92; Gne02; LRM23; MRV21] A very important class of these models are spatio-temporal ones, corresponding to p=2 and  $d_2=1$ . In this case,  $Y(t_1,t_2)$  is said a spatio-temporal functional and  $t_1D_1,t_2D_2$  are respectively the space, time observation windows. An explicit example is the excursion volume of B as the space/time windows grow at different rates. Apart from the separable case previously discussed, there are also many models where C is not separable. A very popular alternative is represented for example by Gneiting covariance functions, introduced in [Gne02]. Some recent works going in this direction, i.e. studying spatio-temporal functionals for Gaussian fields with separable/Gneiting covariance functions, are [LRM23; LRM24].

#### 1.4 New results of the thesis

In this last section, we focus on the main contributions of this thesis, presenting in a concise way the main results of the next four chapters and comparing them with the ones considered in the previous section. Each of the next four subsections is respectively based on Chapters 2, 3, 4, 5, which are themselves reproductions of the following papers/preprints:

- Chapter 2 contains the paper [MN24], "Spectral central limit theorem for additive functionals of isotropic and stationary Gaussian fields", written in collaboration with Ivan Nourdin. *Ann. Probab.* (2024), 52(2), 737 – 763.
- Chapter 3 contains the paper [GMT24], "Fluctuations of polyspectra in spherical and Euclidean random wave models", written in collaboration with Francesco Grotto and Anna Paola Todino. *Electron. Comm. Probab.* (2024), 29, 1-12.

<sup>&</sup>lt;sup>7</sup>This was done in the discrete case in [RST12], but can be translated in the continuous case.

- Chapter 4 contains the paper [Mai24], "Asymptotic covariances for functionals of weakly stationary random fields". *Stoch. Proc. Appl.* (2024), 170, 104297.
- Chapter 5 contains the preprint [Leo+24], "Limit theorems for *p*-domain functionals of stationary Gaussian fields", written in collaboration with Nikolai Leonenko, Ivan Nourdin and Francesca Pistolato. arXiv (2024+): 2402.16701.

For all the missing details in the next subsections, we refer to Chapters 2-5.

# 1.4.1 Chapter 2: Spectral central limit theorem for additive functionals of isotropic and stationary Gaussian fields

In Chapter 2 we deal with Question 1, that is, we study under which conditions the family Y of additive functionals defined in (1.35) satisfies a CLT.

The **binary intuition** forged starting from the seminal works of Breuer, Dobrushin, Major, Taqqu (see Theorem 1.3.1 and Theorem 1.3.2), is that Y satisfies, in general (i.e. except possibly in "special" cases), a **central limit theorem when** B **has short memory**, and a **non-central limit theorem when** B **has long memory and**  $R \geq 2$ .

With "special cases" we mean for example the critical case  $\beta=d/R$  in (1.40), for which we have long memory in the sense of Definition 1.3.1, but usually Gaussian fluctuations (see, e.g., [BM83, Theorem 1'], [NN20, Section 5] and [NNT10, Theorem 1]). Another Gaussian field that may be considered "special" is the Berry random wave model, for which Theorem 1.4.1 provides Gaussian fluctuations in a long memory context (when R=2). Berry's random wave model may be considered special in the sense that it represents Euclidean random waves. Indeed, Gaussian fluctuations in a long memory context were already observed for random waves on the sphere, see e.g. [MW14; Tod19].

The first novelty of Chapter 2 consists in showing that this surprising phenomenon (Gaussian fluctuations in a long memory context with  $R \geq 2$ ) can occur in a large variety of situations. More specifically, we prove in Theorem 1.4.1 that Y satisfies a CLT under an easily verifiable condition on the negative moment of the spectral measure of B, without assumptions on the covariance function. Thanks to it, we construct a family of situations showing that the binary intuition related to memory and limit theorems for Y is wrong in general and not only in special cases. To highlight the fact that the main assumption in Theorem 1.4.1 involves only the spectral measure, we decided to refer to it as a *spectral* CLT.

Note that this also brings to light a second novelty introduced in Chapter 2: we are able to prove central limit theorems for Y even when the covariance function is not available, or has not a convenient mathematical expression.

Let us state the main result of the chapter in a concise and simplified form.

**Theorem 1.4.1** (Spectral CLT). Fix  $d \geq 2$ , let  $B = (B_x)_{x \in \mathbb{R}^d}$  be stationary, isotropic, with covariance function  $C : \mathbb{R}^d \to \mathbb{R}$  and isotropic spectral measure  $\mu$ . Let  $\varphi$  be as in (1.16), with Hermite rank R and second Hermite rank R' (as defined in (2.4)). Suppose that  $\exists k > 0$  with  $a_{2k} \neq 0$  and that  $(R, R') \notin \{(1, 3)\} \cup \{(2k + 1, n) : k \geq 1, n \in \mathbb{N}\}$ .

<sup>&</sup>lt;sup>8</sup>For a class of examples, see Section 2.6.

Finally, when  $R \in \{1, 2\}$ , suppose that D is "nice" (see Theorem 2.1.2). If the following spectral condition holds

$$\int_0^\infty \frac{\mu(ds)}{s^{d/R}} < \infty, \tag{1.42}$$

then Y satisfies a CLT, with the following variance rate

$$\operatorname{Var}(Y_t) \asymp \left\{ \begin{array}{ll} t^d \int_{\{|x| \leq t\}} C^R(x) dx & \text{ if } R \text{ even} \\ t^d \int_{\{|x| \leq t\}} C^{R'}(x) dx & \text{ if } R = 1 \text{ and } R' \in \{2, 4\} \\ t^d & \text{ if } R = 1 \text{ and } R' \geq 5 \end{array} \right..$$

If we consider the 2-dimensional Berry random wave model  $B = (B_x)_{x \in \mathbb{R}^2}$  and R = 2, then the covariance function is neither in  $L^2(\mathbb{R}^d)$  nor regularly varying, but instead has the following oscillating asymptotic behavior (see [Kra14])

$$C(z) = J_0(|z|) = \sqrt{\frac{2}{\pi}} \frac{\cos(|z| - \pi/4)}{\sqrt{|z|}} \left( 1 + O\left(|z|^{-3/2}\right) \right) \qquad |z| \to \infty.$$
 (1.43)

This makes it hard to deal with it in computations, see e.g. (1.31). On the other hand, the isotropic spectral measure of Berry's random wave model is the easiest we can think about: the degenerate measure  $\mu = \delta_1$ . Thus, by applying Theorem 1.4.1 for  $R \in \{1, 2, 4\}$  and Theorem 1.3.1 for  $R \geq 5$ , taking  $\varphi$  and D as in the statement of Theorem 1.4.1, Y satisfies a CLT with variance rates

$$\operatorname{Var}(Y_t) \asymp \begin{cases} t^3 & \text{if } R = 2 \text{ or } (R = 1 \text{ and } R' = 2) \\ t^2 \log(t) & \text{if } R = 4 \text{ or } (R = 1 \text{ and } R' = 4) \\ t^2 & \text{if } R \geq 5 \text{ or } (R = 1 \text{ and } R' \geq 5) \end{cases}$$
 (1.44)

In view of the binary intuition discussed above, a CLT is expected when  $R \geq 4$ , but surprising in the long memory case R=2, and with surprising variance rate in the cases with R=1. This does not happen only for Berry's random wave model because it is "special", but whenever  $C \notin L^R(\mathbb{R}^d)$  and (1.42) holds.

To prove Theorem 1.4.1, we proceed dividing the analysis in two cases:  $R \geq 4$  and  $R \leq 2$  (note that  $R \neq 3$  by assumption). Since if  $C \in L^R$  we can use Breuer-Major Theorem 1.3.1, we assume that  $C \notin L^R(\mathbb{R}^d)$ . Then we can use reduction theorems (see in particular Proposition 2.2.2) to reduce the study from a general  $\varphi$  to  $\varphi = H_R$  (or  $\varphi = H_{R'}$ ,  $R' \geq 2$  if R = 1). At this point, recalling Theorem 1.2.8, (1.28) and (1.31), we need to show that for R even the following converges to R0 for every R1, ..., R1,

$$\frac{\int_{\mathbb{R}^{3d}} C(u)^r C(v)^r C(w)^{R-r} C(u+v+w)^{R-r} \, g_{D_t(-w),D_t(u)}(v) du \, dv \, dw}{(\int_{\mathbb{R}^d} C^R(z) g_{tD}(z) dz)^2} \, .$$

To do this, we study separately the two following cases:

- If  $R \geq 4$  even, then we can show (by using Schoenberg's theorem) that  $C(z)|z|^{d/R}$  is bounded. Thus, we can use techniques exploited e.g. in [NPR19, Lemma 8.1] to show that the previous converges to 0.
- If R=2, then  $C(z)|z|^{d/2}$  is not necessarily bounded, and we exploit Bochner's

and Schoenberg's spectral representations, together with Fourier analysis, to find a bound in terms of spectral measures and the Fourier transforms of cross covariograms, showing that (1.31) converges to 0. To do this, the starting idea is rewriting the "annoying factor" C(u+v+w) using Bochner's theorem,

$$\frac{\int_{\mathbb{R}^d} G(dx) \int_{\mathbb{R}^{2d}} du \, dv \, C(u) \, C(v) e^{i\langle x, u+v\rangle} \int_{\mathbb{R}^d} dw \, C(w) e^{i\langle x, w\rangle} g_{D_t(-w), D_t(u)}(v)}{(\int_{\mathbb{R}^d} C^R(z) g_{tD}(z) dz)^2} \,,$$

recalling that  $g_{D_t(-w),D_t(u)} = \mathbf{1}_{D_t(-w)} * \mathbf{1}_{-D_t(u)}$  and finally using the convolution theorem and other Fourier analysis results to simplify the expression.

• If R=1, we reduce the study from  $\varphi$  to  $\varphi-a_RH_R$  with Hermite rank  $R'\geq 2$ , and use the arguments above.

As we anticipated in section 1.2.1, one may probably extend Theorem 1.4.1 to non-isotropic Gaussian fields with "regular" spherical spectral measure, or maybe to non-stationary harmonizable Gaussian fields, probably under analogous (but different) spectral conditions.

We conclude with an important remark, that also introduces the next subsection.

**Remark 1.4.1.** The reason why we need to exclude some cases in Theorem 1.4.1, assuming for instance that R is not odd, or that the even chaoses are not all null, is the tricky nature of the odd chaoses, due to the possible oscillating behavior of the covariance function (see e.g. the asymptotics (1.43) of the Berry field). Indeed, by (1.27) the variance of additive functionals of Gaussian fields is

$$\operatorname{Var}\left(\int_{tD} \varphi(B_x) dx\right) = \sum_{q=1}^{\infty} a_q^2 \operatorname{Var}\left(\int_D H_q(B_x) dx\right).$$

Thus, since

$$\operatorname{Var}\left(\int_{D} H_{q}(B_{x})dx\right) = q! \int_{\mathbb{R}^{d}} C^{q}(z)g_{tD}(z)dz,$$

one can easily intuit why the variance rate is easier to derive when  $C^q \ge 0$  and harder otherwise, at least if we want to work for general C. Indeed, even in the apparently simple case  $C^q \in L^1(\mathbb{R}^d)$ , reasoning as in (1.38) we have

$$\operatorname{Var}\left(\int_{D} H_{q}(B_{x})dx\right) \sim q!t^{d}\operatorname{Vol}(D)\int_{\mathbb{R}^{d}} C^{q}(z)dz,$$

which is useless without knowing that  $\int_{\mathbb{R}^d} C^q(z)dz \neq 0$ . If B is the Berry random wave model, for example, then the positivity of  $\int_{\mathbb{R}^d} C^q(z)dz$  for every  $q \geq 5$  odd was left as a conjecture in [MW14].

### 1.4.2 Chapter 3: Fluctuations of Polyspectra in Spherical and Euclidean Random Wave Models

In Chapter 3, we focus on Question 2 when  $d \ge 2$ , D = L is a ball and  $A_x = H_q(B_x)$ , where B is the Berry random wave model. As we just discussed in Remark 1.4.1, this problem was still open for general q odd. Moreover, we also deal with the "spherical

version" of this question. More precisely, we study

$$V_{d,R}^{E}(q,\lambda) = \operatorname{Var}\left(\int_{B^{E}(R)} H_{q}(U_{\lambda}(x))dx\right) = \operatorname{Var}\left(\int_{B^{E}(R)} H_{q}(B_{\lambda x})dx\right), \quad \lambda \to \infty$$

and

$$V_{d,R}^S(q,\ell) = \operatorname{Var}\left(\int_{B^S(R)} H_q(T_\ell(x))dx\right), \quad \ell \to \infty,$$

where  $B^E(R), B^S(R)$  are Euclidean/spherical balls of radius R>0 and  $(U_\lambda(x))_{x\in\mathbb{R}^d}$ ,  $(T_\ell(x))_{x\in S^d}$  are Euclidean/spherical random wave models, defined as follows:

- The Euclidean random wave model  $U_{\lambda}(x)$ ,  $\lambda > 0$ ,  $x \in \mathbb{R}^d$ , is  $U_{\lambda}(x) := B_{\lambda x}$ , where B is the Berry random wave model (see Definition 1.1.9).
- The (hyper)spherical random wave model  $T_{\ell}(x)$ ,  $\ell \in \mathbb{N}$ ,  $x \in S^d$ , is the only (in distribution) Gaussian field on  $S^d$  with covariance function

$$\mathbb{E}T_{\ell}(x)T_{\ell}(y) = \mathcal{G}_{d,\ell}(\cos d(x,y)) \quad x, y \in S^d, \tag{1.45}$$

where  $\mathcal{G}_{d,\ell}$  is the Gegenbauer polynomial of degree  $\ell$  (see [Sze39, p. 4.7]).

 $U_{\lambda}$  and  $T_{\ell}$  can be also defined in terms of random linear combinations of Laplace eigenfunctions, similarly to how we defined monochromatic random waves in Example 1.1.3 (see Chapter 3 for more details).

The main result of the chapter is the following theorem, that provides a complete picture of the asymptotic variances of the above polyspectra for  $d \ge 2$  and  $q \ge 2$  (the case q = 1 is apart, because the variances defined above are 0 infinitely many times).

**Theorem 1.4.2.** Let  $d, q \ge 2$ , R > 0. There exist finite **positive** constants  $c_q^d$  such that:

• (Euclidean) as  $\lambda \to \infty$ ,

$$V_{d,R}^E(q,\lambda) = c_q^d \operatorname{Vol}(B^E(R))(1+o_{q,d}(1)) \cdot \begin{cases} \lambda^{1-d} & q=2, \\ \lambda^{-2} \log(\lambda) & q=4, d=2, \\ \lambda^{-d} & \textit{all other } d \geq 2, q \geq 3, \end{cases}$$

• (Hyperspherical) as  $\ell \to \infty$ , if  $R \in (0, \pi)$ ,

$$V_{d,R}^{S}(q,\ell) = c_q^d \sigma_d(B^S(R))(1 + o_{q,d}(1)) \cdot \begin{cases} \ell^{1-d} & q = 2, \\ \ell^{-2} \log(\ell) & q = 4, d = 2, \\ \ell^{-d} & \textit{all other } d \geq 2, q \geq 3, \end{cases}$$

where  $\sigma_d$  is the volume measure on  $S^d$ , and when  $R = \pi$  (that is in the case of polyspectra obtained integrating over the whole  $S^d$ ),

$$V_{d,\pi}^S(q,\ell) = 2c_q^d q! \omega_{d-1} \omega_d (1+o_{q,d}(1)) \cdot \begin{cases} \ell^{1-d} & q=2, \\ \ell^{-2} \log(\ell) & q=4, d=2, \\ 0 & q, \ell \text{ both odd,} \\ \ell^{-d} & \text{all other } d \geq 2, q \geq 3. \end{cases}$$

Some of the previous asymptotic variances were already known in the literature, see Remark 3.1.3. Before Theorem 1.4.2, the missing cases corresponded to  $q \geq 5$  odd. In these situations, we have

$$c_q^d = c_d \int_{\mathbb{R}^d} C^q(z) dz \,,$$

where C is the covariance function of the Berry random wave model and  $c_d$  is a positive constant depending only on d. Our main contribution consists in showing that  $\int_{\mathbb{R}^d} C^q(z) dz$  is positive for every  $q \geq 3$  odd. To prove this, we exploit a relation between the latter and Pearson's random walks. The Pearson's random walk  $X_q^d$  is a random variable defined as the sum of q independent random variables uniformly distributed on  $S^{d-1}$ . Thus, if C is the covariance function of the d-dimensional Berry random wave model, i.e. the characteristic function of  $X_1^d$ , we have that  $C^q$  is the characteristic function of  $X_q^d$ . As a consequence, Kluyver's formula (see Lemma 3.2.1) implies

$$\int_{\mathbb{R}^d} C^q(z)dz = f_q^d(0),$$

where  $f_q^d$  is the density of  $X_q^d$ , i.e. the spectral density associated to  $C^q$ , which exists for  $q \geq 2$ . Thus, the proof is reduced to showing that  $f_q^d(0)$  is positive, which is done relying on technical results in the context of Pearson's random walks.

We conclude with the following remark, that also introduces the next subsection.

Remark 1.4.2. Recall Question 2. Theorem 1.4.2 (Euclidean statement) allows to compute the asymptotic variance of  $\int_{tD} A_x \, dx$  if  $A_x = H_q(B_x)$ , B is the Berry random wave model and D=L is a ball. For  $q\geq 5$ , we have  $C^q\in L^1$ , i.e. A has short memory, and we can obtain the exact rate for  $V_{d,R}^E(q,t)$  combining (1.38) and the positivity of  $\int_{\mathbb{R}^d} C^q(z) dz$  mentioned above. For  $q\in\{2,3,4\}$ ,  $C^q\notin L^1(\mathbb{R}^d)$  and  $C^q$  is not regularly varying. This means that A has long memory, but we can not compute the asymptotic variance as we did in Section 1.3.2. Despite this, we are able to compute the asymptotic variance of  $\int_{tD} H_q(B_x) dx$  with ad-hoc techniques. Since these techniques only partially rely on the specific covariance structure of the Berry random wave model, a natural question arises: can we use analogous techniques to compute the asymptotic (co)variances of other random fields? This question is thoroughly investigated in Chapter 4.

# 1.4.3 Chapter 4: Asymptotic covariances for functionals of weakly stationary random fields

In Chapter 4 we answer to Question 2. More precisely, given a measurable, weakly stationary random field A on  $\mathbb{R}^d$  with covariance function  $K: \mathbb{R}^d \to \mathbb{R}$ , our goal is giving the minimal assumptions on K and  $\mathcal{D}$  (a class of compact sets of  $\mathbb{R}^d$ ) to compute the asymptotic covariances

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} A_x \, dx}{r_t}, \frac{\int_{tL} A_y \, dy}{r_t}\right), \qquad D, L \in \mathcal{D}, \tag{1.46}$$

where  $r_t \to \infty$  is chosen so that the previous limit is finite and not identically 0 for every  $D, L \in \mathcal{D}$ .

To do this, let us consider the **integral covariance function**  $w : \mathbb{R}_+ \to \mathbb{R}$ ,

$$w_t := \int_{\{|z| \le t\}} K(z) dz \,,$$

and note the following facts:

• If  $K \in L^1(\mathbb{R}^d)$  and  $\int_{\mathbb{R}^d} K(z) dz \neq 0$ , then we saw in Subsection 1.3.1 that  $r_t = t^{d/2}$  is a correct choice for computing the asymptotic covariances. Moreover, also  $r_t = t^{d/2} w_t^{1/2}$  is a correct choice and we have  $\forall D, L$  compact

$$\lim_{t \to \infty} \text{Cov}\left(\frac{\int_{tD} A_x \, dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_y \, dy}{t^{d/2} w_t^{1/2}}\right) = \text{Vol}\left(D \cap L\right) \,, \qquad D, L \in \mathcal{D}. \tag{1.47}$$

• If K(z)=k(|z|) is radial and regularly varying with index  $-\beta\in(-d,0)$ , see (1.39), then we saw in Subsection 1.3.2 that  $r_t=t^dk(t)^{1/2}$  is a correct choice for computing the asymptotic covariances. Moreover, by Proposition 4.2.4  $r_t=t^{d/2}w_t^{1/2}$  is also a correct choice, and we have  $\forall\,D,L$  compact

$$\lim_{t \to \infty} \text{Cov}\left(\frac{\int_{tD} A_x \, dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_y \, dy}{t^{d/2} w_t^{1/2}}\right) = \frac{(d-\beta)}{\omega_{d-1}} \int_D \int_L |x-y|^{-\beta} dx dy. \tag{1.48}$$

Thus, in both the classical situations analyzed in Subsections 1.3.1-1.3.2, we have that  $w_t$  is regularly varying (with index 0 in the first case and  $(d-\beta) \in (0,d)$  in the second case) and  $r_t = t^{d/2} w_t^{1/2}$  is a correct choice for computing the asymptotic covariances. For this reason, we may conjecture that if  $w_t$  is regularly varying, then we can compute (1.46) with rate  $r_t = t^{d/2} w_t^{1/2}$ . We prove that this is the case in the two following situations:

Case 1. K is radial and  $\mathcal{D}$  is the class of all compact sets in  $\mathbb{R}^d$  with finite perimeter (recall the definition of perimeter given in (1.29)), i.e. for some  $k : \mathbb{R}_+ \to \mathbb{R}$  we have

$$K(z) = k(|z|), \quad \mathcal{D} = \{D \subseteq \mathbb{R}^d : D \text{ compact}, Per(D) < \infty\}.$$
 (1.49)

Case 2.  $\mathcal{D}$  is the class of closed balls centered at a fixed point  $x_0 \in \mathbb{R}^d$ , namely

$$\mathcal{D} = \left\{ \left\{ x \in \mathbb{R}^d : |x - x_0| \le r \right\} : r \in \mathbb{R}_+ \right\}. \tag{1.50}$$

The main result of Chapter 4 is the following.

**Theorem 1.4.3.** Let  $\mathcal{D}$  be a collection of compact sets in  $\mathbb{R}^d$  and  $A = (A_x)_{x \in \mathbb{R}^d}$  be a measurable, weakly stationary random field with covariance function  $K : \mathbb{R}^d \to \mathbb{R}$ . Furthermore, let K and  $\mathcal{D}$  satisfy the assumption (1.49) or the assumption (1.50), and assume that w is **regularly varying** with index  $\alpha \in (-1, d]$ , that is

$$w_t = \ell(t)t^{\alpha} \quad \forall t \in \mathbb{R}_+ \tag{1.51}$$

where  $\ell: \mathbb{R}_+ \to \mathbb{R}$  is slowly varying. Then, for all  $D, L \in \mathcal{D}$ , we have as  $t \to \infty$ 

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_x dx}{t^{d/2} w_t^{1/2}}\right) \to \frac{1}{\omega_{d-1}} \int_{S^{d-1}} d\theta \int_0^\infty dl \left(-\frac{d}{dl} g_{D,L}(l\theta)\right) l^\alpha. \tag{1.52}$$

Theorem 1.4.3 allows to compute the asymptotic covariances (1.46) under assumptions on K that encompass the classical ones discussed in Subsections 1.3.1-1.3.2. Note that (1.52) is a new expression for the asymptotic covariances that generalizes (1.47)-(1.48), see Remark 4.1.2 for the details.

Moreover, combining Theorem 1.4.1, Theorem 1.4.3 and the Peccati-Tudor Fourth Moment Theorem 1.2.9, we can prove new multi-dimensional central limit theorems for additive functionals of Gaussian fields, assuming only (1.51). If for instance B is Berry's random wave model,  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$  with  $a_2 \neq 0$  in (1.16) and  $A_x = \varphi(B_x)$ , then  $w_t$  is regularly varying with index  $\alpha = 1$  and we have (see Example 4.4.5 for the details)

$$\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_t^{1/2}}\right)_{D \in \mathcal{D}_O} \stackrel{f.d.d.}{\to} G = (G(D))_{D \in \mathcal{D}_O}$$

where  $\overset{f.d.d.}{\to}$  denotes the convergence in finite dimensional distributions,  $\mathcal{D}_O$  is a suitable class of compact sets with finite perimeter and G is a centered Gaussian field on  $\mathcal{D}_O$  with covariance kernel

$$\mathbb{E}\left[G(D)G(L)\right] = \frac{1}{\omega_{d-1}} \int_{D} \int_{L} |x - y|^{1-d} dx dy = \frac{1}{\omega_{d-1}} \int_{\mathbb{R}^d} g_{D,L}(z) |z|^{1-d} dz.$$

For other examples and applications of Theorem 1.4.3, see Sections 4.4-4.5.

# 1.4.4 Chapter 5: Limit theorems for p-domain functionals of stationary Gaussian fields

In the last Chapter 5, we deal with the p-domain generalization of Question 1 introduced in Subsection 1.3.3, that is, we study the fluctuations of

$$Y(t_1, \dots, t_p) := \int_{t_1 D_1 \times \dots \times t_p D_p} \varphi(B_x) dx, \qquad t_1, \dots, t_p \to \infty$$
 (1.53)

where  $2 \leq p \leq d$ , the compact sets  $D_i \subseteq \mathbb{R}^{d_i}$  have positive Lebesgue measure,  $d_1 + \cdots + d_p = d$ , and  $t_1, \ldots, t_p$  can grow possibly at different rates.

We start assuming that the covariance function  $C: \mathbb{R}^d \to \mathbb{R}$  is **separable**, that is

$$C(x_1, ..., x_p) = \prod_{i=1}^{p} C_i(x_i), \quad x_i \in \mathbb{R}^{d_i},$$

where  $C_i : \mathbb{R}^{d_i} \to \mathbb{R}$  are non-negative definite functions on  $\mathbb{R}^{d_i}$ , with  $C_i(0) = 1$ . In this setting, we thoroughly investigate under what conditions the study of the fluctuations of (1.53) can be reduced to that of its marginal functionals  $Y_i(t_i)$ ,

$$Y_i(t_i) := \int_{t_i D_i} \varphi(B_{x_i}^{(i)}) dx_i,$$

where  $B^{(i)}$  is a stationary centered Gaussian field on  $\mathbb{R}^{d_i}$  with covariance function  $C_i$ . The main theorem of Chapter 5 is the following.

**Theorem 1.4.4.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Consider  $\varphi \in L^2(\mathbb{R}, \gamma(dx))$  with Hermite rank  $R \ge 1$ . Let  $C : \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B, assume it is separable, and also that it satisfies, for each i:

$$C^R \geq 0$$
 and  $C_i \in \bigcup_{M=1}^{\infty} L^M(\mathbb{R}^{d_i}).$ 

Let us consider  $\widetilde{Y}$  given by (5.3) and  $\widetilde{Y}_i$  given by (5.6), the rescaled versions of  $Y, Y_i$ . Then, the following two assertions are equivalent:

(a) 
$$\widetilde{Y}_i(t_i) \stackrel{d}{\to} N(0,1)$$
 as  $t_i \to \infty$  for at least one  $i \in \{1,\ldots,p\}$ ;

(b) 
$$\widetilde{Y}(t_1,\ldots,t_p) \stackrel{d}{\to} N(0,1)$$
 as  $t_1,\ldots,t_p \to \infty$ .

The implication  $(a) \implies (b)$  in Theorem 1.4.4 ensures that the p-domain functional (1.53) has Gaussian fluctuations if one of the 1-domain marginal functionals does. This criterion is quite useful because, contrarily to p-domain ones, 1-domain functionals have been extensively studied in the literature, see e.g. Subsections 1.3.1-1.3.2 and the references therein. It is noteworthy that a specific instance of this implication had previously been observed in the papers [RST12] and [PR16]; however, it was restricted to a very specific context – rectangular increments of a fractional Brownian sheet – and was not part of a comprehensive systematic investigation, as we undertake in Chapter 5.

The theorem is proved relying on a p-domain version of Breuer-Major Theorem 1.3.1 when  $C \in L^R$  and reducing the problem to  $\varphi = H_R$  if  $C \notin L^R$ .

Indeed, in the particular case  $\varphi=H_q$ , the equivalence in Theorem 1.4.4 is true without integrability assumptions on C. Moreover, exploiting Theorem 1.2.10 and some inequalities involving the fourth moment in a fixed Wiener chaos, if  $\varphi=H_q$  we prove that (see Proposition 5.3.3)

$$d_{TV}(\tilde{Y}(t_1, \dots, t_p), N) \le c_q \prod_{i=1}^p \sqrt{\mathbb{E}[\tilde{Y}_i(t_i)^4] - 3}$$
 (1.54)

This inequality also allows to improve some bounds for the Hermite variations of the rectangular increments of the fractional Brownian sheet obtained in [RST12], see Example 5.4.1 for the details.

Note that Theorem 1.4.4 also brings to light another large class of Gaussian fields for which the intuition "long memory and  $R \geq 2$  imply Gaussian fluctuations" is wrong (recall the discussion at the beginning of Subsection 1.4.1). Indeed, if we take for instance R=2,  $C_1 \in L^R(\mathbb{R}^d)$  and  $C_i$  of the form (1.39) if  $i \neq 1$ , then combining Theorem 1.3.1 on  $Y_1$  and Theorem 1.4.4 we have Gaussian fluctuations even if we are in a long memory setting.

Conversely, when  $C_i$  is of the form (1.39) for every i, we are able to prove a generalization of Theorem 1.3.2 (see Theorem 5.1.2) for p-domain functionals, ensuring non-Gaussian fluctuations as soon as  $R \ge 2$ .

It is worth noting that all our results obtained assuming that C is separable may be adapted to situations where some of the domains are fixed, see Remark 5.1.4.

When C is **not separable**, it is easy to construct examples illustrating that the normal convergence of the functionals  $\widetilde{Y}_i(t_i)$  is in general not enough to determine the behavior of  $\widetilde{Y}(t_1,\ldots,t_p)$  (see Example 5.5.1).

For this reason, we go beyond separability, studying two non-separable classes of covariance functions (in the specific case  $\varphi = H_q$ ):

#### • Gneiting covariance functions, of the form

$$C(x_1, x_2) = C_2(x_2)C_1\left(x_1C_2(x_2)^{2/d_1}\right), \qquad x_1 \in \mathbb{R}^{d_1}, x_2 \in \mathbb{R}^{d_2},$$

where  $C_1$ ,  $C_2$  have "nice" properties. In this case, C can be wedged between two separable covariance functions. This fact allows to prove Theorem 5.5.1, that brings to light a comparable phenomenon to the one observed in the separable case.

#### Additively separable covariance functions, of the form

$$C(x_1, x_2) = K_1(x_1) + K_2(x_2), \qquad x_1 \in \mathbb{R}^{d_1}, x_2 \in \mathbb{R}^{d_2},$$

where  $K_1$ ,  $K_2$  are positive covariance functions on  $\mathbb{R}^{d_1}$ ,  $\mathbb{R}^{d_2}$ . In this case, we are still able to prove a kind of "reduction" theorem (see Theorem 5.5.2), but with marginal functionals completely different from the  $Y_i$ 's of Theorem 1.4.4.

In particular, a remarkable fact observed in the additively separable case (see Example 5.5.2) is that  $Y(t_1, t_2)$  has (or has not) Gaussian fluctuations depending on the growth rates  $t_1, t_2$ . This is very different from what we observe in the separable case, where the interplay between the growth rates of the domains plays no role.

# CHAPTER 2

# Spectral central limit theorem for additive functionals of isotropic and stationary Gaussian fields

This chapter contains the paper [MN24], "Spectral central limit theorem for additive functionals of isotropic and stationary Gaussian fields", written in collaboration with Ivan Nourdin. *Ann. Probab.* (2024), 52(2), 737 – 763.

#### 2.1 Introduction

Fix a dimension  $d \geq 2$ , and consider a real-valued almost surely continuous Gaussian field  $(B_x)_{x \in \mathbb{R}^d}$  defined on  $\mathbb{R}^d$ . Assume furthermore that B is **stationary**, that is, there is a function  $C : \mathbb{R}^d \to \mathbb{R}$  such that

$$Cov(B_x, B_y) = C(x - y), \quad x, y \in \mathbb{R}^d,$$
(2.1)

and suppose that  $B_x \sim N(0,1)$  for all  $x \in \mathbb{R}^d$  or, equivalently, that  $\mathbb{E}[B_x] = 0$  and C(0) = 1.

As a second ingredient, consider a measurable function  $\varphi : \mathbb{R} \to \mathbb{R}$  such that

$$\mathbb{E}\left[\varphi(N)^2\right] < \infty, \quad \text{for } N \sim N(0, 1).$$
 (2.2)

Our object of interest in this paper is

$$Y_t = \int_{tD} \varphi(B_x) dx, \quad t > 0, \tag{2.3}$$

where  $D \subset \mathbb{R}^d$  is compact with Vol(D) > 0 and  $tD := \{tx | x \in D\}$ . Well-posedness of (2.3) as a random variable in  $L^2(\Omega)$  is ensured by Proposition 2.2.1 and the almost sure continuity of  $(B_x)_{x\in\mathbb{R}^d}$ . We also observe that the continuity of B, together with its stationarity, implies the continuity of its covariance function C, a property that will be needed to evoke Bochner's theorem later in (2.14).

Studying the asymptotic behavior of functionals of the form (2.3) dates back from the eighties, with seminal works by Breuer and Major [BM83], Dobrushin and Major [DM79], Rosenblatt [Ros60] and Taqqu [Taq79]. Since then, limit theorems for (2.3) have been constantly investigated, and represent nowadays a central theme in the modern probability theory.

We note that many interesting geometric quantities associated with the Gaussian field  $(B_x)_{x\in\mathbb{R}^d}$  can be represented as functionals of the form (2.3). For instance, the choice  $\varphi = \mathbf{1}_{(-\infty,u]}$  (resp.  $\varphi = \mathbf{1}_{[u,\infty)}$ ),  $u \in \mathbb{R}$  corresponds to the volume of the lower (resp. upper) level sets of  $(B_x)_{x \in \mathbb{R}^d}$ .

Since (2.2) holds, we can decompose  $\varphi$  in Hermite polynomials (see, e.g., [NP12, Section 1.4]) as

$$\varphi = \mathbb{E}[\varphi(N)] + \sum_{q=R}^{\infty} a_q H_q, \quad \text{with } R \ge 1 \text{ such that } a_R \ne 0,$$
 (2.4)

where  $H_q$  denotes the qth Hermite polynomial and  $a_q = a_q(\varphi) = \frac{1}{d} \mathbb{E} \left[ \varphi(N) H_q(N) \right] \in$  $\mathbb{R}$ . The integer  $R \geq 1$  is called the **Hermite rank** of  $\varphi$ . We also define the **second Hermite rank**  $R' \geq 2$  of  $\varphi$  as the Hermite rank of  $\varphi(x) - \mathbb{E}[\varphi(N)] - a_R H_R(x)$  (if  $\varphi(x) = \mathbb{E}[\varphi(N)] + a_R H_R(x)$ , we set  $R' = \infty$ ).

In the present paper, we are more specifically interested in the asymptotic behavior of

$$\frac{Y_t - m_t}{\sigma_t}, \quad t \to \infty,$$
 (2.5)

where we have  $Y_t \in L^2(\Omega)$  for all t, and where we note  $m_t = \mathbb{E}[Y_t]$  and  $\sigma_t =$  $\sqrt{\operatorname{Var}(Y_t)} > 0$ . For simplicity, to ensure that  $\sigma_t > 0$  for all t, we will assume for the rest of the paper the existence of some  $k \ge 1$  such that  $a_{2k} \ne 0^2$ , or equivalently that  $\varphi$  is **not an odd function**. As an illustration of what may happen when  $\varphi$  is odd, see (2.22).

Throughout all the paper, for two functions  $f, g : \mathbb{R}_+ \to \mathbb{R}_+$  we write

$$f(t) \approx q(t) \tag{2.6}$$

to indicate that f(t) = O(g(t)) and g(t) = O(f(t)) as  $t \to \infty$ .

Indeed, since  $|B_0(B_{x+h}-B_x)| \leq B_0^2 + \frac{1}{2}B_x^2 + \frac{1}{2}B_{x+h}^2$ , by applying the generalized dominated convergence theorem we have  $C(x+h)-C(x)=\mathbb{E}\left[B_0(B_{x+h}-B_x)\right]\to 0$  as  $h\to 0$ . This comes from the fact that  $\sigma_t^2=\operatorname{Var}(Y_t)=\sum_{q=R}^\infty q!a_q^2\int_{(tD)^2}C^q(x-y)dxdy$ .

2.1. Introduction 39

#### 2.1.1 Previous results

Given its importance in our paper, we start with the celebrated Breuer-Major theorem, stated here in its continuous form.

**Theorem 2.1.1** (Breuer, Major [BM83]). Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , assumed to be stationary and to have unit-variance. Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be such that  $\mathbb{E}[\varphi(N)^2] < \infty$  with  $N \sim N(0,1)$ , let R be the Hermite rank of  $\varphi$ , consider  $Y_t$  defined by (2.3), and recall the definition (2.1) of the covariance function C.

If 
$$\int_{\mathbb{R}^d} |C(x)|^R dx < \infty$$
 , then  $t^{-\frac{d}{2}}(Y_t - \mathbb{E}[Y_t]) \stackrel{\mathrm{law}}{\to} N(0, \sigma^2)$  where

$$\sigma^2 = \operatorname{Vol}(D) \sum_{q=R}^{\infty} q! a_q^2 \int_{\mathbb{R}^d} C(z)^q dz \ge 0.$$
 (2.7)

In particular, if  $\varphi$  is not odd, then  $\sigma^2>0$ ,  $\sigma^2_t symp t^d$  and

$$\frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0, 1).$$

To describe the asymptotic behavior in the case where  $C \notin L^R(\mathbb{R}^d)$  (that is, when we cannot apply the Breuer-Major Theorem 2.1.1), we have to be more precise on the behavior of C at infinity. In the papers studying limit theorem for (2.5) in a *general* framework (i.e. not for a *particular* model), it is often (if not always) assumed that

$$C(x) = |x|^{-\beta} L(|x|),$$
 (2.8)

with  $\beta \in (0, \infty)$  and  $L : \mathbb{R}_+ \to \mathbb{R}$  slowly varying (that is, satisfying  $L(\lambda r)/L(\lambda) \to 1$  as  $\lambda \to \infty$ , for every fixed r > 0). The following three different situations (i)-(ii)-(iii) then occur:

- 1. If  $\beta>\frac{d}{R}$  then  $C\in L^R(\mathbb{R}^d)$  and we say that we are in the **short-memory case**. We deduce from Breuer-Major Theorem 2.1.1 that  $t^{-d/2}(Y_t-\mathbb{E}[Y_t])\stackrel{\mathrm{law}}{\to} N(0,\sigma^2)$ .
- 2. If  $\beta = \frac{d}{R}$  we are in a **critical case**. Although  $C \notin L^R(\mathbb{R}^d)$ , fluctuations of  $Y_t$  around its mean are still asymptotically Gaussian (see, e.g., [BM83, Theorem 1'], [NN20, Section 5] and [NNT10, Theorem 1]).
- 3. If  $\beta < \frac{d}{R}$  then  $C \not\in L^R(\mathbb{R}^d)$  and we say that we are in the **long-memory case**. A theorem of Dobrushin and Major [DM79] asserts that  $t^{-(d-\frac{R\beta}{2})}L^{-R/2}(t)(Y_t-\mathbb{E}[Y_t])\stackrel{\text{law}}{\to} Z$  where, up to a multiplicative constant, Z is the Hermite distribution of order R and self-similarity index  $H \in (0,1)$  (with H depending only on  $\beta$ ). Since we do not use it in the sequel, we do not give its precise definition here. (The interested reader can e.g. consult [Tud13].) Let us only stress here that the Hermite distribution of order R belongs to the Rth Wiener chaos, and so is **not Gaussian** as soon as  $R \geq 2$ .

#### 2.1.2 Motivating examples

The intuition we can naturally develop from the previous (i)-(ii)-(iii) (and that represents the common intuition forged by the papers written on the subject over the last forty years) is that  $Y_t$  defined by (2.3) displays *Gaussian* (resp. *non-Gaussian*) fluctuations when the Gaussian field B has *short* (resp. *long*) memory, and this whatever the function  $\varphi$  with Hermite rank  $R \ge 2$ . (The case R = 1 is apart.<sup>3</sup>)

As anticipated, we will show in this paper that this intuition can be wrong, and not only marginally or in critical cases. We will indeed state a central limit theorem (Theorem 2.1.2 below) whose conclusion is valid provided that a certain spectral condition is satisfied. As we will see, this may lead to Gaussian fluctuations in a long memory context, in total contrast with Dobrushin and Major [DM79] (see (iii) above).

Recently, Berry's Random Wave Model (BRWM) has attracted a lot of attention. It is defined as follows. Choose the dimension d=2 and consider the centered continuous Gaussian field B on  $\mathbb{R}^2$  with covariance  $\mathbb{E}[B_xB_y]=C(x-y)=J_0(|x-y|)$ , with  $J_0$  the Bessel function of the first kind of order 0 (see Section 2.2.2 for its definition and some properties); in particular, we have

$$C(x) = \sqrt{\frac{2}{\pi}} |x|^{-\frac{1}{2}} \cos\left(|x| - \frac{\pi}{4}\right) + O\left(|x|^{-\frac{3}{2}}\right) \quad \text{as } |x| \to \infty,$$
 (2.9)

see e.g. [Kra14, Theorem 4]. This field, called in this way in honor of Berry who introduced it in the seminal paper [Ber77], can be seen as a universal Gaussian field emerging as the local scaling limit of a number of random fields on two-dimensional manifolds, see e.g. [Die+23] and the references therein. It is widely regarded as a popular model for the Laplacian eigenfunctions (with large eigenvalue  $t^2$ ) of classically chaotic billiards, hence its importance in quantum mechanics. Indeed, integrating  $\varphi(B_x)$  over tD is the same as integrating  $\varphi(B_{tx})$  over the fixed domain D, after a change of variable. As we will see, our Theorem 2.1.2 will allow to prove the Gaussian fluctuations of  $\int_D \varphi(B_{tx}) dx$  for many functionals that were never investigated in the literature, in particular for the cases R=2 and R=1 in (2.10), enriching the knowledge on the geometric properties of the Berry's random wave model. Indeed, to the best of our knowledge, so far Gaussian fluctuations were proved only for (2.3) under the assumption  $R \geq 4$  (see e.g. [Not21]), or for nodal set volumes (see [NPR19]), where the latter cannot be expressed in the form (2.3).

If we compare (2.9) with (2.8), it is like we have  $\beta=\frac{1}{2}$  and that we had replaced the slowly varying function L in (2.8) by the bounded and oscillatory function  $\cos(\cdot-\frac{\pi}{4})$ . As we will see, this replacement is all but insignificant in the presence of long memory. Indeed, taking for example D compact such that  $D=\frac{\vec{D}}{D}$  and with smooth  $\partial D$  and non-vanishing Gaussian curvature, if we apply our main result (Theorem 2.1.2 below) for  $R\in\{1,2,4\}$  and Breuer-Major theorem for  $R\geq 5$ , we obtain (recalling that R, resp. R', denotes the Hermite rank, resp. the second Hermite

 $<sup>^{3}</sup>$ The case where R=1 is apart since, whatever the memory, we generally obtain Gaussian fluctuations. But this is for different reasons that we understand better in the functional version: in the short memory case the limit is a standard Brownian motion, while in the long memory case the limit is a fractional Brownian motion.

2.1. Introduction 41

rank, of the **non-odd** function  $\varphi$ ):

$$\sigma_t^2 \asymp \begin{cases} t^3 & \text{if } R = 2 \text{ or } (R = 1 \text{ and } R' = 2) \\ t^2 \log(t) & \text{if } R = 4 \text{ or } (R = 1 \text{ and } R' = 4) \\ t^2 & \text{if } R \ge 5 \text{ or } (R = 1 \text{ and } R' \ge 5) \end{cases}$$
 (2.10)

and

$$\frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0, 1) \quad \text{as } t \to \infty.$$
 (2.11)

Indeed, Gaussian fluctuations (2.11) and asymptotics (2.10) follow directly from (2.9) and the Breuer-Major Theorem 2.1.1 when  $R \geq 5$ , because in this case we have  $\int_{\mathbb{R}^d} |C(x)|^R dx < \infty$ . The case R=4 is comparable with the situation (ii) in the model (2.8), since  $\beta = \frac{1}{2} = \frac{d}{R}$ ; Gaussian fluctuations (2.11) and asymptotic (2.10) displaying a logarithmic correction are then not surprising, since in agreement with what is usually observed in critical cases. In constrast, the fact that we still have Gaussian fluctuations in (2.11) with unconventional rate when  $R \in \{1,2\}$  is very surprising. Indeed, since  $\beta = \frac{1}{2} < 1 \leq \frac{d}{R}$  in this case, we are in the long-memory case where non-Gaussian fluctuations are usually the rule, see (iii). This last case turns out to be the most important, since R=1 and R=2 are the most common values in applications.

As we will see, what we just described (Gaussian fluctuations in a long memory framework) for BRWM is not an isolated phenomenon. In fact, we are going to state and prove Theorem 2.1.2 (our main result), which not only explains in a clear way the phenomenon observed for BRWM, but also bring to light an easy-to-check condition on the spectral measure of a general Gaussian field B (see (2.20) below) to imply Gaussian fluctuations for  $Y_t$ .

We conclude this section by emphasing that Gaussian fluctuations in presence of long memory have already been observed in the literature in other (non-Euclidean) contexts, in particular for integral functionals of random Laplace eigenfunctions on the sphere  $\mathbb{S}^2$ , see [MW14] for more details.

#### 2.1.3 Main result

In order to state Theorem 2.1.2, we have to introduce a certain number of further quantities and notations. We continue to let B,  $\varphi$  and  $Y_t$  be as described in (2.1), (2.2) and (2.3) respectively, and we recall the Hermite decomposition (2.4) of  $\varphi$  defining its Hermite rank R and its second Hermite rank R'.

Fix t > 0 and  $q \ge R$ , and set

$$Y_{q,t} = \int_{tD} H_q(B_x) dx. \tag{2.12}$$

Using (2.4), we immediately get that

$$Y_t = \mathbb{E}[Y_t] + \sum_{q=R}^{\infty} a_q Y_{q,t}, \quad t \ge 0.$$
 (2.13)

Moreover, a direct computation (making use of the isometry properties of Hermite

polynomials, see, e.g., [NP12, Section 1.4]) yields that

$$Var(Y_{q,t}) = q!t^d v_{q,t},$$

where

$$v_{q,t} = \int_{\mathbb{R}^d} C(z)^q g_D\left(\frac{z}{t}\right) dz = \int_{\{|z| \le \operatorname{diam}(D)t\}} C(z)^q g_D\left(\frac{z}{t}\right) dz \ge 0,$$

with  $g_D(x)$  the **covariogram** of D at  $x \in \mathbb{R}^d$ , defined as the Lebesgue measure of  $D \cap (x+D)$ , and  $\operatorname{diam}(D) := \sup\{|x-y| : x, y \in D\} < \infty$  since D is compact. When  $C \in L^q(\mathbb{R}^d)$ , we deduce by dominated convergence (using that  $\|g_D\|_{\infty} \leq \operatorname{Vol}(D)$  and  $g_D(\frac{z}{t}) \to \operatorname{Vol}(D)$  as  $t \to \infty$  for all fixed  $z \in \mathbb{R}^d$ ) that

$$v_{q,t} \to \operatorname{Vol}(D) \int_{\mathbb{R}^d} C(z)^q dz \ge 0.$$

Since our field B is stationary and its covariance function C is continuous, Bochner's theorem yields the existence of a finite measure G on  $\mathbb{R}^d$  such that

$$C(x) = \int_{\mathbb{R}^d} e^{i\langle \lambda, x \rangle} G(d\lambda), \qquad (2.14)$$

with  $\langle \cdot, \cdot \rangle$  the usual scalar product on  $\mathbb{R}^d$ . The measure G is called the **spectral** measure of B, and it will be our gateway towards the Fourier analysis techniques developed in the sequel.

At this stage, let us make a further assumption on B, by supposing that it is also **isotropic**. In our framework, this is equivalent to suppose that the quantity C(x) only depends on the norm |x|, namely that there is a function  $\rho : \mathbb{R}_+ \to \mathbb{R}$  such that

$$C(x) = \rho(|x|), \quad x \in \mathbb{R}^d.$$

Now, set

$$\mu(s) = G(\{|x| \le s\}), \quad s \in (0, \infty).$$
 (2.15)

Since  $\mu$  is increasing and bounded, it defines a finite measure on  $\mathbb{R}_+$ , called the **isotropic spectral measure** of B. Because C(x) only depends of |x|, we can write, with  $w_d = \int_{S^{d-1}} d\xi$  ( $S^{d-1}$  being the unit sphere of  $\mathbb{R}^d$ ):

$$\rho(r) = \frac{1}{w_d} \int_{S^{d-1}} C(r\xi) d\xi = \frac{1}{w_d} \int_{S^{d-1}} d\xi \int_{\mathbb{R}^d} e^{i\langle \lambda, r\xi \rangle} G(d\lambda)$$
$$= \int_{\mathbb{R}^d} b_d(r|\lambda|) G(d\lambda) = \int_0^\infty b_d(rs) \mu(ds), \tag{2.16}$$

where

$$b_d(|\lambda|) = \frac{1}{w_d} \int_{S^{d-1}} e^{i\langle\lambda,\xi\rangle} d\xi.$$
 (2.17)

By d-dimensional polar coordinates, we readily find (see [Sch38, page 815]) that

$$b_d(r) = c_d r^{1 - \frac{d}{2}} J_{\frac{d}{2} - 1}(r), \quad r > 0,$$
 (2.18)

2.1. Introduction 43

with  $J_{\nu}$  denoting the Bessel function of the first kind of order  $\nu$  (see Section 2.2.2) and  $c_d > 0$  a constant depending only of d.

We are at last in a position to state our main result, which we have decided to call the **spectral central limit theorem**, because it leads to Gaussian fluctuations on the one hand (hence 'central') and the main assumption we have to check is the spectral condition (2.20) on the other hand (hence 'spectral').

**Theorem 2.1.2** (Spectral CLT). Fix  $d \geq 2$ , let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary, isotropic and has unit-variance. Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be **not odd** and such that  $\mathbb{E}[\varphi(N)^2] < \infty$  with  $N \sim N(0,1)$ , let R (resp. R') be the Hermite rank (resp. second Hermite rank) of  $\varphi$ , where  $(R,R') \notin \{(1,3)\} \cup \{(2k+1,n) : k \geq 1, n \in \mathbb{N}\}$ . Consider  $Y_t$  defined by (2.3), where  $D \subset \mathbb{R}^d$  compact and  $\operatorname{Vol}(D) > 0$ . Set  $m_t = \mathbb{E}[Y_t]$  and  $\sigma_t = \sqrt{\operatorname{Var}(Y_t)} > 0$ , and recall the definition (2.15) of the isotropic spectral measure  $\mu$ . Set

$$w_{q,t} = \int_{\{|z| < t\}} C(z)^q dz, \tag{2.19}$$

and assume that the following spectral condition holds

$$\int_0^\infty s^{-\frac{d}{R}} \mu(ds) < \infty. \tag{2.20}$$

Finally, when R=2 assume that  $|\mathcal{F}[\mathbf{1}_D](x)|=O\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x|\to\infty$ , and when R=1 assume that  $|\mathcal{F}[\mathbf{1}_D](x)|=o\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x|\to\infty$ , with  $\mathcal{F}$  the Fourier transform (since we assumed D to be compact, these two assumptions are satisfied for example when  $D=\overset{\circ}{D}$  and  $\partial D$  is smooth with non-vanishing Gaussian curvature, see e.g. [BHI03]. Then we have:

$$\sigma_t^2 \asymp \left\{ \begin{array}{ll} t^d w_{R,t} & \text{ if } R \text{ even} \\ t^d w_{R',t} & \text{ if } R = 1 \text{ and } R' \in \{2,4\} \\ t^d & \text{ if } R = 1 \text{ and } R' \geq 5 \end{array} \right.$$

and

$$\frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0, 1) \quad \textit{as } t \to \infty.$$

By a simple Fubini argument, we observe that the spectral condition (2.20) is equivalent to

$$\int_0^\infty s^{-1-\frac{d}{R}}\mu(s)ds < \infty. \tag{2.21}$$

When  $\rho(r)=r^{-\beta}L(r)$  with  $\beta\in(0,\frac{d}{R})$  and L slowly varying as  $r\to\infty$ , we have  $\mu(s)\sim s^\beta L(s^{-1})$  as  $s\to0$ , see [Leo99, Thm 1.4.3]. We deduce that (2.20)-(2.21) is not satisfied, which is of course perfectly consistent with the conclusion of the Dobrushin-Major noncentral limit theorem [DM79].

Also, let us observe that the assumption " $\varphi$  is not odd" is equivalent to " $\exists \ k \geq 1$  such that  $a_{2k} \neq 0$ ", and so is needed only in the case where R=1 and  $R' \geq 5$ . Moreover, the cases not covered by our Theorem 2.1.2 are  $R \geq 3$  odd, (R,R')=(1,3) and  $\varphi$  odd. In these situations, very peculiar things can happen. Consider for example  $\varphi(x)=x$ , which gives  $Y_t=\int_{tD}B_xdx$ . Using (2.29), (2.14) and Fubini we have, with

 $\mathcal{F}$  the Fourier transform,

$$\operatorname{Var}(Y_t) = \int_{\mathbb{R}^d} C(z) g_{tD}(z) dz = \int_{\mathbb{R}^d} \mathcal{F}[g_{tD}](\lambda) G(d\lambda).$$

Using that  $g_{tD} = \mathbf{1}_{tD} * \mathbf{1}_{-tD}$  we obtain

$$\operatorname{Var}(Y_t) = \operatorname{const} \int_{\mathbb{R}^d} \frac{t^d}{|\lambda|^d} |t\lambda|^d |\mathcal{F}[\mathbf{1}_D](t\lambda)|^2 G(d\lambda). \tag{2.22}$$

In particular, for  $D = \{|z| \le 1\} \subset \mathbb{R}^d$  and Berry's Random Wave Model  $B = (B_x)_{x \in \mathbb{R}^2}$  we get (using (2.26))

$$\sigma_t^2 = \text{Var}(\mathbf{Y_t}) = \text{const } t^2 J_1^2(t);$$

although  $Y_t$  is Gaussian the conclusion of Theorem 2.1.2 cannot hold in this case, the limit of  $\frac{Y_t-m_t}{\sigma_t}$  being ill-defined due to the fact that  $\operatorname{Card}\{t: \sigma_t=0\} \cap [T,\infty) = +\infty$  for all T>0.

To understand the significance of our spectral CLT, let us go back to BRWM and explain how Theorem 2.1.2 together with Breuer-Major theorem allow to prove (2.11). Since  $C(x) = \rho(|x|) = J_0(|x|)$ , it follows immediately from the representation (2.16) that the isotropic spectral measure associated with BRWM  $B = (B_x)_{x \in \mathbb{R}^2}$  is  $\mu = \delta_1$ , with  $\delta_1$  the Dirac mass at 1. In particular, the spectral condition (2.20) is obviously satisfied, whatever the value of R. The convergence (2.11) thus follows from the Breuer-Major Theorem 2.1.1 (see also (2.9)) if  $R \geq 5$  and from Theorem 2.1.2 in all the other cases in (2.10).

#### 2.1.4 Possible natural extensions of Theorem 2.1.2

As a natural extension of the present work, it would be interesting to study the *joint convergence* associated with Theorem 2.1.2. More precisely, taking  $D_1, \ldots, D_n$  compact domains in  $\mathbb{R}^d$ , is it possible to identify conditions that resemble those in Theorem 2.1.2 ensuring that the random vector

$$\left(\int_{tD_1} \varphi(B_x) dx, \dots, \int_{tD_n} \varphi(B_x) dx\right)$$

converges, after proper normalization, to a Gaussian random vector? In other words, can we prove a *multivariate* spectral central limit theorem? Such an extension is not immediate, because the spectral condition (2.20) alone looks too general to capture the asymptotic behavior of the covariances between the components of the random vector. A partial answer to this question (covering also Berry's model) will be given in the forthcoming work [Mai24] by the first author. Note that a multivariate result was obtained in the particular case of the nodal length of Berry's model restricted to a finite collection of smooth compact domains  $D_1, \ldots, D_n$  of the plane in the recent paper [PV20].

A further stronger generalization of Theorem 2.1.2 would be to prove a *functional* spectral central limit theorem. This problem has been investigated in the particular case of nodal set volumes in [NPV23] by ad hoc techniques. In our general framework, we believe however that proving such a functional extension would require novel

ideas that go beyond the scope of the present paper.

#### 2.1.5 Plan of the paper

Apart from Section 2.6 that illustrates a use of Theorem 2.1.2, the rest of the paper is fully devoted to the proof of this latter. More precisely, after a few needed preliminaries given in Section 2.2,

- 1. in Section 2.3 we will first consider the situation where the Hermite rank R of  $\varphi$  is even and bigger or equal than 4. As we will see, in this case we have that  $\int_{tD} |C(x)|^R dx = O(\log t)$ , meaning that we are somehow in the 'domain of attraction' of the Breuer-Major theorem 2.1.1.;
- 2. we will then deal with the remaining cases, namely R=2 in Section 2.4 and  $(R=1,\,\varphi \text{ non-odd} \text{ and } R'\neq 3)$  in Section 2.5. We regard this part as the most important contribution of this paper, especially since among the Hermite ranks, R=1 and R=2 are the most common values encountered in practice. They turn out to be also the most difficult cases. To deal with them, we will have to introduce novel ideas, by making heavy use of Fourier techniques in a way that, to the best of our knowledge, has never been introduced before this work.

#### 2.2 A few preliminaries for the proof of Theorem 2.1.2

#### **2.2.1** Well-posedness of $Y_t$

The following proposition explains why the random variable  $Y_t$  defined by (2.3) is well-defined for all t.

**Proposition 2.2.1.** Let  $B=(B_x)_{x\in\mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , assumed to be stationary and to have unit-variance. Let  $D\subset\mathbb{R}^d$  be compact with  $\operatorname{Vol}(D)>0$  and let  $\varphi:\mathbb{R}\to\mathbb{R}$  be such that  $\mathbb{E}[\varphi(N)^2]<\infty$  with  $N\sim N(0,1)$ . Consider the Hermite expansion of  $\varphi$  given by (2.4), and set

$$\varphi_n = \mathbb{E}[\varphi(N)] + \sum_{q=R}^n a_q H_q, \quad n \ge R.$$

For each fixed t > 0, the sequence  $\int_{tD} \varphi_n(B_x) dx$ ,  $n \geq R$ , is a.s. well-defined and converges in  $L^2(\Omega)$ . The limit is noted  $Y_t$  and we write, possibly with a slight abuse of notation:

$$Y_t = \int_{tD} \varphi(B_x) dx.$$

*Proof.* That  $\int_{tD} \varphi_n(B_x) dx$  is a.s. well-defined is because tD is compact,  $\varphi_n$  is a polynomial and B is continuous. For any  $n, m \geq R$  and  $q \in \mathbb{N}$ , we have (since  $|C(z)| \leq C(0) = 1$ ,  $a_0 = \mathbb{E}[\varphi(N)]$  and  $a_q = 0$  if  $q \in \{1, \dots, R-1\}$ )

$$\left|\mathbf{1}_{[0,n\wedge m]}(q)\int_{(tD)^2}q!a_q^2C(x-y)^qdxdy\right|\leq \operatorname{Vol}(tD)^2q!a_q^2,$$

with  $\sum_{q=0}^{\infty} \operatorname{Vol}(tD)^2 q! a_q^2 = \operatorname{Vol}(tD)^2 \mathbb{E}[\varphi(N)^2] < \infty$  by assumption. Then, by dominated convergence we obtain as  $n, m \to \infty$ 

$$\mathbb{E}\left[\int_{tD} \varphi_n(B_x) dx \int_{tD} \varphi_m(B_y) dy\right] = \sum_{q=0}^{n \wedge m} q! a_q^2 \int_{(tD)^2} C(x-y)^q dx dy$$

$$\to \mathbb{E}[\varphi(N)]^2 \text{Vol}(tD)^2 + \sum_{q=R}^{\infty} q! a_q^2 \int_{(tD)^2} C(x-y)^q dx dy.$$

That is,  $\{\int_{tD} \varphi_n(B_x) dx\}_{n \geq R}$  is an  $L^2(\Omega)$ -Cauchy sequence, and the desired conclusion follows.

## 2.2.2 Bessel functions and Fourier transform of the indicator of the unit ball

The Bessel function  $J_{\nu}$ ,  $\nu \geq 0$ , is defined by the series

$$J_{\nu}(t) = \left(\frac{t}{2}\right)^{\nu} \sum_{j=0}^{\infty} (-1)^{j} \frac{(t^{2}/4)^{j}}{j!\Gamma(j+\nu+1)}, \quad t \in \mathbb{R},$$

and it is a solution to the ODE

$$t^2 J_{\nu}''(t) + t J_{\nu}'(t) + (t^2 - \nu^2) J_{\nu}(t) = 0.$$

It satisfies the classical Schläfli's representation

$$J_{\nu}(t) = \frac{1}{\pi} \int_{0}^{\pi} \cos(\nu\theta - t\sin\theta) d\theta - \frac{\sin(\nu\pi)}{\pi} \int_{0}^{\infty} e^{-\nu\theta - t\sinh\theta} d\theta, \qquad (2.23)$$

as well as the Mehler-Sonine formula

$$J_{\nu}(t) = \frac{(t/2)^{\nu}}{\sqrt{\pi} \Gamma(\nu + \frac{1}{2})} \int_{-1}^{1} e^{its} (1 - s^{2})^{\nu - \frac{1}{2}} ds.$$
 (2.24)

Schläfli's representation (2.23) immediately implies that  $J_{\nu}$  is bounded on  $\mathbb{R}$ . This, combined with an inequality found e.g. in [Kra14] (see also the references therein), leads to

$$\sup_{t \in \mathbb{R}_+} \sqrt{t} |J_{\nu}(t)| < \infty \quad \text{for all } \nu \ge 0, \tag{2.25}$$

a simple property that we will use several times in the forthcoming proof of Theorem 2.1.2.

The Fourier transform of the indicator of the unit ball is given by

$$\mathcal{F}\left[\mathbf{1}_{\{|x|\leq 1\}}\right](y) = \int_{\{|x|<1\}} e^{i\langle x,y\rangle} dx.$$

Since  $\mathcal{F}\left[\mathbf{1}_{\{|x|\leq 1\}}\right]$  is rotationally symmetric, we can write, with  $lpha_d$  the volume of the

unit ball  $\{|x| \le 1\} \subseteq \mathbb{R}^d$ ,

$$\mathcal{F}\left[\mathbf{1}_{\{|x|\leq 1\}}\right](y) = \mathcal{F}\left[\mathbf{1}_{\{|x|\leq 1\}}\right]((0,\ldots,0,|y|)) 
= \int_{-1}^{1} (1-x_d^2)^{\frac{d-1}{2}} \alpha_{d-1} e^{ix_d|y|} dx_d 
= \alpha_{d-1}\sqrt{\pi} \Gamma\left(\frac{d+1}{2}\right) 2^{\nu} |y|^{-\frac{d}{2}} J_{d/2}(|y|),$$
(2.26)

the last equality being a consequence of (2.24).

#### **2.2.3** Reduction to the *R*th chaos

In the short memory case, that is when  $C \in L^R(\mathbb{R}^d)$ , the Breuer-Major theorem 2.1.1 yields Gaussian fluctuations for  $Y_t$ . In its modern proof given by [NPP11] (see also [NP12, Chapter 7]) the chaotic expansion of  $Y_t - m_t$  is considered, namely

$$Y_t - m_t = \sum_{q=R}^{\infty} a_q Y_{q,t},$$
 (2.27)

and the proof goes as follows. It is first shown that  $t^{-d}\sigma_t^2 \to \sigma^2$  in (2.7) by means of the isometry property of Hermite polynomials. Then, it is proved using the Fourth Moment Theorem (see [NP12, Theorem 5.2.7]) that  $t^{-d/2}Y_{q,t} \stackrel{\text{law}}{\to} N(0, \sigma_q^2)$  for all  $q \geq R$  from which it is deduced that  $t^{-\frac{d}{2}}(Y_t - m_t) \stackrel{\text{law}}{\to} N(0, \sum_{q=R}^{\infty} a_q^2 \sigma_q^2)$  thanks to [NP12, Theorem 5.2.7]. In particular, we observe that no term is asymptotically dominant in (2.27): they all contribute to the limit.

As the following result will show, the situation is totally opposite in the critical and long memory cases (when R is even): here, it is the term  $Y_{R,t}$  alone which is responsible of the limit.

**Proposition 2.2.2** (Reduction to the Rth chaos). Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary and has unit-variance (note that  $d \geq 2$  and isotropy are not required here). Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be such that  $\mathbb{E}[\varphi(N)^2] < \infty$  (with  $N \sim N(0,1)$ ) and have Hermite decomposition (2.4) and Hermite rank R, and consider  $Y_t$  defined by (2.3), where  $D \subset \mathbb{R}^d$  compact and  $\operatorname{Vol}(D) > 0$ . Set  $m_t = \mathbb{E}[Y_t]$ ,  $\sigma_t = \sqrt{\operatorname{Var}(Y_t)} > 0$ , and recall the definition (2.1) of C(x) and the definition (2.12) of  $Y_{q,t}$ .

If R is even, if  $t^{-d}\mathrm{Var}(Y_{R,t})\to\infty$  and if  $\int_{\mathbb{R}^d}|C(x)|^Mdx<\infty$  for some  $M\geq R+1$  then, as  $t\to\infty$ , we have  $\sigma_t^2\asymp\mathrm{Var}(Y_{R,t})$  and

$$Y_{R,t}/\sqrt{\operatorname{Var}(Y_{R,t})} \stackrel{\text{law}}{\to} N(0,1) \implies \frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0,1).$$
 (2.28)

Proof. We have

$$Var(Y_{q,t}) = \int_{(tD)^2} \mathbb{E}[H_q(B_y)H_q(B_x)]dxdy = q! \int_{(tD)^2} C(x-y)^q dxdy.$$

Applying the change of variable z = y - x and then Fubini, we obtain that

$$Var(Y_{q,t}) = q!t^d v_{q,t},$$

where

$$v_{q,t} = \int_{\{|z| \le \operatorname{diam}(D)t\}} C(z)^q g_D\left(\frac{z}{t}\right) dz, \tag{2.29}$$

where diam $(D) = \sup\{|x - y| : x, y \in D\} < \infty$  (because D is compact) and  $g_D(x)$  is the covariogram of D at  $x \in \mathbb{R}^d$ , that is,  $g_D(x)$  is the Lebesgue measure of  $D \cap (x + D)$ . Also, set

$$\widetilde{v}_{q,t} = \int_{\{|z| \le \operatorname{diam}(D)t\}} |C(z)|^q g_D\left(\frac{z}{t}\right) dz, \tag{2.30}$$

and observe that  $v_{R,t} = \tilde{v}_{R,t} > 0$  since R is even.

For any q > R, we have

$$\frac{\text{Var}(Y_{q,t})}{\text{Var}(Y_{R,t})} = \frac{q! v_{q,t}}{R! v_{R,t}}.$$
(2.31)

Applying Cauchy-Schwarz inequality we obtain, for q > R,

$$\begin{split} \frac{|v_{q,t}|}{v_{R,t}} & \leq \frac{\int_{\{|z| \leq \operatorname{diam}(D)t\}} |C(z)|^q g_D(\frac{z}{t}) dz}{v_{R,t}} \\ & = \frac{\int_{\{|z| \leq \operatorname{diam}(D)t\}} |C(z)|^{q-\frac{R}{2}} |C(z)|^{\frac{R}{2}} g_D\left(\frac{z}{t}\right) dz}{v_{R,t}} \\ & \leq \left(\frac{\int_{\{|z| \leq \operatorname{diam}(D)t\}} |C(z)|^{2q-R} g_D\left(\frac{z}{t}\right) dz}{v_{R,t}}\right)^{1/2} = \left(\frac{\widetilde{v}_{2q-R,t}}{v_{R,t}}\right)^{1/2}. \end{split}$$

Applying Cauchy-Schwarz again, but this time with 2q - R > R instead of q > R, we obtain

$$\left(\frac{\widetilde{v}_{2q-R,t}}{v_{R,t}}\right)^{1/2} \le \left(\frac{\widetilde{v}_{4q-3R,t}}{v_{R,t}}\right)^{\frac{1}{4}}.$$

By iterating the process, we get, for every  $n \ge 3$ 

$$\frac{|v_{q,t}|}{v_{R,t}} \le \left(\frac{\widetilde{v}_{2q-R,t}}{v_{R,t}}\right)^{1/2} \le \left(\frac{\widetilde{v}_{4q-3R,t}}{v_{R,t}}\right)^{\frac{1}{4}} \le \dots \le \left(\frac{\widetilde{v}_{R+2^n(q-R),t}}{v_{R,t}}\right)^{\frac{1}{2^n}}.$$

When q>R, we have  $R+2^n(q-R)\geq 2^n$ , so we may and will choose n large enough so that  $R+2^n(q-R)\geq M$  for all q>R (recall from the statement of Proposition 2.2.2 that M is an integer supposed to be such that  $\int_{\mathbb{R}^d}|C(x)|^Mdx<\infty$ ). Using  $|C(x)|\leq C(0)=1$  and  $g_D(x)\leq \mathrm{Vol}(D)$  for all  $x\in\mathbb{R}^d$ , we deduce that  $\widetilde{v}_{R+2^n(q-R),t}\leq \mathrm{Vol}(D)\int_{\mathbb{R}^d}\left|C(x)\right|^Mdx$ . Combining all these facts together, we finally get that

$$\frac{|v_{q,t}|}{v_{R,t}} \le \left( \text{Vol}(D) \int_{\mathbb{R}^d} |C(x)|^M dx \right)^{\frac{1}{2^n}} v_{R,t}^{-\frac{1}{2^n}} \quad \text{for all } q > R.$$
 (2.32)

Since  $Var(Y_t) = \sum_{q=R}^{\infty} a_q^2 Var(Y_{q,t})$ , we deduce

$$\left| \frac{\operatorname{Var}(Y_t)}{\operatorname{Var}(Y_{R,t})} - a_R^2 \right| \le \frac{1}{R!} \left( \operatorname{Vol}(D) \int_{\mathbb{R}^d} |C(x)|^M dx \right)^{\frac{1}{2^n}} \left( \sum_{q=R+1}^{\infty} a_q^2 q! \right) v_{R,t}^{-\frac{1}{2^n}}, \quad (2.33)$$

from which it comes that  $\mathrm{Var}(Y_t) \sim a_R^2 \mathrm{Var}(Y_{R,t})$  as  $t \to \infty$ , since  $v_{R,t} = \frac{1}{R!} t^{-d} \mathrm{Var}(Y_{R,t}) \to \infty$  by assumption.

To conclude the proof of Proposition 2.2.2, it remains to prove (2.28). Using the decomposition

$$\frac{Y_t - m_t}{\sqrt{\operatorname{Var}(Y_t)}} - \operatorname{sgn}(a_R) \frac{Y_{R,t}}{\sqrt{\operatorname{Var}(Y_{R,t})}}$$

$$= \frac{\operatorname{sgn}(a_R)(Y_t - m_t - a_R Y_{R,t})}{a_R \sqrt{\operatorname{Var}(Y_{R,t})}} + \frac{Y_t - m_t}{\sqrt{\operatorname{Var}(Y_t)}} \left\{ 1 - \frac{1}{|a_R|} \sqrt{\frac{\operatorname{Var}(Y_t)}{\operatorname{Var}(Y_{R,t})}} \right\}$$

we get that

$$\mathbb{E}\left[\left(\frac{Y_t - m_t}{\sqrt{\operatorname{Var}(Y_t)}} - \operatorname{sgn}(a_R) \frac{Y_{R,t}}{\sqrt{\operatorname{Var}(Y_{R,t})}}\right)^2\right] \\
\leq 2 \frac{\mathbb{E}[(Y_t - m_t - a_R Y_{R,t})^2]}{a_R^2 \operatorname{Var}(Y_{R,t})} + 2\left(1 - \frac{1}{|a_R|} \sqrt{\frac{\operatorname{Var}(Y_t)}{\operatorname{Var}(Y_{R,t})}}\right)^2.$$
(2.34)

Since  $\mathbb{E}\left[\left(Y_t-m_t-a_RY_{R,t}\right)^2\right]=\sum_{q=R+1}^{\infty}a_q^2\operatorname{Var}(Y_{q,t})$ , we deduce from (2.31) and (2.32) that

$$\frac{\mathbb{E}[(Y_t - m_t - a_R Y_{R,t})^2]}{a_R^2 \text{Var}(Y_{R,t})} \le \frac{1}{R! a_R^2} \left( \text{Vol}(D) \int_{\mathbb{R}^d} |C(x)|^M dx \right)^{\frac{1}{2^n}} \left( \sum_{q=R+1}^{\infty} q! a_q^2 \right) v_{R,t}^{-\frac{1}{2^n}},$$

and this tends to zero as  $t \to \infty$ . By plugging this into (2.34) and taking into account that (2.33) holds, we deduce that

$$\frac{Y_t}{\sqrt{\operatorname{Var}(Y_t)}} - \operatorname{sgn}(a_R) \frac{Y_{R,t}}{\sqrt{\operatorname{Var}(Y_{R,t})}} \stackrel{L^2(\Omega)}{\to} 0 \quad \text{as } t \to \infty,$$

from which the implication (2.28) now follows easily.

#### 2.2.4 Elements of Malliavin calculus and the Fourth Moment theorem

To obtain the N(0,1) distribution in the limit in Theorem 2.1.2, we rely on the Fourth Moment Theorem of Nualart and Peccati (see [NP12, Theorem 5.2.7]). Before stating and reformulating it in our framework, we start with some notions on Malliavin calculus. For all the missing details we refer to [NP12] and [Nua06].

#### The Wiener-Itô integral

Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary with unit-variance. Define

$$\mathcal{H} = \overline{\operatorname{span}\{B_x, x \in \mathbb{R}^d\}}^{L^2(\Omega)}.$$

Since  $\mathcal{H}$  is a real, separable Hilbert space, there is an isometry  $\Phi: \mathcal{H} \to L^2(\mathbb{R}_+)$ . If we set  $e_x := \Phi(B_x)$  for every  $x \in \mathbb{R}^d$ , we have

$$\mathbb{E}[B_x B_y] = C(x - y) = \langle e_x, e_y \rangle_{L^2(\mathbb{R}_+)}.$$

Consider now the Gaussian noise  $W = \{W(h), h \in L^2(\mathbb{R}_+)\}$ , i.e. a family of centered Gaussian random variables with covariance given by

$$\mathbb{E}[W(h)W(g)] = \langle h, g \rangle_{L^2(\mathbb{R}_+)}.$$

Since in this paper we are only interested in distributions, we can assume without loss of generality that  $(B_x)_{x \in \mathbb{R}^d} = (W(e_x))_{x \in \mathbb{R}^d}$ .

For  $q \geq 1$ , we also define the qth Wiener chaos as the linear subspace of  $L^2(\Omega)$  generated by  $\{H_q(W(h)), h \in L^2(\mathbb{R}_+)\}$ . For every  $h \in L^2(\mathbb{R}_+)$  with  $\|h\|_{L^2(\mathbb{R}_+)} = 1$ , we define the qth Wiener-Itô integral

$$I_q(h^{\otimes q}) = H_q(W(h)),$$

where  $h^{\otimes q}: \mathbb{R}^q_+ \to \mathbb{R}$  is defined by

$$h^{\otimes q}(x_1,\ldots,x_q) = \prod_{r=1}^q h(x_r).$$

Note that the definition of  $I_q$  can be extended to every function in the space  $L^2_s(\mathbb{R}^q_+)$  of symmetric functions in  $L^2(\mathbb{R}^q_+)$  so that  $I_q:L^2_s(\mathbb{R}^q_+)\to L^2(\Omega)$  is a linear map, because  $\mathrm{span}\{h^{\otimes q},h\in L^2(\mathbb{R}_+)\}$  is dense in  $L^2_s(\mathbb{R}^q_+)$  (see, e.g., [Flo97]).

#### **Contractions and the Fourth Moment Theorem**

For  $q \in \mathbb{N}$ ,  $r \in \{1, \dots, q-1\}$  and  $h_1, h_2$  symmetric functions with unit norm in  $L^2(\mathbb{R}_+)$ , we can define the rth **contraction** of  $h_1^{\otimes q}$  and  $h_2^{\otimes q}$  as the (non-symmetric) element of  $L^2(\mathbb{R}_+^{2q-2r})$  given by

$$h_1^{\otimes q} \otimes_r h_2^{\otimes q} = \langle h_1, h_2 \rangle_{L^2(\mathbb{R}_+)}^r h_1^{\otimes q-r} \otimes h_2^{\otimes q-r}.$$

Here again this definition can be extended (taking the closure in  $L^2(\mathbb{R}^q_+)$ ) to every  $h_1,h_2$  in  $L^2_s(\mathbb{R}^q_+)$ . We will denote the norm in the space  $L^2(\mathbb{R}^q_+)$  as  $\|\cdot\|_q$ . We can finally state the celebrated **Fourth Moment Theorem**, proved in [NP05] by Nualart and Peccati.

**Theorem 2.2.3** (Fourth Moment Theorem). Fix  $q \ge 2$ , consider  $(h_t)_{t>0} \subset L^2_s(\mathbb{R}^q_+)$  and assume that  $\mathbb{E}[I_q(h_t)^2] \to 1$  as  $t \to \infty$ . Then the following assertions are equivalent:

- $I_q(h_t)$  converges in distribution to a standard Gaussian  $N \sim N(0,1)$ .
- $\mathbb{E}[I_a(h_t)^4] \to 3 = \mathbb{E}[N^4]$ , where  $N \sim N(0, 1)$ .
- $||h_t \otimes_r h_t||_{2q-2r} \to 0$  as  $t \to \infty$ , for all  $r = 1, \ldots, q-1$ .

An important consequence of the previous result (in our framework) is the following.

**Theorem 2.2.4.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary and has unit-variance. Assume that  $D \subset \mathbb{R}^d$  is compact with  $\operatorname{Vol}(D) > 0$ . Recall the definition (2.1) of C. Fix also  $q \geq 2$ , recall the definition (2.12) of  $Y_{q,t}$  and assume  $\operatorname{Var}(Y_{q,t}) > 0$  for all t large enough. If we have, for any  $r \in \{1, \ldots, q-1\}$ , that

$$\frac{t^d}{\operatorname{Var}^2(Y_{q,t})} \int_{\{|u| \le diam(D)t\}^3} |C(x)|^r |C(y)|^r |C(z)|^{q-r} |C(x+y+z)|^{q-r} dx dy dz \quad \textbf{(2.35)}$$

converges to 0 as  $t \to \infty$ , then  $Y_{q,t}/\sqrt{\operatorname{Var}(Y_{q,t})} \stackrel{law}{\to} N(0,1)$  as  $t \to \infty$ .

*Proof.* Let us first write  $Y_{q,t}$  as a q-th multiple Wiener-Itô integral with respect to B:

$$Y_{q,t} = \int_{tD} H_q(B_x) dx = \int_{tD} H_q(W(e_x)) dx = \int_{tD} I_q(e_x^{\otimes q}) dx = I_q(f_{t,q}),$$

where

$$f_{t,q} = \int_{tD} e_x^{\otimes q} dx.$$

Since by definition

$$f_{t,q} \otimes_r f_{t,q} = \int_{AD} \int_{AD} C^r(x-y) e_x^{\otimes q-r} e_y^{\otimes q-r} dx dy,$$

we obtain

$$||f_{t,q} \otimes_r f_{t,q}||_{2q-2r}^2$$

$$= \int_{(tD)^4} C(x_1 - x_3)^r C(x_2 - x_4)^r C(x_1 - x_2)^{q-r} \times C(x_3 - x_4)^{q-r} dx_1 dx_2 dx_3 dx_4.$$

Applying the change of variable  $x = x_3 - x_1$ ,  $y = x_2 - x_4$ ,  $z = x_1 - x_2$ ,  $a = x_4$  (whose Jacobian is equal to 1) and using the symmetry of C, we get that

$$||f_{t,q} \otimes_r f_{t,q}||_{2q-2r}^2 \le \text{Vol}(D)$$

$$t^d \int_{\{|y| \le \text{diam}(D)t\}^3} |C(x)|^r |C(y)|^r |C(z)|^{q-r} |C(x+y+z)|^{q-r} dx dy dz.$$

The Fourth Moment Theorem asserts that  $Y_{q,t}/\sqrt{\operatorname{Var}(Y_{q,t})}$  converges in distribution to N(0,1) if (and only if)  $\|f_{t,q}\otimes_r f_{t,q}\|_{2q-2r}/\operatorname{Var}(Y_{q,t})\to 0$  for all  $r=1,\ldots,q-1$ . The desired conclusion thus follows from (2.35) and the previous bound for  $\|f_{t,q}\otimes_r f_{t,q}\|_{2q-2r}^2$ .

#### **2.3** Proof of Theorem **2.1.2** when $R \ge 4$ even

This section is devoted to the proof of Theorem 2.1.2 when  $R \ge 4$  is even, which is equivalent to say that  $\frac{d}{R} \le \frac{d-1}{2}$  and R even. It represents the 'easy' part of Theorem 2.1.2.

To ease the exposition, we write in the following proposition the statement obtained when, in Theorem 2.1.2, we suppose that  $\frac{d}{R} \leq \frac{d-1}{2}$  and R even.

**Proposition 2.3.1.** Fix  $d \geq 2$ , let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary, isotropic and has unit-variance. Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be such that  $\mathbb{E}[\varphi(N)^2] < \infty$  with  $N \sim N(0,1)$ , let R be the Hermite rank of  $\varphi$  and consider  $Y_t$  defined by (2.3), where D compact and  $\operatorname{Vol}(D) > 0$ . Set  $m_t = \mathbb{E}[Y_t]$  and  $\sigma_t = \sqrt{\operatorname{Var}(Y_t)} > 0$ , and recall the definition (2.15) of the isotropic spectral measure  $\mu$  and the definition (2.19) of  $w_{R,t}$ :

$$w_{R,t} = \int_{\{|z| \le t\}} C^R(z) dz.$$

If (2.20) holds, if R even and  $\frac{d}{R} \leq \frac{d-1}{2}$  (i.e.  $R \geq 4$  and R even), then  $\sigma_t^2 \asymp t^d w_{R,t}$  and

$$\frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0, 1)$$
 as  $t \to \infty$ .

The goal of this Section 2.3 is to prove Proposition 2.3.1. As we will see, it will be a direct consequence of Lemma 2.3.2 and Proposition 2.3.3 below.

We start with Lemma 2.3.2.

**Lemma 2.3.2.** Let  $\rho$  and  $\mu$  be associated as in (2.16) and consider an exponent  $\beta \in (0, \frac{d-1}{2}]$ . If  $\int_0^\infty s^{-\beta} \mu(ds) < \infty$ , then  $\sup_{r \in \mathbb{R}_+} r^{\beta} |\rho(r)| < \infty$ .

*Proof.* Since  $\rho(|x|) = \mathbb{E}[B_0B_x]$  for all  $x \in \mathbb{R}^d$ , we deduce from Cauchy-Schwarz that  $|\rho|$  is bounded by 1, and thus  $\sup_{r \in [0,T]} r^{\beta} |\rho(r)| < \infty$  for all fixed T > 0.

On the other hand, using the representation (2.16), we can write

$$r^{\beta} 
ho(r) = \int_0^{\infty} (rs)^{\beta} b_d(rs) \frac{\mu(ds)}{s^{\beta}}.$$

But  $J_{\frac{d}{2}-1}$  is bounded and satisfies  $J_{\frac{d}{2}-1}(u)=O(u^{-1/2})$  as  $u\to\infty$  (see (2.25)). So  $\sup_{u\in\mathbb{R}_+}u^{\beta}|b_d(u)|<\infty$ , and the desired conclusion follows.

Now, let us state and prove Proposition 2.3.3, which may be of independent interest.

**Proposition 2.3.3.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary and has unit-variance (note that we did not assume isotropy and  $d \geq 2$ ). Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be such that  $\mathbb{E}[\varphi(N)^2] < \infty$  with  $N \sim N(0,1)$ , let R be the Hermite rank of  $\varphi$  and consider  $Y_t$  defined by (2.3), where D compact and  $\operatorname{Vol}(D) > 0$ . Set  $m_t = \mathbb{E}[Y_t]$  and  $\sigma_t = \sqrt{\operatorname{Var}(Y_t)} > 0$ , and recall the definition (2.1) of the covariance C and the definition (2.19) of  $w_{R,t}$ .

If R is even, if  $\lim_{t\to\infty} w_{R,t} = \infty$  and if

$$\sup_{x \in \mathbb{R}^d} |x|^{d/R} |C(x)| < \infty, \tag{2.36}$$

then  $\sigma_t^2 symp t^d w_{R,t}$  and

$$\frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0, 1) \quad \text{as } t \to \infty.$$
 (2.37)

*Proof.* The proof is divided into several steps. Recall the definition (2.12) of  $Y_{q,t}$ .

Step 1. We claim that if R even, then

$$v_{R,t} \asymp w_{R,t},\tag{2.38}$$

where  $v_{R,t}$  is defined by (2.29). To prove it, below we let c > 0 denote a constant independent of t whose value can change from one instance to another. Using (2.29), we have on the one hand

$$v_{R,t} = \frac{1}{R!} t^{-d} \operatorname{Var}(Y_{R,t}) = \int_{\{|z| \le \operatorname{diam}(D)t\}} C(z)^R g_D\left(\frac{z}{t}\right) dz$$
  
$$\le \operatorname{Vol}(D) w_{R,\operatorname{diam}(D)t} \le c w_{R,t}$$

where the last equality follows from the so-called doubling conditions at the origin for non-negative positive definite functions (see [GT19]).

On the other hand,  $g_D$  is uniformly continuous (in particular continuous in 0) according to [Gal11]. We deduce that  $g_D\left(\frac{z}{t}\right) \geq \operatorname{Vol}(D) - \frac{1}{2}\operatorname{Vol}(D) = \frac{1}{2}\operatorname{Vol}(D)$  for all  $z \in \{|z| \leq \delta_D t\}$  for some  $\delta_D > 0$  depending only on D and for every t > 0. As a result,

$$v_{R,t} = \frac{1}{R!} t^{-d} \operatorname{Var}(Y_{R,t}) \ge \int_{\{|z| \le \delta_D t\}} C(z)^R g_D\left(\frac{z}{t}\right) dz$$
$$\ge \frac{1}{2} \operatorname{Vol}(D) w_{R,\delta_D t} \ge c w_{R,t},$$

where the last inequality follows again from the doubling conditions proved in [GT19]. The announced claim (2.38) follows.

Step 2. Since  $w_{R,t} \to \infty$ , it follows from Step 1 that  $t^{-d}\mathrm{Var}(Y_{R,t}) \to \infty$ . Moreover, since (2.36) holds, we have that  $\int_{\mathbb{R}^d} |C(x)|^{R+1} dx < \infty$ . Applying Proposition 2.2.2, we obtain that  $\sigma_t^2 \asymp t^d w_{R,t}$  and also that (2.37) will follow if we prove that  $Y_{R,t}/\sqrt{\mathrm{Var}(Y_{R,t})} \to N(0,1)$ .

Step 3. In this last step, we prove that  $Y_{R,t}/\sqrt{\mathrm{Var}(Y_{R,t})} \to N(0,1)$ , which will complete the proof of Proposition 2.3.3, see the conclusion of Step 2. To do this, we use Theorem 2.2.4, following the same approach as in [NPR19, Lemma 8.1]. Fix a contraction index  $r \in \{1, \dots, R-1\}$ . Using the inequality  $u^r v^{R-r} \leq u^R + v^R$  for

 $u, v \in \mathbb{R}_+$ , we have

$$\int_{\{|u| \le \operatorname{diam}(D)t\}^{3}} |C(x)|^{r} |C(y)|^{r} |C(z)|^{R-r} |C(x+y+z)|^{R-r} dx dy dz 
\le 2 \int_{\{|u| \le \operatorname{diam}(D)t\}^{3}} |C(x)|^{r} |C(y)|^{R} |C(x+y+z)|^{R-r} dx dy dz 
\le c \widetilde{w}_{r,t} w_{R,t} \widetilde{w}_{R-r,t}$$

where the last inequality follows from the change of variable a = x + y + z and doubling conditions in [GT19], and

$$\widetilde{w}_{q,t} = \int_{\{|z| \le t\}} |C(z)|^q dz.$$

We deduce that (2.35) is bounded by

$$\frac{c}{t^d} \frac{\widetilde{w}_{r,t} w_{R,t} \widetilde{w}_{R-r,t}}{v_{R,t}^2} = O\left(t^{-d} \frac{\widetilde{w}_{r,t} \widetilde{w}_{R-r,t}}{w_{R,t}}\right),\,$$

where the big O comes from (2.38). We deduce from (2.36) that  $\widetilde{w}_{q,t} = O\left(t^{d-d\frac{q}{R}}\right)$  for q < R. Using this last fact, we get

$$O\left(t^{-d}\frac{\widetilde{w}_{r,t}\widetilde{w}_{R-r,t}}{w_{R,t}}\right) = O\left(t^{-d}\frac{t^{d-d}\frac{r}{R}t^{d-d}\frac{R-r}{R}}{w_{R,t}}\right) = O\left(\frac{1}{w_{R,t}}\right)$$

and then (2.35) converges to 0 as  $t \to \infty$ . Therefore, the convergence  $Y_{R,t}/\sqrt{\operatorname{Var}(Y_{R,t})} \to N(0,1)$  follows from Theorem 2.2.4, and completes the proof of Proposition 2.3.3.  $\square$ 

We can now proceed with the proof of Proposition 2.3.1, that is, the proof of Theorem 2.1.2 when  $R \geq 4$  and R even (or equivalently,  $\frac{d}{R} \leq \frac{d-1}{2}$  and R even).

*Proof of Proposition 2.3.1.* If  $w_{R,t}$  is convergent, then the result follows applying the Breuer-Major theorem 2.1.1. So we can assume that  $w_{R,t} \to \infty$ . Now, by comparing the statements of Proposition 2.3.1 and Proposition 2.3.3, we see that we are left to check that, if (2.20) holds with  $\frac{d}{R} \leq \frac{d-1}{2}$ , then (2.36) holds. Since this is a mere application of Lemma 2.3.2 with  $\beta = \frac{d}{R}$ , the proof of Proposition 2.3.1 is complete.

#### **2.4** Proof of Theorem **2.1.2** when R=2

This section is devoted to the proof of Theorem 2.1.2 when R=2. It requires the introduction of novel ideas with respect to the existing literature, mainly Fourier arguments.

To ease the exposition, we write in the following proposition the statement obtained when, in Theorem 2.1.2, we additionally suppose that R=2 and  $|\mathcal{F}[\mathbf{1}_D](x)|=O\left(\frac{1}{|x|^{d/2}}\right)$ .

**Proposition 2.4.1.** Fix  $d \ge 2$ , let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary, isotropic and has unit-variance.

Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be such that  $\mathbb{E}[\varphi(N)^2] < \infty$  with  $N \sim N(0,1)$ , and assume that  $\varphi$  has Hermite rank R = 2. Consider  $Y_t$  defined by (2.3), where D compact and  $\operatorname{Vol}(D) > 0$ . Set  $m_t = \mathbb{E}[Y_t]$  and  $\sigma_t = \sqrt{\operatorname{Var}(Y_t)} > 0$ , and recall the definition (2.15) of the isotropic spectral measure  $\mu$  and the definition (2.19) of  $w_{q,t}$ .

If  $|\mathcal{F}[\mathbf{1}_D](x)| = O\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x| \to \infty^4$ , with  $\mathcal{F}$  the Fourier transform, and if the spectral condition holds

$$\int_0^\infty s^{-\frac{d}{2}}\mu(ds) < \infty \tag{2.39}$$

then  $\sigma_t^2 \asymp t^d w_{2,t}$  and

$$\frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0, 1) \quad \textit{as } t \to \infty.$$

Before proving Proposition 2.4.1 (which is the goal of this section), let us recall the definition (2.12) of  $Y_{2,t}$  and let us state and prove some preliminary results. We start with Lemma 2.4.2, reformulating in a spectral form the norm of the contractions introduced in section 2.2.4. Note that the following result has an analogous version for a general rth contraction in the case  $Y_{q,t}$ , but we skip this unnecessary extension for the sake of brevity.

**Lemma 2.4.2.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary and has unit-variance (note that we did not assume isotropy and  $d \geq 2$ ). Assume that  $D \subset \mathbb{R}^d$  is compact with  $\operatorname{Vol}(D) > 0$ , recall the notions introduced in section 2.2.4 and write

$$Y_{2,t} = \int_{tD} H_2(B_x) dx = I_2(f_t),$$

where

$$f_t = \int_{tD} e_x^{\otimes 2} dx.$$

Recall the definition (2.1) of C and define

$$C_t(u) := C(u) \mathbf{1}_{\{|u| \le \text{diam}(D)t\}}(u), \quad t > 0, \quad u \in \mathbb{R}^d,$$
 (2.40)

and for  $D \subset \mathbb{R}^d$  compact

$$D_t(u) := tD \cap (tD + u), \quad u \in \mathbb{R}^d. \tag{2.41}$$

Finally, recall the definition (2.14) of G. Then

$$\|f_t \otimes_1 f_t\|_2^2 = \int_{\mathbb{D}^d} G(dx) \int_{\mathbb{D}^d} dy \mathcal{F}[C_t](x-y) \left| \int_{\mathbb{D}^d} C_t(w) e^{i\langle x, w \rangle}, \mathcal{F}\left[\mathbf{1}_{D_t(-w)}\right](y) dw \right|^2 (2.42)$$

where  $\mathcal{F}$  is the Fourier transform, and

$$\int_{\mathbb{R}^d} C_t(w) e^{i\langle x, w \rangle} \mathcal{F}\left[\mathbf{1}_{D_t(-w)}\right](y) dw = \int_{\mathbb{R}^d} \mathcal{F}[C_t](x-z) \mathcal{F}[\mathbf{1}_{tD}](y-z) \mathcal{F}[\mathbf{1}_{tD}](z) dz.$$
(2.43)

<sup>&</sup>lt;sup>4</sup>Since we assumed D to be compact, this happens for example when  $D = \hat{D}$  and  $\partial D$  is smooth with non-vanishing Gaussian curvature, see e.g. [BHI03].

Proof. Proceeding exactly as in the proof of Theorem 2.2.4, we have

$$||f_t \otimes_1 f_t||_2^2 = \int_{(tD)^4} C(x_1 - x_3)C(x_2 - x_4)C(x_1 - x_2) \times C(x_3 - x_4)dx_1dx_2dx_3dx_4.$$

Applying the change of variable  $u=x_1-x_3$ ,  $v=x_3-x_4$ ,  $w=x_4-x_2$ ,  $z=x_2$  we have:  $x_2=z$ ,  $x_4=w+x_2=w+z$ ,  $x_3=v+x_4=v+w+z$  and  $x_1=u+x_3=u+v+w+z$ . Then

$$\begin{aligned} &\|f_{t} \otimes_{1} f_{t}\|_{2}^{2} \\ &= \int_{\mathbb{R}^{3d}} C(u)C(v)C(w)C(u+v+w) \\ &\times \left(\int_{\mathbb{R}^{d}} \mathbf{1}_{tD}(z)\mathbf{1}_{tD}(w+z)\mathbf{1}_{tD}(v+w+z)\mathbf{1}_{tD}(u+v+w+z)dz\right) dudvdw \\ &= \int_{\mathbb{R}^{3d}} C(u)C(v)C(w)C(u+v+w) \\ &\times \left(\int_{\mathbb{R}^{d}} \mathbf{1}_{tD}(z)\mathbf{1}_{tD}(-w+z)\mathbf{1}_{tD}(-v-w+z)\mathbf{1}_{tD}(-u-v-w+z)dz\right) dudvdw \\ &= \int_{\mathbb{R}^{3d}} C(u)C(v)C(w)C(u+v+w) \\ &\times \left(\int_{\mathbb{R}^{d}} \mathbf{1}_{tD\cap(tD+w)\cap(tD+w+v)\cap(tD+u+v+w)}(z)dz\right) dudvdw \\ &= \int_{\mathbb{R}^{3d}} C_{t}(u)C_{t}(v)C_{t}(w)C(u+v+w) \\ &\times \operatorname{Vol}((tD-w)\cap tD\cap(tD+v)\cap(tD+u+v)) dudvdw, \end{aligned}$$

where in the last equality we used the translation invariance of the Lebesgue measure (subtracting w) and the definition (2.40), justified by the fact that  $tD \cap (tD+a)$  is empty when  $|a| > \operatorname{diam}(D)t$ . Now recall the definition (2.41). We have

$$||f_{t} \otimes_{1} f_{t}||_{2}^{2}$$

$$= \int_{\mathbb{R}^{3d}} C_{t}(u)C_{t}(v)C_{t}(w)C(u+v+w)\operatorname{Vol}(D_{t}(-w)\cap(D_{t}(u)+v))dudvdw$$

$$= \int_{\mathbb{R}^{3d}} C_{t}(u)C_{t}(v)C_{t}(w)C(u+v+w)\left(\mathbf{1}_{D_{t}(-w)}*\mathbf{1}_{-D_{t}(u)}\right)(v)dudvdw,$$

where in the last expression we used that

$$\operatorname{Vol}(D_{t}(-w) \cap (D_{t}(u) + v))$$

$$= \int_{\mathbb{R}^{d}} \mathbf{1}_{D_{t}(-w)}(z) \mathbf{1}_{D_{t}(u)}(z - v) dz = \left(\mathbf{1}_{D_{t}(-w)} * \mathbf{1}_{-D_{t}(u)}\right)(v).$$

Now, using the spectral representation (2.14)

$$C(u+v+w) = \int_{\mathbb{R}^d} e^{i\langle x, u+v+w\rangle} G(dx),$$

we have that

$$||f_{t} \otimes_{1} f_{t}||_{2}^{2}$$

$$= \int_{\mathbb{R}^{d}} G(dx) \int_{\mathbb{R}^{2d}} du dw C_{t}(u) C_{t}(w) e^{i\langle x, u+w \rangle} \int_{\mathbb{R}^{d}} e^{i\langle x, v \rangle} C_{t}(v)$$

$$\times \left( \mathbf{1}_{D_{t}(-w)} * \mathbf{1}_{-D_{t}(u)} \right) (v) dv.$$
(2.44)

Fix x, u and w, and let us focus in (2.44) on the integral with respect to v. Using the properties of the Fourier transform  $\mathcal{F}$  with respect to convolution and products, we can write

$$\int_{\mathbb{R}^d} e^{i\langle x,v\rangle} C_t(v) (\mathbf{1}_{D_t(-w)} * \mathbf{1}_{-D_t(u)})(v) dv = \mathcal{F}[C_t \cdot (\mathbf{1}_{D_t(-w)} * \mathbf{1}_{-D_t(u)})](x)$$

$$= \left(\mathcal{F}[C_t] * \mathcal{F}[\mathbf{1}_{D_t(-w)} * \mathbf{1}_{-D_t(u)}]\right)(x) = \left(\mathcal{F}[C_t] * \left(\mathcal{F}[\mathbf{1}_{D_t(-w)}]\mathcal{F}[\mathbf{1}_{-D_t(u)}]\right)\right)(x)$$

$$= \int_{\mathbb{R}^d} \mathcal{F}[C_t](x-y) \mathcal{F}[\mathbf{1}_{D_t(-w)}](y) \mathcal{F}[\mathbf{1}_{-D_t(u)}](y) dy.$$

Putting everything in (2.44), we get

$$||f_t \otimes_1 f_t||_2^2 = \int_{\mathbb{R}^d} G(dx) \int_{\mathbb{R}^{2d}} du dw C_t(u) C_t(w) e^{i\langle x, u+w \rangle}$$

$$\times \int_{\mathbb{R}^d} dy \mathcal{F}[C_t](x-y) \mathcal{F}[\mathbf{1}_{D_t(-w)}](y) \mathcal{F}[\mathbf{1}_{-D_t(u)}](y).$$
(2.45)

Exchanging integrals in (2.45) yields

$$\begin{aligned} &\|f_{t} \otimes_{1} f_{t}\|_{2}^{2} \\ &= \int_{\mathbb{R}^{d}} G(dx) \int_{\mathbb{R}^{d}} dy \mathcal{F}[C_{t}](x-y) \\ &\times \left( \int_{\mathbb{R}^{d}} dw C_{t}(w) e^{i\langle x,w \rangle} \mathcal{F}[\mathbf{1}_{D_{t}(-w)}](y) \right) \left( \int_{\mathbb{R}^{d}} du C_{t}(u) e^{i\langle x,u \rangle} \mathcal{F}[\mathbf{1}_{-D_{t}(u)}](y) \right) \\ &= \int_{\mathbb{R}^{d}} G(dx) \int_{\mathbb{R}^{d}} dy \mathcal{F}[C_{t}](x-y) \left| \int_{\mathbb{R}^{d}} C_{t}(w) e^{i\langle x,w \rangle} \mathcal{F}[\mathbf{1}_{D_{t}(-w)}](y) dw \right|^{2}, \end{aligned}$$

which is exactly (2.42). Now, let us focus on the term  $\int_{\mathbb{R}^d} C_t(w) e^{i\langle x,w\rangle} \mathcal{F}[\mathbf{1}_{D_t(-w)}](y) dw$  in (2.42). Using again basic Fourier analysis and Fubini theorem, we obtain

$$\int_{\mathbb{R}^{d}} C_{t}(w)e^{i\langle x,w\rangle} \mathcal{F}[\mathbf{1}_{D_{t}(-w)}](y)dw = \int_{\mathbb{R}^{d}} C_{t}(w)e^{i\langle x,w\rangle} \mathcal{F}[\mathbf{1}_{tD}\mathbf{1}_{tD-w}](y)dw$$

$$= \int_{\mathbb{R}^{d}} C_{t}(w)e^{i\langle x,w\rangle} \left( \int_{\mathbb{R}^{d}} \mathcal{F}[\mathbf{1}_{tD}](y-z)\mathcal{F}[\mathbf{1}_{tD-w}](z)dz \right)dw$$

$$= \int_{\mathbb{R}^{d}} C_{t}(w)e^{i\langle x,w\rangle} \left( \int_{\mathbb{R}^{d}} \mathcal{F}[\mathbf{1}_{tD}](y-z)\mathcal{F}[\mathbf{1}_{tD}](z)e^{i\langle -w,z\rangle}dz \right)dw$$

$$= \int_{\mathbb{R}^{d}} \mathcal{F}[\mathbf{1}_{tD}](y-z)\mathcal{F}[\mathbf{1}_{tD}](z) \left( \int_{\mathbb{R}^{d}} C_{t}(w)e^{i\langle x-z,w\rangle}dw \right)dz$$

$$= \int_{\mathbb{R}^{d}} \mathcal{F}[C_{t}](x-z)\mathcal{F}[\mathbf{1}_{tD}](y-z)\mathcal{F}[\mathbf{1}_{tD}](z)dz,$$

which is exactly (2.43).

**Lemma 2.4.3.** Fix  $d \geq 2$ , let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is isotropic, stationary and has unit-variance. Assume that  $D \subset \mathbb{R}^d$  is compact with  $\operatorname{Vol}(D) > 0$ . Recall the definition (2.1) of C, the definition (2.14) of G, the definition (2.19) of  $w_{2,t}$ , the definition (2.40) of  $C_t$  and the definition (2.41) of  $D_t(u)$ . If the spectral condition (2.39) holds, then

$$\int_{\mathbb{R}^d} G(dx) \int_{|x-y| \ge |x|/2} dy |\mathcal{F}[C_t](x-y)| \left| \int_{\mathbb{R}^d} C_t(w) e^{i\langle x, w \rangle} \mathcal{F}[\mathbf{1}_{D_t(-w)}](y) dw \right|^2$$

$$\le \operatorname{const} w_{2,t}^{3/2} t^{2d}.$$

Proof. First, call

$$I := \int_{\mathbb{R}^d} G(dx) \int_{|x-y| > |x|/2} dy |\mathcal{F}[C_t](x-y)| \left| \int_{\mathbb{R}^d} C_t(w) e^{i\langle x, w \rangle} \mathcal{F}[\mathbf{1}_{D_t(-w)}](y) dw \right|^2.$$

Using polar coordinates, we can write

$$\mathcal{F}[C_t](x) = \int_{\{|u| \le \operatorname{diam}(D)t\}} C(u)e^{i\langle x, u\rangle} du = t^d \int_0^{\operatorname{diam}(D)} dr \rho(rt)r^{d-1} \int_{S^{d-1}} e^{irt\langle x, \theta\rangle} d\theta.$$

Thanks to (2.17)-(2.18) and (2.25) we have for |x| > 1

$$\left| \int_{S^{d-1}} e^{i \langle x, \theta \rangle} d\theta \right| = \text{const} \, |x|^{1 - \frac{d}{2}} \, \left| J_{\frac{d}{2} - 1}(|x|) \right| \leq \text{const} \, |x|^{1 - \frac{d}{2}} |x|^{-\frac{1}{2}} \leq \frac{\text{const}}{\sqrt{|x|}}.$$

We deduce

$$\begin{split} |\mathcal{F}[C_t](x)| &\leq \frac{\text{const}}{\sqrt{|x|}} t^d \int_0^{\text{diam}(D)} |\rho(rt)| \frac{r^{d-1}}{\sqrt{rt}} dr = \frac{\text{const}}{\sqrt{|x|}} \int_0^{\text{diam}(D)t} |\rho(r)| r^{\frac{d-1}{2} + \frac{d}{2} - 1} dr \\ &\leq \frac{\text{const}}{\sqrt{|x|}} t^{\frac{d}{2} - 1} \int_0^{\text{diam}(D)t} |\rho(r)| r^{\frac{d-1}{2}} dr \leq \frac{\text{const}}{\sqrt{|x|}} t^{\frac{d-1}{2}} \sqrt{\int_0^{\text{diam}(D)t} \rho^2(r) r^{d-1} dr} \\ &= \frac{\text{const}}{\sqrt{|x|}} t^{\frac{d-1}{2}} w_{R,\text{diam}(D)t}^{1/2} \leq \frac{\text{const}}{\sqrt{|x|}} t^{\frac{d-1}{2}} w_{R,t}^{1/2} \end{split}$$

where the last inequality follows from the doubling conditions in [GT19]. With this

estimate, we have that I satisfies:

$$I \leq \int_{\mathbb{R}^{d}} G(dx) \int_{|y-x| \geq |x|/2} dy \, |\mathcal{F}[C_{t}](x-y)|$$

$$\times \left| \int_{\mathbb{R}^{d}} dw C_{t}(w) e^{i\langle x,w \rangle} \mathcal{F}[\mathbf{1}_{D_{t}(-w)}](y) \right|^{2}$$

$$\leq \operatorname{const} \cdot t^{\frac{d-1}{2}} w_{R,t}^{1/2} \int_{\mathbb{R}^{d}} \frac{G(dx)}{\sqrt{|x|}} \int_{\mathbb{R}^{d}} dy \, \left| \int_{\mathbb{R}^{d}} dw C_{t}(w) e^{i\langle x,w \rangle} \mathcal{F}[\mathbf{1}_{D_{t}(-w)}](y) \right|^{2}$$

$$= \operatorname{const} \cdot t^{\frac{d-1}{2}} w_{R,t}^{1/2} \int_{\mathbb{R}^{2d}} du dw C_{t}(u) C_{t}(w)$$

$$\times \underbrace{\left( \int_{\mathbb{R}^{d}} \frac{G(dx)}{\sqrt{|x|}} e^{i\langle x,u+w \rangle} \right)}_{:=\bar{C}(u+w)} \int_{\mathbb{R}^{d}} dy \mathcal{F}[\mathbf{1}_{D_{t}(-w)}](y) \mathcal{F}[\mathbf{1}_{-D_{t}(u)}](y)$$

$$\leq \operatorname{const} \cdot t^{\frac{d-1}{2}} w_{R,t}^{1/2} \int_{\mathbb{R}^{2d}} du dw |C_{t}(u) C_{t}(w) \bar{C}(u+w)|$$

$$\times \sqrt{\operatorname{Vol}(tD \cap (tD+w))} \sqrt{\operatorname{Vol}(tD \cap (tD+u))}$$

$$\leq \operatorname{const} \cdot t^{\frac{3}{2}d-\frac{1}{2}} w_{R,t}^{1/2} \int_{\mathbb{R}^{2d}} du dw |C_{t}(u) C_{t}(w) \bar{C}(u+w)|,$$

where in the third inequality we used Cauchy-Schwarz inequality and Plancherel theorem.

Recall now that  $\int_{\mathbb{R}^d} |x|^{-\frac{d}{2}} G(dx) = \int_0^\infty s^{-\frac{d}{2}} \mu(ds) < \infty$  by assumption, see (2.39). By definition, the spectral measure of  $\bar{C}$  is  $\frac{G(d\lambda)}{\sqrt{|\lambda|}}$ , see (2.14), and the associated isotropic spectral measure is  $\bar{\mu}(ds) = \frac{\mu(ds)}{\sqrt{s}}$ . Since  $\int_0^\infty s^{-\frac{d}{2}} \mu(ds) = \int_0^\infty s^{-\frac{d-1}{2}} \bar{\mu}(ds) < \infty$ , we deduce from Lemma 2.3.2 that  $\sup_{r \in \mathbb{R}_+} r^{\frac{d-1}{2}} |\bar{\rho}(r)| = \sup_{u \in \mathbb{R}^d} |u|^{\frac{d-1}{2}} |\bar{C}(u)| < \infty$ , that is,  $\bar{C}(u) \leq \operatorname{const} |u|^{-\frac{d-1}{2}}$ . Using the inequality  $|ab| \leq a^2 + b^2$ , this yields

$$I \leq \operatorname{const} \cdot t^{\frac{3}{2}d - \frac{1}{2}} w_{R,t}^{1/2} \int_{\{|u|,|w| \leq \operatorname{diam}(D)t\}} du dw C^{2}(u) |\bar{C}(u+w)|$$

$$\leq \operatorname{const} \cdot t^{\frac{3}{2}d - \frac{1}{2}} w_{R,t}^{1/2} w_{R,\operatorname{diam}(D)t} \int_{\{|z| \leq 2\operatorname{diam}(D)t\}} |\bar{C}(z)| dz$$

$$\leq \operatorname{const} \cdot t^{\frac{3}{2}d - \frac{1}{2}} w_{R,t}^{3/2} \int_{0}^{2\operatorname{diam}(D)t} r^{\frac{d-1}{2}} dr = \operatorname{const} \cdot t^{2d} w_{R,t}^{3/2}.$$

where in the last inequality we used again doubling conditions in [GT19].  $\Box$ 

**Lemma 2.4.4.** Let  $B=(B_x)_{x\in\mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary and has unit-variance (note that we did not assume isotropy and  $d\geq 2$ ). Assume that  $D\subset\mathbb{R}^d$  is compact with  $\operatorname{Vol}(D)>0$ . Recall the definition (2.1) of C, the definition (2.14) of C, the definition (2.19) of  $C_t$ , the definition (2.40) of  $C_t$  and the definition (2.41) of  $C_t$  (1). If  $|\mathcal{F}[\mathbf{1}_D](x)| = O\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x|\to\infty^5$ , with  $\mathcal{F}$  the Fourier transform, and if the analogous of the spectral condition (2.39) holds (note that here the isotropic spectral measure  $\mu$  is not defined, because  $C_t$  is

<sup>&</sup>lt;sup>5</sup>Since we assumed D to be compact, this happens for example when  $D = \overline{\hat{D}}$  and  $\partial D$  is smooth with non-vanishing Gaussian curvature, see e.g. [BHI03].

not necessarily isotropic, but G is defined)

$$\int_{\mathbb{R}^d} |x|^{-\frac{d}{2}} G(dx) < \infty,$$

then

$$\int_{\mathbb{R}^d} G(dx) \int_{|y| \ge |x|/2} dy |\mathcal{F}[C_t](x-y)| \left| \int_{\mathbb{R}^d} C_t(w) e^{i\langle x, w \rangle} \mathcal{F}[\mathbf{1}_{D_t(-w)}](y) dw \right|^2$$

$$\le \operatorname{const} w_{2,t}^{3/2} t^{2d}.$$

Proof. First, by (2.43) in Lemma 2.4.2, we can write

$$II: = \int_{\mathbb{R}^d} G(dx) \int_{|y| \ge |x|/2} dy |\mathcal{F}[C_t](x-y)| \left| \int_{\mathbb{R}^d} C_t(w) e^{i\langle x, w \rangle} \mathcal{F}[\mathbf{1}_{D_t(-w)}](y) dw \right|^2$$

$$= \int_{\mathbb{R}^d} G(dx)$$

$$\int_{|y| > |x|/2} dy |\mathcal{F}[C_t](x-y)| \left| \int_{\mathbb{R}^d} \mathcal{F}[C_t](x-z) \mathcal{F}[\mathbf{1}_{tD}](y-z) \mathcal{F}[\mathbf{1}_{tD}](z) dz \right|^2$$

Using Cauchy-Schwarz inequality two times (first with respect to y, then with respect to z) and then Plancherel theorem, II satisfies, with  $\|\cdot\|_2$  the  $L^2$ -norm:

$$II \leq \|\mathcal{F}[C_{t}]\|_{2} \int_{\mathbb{R}^{d}} G(dx)$$

$$\times \left( \int_{|y| \geq |x|/2} dy \left| \int_{\mathbb{R}^{d}} dz \mathcal{F}[C_{t}](x-z) \mathcal{F}[\mathbf{1}_{tD}](y-z) \mathcal{F}[\mathbf{1}_{tD}](z) \right|^{4} \right)^{1/2}$$

$$\leq \|\mathcal{F}[C_{t}]\|_{2} \int_{\mathbb{R}^{d}} G(dx)$$

$$\times \left( \int_{|y| \geq |x|/2} dy \|\mathcal{F}[C_{t}]\|_{2}^{4} \left| \int_{\mathbb{R}^{d}} dz |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y-z) |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (z) \right|^{2} \right)^{1/2}$$

$$= \|\mathcal{F}[C_{t}]\|_{2}^{3} \int_{\mathbb{R}^{d}} G(dx) \left( \int_{|y| \geq |x|/2} dy ||\mathcal{F}[\mathbf{1}_{tD}]|^{2} * |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y) \right|^{2} \right)^{1/2}$$

$$\leq \|\mathcal{F}[C_{t}]\|_{2}^{3} \int_{\mathbb{R}^{d}} G(dx) \left( \sup_{|y| \geq |x|/2} |\mathcal{F}[\mathbf{1}_{tD}]|^{2} * |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y) \right)^{1/2}$$

$$\times \left( \int_{\mathbb{R}^{d}} dy ||\mathcal{F}[\mathbf{1}_{tD}]|^{2} * |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y) \right)^{1/2}$$

$$\leq \|\mathcal{F}[C_{t}]\|_{2}^{3} \int_{\mathbb{R}^{d}} G(dx) \left( \sup_{|y| \geq |x|/2} |\mathcal{F}[\mathbf{1}_{tD}]|^{2} * |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y) \right)^{1/2}$$

$$\times \left( \int_{\mathbb{R}^{d}} |\mathcal{F}[\mathbf{1}_{tD}](y)|^{2} dy \right)$$

$$= \operatorname{const} \cdot t^{d} \|\mathcal{F}[C_{t}]\|_{2}^{3} \int_{\mathbb{R}^{d}} G(dx) \left( \sup_{|y| \geq |x|/2} |\mathcal{F}[\mathbf{1}_{tD}]|^{2} * |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y) \right)^{1/2},$$

where the last inequality comes from Young convolution inequalities and the last equality from Plancherel theorem.

Since by assumption  $|\mathcal{F}[\mathbf{1}_D](x)| = O\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x| \to \infty$ , in particular  $\sup_{x \in \mathbb{R}^d} |x|^{d/2} |\mathcal{F}[\mathbf{1}_D](x)| < \infty$ . We deduce, for all t > 0 and all  $y \in \mathbb{R}^d$ , that

$$|\mathcal{F}[\mathbf{1}_{tD}](y)|^2 = t^{2d}|\mathcal{F}[\mathbf{1}_D](ty)|^2 \le \operatorname{const} t^d |y|^{-d}.$$

This implies

$$\sup_{|y| \ge |x|/2} |\mathcal{F}[\mathbf{1}_{tD}]|^{2} * |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y) = \sup_{|y| \ge |x|/2} \left| \int_{\mathbb{R}^{d}} |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y-z) |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (z) dz \right|$$

$$\le \operatorname{const} \cdot \frac{t^{d}}{|x|^{d}} \sup_{|y| \ge |x|/2} \left| \int_{|z| \ge |x|/4} |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (y-z) dz + \int_{|y-z| \ge |x|/4} |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (z) dz \right|$$

$$\le \operatorname{const} \cdot \frac{t^{d}}{|x|^{d}} \int_{\mathbb{R}^{d}} |\mathcal{F}[\mathbf{1}_{tD}]|^{2} (z) dz = \operatorname{const} \frac{t^{2d}}{|x|^{d}}.$$

We deduce that

$$\begin{split} II & \leq & \operatorname{const} \cdot t^{2d} \, \| \mathcal{F}[C_t] \|_2^3 \int_{\mathbb{R}^d} \frac{G(dx)}{|x|^{d/2}} \\ & = & \operatorname{const} \cdot t^{2d} w_{R, \operatorname{diam}(D)t}^{3/2} \int_{\mathbb{R}^d} \frac{G(dx)}{|x|^{d/2}} \leq \operatorname{const} \cdot t^{2d} w_{R,t}^{3/2} \int_{\mathbb{R}^d} \frac{G(dx)}{|x|^{d/2}}, \end{split}$$

where the last equality comes from Plancherel theorem and the last inequality from doubling conditions in [GT19].

Since  $\int_{\mathbb{R}^d} |x|^{-\frac{d}{2}} G(dx) < \infty$  by assumption, our bound for II is

$$II \leq \operatorname{const} \cdot t^{2d} w_{R_t}^{3/2}$$

and the proof is concluded.

Now we can proceed with the proof of Proposition 2.4.1.

Proof of Proposition 2.4.1. First, we assume without loss of generality that  $w_{2,t} \to \infty$ , since otherwise the statement follows from theorem 2.1.1. Moreover, throughout all the proof we freely use that  $v_{2,t} \times w_{2,t}$ , see (2.38).

Starting from now, the proof is divided into three steps.

Step 1: Reduction of the proof. We claim that it is enough to check that  $Y_{2,t}/\sqrt{\mathrm{Var}(Y_{2,t})} \to N(0,1)$  in order to prove Proposition 2.4.1. Indeed, since (2.39) holds and given that  $\frac{d}{2} > \frac{d-1}{2}$ , we deduce from Lemma 2.3.2 that  $\sup_{r \in \mathbb{R}_+} \{r^{\frac{d-1}{2}}\rho(r)\} < \infty$ . Proposition 2.2.2 implies the statement on  $\sigma_t^2$  and justifies that we are left to prove that  $Y_{2,t}/\sqrt{\mathrm{Var}(Y_{2,t})} \to N(0,1)$ .

As done in the proof of Proposition 2.3.1, in order to prove the convergence  $Y_{2,t}/\sqrt{\mathrm{Var}(Y_{2,t})} \to N(0,1)$  we make use of the Fourth Moment Theorem ([NP12, Theorem 5.2.7]); this requires checking that the only involved contraction goes to zero. To do this, we will use the novel ideas from Fourier analysis introduced in

Lemma 2.4.2, Lemma 2.4.3 and Lemma 2.4.4.

As in the proof of Theorem 2.2.4, we can first rewrite  $Y_{2,t}$  as a double Wiener-Itô integral with respect to B:

$$Y_{2,t} = I_2(f_t)$$
, where  $f_t = \int_{tD} e_x^{\otimes 2} dx$ ,

with  $e_x$  such that  $B_x = I_1(e_x)$ . We know from the Fourth Moment Theorem stated in section 2.2.4 that  $Y_{2,t}/\sqrt{\mathrm{Var}(Y_{2,t})} \to N(0,1)$  if and only if  $\|f_t \otimes_1 f_t\|_2/\mathrm{Var}(Y_{2,t}) \to 0$ .

Step 2: A two-term error bound for the norm of the contraction. Here we apply Lemma 2.4.2. Noting that for  $x, y \in \mathbb{R}^d$  either  $|x - y| \ge |x|/2$  or  $|y| \ge |x|/2$ , we deduce from (2.42) and (2.43) the following two terms error bound:

$$\begin{aligned} & \|f_{t} \otimes_{1} f_{t}\|_{2}^{2} \\ & \leq \int_{\mathbb{R}^{d}} G(dx) \int_{|x-y| \geq |x|/2} dy \, |\mathcal{F}[C_{t}](x-y)| \left| \int_{\mathbb{R}^{d}} C_{t}(w) e^{i\langle x, w \rangle} \mathcal{F}[\mathbf{1}_{D_{t}(-w)}](y) dw \right|^{2} \\ & + \int_{\mathbb{R}^{d}} G(dx) \int_{|y| \geq |x|/2} dy \, |\mathcal{F}[C_{t}](x-y)| \left| \int_{\mathbb{R}^{d}} C_{t}(w) e^{i\langle x, w \rangle} \mathcal{F}[\mathbf{1}_{D_{t}(-w)}](y) dw \right|^{2} \\ & = I + II. \end{aligned}$$

Step 3: The norm of the contraction divided by the variance goes to 0. To conclude the proof of Proposition 2.4.1, we will now check (see the conclusion of Step 1) that  $||f_t \otimes_1 f_t||_2 / \text{Var}(Y_{2,t}) \to 0$ . From (2.38), we have

$$\operatorname{Var}(Y_{2,t}) \approx t^d w_{R,t}. \tag{2.46}$$

We deduce from Step 2, Lemma 2.4.3 and Lemma 2.4.4, that

$$\frac{\|f_t \otimes_1 f_t\|_2^2}{\operatorname{Var}(Y_{2,t})^2} \le \operatorname{const} \cdot \frac{w_{R,t}^{3/2}}{w_{R,t}^2} = \operatorname{const} \cdot \frac{1}{w_{R,t}^{1/2}},$$

and the right-hand size goes to 0, since we assumed at the beginning of the proof that  $w_{2,t} \to \infty$ .

#### **2.5** Proof of theorem **2.1.2** when R=1

This section is devoted to the proof of Theorem 2.1.2 in the remaining cases, namely  $\varphi$  non-odd, R=1 and  $R'\neq 3$ .

To ease the exposition, we write in the following proposition the statement obtained when, in Theorem 2.1.2, we additionally suppose that we are in the cases just mentioned above.

**Proposition 2.5.1.** Fix  $d \geq 2$ , let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued continuous centered Gaussian field on  $\mathbb{R}^d$ , and assume that B is stationary, isotropic and has unit-variance. Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be not odd and such that  $\mathbb{E}[\varphi(N)^2] < \infty$  with  $N \sim N(0,1)$ . Assume

 $\varphi$  has Hermite rank R=1 and let  $R'\geq 2$ ,  $R'\neq 3$ , be its second Hermite rank. Consider  $Y_t$  defined by (2.3), where D is compact and  $\operatorname{Vol}(D)>0$ . Set  $m_t=\mathbb{E}[Y_t]$  and  $\sigma_t=\sqrt{\operatorname{Var}(Y_t)}>0$ , and recall the definition (2.15) of the isotropic spectral measure  $\mu$ , the definition (2.12) of  $Y_{q,t}$  and the definition (2.19) of  $w_{q,t}$ . Assume that the spectral condition holds

$$\int_0^\infty s^{-d}\mu(ds) < \infty. \tag{2.47}$$

and that  $|\mathcal{F}[\mathbf{1}_D](x)| = o\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x| \to \infty$  6, with  $\mathcal{F}$  the Fourier transform. Then

$$\sigma_t^2 \asymp \left\{ \begin{array}{ll} t^d w_{R',t} & \text{if } R' \in \{2,4\} \\ t^d & \text{if } R' \ge 5 \end{array} \right.$$

and

$$\frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0, 1)$$
 as  $t \to \infty$ .

*Proof.* Set  $\widehat{\varphi}(x) = \varphi(x) - a_0 - a_1 x$ . By the very definition of R', the function  $\widehat{\varphi}$  has Hermite rank R'. Let us define

$$\widehat{Y}_{t} = \int_{tD} \widehat{\varphi}(B_{x}) dx = \sum_{q=R'}^{\infty} a_{q} Y_{q,t} = Y_{t} - \mathbb{E}[Y_{t}] - a_{1} Y_{1,t}$$
 (2.48)

and its variance

$$\widehat{\sigma}_t^2 = \operatorname{Var}(\widehat{Y}_t) = \sum_{q=R'}^{\infty} a_q^2 \operatorname{Var}(Y_{q,t}) = \sigma_t^2 - a_1^2 \operatorname{Var}(Y_{1,t}). \tag{2.49}$$

The proof is divided into two steps.

Step 1: CLT for  $\hat{Y}_t$ . We claim that  $\hat{\sigma}_t^2 \approx t^d w_{R',t}$  and

$$\frac{\widehat{Y}_t - m_t}{\widehat{\sigma}_t} \stackrel{\text{law}}{\to} N(0, 1) \quad \text{as } t \to \infty.$$
 (2.50)

First of all, observe that (2.47) implies (2.20) for  $R' \ge 2$ . Moreover, note that we can have three different situations:

- If  $R'\geq 5$ , then by (2.47) and Lemma 2.3.2 we have  $\sup_{r\in\mathbb{R}_+} r^{\frac{d-1}{2}} |\rho(r)| = \sup_{u\in\mathbb{R}^d} |u|^{\frac{d-1}{2}} |C(u)| < \infty, \ C\in L^{R'}(\mathbb{R}^d)$ , and the claim follows immediately by Theorem 2.1.1 and the fact that  $\varphi$  is not odd.
- If R' = 4, then the claim follows by Proposition 2.3.1.
- If R'=2, since  $|\mathcal{F}[\mathbf{1}_D](x)|=o\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x|\to\infty$  implies  $|\mathcal{F}[\mathbf{1}_D](x)|=O\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x|\to\infty$ , then the claim follows by Proposition 2.4.1.

<sup>&</sup>lt;sup>6</sup>Since we assumed D to be compact, this happens for example when  $D = \overline{\hat{D}}$  and  $\partial D$  is smooth with non-vanishing Gaussian curvature, see e.g. [BHI03].

Step 2: CLT for  $Y_t$ . We claim that  $\sigma_t \sim \hat{\sigma}_t$  as  $t \to \infty$  and

$$\mathbb{E}\left[\left(\frac{Y_t - m_t}{\sigma_t} - \frac{\widehat{Y}_t}{\widehat{\sigma}_t}\right)^2\right] \to 0. \tag{2.51}$$

The proof of Proposition 2.5.1 thus follows as soon as these two claims are shown to be true. From (2.22) we have

$$t^{-d}\operatorname{Var}(Y_{1,t}) = \operatorname{const} \int_{\mathbb{R}^d} \frac{G(d\lambda)}{|\lambda|^d} |t\lambda|^d |\mathcal{F}[\mathbf{1}_D](t\lambda)|^2.$$

Then, combining (2.47), dominated convergence theorem and the fact that  $|\mathcal{F}[\mathbf{1}_D](x)| = o\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x| \to \infty$ , we have that  $\mathrm{Var}(Y_{1,t}) = o(t^d)$  as  $t \to \infty$ . Since in Step 1 we proved that  $\widehat{\sigma}_t^2 \asymp t^d$  or  $\widehat{\sigma}_t^2 \asymp t^d w_{R',t}$ , we have by (2.49) that  $\widehat{\sigma}_t \sim \sigma_t$ . It remains to prove (2.51). By orthogonality of chaotic projections we have

$$\mathbb{E}\left[\left(\frac{Y_t - m_t}{\sigma_t} - \frac{\widehat{Y}_t}{\widehat{\sigma}_t}\right)^2\right] = a_1^2 \frac{\operatorname{Var}(Y_{1,t})}{\operatorname{Var}(Y_t)} + \left(\frac{\widehat{\sigma}_t}{\sigma_t} - 1\right)^2.$$

Since  $Var(Y_{1,t}) = o(t^d)$ , the first addend converges to 0. On the other hand, the second term vanishes because  $\hat{\sigma}_t \sim \sigma_t$  as  $t \to \infty$ .

#### 

# 2.6 An example of application of Theorem 2.1.2

In this section, we illustrate a possible use of Theorem 2.1.2. In order to introduce our class of fields of interest, recall the definition of the Bessel function  $J_{\nu}$  given in Section 2.2.2 and define, for every  $\nu \geq 0$ , the normalized Bessel function function  $\rho_{\nu}: [0,\infty) \to \mathbb{R}$  as

$$\rho_{\nu}(r) = c_{\nu} \frac{J_{\nu}(r)}{r^{\nu}},$$

where  $c_{\nu}$  is chosen so that  $\rho_{\nu}(0) = 1$ .

Note that  $\rho_{\frac{d}{2}-1}$  is equal to the function  $b_d$  defined in (2.18), and in particular  $\rho_0=b_2=J_0$  when d=2.

Throughout this section, we define the Bessel Gaussian field of order  $\nu$  and dimension d as the real-valued continuous centered Gaussian field  $B^{\nu}=(B_{x}^{\nu})_{x\in\mathbb{R}^{d}}$  with covariance function

$$\mathbb{E}[B_x^{\nu}B_y^{\nu}] = \rho_{\nu}(|x-y|).$$

In particular, d-dimensional Berry's Random Wave Model is defined as the Bessel Gaussian field of order  $\frac{d}{2}-1$  and dimension d.

The existence of the Bessel Gaussian field of order  $\nu$  and dimension  $d \geq 2$  is neither obvious, nor always true. The following result provides a complete picture, and also shows that d-dimensional Berry's Random Wave Model appears to be the critical case for the existence of the Bessel Gaussian field.

**Proposition 2.6.1.** Fix  $d \geq 2$  and  $\nu \geq 0$ . There exists a Bessel Gaussian field  $B^{\nu} = (B_x^{\nu})_{x \in \mathbb{R}^d}$  of order  $\nu$  and dimension d if and only if  $\nu \geq \frac{d}{2} - 1$ . In this case, the isotropic spectral measure associated to  $B^{\nu}$  (see (2.16)) is

$$\mu_{\nu}(ds) = \begin{cases} c_{d,\nu} s^{d-1} (1 - s^2)^{\nu - \frac{d}{2}} \mathbf{1}_{(0,1)} ds & \text{if } \nu > \frac{d}{2} - 1\\ \delta_1(ds) & \text{if } \nu = \frac{d}{2} - 1 \end{cases}, \tag{2.52}$$

where  $c_{d,\nu} > 0$  is chosen so that  $\mu_{\nu}$  is a probability measure.

*Proof.* Observe that  $\rho_{\nu}$  has representation

$$\rho_{\nu}(r) = c_{\nu} \frac{J_{\nu}(r)}{r^{\nu}} = \text{const} \times \sum_{j=0}^{\infty} (-1)^{j} \frac{\Gamma(\nu+1)(r^{2}/4)^{j}}{j!\Gamma(j+\nu+1)}.$$

Combining this representation with [GMO18, Proposition 2.2], one has that  $\rho_{\nu}$  is not positive definite when  $\nu < \frac{d}{2} - 1$ , showing that the Bessel Gaussian field does not exist in this case.

The statement for the critical case  $\nu=\frac{d}{2}-1$  immediately follows from (2.16) and (2.18), after observing that  $\rho_{\frac{d}{2}-1}=b_d$ .

For  $\nu>\frac{d}{2}-1$ , one can actually check that  $\mu_{\nu}$  given in the statement is the distribution of the square root of the Beta random variable  $\beta(\frac{d}{2},\nu-\frac{d}{2}+1)$ . Then, using Fubini and  $b_d=\rho_{\frac{d}{2}-1}$  we have

$$\int_{0}^{1} b_{d}(rt)\mu_{\nu}(dr) = \operatorname{const} \int_{0}^{1} \left( \sum_{k=0}^{\infty} \frac{(-1)^{k} \Gamma(d/2)}{\Gamma(k+d/2)k!} ((rt)^{2}/4)^{k} \right) \mu_{\nu}(dr)$$

$$= \operatorname{const} \sum_{k=0}^{\infty} \frac{(-1)^{k} \Gamma(d/2)}{\Gamma(k+d/2)k!} \left( \frac{t}{2} \right)^{2k} \int_{0}^{1} r^{2k} \mu_{\nu}(dr)$$

$$= \operatorname{const} \sum_{k=0}^{\infty} \frac{\Gamma(\nu+1)(-1)^{k}}{\Gamma(\nu+k+1)k!} \left( \frac{t}{2} \right)^{2k} = \rho_{\nu}(t),$$

where the second-last equality comes from the fact that the k-th moment of a random variable  $\beta(\frac{d}{2}, \nu - \frac{d}{2} + 1)$  is

$$\frac{\Gamma(d/2+k)\Gamma(\nu+1)}{\Gamma(d/2)\Gamma(\nu+k+1)}.$$

This means (see (2.16)) that  $\rho_{\nu}$  defines a Bessel Gaussian field when  $\nu > \frac{d}{2} - 1$  and that its spectral measure is  $\mu_{\nu}$  as defined in the statement.

The fact that any Bessel Gaussian field is continuous can be proved using standard results, see e.g. [AT07, Theorem 1.4.1].  $\Box$ 

Now let us apply our Theorem 2.1.2 to the Bessel Gaussian field with parameter  $\nu \geq \frac{d}{2} - 1$  and dimension  $d \geq 2$ .

**Corollary 2.6.2** (Spectral CLT applied to  $B^{\nu}$ ). Fix  $d \geq 2$  and  $\nu \geq \frac{d}{2} - 1$ . Let  $B^{\nu} = (B^{\nu}_x)_{x \in \mathbb{R}^d}$  be a Bessel Gaussian field on  $\mathbb{R}^d$ . Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be not odd and such that  $\mathbb{E}[\varphi(N)^2] < \infty$  with  $N \sim N(0,1)$ , let R be the Hermite rank of  $\varphi$  and consider  $Y_t$  defined by (2.3), where  $B = B^{\nu}$  and D is compact, Vol(D) > 0. Set  $m_t = \mathbb{E}[Y_t]$  and  $\sigma_t = \sqrt{\mathrm{Var}(Y_t)} > 0$ . If R = 2, assume that  $|\mathcal{F}[\mathbf{1}_D](x)| = O\left(\frac{1}{|x|^{d/2}}\right)$  as  $|x| \to \infty$ . Then,

if R=2 or  $R\geq 4$ , we have that  $\sigma_t^2\asymp t^dw_{R,t}$  and

$$\frac{Y_t - m_t}{\sigma_t} \stackrel{\text{law}}{\to} N(0, 1)$$
 as  $t \to \infty$ .

Proof. Using standard asymptotics for Bessel functions (see e.g. [Kra14]), one has

$$\rho_{\nu}(r) = c_{\nu} \frac{\cos(r - a_{\nu})}{r^{\nu + 1/2}} + O\left(\frac{1}{r^{\nu + 3/2}}\right) \quad \text{as } r \to \infty$$

for some constants  $c_{\nu}$  and  $a_{\nu}$ . Then, it follows that

$$\int_0^{2t} |\rho_{\nu}(r)|^q r^{d-1} dr = O\left(\int_1^{2t} \frac{dr}{r^{q\nu + \frac{q}{2} + 1 - d}}\right) \quad \text{as } t \to \infty.$$

Since  $\nu \geq \frac{d}{2}-1$ , when  $R \geq 5$  we have that  $|\rho|^q(r)r^{d-1}$  is integrable, and the result follows by Breuer Major Theorem 2.1.1. For R=2 and R=4, the result follows by a direct application of Theorem 2.1.2, since the negative moment of order d/R of  $\mu_{\nu}$  exists for every  $d \geq 2$  and  $R \geq 2$ .

**Remark 2.6.1.** For simplicity, in the previous statement we did not consider all the cases which can be solved applying our Theorem 2.1.2. For example, the spectral condition (2.20) for R=1 is not verified by  $\mu_{\nu}$  if  $\nu>\frac{d}{2}-1$ , but holds in the d-dimensional Berry case  $\nu=\frac{d}{2}-1$ . Moreover, relying on a slight variation of the proof of Proposition 2.5.1, it would have not been very difficult to also cover the non-critical cases where  $\nu\in[\frac{d}{2}-1,\frac{d}{2}-\frac{1}{2})$ . We skip the details for the sake of brevity.

# CHAPTER 3

# Fluctuations of polyspectra in spherical and Euclidean random wave models

This chapter contains the paper [GMT24], "Fluctuations of polyspectra in spherical and Euclidean random wave models", written in collaboration with Francesco Grotto and Anna Paola Todino. *Electron. Comm. Probab.* (2024), 29, 1-12.

#### 3.1 Introduction and Main Results

A Random Wave model (RW) can be defined as a random field on a Riemannian manifold given by a random linear combination of eigenfunctions of the Laplace-Beltrami operator at a fixed frequency, or within a small frequency bandwidth. The study of such stochastic objects dates back to the works of Berry and Zelditch [Ber77; Zel09] and it is now a well-established area of research, in which properties of nodal sets of RWs and the comparison with their deterministic counterpart have garnered significant focus (cf. [Wig23] for a recent survey). In homogeneous spaces such as  $\mathbb{R}^d$ , the hypersphere  $S^d$  or the hyperbolic space  $\mathbb{H}^d$  (cf. respectively [GP23; MN24; MP11] for specific discussions), RWs also appear naturally in the spectral decomposition of isotropic Gaussian random fields, that is Gaussian fields whose law (thus, its covariance function) is invariant under isometries of the underlying manifold, as detailed in [CL12].

In this note, we focus on Gaussian RWs on Euclidean space  $\mathbb{R}^d$  and on the (hyper)sphere  $S^d \subset \mathbb{R}^{d+1}$ . Our main result, Theorem 3.1.1, establishes asymptotics

for (variances of) polynomial transforms of those RWs. In particular, we complete the high-frequency asymptotic description of the so-called *polyspectra*, that is integrals of Hermite polynomials of the RWs on fixed space domains: the case of Hermite polynomials of large odd order was left as a conjecture on hyperspheres in [MR15; MW14] and was not discussed in works (such as [MN24; NPR19]) in the Euclidean case (see Remark 3.1.3 for a precise comparison with previous results). Polyspectra constitute the elementary objects in the Wiener chaos decomposition of functionals of Gaussian RWs, therefore their asymptotics play a crucial role in the Wiener chaos approach to limit theorems.

In order to prove Theorem 3.1.1 we will exploit a peculiar relation with uniform random walks on Euclidean space, and the well-established theory of those random processes. To some extent, this resembles a connection with short random walks considered in the case of random spherical harmonics in [MP08; MP10].

#### **Notation**

We regard the d-dimensional sphere  $S^d=\{x\in\mathbb{R}^{d+1}:|x|=1\}$  as an embedded manifold of codimension 1; we denote by  $\sigma_d$  the induced volume measure and by  $\omega_d=\sigma_d(S^d)=2\pi^{(d+1)/2}/\Gamma\left(\frac{d+1}{2}\right)$  the total volume. The geodesic distance on  $S^d$  is given by  $d(x,y)=\cos^{-1}(x\cdot y)$ ,  $x\cdot y$  denoting the canonical scalar product in  $\mathbb{R}^{d+1}$ .

We denote by  $B_d^E(x,R) \subset \mathbb{R}^d$  the ball of radius R in  $\mathbb{R}^d$  centered at the point x. When the context allows it, we lighten notation omitting dependencies, writing for instance  $B_R$  to denote a ball of radius R whose center can be chosen arbitrarily. The same applies to geodesic balls  $B_d^S(x,R) \subset S^d$ . The symbol  $\chi_A$  denotes the indicator function of a set A.

The symbol C denotes a *positive* constant, possibly differing in any of its occurrences, and depending only on eventual subscripts, as in  $C_{a,b}$ . Landau's O and o symbols have their usual meaning, with constants involved in upper and lower bounds depending again only on eventual subscripts, as in  $O_{a,b}(1)$ .

#### 3.1.1 Random Wave Models

*Berry's model*, that is the RW model on Euclidean space  $\mathbb{R}^d$ , is the centered Gaussian random field  $U_{\lambda}(x)$ ,  $\lambda > 0$ ,  $x \in \mathbb{R}^d$ , with covariance function

$$\mathbb{E}\left[U_{\lambda}(x)U_{\lambda}(y)\right] = j_d(\lambda|x - y|), \quad x, y \in \mathbb{R}^d,$$
(3.1)

where  $j_d$  is (a multiple of) the Fourier transform of the volume measure  $\sigma_{d-1}$  of  $S^{d-1}$ , regarded as a generalized function on the ambient space  $\mathbb{R}^d$ ,

$$j_d(|x|) = \frac{1}{\omega_{d-1}} \int_{S^{d-1}} e^{ix \cdot \theta} d\sigma_{d-1}(\theta) = \frac{\nu! 2^{\nu}}{(|x-y|)^{\nu}} J_{\nu}(|x-y|), \quad \nu = \nu_d \coloneqq \frac{d}{2} - 1, \quad (3.2)$$

 $J_{\nu}$  being the Bessel function of first kind and order  $\nu$ . In particular,  $j_2(r)=J_0(r)$  and  $j_3(r)=\mathrm{sin}(r)=\frac{\sin(r)}{r}$ . Notation  $\nu=d/2-1$  will remain valid throughout the paper.

The Gaussian field  $U_{\lambda}$  is named after M. Berry, who introduced it in order to describe local behavior of high frequency eigenstates in quantum billiards [Ber77;

Ber02]. It arises as a central limit of random linear combinations of planar waves on  $\mathbb{R}^d$  of the form

$$\frac{1}{\sqrt{N}} \sum_{i=1}^{N} \cos(\lambda x \cdot y_i + \varphi_i), \quad x \in \mathbb{R}^d, \quad N \to \infty,$$

with the  $y_i$ 's and  $\varphi_i$ 's being i.i.d. uniform random variables respectively in  $S^{d-1}$  and  $[0, 2\pi]$  (cf. [Ber02; NPR19]). Samples of  $U_{\lambda}$  are themselves smooth  $\lambda^2$ -eigenfunctions of the Laplace operator.

Random (Hyper)Spherical Harmonics, that is the RW model on the sphere  $S^d$ , consist in the centered Gaussian random field  $T_{\ell}(x)$ ,  $\ell \in \mathbb{N}$ ,  $x \in S^d$ , with covariance

$$\mathbb{E}\left[T_{\ell}(x)T_{\ell}(y)\right] = \mathcal{G}_{d,\ell}(\cos d(x,y)) = \binom{\ell+\nu}{\ell}^{-1} P_{\ell}^{\nu,\nu}(\cos d(x,y)) \quad x, y \in S^d, \quad (3.3)$$

where  $\mathcal{G}_{d,\ell}$  is the normalized Gegenbauer polynomial of degree  $\ell$  (see [Sze39, p. 4.7]), the right-hand side providing an alternative representation in terms of Jacobi polynomials. This random field can be regarded as the analog of Berry's model on  $\mathbb{R}^d$ . Indeed, the role of Euclidean planar waves is played on the sphere  $S^d$  by hyperspherical harmonics, that is Laplace-Beltrami eigenfunctions

$$\{Y_{\ell,m,d}\}_{m=1,\dots,\eta_{\ell,d}} \subset C^{\infty}(S^d), \ \ell \in \mathbb{N}, \quad \Delta_{S^d}Y_{\ell,m,d} + \ell(\ell+d-1)Y_{\ell,m,d} = 0,$$

forming a complete orthonormal basis of  $L^2(S^d, \sigma_d)$ , spanning at each fixed  $\ell \in \mathbb{N}$  a distinct eigenspace<sup>1</sup> of dimension  $\eta_{\ell,d} = \frac{2\ell + d - 1}{\ell} \binom{\ell + d - 2}{\ell - 1}$ . The Gaussian field  $T_\ell$  has the law of a random superposition of waves at a fixed wavenumber:

$$T_{\ell}(x) \sim \sum_{m=1}^{\eta_{\ell,d}} a_{\ell,m,d} Y_{\ell,m,d}(x), \quad x \in S^d,$$

with  $\{a_{\ell,m,d}\}_{m=1}^{\eta_{\ell,d}}$  being i.i.d. Gaussian variables N  $(0,\omega_d/\eta_{\ell,d})$ . The index  $\ell$  parametrizes the spectrum of Laplace-Beltrami operator on  $S^d$  through  $\ell(\ell+d-1)$ , thus in the limit  $\ell\to\infty$  it plays the same asymptotic role of  $\lambda\to\infty$  in the Euclidean setting, where  $\lambda^2$  is the eigenvalue of the wave  $x\mapsto e^{i\lambda x\cdot\theta}$ .

#### 3.1.2 Polyspectra

Relevant geometric functionals of  $U_{\lambda}$ , such as the (d-1)-dimensional volume of the nodal set  $\{U_{\lambda}=0\}$ , depend in general also on derivatives (i.e. the gradient) of the random field. However, already in the case of integral functionals of the form

$$\mathcal{F}_{\lambda} = \int_{D} F(U_{\lambda}(x)) dx, \quad F: \mathbb{R} \to \mathbb{R}, \ \lambda > 0, \ D \subset \mathbb{R}^{d}, \tag{3.4}$$

(and the analog for spherical RWs on measurable subsets  $D \subseteq S^d$ ) the description of the asymptotic behavior as  $\lambda \to \infty$  is not trivial, essentially because covariance

 $<sup>^1</sup>$ In fact, the arbitrary choice of an orthogonal basis of the eigenspace relative to  $\ell(\ell+d-1)$  is irrelevant in our scope.

functions do not satisfy integrability conditions allowing direct applications of basic results (e.g. [BM83]), and they oscillate between positive and negative values.

**Remark 3.1.1.** The covariance function (3.1) depends on  $\lambda |x-y|$ , therefore the high-frequency limit for functionals of the form (3.4) is equivalent to a (fixed-frequency) large domain limit. Indeed, the random fields  $B_{\lambda}(\cdot) \sim B_1(\lambda \cdot)$  have the same law, and thus integral functionals

$$\int_D F(U_{\lambda}(x))dx \sim \frac{1}{\lambda^d} \int_{\lambda D} F(U_1(x))dx,$$

are also equidistributed. There is no equivalent infinite-domain limit on the compact  $S^d$ .

Since the RWs we consider are Gaussian random fields, functionals of the form (3.4) can be studied considering their Wiener chaos expansion. Hermite polynomials,  $H_n(t) = (-1)^n \varphi^{(n)}(t)/\varphi(t)$ ,  $n \in \mathbb{N}$ , form an orthogonal basis of  $L^2(\mathbb{R}, \varphi(t)dt)$ ,  $\varphi$  being the p.d.f. of the standard Gaussian variable. If  $F \in L^2(\mathbb{R}; \varphi(x)dx)$ , the Wiener chaos decomposition of  $\mathcal{F}_{\lambda}$  is given by (cf. [NPR19])

$$\mathcal{F}_{\lambda} = \sum_{q=0}^{\infty} \frac{1}{q!} \left( \int_{\mathbb{R}} F(t) H_q(t) \varphi(t) dt \right) \left( \int_{D} H_q(U_{\lambda}(x)) dx \right).$$

The stochastic terms  $\int_D H_q(U_\lambda(x)) dx$  in the decomposition are called *polyspectra*: Fourth Moment Theorems –by now a standard tool in Gaussian analysis, *cf.* [NP12]–allow to deduce Central Limit Theorems for single polyspectra and functionals of the above form as  $\lambda \to \infty$ , provided that asymptotics of variance and fourth cumulants are available.

#### 3.1.3 Variance Asymptotics

For  $d, q \ge 2$ , R > 0, we will write:

$$\begin{split} V^E_{d,R}(q,\lambda) &= \operatorname{Var}\left(\int_{B^E(x_0,R)} H_q(U_\lambda(x)) dx\right), \quad \lambda > 0, \\ V^S_{d,R}(q,\ell) &= \operatorname{Var}\left(\int_{B^S(x_0,R)} H_q(T_\ell(x)) d\sigma_d(x)\right), \quad \ell \in \mathbb{N} \,. \end{split}$$

**Remark 3.1.2.** The case q=0 needs no discussion. As for q=1, it turns out that both  $V_{d,R}^E(1,\lambda)$  and  $V_{d,R}^S(1,\ell)$  have an oscillatory behavior as  $\lambda,\ell\to\infty$  and they can vanish (see respectively [MN24; Tod19] in the two geometrical settings). Anyways, when q=1 the polyspectrum is a Gaussian variable, and the study of its variance is simpler, so we will omit that case in our discussion.

Our main result is the following:

**Theorem 3.1.1.** Let  $d, q \ge 2$ , R > 0. There exist finite **positive** constants  $c_q^d$  such that:

• (Euclidean) as  $\lambda \to \infty$ ,

$$V^E_{d,R}(q,\lambda) = c^d_q q! \omega_d \omega_{d-1} R^d (1+o_{q,d}(1)) \cdot \begin{cases} \lambda^{1-d} & q=2 \\ \lambda^{-2} \log(\lambda) & q=4, d=2 \\ \lambda^{-d} & \textit{all other } d \geq 2, q \geq 3 \end{cases},$$

• (Hyperspherical) as  $\ell \to \infty$ , if  $R \in (0, \pi)$ ,

$$V_{d,R}^S(q,\ell) = c_q^d q! \omega_{d-1} \sigma_d(B_R^S) (1 + o_{q,d}(1)) \cdot \begin{cases} \ell^{1-d} & q = 2, \\ \ell^{-2} \log(\ell) & q = 4, d = 2, \\ \ell^{-d} & \textit{all other } d \geq 2, q \geq 3, \end{cases}$$

and when  $R = \pi$  (that is in the case of polyspectra obtained integrating over the whole  $S^d$ ),

$$V_{d,\pi}^S(q,\ell) = 2c_q^d q! \omega_{d-1} \omega_d (1+o_{q,d}(1)) \cdot \begin{cases} \ell^{1-d} & q=2, \\ \ell^{-2} \log(\ell) & q=4, d=2, \\ 0 & q, \ell \text{ both odd,} \\ \ell^{-d} & \text{all other } d \geq 2, q \geq 3. \end{cases}$$

The proof of Theorem 3.1.1 is the content of the forthcoming Section 3.3.

**Remark 3.1.3.** Let us clarify the relation of Theorem 3.1.1 with the previous literature.

- (Euclidean case) [NPR19] proved the result in dimension d=2 for even  $q\geq 2$  and upper bounds for odd  $q\geq 3$ , for integration domains  $D\subset \mathbb{R}^2$  with boundary of class  $C^1$ ; [Not21] generalizes the discussion to arbitrary  $d\geq 2$  and even  $q\geq 2$ ; for  $d\geq 2$  and even  $q\geq 2$  [Mai24] considers compact domains  $D\subset \mathbb{R}^d$  with finite perimeter, treating the larger class of weakly stationary random fields and establishing asymptotics for covariances of functionals;
- (case of  $S^2$ ) [MW11; MW14] proved the above asymptotics on the whole  $S^2$  (case  $R=\pi$ ) leaving as an open problem the positivity of  $c_q^2$  for odd  $q\geq 5$ ; [Tod19] discussed the case  $R<\pi$  for q=2 and provided upper bounds for larger  $q\geq 3$ ;
- (Hyperspherical case) concerning polyspectra over the whole  $S^d$  ( $R=\pi$ ), in larger dimension the above asymptotic was established for all even  $q \geq 2$  in [MR15] and for q=3 [Ros19], leaving positivity of  $c_q^d$  for odd  $q \geq 5$  as a conjecture (see [MR15, p. 2386]).

The remainder of this note is organized as follows. Section 3.2 details the relation between variance of polyspectra and the density of uniform random walks, and introduces results on the latter that we can exploit to control the former. We prove Theorem 3.1.1 in Section 3.3.

#### 3.2 Pearson's Random Walk

Let  $X_n^d$ ,  $n=1,2,\ldots$  denote the uniform random walk on  $\mathbb{R}^d$ , that is

$$X_n^d = U_1^d + \dots + U_n^d,$$

with  $(U_k^d)_{k\in\mathbb{N}}$  a sequence of i.i.d. uniform random variables on  $S^{d-1}\subset\mathbb{R}^d$ . The stochastic process  $X_n^d$  is known as *Pearson's Random Walk*, because of a very early mention in [Pea05].

Much is known on Pearson's random walk: in particular, its distribution and moments for a small number of steps n are of interest in Number Theory because of their relation with Mahler measures. We can thus rely on the series of works [BSV16; BSW13; Bor+11; Bor+12], to which we refer for a comprehensive background on the matter and the related literature.

The basic result on the distribution of  $X_n$  for small n is a representation formula for the p.d.f. dating back to the seminal work [Klu06].

**Lemma 3.2.1** (Kluyver's Formula). For  $n \geq 2$  the law  $\mu_n^d$  of  $X_n^d$  is absolutely continuous with respect to Lebesgue's measure on  $\mathbb{R}^d$ , and its density is a radial function, i.e.  $\mu_q^d(dx) = f_q^d(\|x\|) dx$  with  $f_q^d(\|\cdot\|) \in L^1(\mathbb{R}^d)$ .

The (Euclidean) distance from the origin  $\|X_n^d\|$  is absolutely continuous with respect to Lebesgue's measure on  $(0, \infty)$ , and its density is given by

$$\rho_n^d(r) = \frac{1}{(\nu!)^2 4^{\nu}} \int_0^\infty (tr)^{2\nu+1} j_d(tr) j_d(t)^n dt, \quad r > 0, \quad \nu = \frac{d}{2} - 1.$$

To be precise, the integral in Kluyver's formula has to be regarded as an improper integral that either converges to a non-negative number (for large enough d,n even absolutely) or diverges to  $+\infty$  (see Remark 3.2.2 below). We refer to [BSV16, Theorem 2.1] for a proof of Lemma 3.2.1 and a thorough general discussion. Let us note that  $f_q^d$  and  $\rho_q^d$  are related by

$$\omega_{d-1} \int_0^s r^{d-1} f_q^d(r) dr = \int_{B_s} \mu_q^d(dx) = \mathbb{P}\left(X_q^d \in B_s\right) = \mathbb{P}\left(\|X_q^d\| \le s\right) = \int_0^s \rho_q^d(r) dr.$$
(3.5)

#### 3.2.1 Random Walks and Polyspectra

Before we outline further properties of Pearson's random walk to be used in the last Section, let us explain heuristically how one is led to consider this apparently unrelated object in studying RW's polyspectra. The following computation concerns the Euclidean case only, and has no direct analog in the case of hyperspherical RWs. Nevertheless, in the high-frequency limiting regime the Euclidean and hyperspherical RW models turn out to exhibit the same behavior. It holds

$$V_{d,R}^{E}(q,\lambda) = q! \int_{B_R} \int_{B_R} j_d(\lambda|x-y|)^q dx dy,$$

since  $j_d(\lambda|x-y|)$  is the covariance function of  $U_{\lambda}$ , and if X,Y are two centered jointly Gaussian variables  $\mathbb{E}[H_q(X)H_q(Y)] = q!\mathbb{E}[XY]^q$ . By (3.2), the function

$$j_d(|x|) = \mathcal{F}[\mu_1^d](x) := \int_{\mathbb{R}^d} e^{ix \cdot y} \mu_1^d(dy)$$
 (3.6)

can be represented as the Fourier transform of  $\mu_1^d$ . Hence we can write

$$\begin{split} V_{d,R}^{E}(q,\lambda) &= q! \int_{\mathbb{R}^d} \int_{\mathbb{R}^d} \frac{\chi_{B_{\lambda R}}(x)\chi_{B_{\lambda R}}(y)}{\lambda^{2d}} \mathcal{F}[\mu_1^d](x-y)^q dx dy \\ &= q! \int_{\mathbb{R}^d} \int_{\mathbb{R}^d} \frac{\chi_{B_{\lambda R}}(x)\chi_{x+B_{\lambda R}}(z)}{\lambda^{2d}} \mathcal{F}[\underbrace{\mu_1^d * \cdots * \mu_1^d}_{q \text{ times}}](z) dx dz, \end{split}$$

where we recognize the q-fold convolution of  $\mu_1^d$  with itself as the law  $\mu_q^d$  of  $X_q^d$ . As  $\lambda \to \infty$ , indicator functions in the last displayed equation converge pointwise to 1 on  $\mathbb{R}^d$ , therefore we expect —at least intuitively— that the limiting integral

$$\int_{\mathbb{R}^d} \mathcal{F}[\mu_q^d](z) dz = f_q^d(0)$$

plays a relevant role in determining the asymptotic behavior of  $V_{d,R}^E(q,\lambda)$ . By (3.5) and Kluyver's formula we can also write

$$\omega_{d-1} f_q^d(0) = \lim_{r \to 0} \frac{\rho_q^d(r)}{r^{d-1}} = \frac{1}{(\nu!)^2 4^{\nu}} \int_0^\infty j_d(t)^q t^{d-1} dt = \rho_{q-1}^d(1),$$

therefore the crucial objects appear to be the integrals

$$I_q^d := \int_0^\infty j_d(t)^q t^{d-1} dt = \frac{\rho_{q-1}^d(1)}{(\nu!)^2 4^{\nu}},\tag{3.7}$$

and these numbers can a priori take any value in  $[0, \infty]$ .

In fact,  $I_q^d$  coincides with the coefficient  $c_q^d$  appearing in Theorem 3.1.1 in the cases  $d \geq 2, q \geq 3$  except d=2, q=4. Before we prove it in the last Section, we need to discuss finiteness and positivity of the  $I_q^{d}$ 's, that is the task for which we need to resort to previous results on Pearson's random walk.

#### 3.2.2 Density of short random walks

We report an explicit expression for the densities  $\rho_2^d$  and a recursion formula allowing to treat the density at higher n. Representations for  $\rho_n^d$ , n=3,4, in terms of hypergeometric functions are available in [BSV16, Equations (74), (79)]. We recall that  $\nu=d/2-1$  in all its occurences.

**Lemma 3.2.2.** [BSV16, Lemma 4.1] For  $d \ge 2$  and 0 < r < 2 it holds

$$\rho_2^d(r) = \frac{2}{\pi \binom{2\nu}{\nu}} r^{2\nu} (4 - r^2)^{\nu - 1/2} \chi_{(0,2)}(r).$$

**Lemma 3.2.3.** [BSV16, Theorem 2.9] Let  $d, n \geq 2$ , r > 0 and define  $\psi_n^d(r) =$ 

 $\rho_n^d(r)/r^{d-1}$ . It holds, for  $n \geq 3$ ,

$$\psi_n^d(r) = \frac{(\nu!)^2 4^{\nu}}{\pi (2\nu)!} \int_{-1}^1 \psi_{n-1}^d \left(\sqrt{1 + 2sr + r^2}\right) (1 - s^2)^{\nu - 1/2} ds, \quad r \in (0, n).$$
 (3.8)

**Corollary 3.2.4.** For any  $d \ge 2$  and  $n \ge 3$ ,  $\psi_n^d$  is strictly positive (possibly infinite) on (0,n).

*Proof.* By induction on n, the case n=2 follows by definition of  $\psi_2^d$  and Lemma 3.2.2. Assume now that  $\psi_{n-1}^d(r)>0$  for  $r\in(0,n-1)$  (by definition it vanishes for r>n-1); by (3.8) if the open set

$$\{s \in (-1,1): \psi_{n-1}^d \left(\sqrt{1+2sr+r^2}\right) > 0\} = \{s \in (-1,1): 1+2sr+r^2 < (n-1)^2\}$$

is not empty, then  $\psi_n^d(r) > 0$ . This is in fact the case, because  $p(s) = 1 + 2sr + r^2$  is increasing and continuous and  $p(-1) = 1 - 2r + r^2 = (r-1)^2 < (n-1)^2$  for r < n.

Moving back to the specific integrals  $I_q^d$  we are interested in, we combine the latter Corollary and the following approximation result on Bessel functions to show that they are both positive and finite.

**Lemma 3.2.5.** [Kra14, Theorem 4, Equation 6] Let  $d \ge 2$ . There exist constants  $\varphi_d > 0$  such that

$$0 < \sup_{r \ge 0} r^{3/2} \left| J_{\nu}(r) - \sqrt{\frac{2}{\pi r}} \cos(r - \varphi_d) \right| < \infty.$$
 (3.9)

In particular by (3.2),

$$j_d(r) = C_d r^{-(d-1)/2} \cos(r - \varphi_d) + O_d(r^{-(d+1)/2}), \quad r \to \infty.$$
 (3.10)

**Lemma 3.2.6.** For any  $d \geq 2, q \geq 3$  except the single case  $d = 2, q = 4, I_q^d \in (0, \infty)$  is positive and finite.

*Proof.* By Corollary 3.2.4, for all  $d \ge 2, q \ge 3$  it holds

$$I_n^d = \frac{\rho_{n-1}^d(1)}{(\nu!)^2 4^{\nu}} = \frac{\psi_{n-1}^d(1)}{(\nu!)^2 4^{\nu}} > 0.$$

Concerning finiteness, by (3.10) and (3.7) we have

$$I_q^d \le C_{d,q} \int_1^\infty r^{(d-1)(1-q/2)} dr;$$

in particular, the integral in (3.7) is absolutely convergent if  $q>\frac{d}{d-1}$ . We are left to discuss the cases d=2,3,q=3, since we are excluding d=2,q=4 (see Remark 3.2.2 below). A classical result on Bessel functions (cf. [Wat95, Section 13.46]) implies  $I_3^2=\frac{2}{\pi\sqrt{3}}$ , and  $I_3^3=\int_0^\infty t^{-1}\sin(t)^3dt=\pi/4$  is easily computed reducing it to Dirichlet's integral.

Remark 3.2.1. The following exact expressions

$$I_3^d = \frac{2}{\pi\sqrt{3}} \cdot \frac{12^{\nu}(\nu!)^4}{(2\nu)!}, \ d \ge 2, \qquad I_5^2 = \frac{\sqrt{5}\Gamma\left(\frac{1}{15}\right)\Gamma\left(\frac{2}{15}\right)\Gamma\left(\frac{4}{15}\right)\Gamma\left(\frac{8}{15}\right)}{40\pi^4},$$

can be deduced respectively from [BSV16, Proposition 4.3] and [Bor+12, Theorem 5.1] (concerning the asymptotics of  $\rho_n^d(r)$  as  $r \to 0$ ). To the best of our knowledge, no representation is available in the general case.

Remark 3.2.2. In the case d=2, q=4 excluded from Lemma 3.2.6, one can prove by means of Lemma 3.2.5 that  $I_4^2=\int_0^\infty J_0(t)^4tdt=+\infty$ , coherently with the fact that the density  $\rho_2^3(1)=I_2^4=+\infty$  is not finite. This is actually the only case of a density  $\rho_n^d$ ,  $d,n\geq 2$ , taking value  $+\infty$  at some point, as for larger n the regularizing effect of the multiple convolution defining the law of Pearson's random walk prevails (cf. again [BSV16]).

### 3.3 Fluctuation Asymptotics and Integration Domains

#### 3.3.1 Variance of Euclidean Polyspectra

We first rewrite the variances we are interested in as integrals of a single variable.

**Lemma 3.3.1.** Let R > 0,  $B_R = B^E(x_0, R) \subseteq \mathbb{R}^d$  a ball of radius R centered at a (fixed, arbitrary) point  $x_0 \in \mathbb{R}^d$ , and  $f \in L^1_{loc}(\mathbb{R})$ ; it holds

$$\int_{B_R} \int_{B_R} f(|x-y|) dx dy = \int_0^{2R} f(r) W_{d,R}(r) r^{d-1} dr,$$
(3.11)

where, denoting by  $m_d$  Lebesgue's measure on  $\mathbb{R}^d$ ,

$$W_{d,R}(r) = \omega_{d-1} m_d(B(x,R) \cap B(y,R)), \quad |x-y| = r,$$
(3.12)

whose value does not depend on the choice of points  $x, y \in \mathbb{R}^d$ .

The proof is a consequence of the fact that translations on  $\mathbb{R}^d$  are isometries of the whole space: it holds

$$\int_{\mathbb{R}^d} \int_{\mathbb{R}^d} \chi_{B_R}(x) \chi_{B_R}(y) f(|x-y|) dx dy = \int_{\mathbb{R}^d} \int_{\mathbb{R}^d} \chi_{B_R}(x) \chi_{B_R+z}(x) f(|z|) dx dz,$$

from which (3.11) follows moving to polar coordinates for z. It is easy to observe that  $W_{d,R}:[0,\infty)\to[0,\infty)$  is supported by [0,2R], it is differentiable on  $(0,\infty)\setminus\{2R\}$  and  $W_{d,R}=\omega_{d-1}m_d(B_R), W_{d,R}(2R)=0$ .

*Proof of Theorem 3.1.1, Euclidean part.* In sight of Lemma 3.3.1 it holds,

$$V_{d,R}^{E}(q,\lambda) = q! \int_{B_R} \int_{B_R} j_d(\lambda|x-y|)^q dx dy = q! \int_0^{2R} j_d(\lambda r)^q W_{d,R}(r) r^{d-1} dr.$$
 (3.13)

For q = 2, (3.10) implies that, as  $\lambda \to \infty$ ,

$$V_{d,R}^{E}(2,\lambda) = C_{d} \int_{0}^{2R} \frac{\cos(\lambda r - \varphi_{d})^{2}}{\lambda^{d-1}} W_{d,R}(r) dr + O_{d,R}\left(\frac{1}{\lambda^{d}}\right) = C_{d,R} \frac{1 + o_{d,R}(1)}{\lambda^{d-1}},$$

the second step following by expanding  $2\cos(\theta)^2 = 1 + \cos(2\theta)$  and applying Riemann-Lebesgue lemma to oscillatory terms.

We also treat separately the case d=2, q=4, since it involves a logarithmic discrepancy with the general case  $q\geq 3$ . (3.10) implies that, as  $\lambda\to\infty$ ,

$$V_{2,R}^{E}(4,\lambda) = C_d \int_0^{2R} \frac{\cos(\lambda r - \varphi_d)^4}{\lambda^2 r} W_{d,R}(r) dr + O_R\left(\frac{1}{\lambda^3}\right) = C_R \frac{\log(\lambda) + o_R(1)}{\lambda^2},$$

where, as in the case q=2, the last equality follows by  $8\cos(\theta)^4=3+4\cos(2\theta)+\cos(4\theta)$  and Riemann-Lebesgue lemma.

Finally, we consider the general case  $q \ge 3$ , excluding d = 2, q = 4. By (3.13),

$$\begin{split} V_{d,R}^{E}(q,\lambda) &= q! \int_{0}^{2R} j_{d}(\lambda r)^{q} r^{d-1} \left( \int_{r}^{2R} -W_{d,R}'(s) ds \right) dr \\ &= -q! \int_{0}^{2R} W_{d,R}'(s) \left( \int_{0}^{s} j_{d}(\lambda r)^{q} r^{d-1} dr \right) ds \\ &= \frac{q!}{\lambda^{d}} \int_{0}^{2R} -W_{d,R}'(s) \left( \int_{0}^{s\lambda} j_{d}(r)^{q} r^{d-1} dr \right) ds, \end{split}$$

integrating by parts in the second step. Since  $|W'_{d,R}(r)|$  is bounded and supported by [0,2R], the thesis follows by dominated convergence and Lemma 3.2.6 (implying convergence of the inner integral on the right-hand side as  $\lambda \to \infty$ ).

#### 3.3.2 Variance of Hyperspherical Polyspectra

The forthcoming computations are in fact close analogs of the ones in the previous paragraph, only in a perhaps less familiar geometrical setting. The change of variables of Lemma 3.3.1 takes on  $S^d$  the following form, with flat translations of  $\mathbb{R}^d$  being replaced by isometries of  $S^d$ .

**Lemma 3.3.2.** Let  $R \in [0, \pi]$ ,  $B_R = B^S(x_0, R) \subseteq S^d$  a ball of radius R centered at a (fixed, arbitrary) point  $x_0 \in S^d$ , and  $f \in L^1(\mathbb{R})$ ; it holds

$$\int_{B_R} \int_{B_R} f(d(x,y)) d\sigma_d(x) d\sigma_d(y) = \int_0^{\pi} f(r) \tilde{W}_{d,R}(r) \sin(r)^{d-1} dr,$$
 (3.14)

where

$$\tilde{W}_{d,R}(r) = \omega_{d-1}\sigma_d(B(x,R) \cap B(y,R)), \quad d(x,y) = r,$$
 (3.15)

whose value does not depend on the choice of points  $x, y \in S^d$ .

The proof is conceptually analogous to the one of Lemma 3.3.1. First, let us recall that the isometry group of  $S^d$  can be identified with that of linear orthogonal transformations of the ambient  $R^{d+1}$ , and it acts transitively on  $S^d$ . The volume  $\sigma_d$  and the spherical geodesic distance are invariant under isometries, that is  $I_{\#}\sigma_d =$ 

 $\sigma_d \circ I = \sigma_d$  and d(a,b) = d(I(a),I(b)) for any isometry  $I:S^d \to S^d$  and points  $a,b \in S^d$ .

*Proof.* We first observe that for any isometry  $I: S^d \to S^d$  it holds

$$\begin{split} \sigma_d(B(Ix,R) \cap B(Iy,R)) &= \sigma_d \left\{ z : d(Ix,z), d(Iy,z) \leq R \right\} \\ &= \sigma_d \left\{ z : d(x,I^{-1}z), d(y,I^{-1}z) \leq R \right\} = \sigma_d \left\{ Iw : d(x,w), d(y,w) \leq R \right\} \\ &= I_\#^{-1} \sigma_d \left\{ w : d(x,w), d(y,w) \leq R \right\} = \sigma_d(B(x,R) \cap B(y,R)), \end{split}$$

therefore the right-hand side only depends on d(x,y), as it is a function of x,y invariant under isometries. In particular,  $W_{d,R}(r)$  in (3.15) is well-defined.

For any two points  $x, y \in S^d$  we can choose isometries  $I_x, I_y : S^d \to S^d$  such that  $I_x x_0 = x$  and  $I_y x_0 = y$ . Letting  $z = I_y I_x^{-1} x_0$ , by Fubini-Tonelli Theorem it holds

$$\begin{split} \int_{S^d} \int_{S^d} \chi \left\{ x \in B_R \right\} \chi \left\{ y \in B_R \right\} f(d(x,y)) d\sigma_d(x) d\sigma_d(y) \\ &= \int_{S^d} d\sigma_d(z) f(d(x_0,z)) \int_{S^d} d\sigma_d(x) \chi \left\{ x \in B_R \right\} \chi \left\{ I_x z \in B_R \right\}, \end{split}$$

where the inner integral can be rewritten as follows: since  $I_x^{-1}x_0 = I_x^{-2}x$ ,

$$\int_{S^d} \chi \left\{ x \in B_R \right\} \chi \left\{ I_x z \in B_R \right\} d\sigma_d(x) = \sigma_d \left\{ x : d(x, x_0), d(I_x z, x_0) \le R \right\}$$

$$= I_\#^2 \sigma_d \left\{ x : d(x_0, I_x^{-1} x_0), d(z, I_x^{-1} x_0) \le R \right\}$$

$$= \sigma_d \left\{ I_x^{-2} x : d(x_0, I_x^{-1} x_0), d(z, I_x^{-1} x_0) \le R \right\}$$

$$= \sigma_d \left\{ w : d(x_0, w), d(z, w) \le R \right\},$$

thus in particular it only depends on  $d(x_0, z)$ , thanks to the argument at the beginning of the proof. Since the volume element of  $S^d$  reduces to

$$\int_{S^d} h(d(x_0, x)) d\sigma_d(x) = \omega_{d-1} \int_0^{\pi} h(r) \sin(r)^{d-1} dr, \quad h \in L^1([0, \pi]),$$

for radial functions, combining the above equations concludes the proof.  $\Box$ 

The function  $\tilde{W}_{d,R}:[0,\pi]\to[0,\infty)$  is differentiable, and attains its maximum value in  $\tilde{W}_{d,R}(0)=\omega_{d-1}\sigma_d(B(x_0,R))$  and minimum in  $\tilde{W}_{d,R}(\pi)$  (which, unlike the Euclidean  $W_{d,R}(2R)$ , does not vanish in general). We can now proceed to complete the proof of Theorem 3.1.1 by reducing ourselves to the computations performed in the Euclidean case thanks to the following:

**Lemma 3.3.3** (Hilb's Asymptotic Formula). [Sze39, Theorem 8.21.12] Let  $d \geq 2$ ,  $\ell \in \mathbb{N}$ . Given  $a \in (0, \pi)$ , for all  $\theta \in (0, a)$  it holds:

$$(\sin \theta)^{\nu} \mathcal{G}_{d,\ell}(\cos \theta) = \frac{2^{\nu}}{\binom{\ell+\nu}{\ell}} \left( \frac{\Gamma(\ell+d/2)}{L^{\nu}\ell!} \left( \frac{\theta}{\sin \theta} \right)^{1/2} J_{\nu}(L\theta) + \delta(\theta) \right), \tag{3.16}$$

where  $L = \ell + \frac{d-1}{2}$  and  $\delta(\theta) \leq C_{d,a}(\chi_{(0,1/\ell)}(\theta)\theta^{\nu+2}\ell^{\nu} + \chi_{(1/\ell,a)}(\theta)\sqrt{\theta}\ell^{-3/2}).$ 

**Corollary 3.3.4.** Let  $d \geq 2$ ,  $\ell \in \mathbb{N}$ . For all  $a \in (0, \pi)$  it holds, as  $\ell \to \infty$ ,

$$\int_{0}^{a} \mathcal{G}_{d,\ell}(\cos r)^{q} (\sin r)^{d-1} dr = (1+o(1)) \begin{cases} C_{d} \ell^{1-d} & q=2 \\ C \log(\ell)/\ell^{2} & q=4, d=2 \\ I_{q}^{d}/\ell^{d} & \text{all other } d \geq 2, q \geq 3 \end{cases}$$
(3.17)

with  $I_q^d$  being as in Lemma 3.2.6 and  $C, C_d > 0$  positive constants.

**Remark 3.3.1.** The limit case  $a=\pi$  can be reduced to the one with  $a=\pi/2$  splitting the integration domain in half and exploiting the following symmetry property: for all  $d \geq 2$ ,

$$\mathcal{G}_{d,\ell}(t) = (-1)^{\ell} \mathcal{G}_{d,\ell}(-t), \quad t \in \mathbb{R}, \ell \in \mathbb{N}.$$

In particular, the integral in Corollary 3.3.4 identically vanishes if  $a=\pi$  and both  $q,\ell$  are odd. If this is not the case, then the thesis of Corollary 3.3.4 holds with an additional factor 2 on the right-hand side.

Corollary 3.3.4 actually collects computations already appeared in [MR15; MW11; Wig10]. Because of this and the similarity with computations in the previous paragraph, we only present a sketch of the proof.

*Proof.* We first observe that as  $\ell \to \infty$ ,

$$\frac{\Gamma(\ell + d/2)}{L^{\nu}\ell!} = 1 + o(1), \quad \binom{\ell + \nu}{\ell} = \frac{\ell^{\nu}}{\nu!} (1 + o(1)). \tag{3.18}$$

When q = 2, Lemma 3.3.3 and (3.18) imply

$$\begin{split} \int_0^a \mathcal{G}_{d,\ell}(\cos r)^2 (\sin r)^{d-1} dr &= \frac{4^{\nu} (\nu !)^2}{\ell^{2\nu}} (1+o(1)) \int_0^a J_{\nu}(Lr)^2 r dr \\ &\quad + O\left(\frac{1}{\ell^{d-2}} \int_0^a |J_{\nu}(Lr)| \delta(r) \sin(r) dr\right). \end{split}$$

The thesis now follows by applying Lemma 3.2.5 in the same fashion of the Euclidean case (the remainder term on the right-hand side is proved to be  $O(\ell^{-d})$ ). The analysis of the case d=2, q=4 is similar, we refer again to [Wig10] for full details.

As for the remaining cases, by Lemma 3.3.3, (3.18) and the definition of  $j_d$  in (3.2),

$$\int_0^a \mathcal{G}_{d,\ell}(\cos r)^q (\sin r)^{d-1} dr = (1+o(1)) \int_0^a \left(\frac{r}{\sin r}\right)^{q\nu+q/2-(d-1)} j_d(Lr)^q r^{d-1} dr + O\left(\frac{1}{\ell^{q\nu}} \int_0^a (\sin r)^{-q\nu} |J_{\nu}(Lr)|^{q-1} \delta(r) (\sin r)^{d-1} dr\right).$$

The remainder term can be shown to be  $O(\ell^{-(d+2)\wedge(q(d/2-1/2)-1)})=o(\ell^{-d})$  by means of Lemma 3.2.5 (changing variables and splitting the integration domain similarly to the previous case). The thesis now follows if

$$\int_0^a \left(\frac{r}{\sin r}\right)^{q\nu + q/2 - (d-1)} j_d(Lr)^q r^{d-1} dr = (1 + o(1)) \int_0^\infty j_d(r)^q r^{d-1} dr, \quad \ell \to \infty,$$

which can be proved with the same computation exploiting integration by parts we used at the end of the proof of Theorem 3.1.1, Euclidean case, that is replacing  $W_{d,R}$  of that computation with  $\chi_{[0,a]}(r)(r/\sin r)^{q\nu+q/2-(d-1)}$ .

*Proof of Theorem 3.1.1, Hyperspherical part.* Let  $R \in (0, \pi)$  (the case  $R = \pi$  is dealt with similarly, by Remark 3.3.1). By Lemma 3.3.2,

$$V_{d,R}^{S}(q,\ell) = \operatorname{Var}\left(\int_{B_R} H_q(T_\ell(x)) d\sigma_d(x)\right)$$
$$= q! \int_{B_R} \int_{B_R} \mathcal{G}_{d,\ell}(x,y)^q d\sigma_d(x) d\sigma_d(y) = q! \int_0^{\pi} \mathcal{G}_{d,\ell}(\cos r)^q \sin(r)^{d-1} \tilde{W}_{d,R}(r) dr.$$

Integrating by parts as in Section 3.3.1 we rewrite:

$$\int_{0}^{\pi} \mathcal{G}_{d,\ell}(\cos r)^{q} (\sin r)^{d-1} \tilde{W}_{d,R}(r) dr = \tilde{W}_{d,R}(\pi) \int_{0}^{\pi} \mathcal{G}_{d,\ell}(\cos r)^{q} (\sin r)^{d-1} dr - \int_{0}^{\pi} \left( \int_{0}^{r} \mathcal{G}_{d,\ell}(\cos s)^{q} (\sin s)^{d-1} ds \right) \tilde{W}'_{d,R}(r) dr.$$

The thesis now follows from Corollary 3.3.4. We observe in particular that when we are not in the cases q=2 nor in the one d=2, q=4, by Corollary 3.3.4 and dominated convergence we have

$$\ell^{d} \int_{0}^{\pi} \mathcal{G}_{d,\ell}(\cos r)^{q} (\sin r)^{d-1} \tilde{W}_{d,R}(r) dr$$

$$\xrightarrow{\ell \to \infty} I_{q}^{d} \tilde{W}_{d,R}(\pi) - I_{q}^{d} \int_{0}^{\pi} \tilde{W}_{d,R}(r)' dr = I_{q}^{d} \tilde{W}_{d,R}(0). \quad \Box$$

# Asymptotic covariances for functionals of weakly stationary random fields

This chapter contains the paper [Mai24], "Asymptotic covariances for functionals of weakly stationary random fields". *Stoch. Proc. Appl.* (2024), 170, 104297.

#### 4.1 Introduction

Fix a probability space  $(\Omega, \mathcal{F}, \mathbb{P})$ , a dimension  $d \geq 1$  and let  $A = (A_x)_{x \in \mathbb{R}^d}$  be a **measurable** random field on  $(\Omega, \mathcal{F}, \mathbb{P})$ , that is,

$$A: (\Omega, \mathcal{F}) \times (\mathbb{R}^d, \mathcal{B}(\mathbb{R}^d)) \longrightarrow (\mathbb{R}, \mathcal{B}(\mathbb{R})), \qquad (\omega, x) \longmapsto A_x(\omega),$$

is a measurable function, where  $\mathcal{B}(\mathbb{R}^d)$  denotes the Borel sigma-algebra on  $\mathbb{R}^d$ . In addition, assume that A is **weakly stationary**, i.e.

$$\mathbb{E}[A_x] = \mathbb{E}[A_y], \quad \operatorname{Cov}(A_x, A_y) = K(x - y), \quad \forall x, y \in \mathbb{R}^d,$$

with **covariance function**  $K: \mathbb{R}^d \to \mathbb{R}$ . Note that under the above assumptions K is measurable and bounded, hence locally integrable. Therefore, we can define the **integral covariance function**  $w: \mathbb{R}_+ \to \mathbb{R}$  as

$$w_t := \int_{\{|z| \le t\}} K(z)dz, \qquad t > 0,$$
 (4.1)

where  $|\cdot|$  denotes the Euclidean norm on  $\mathbb{R}^d$ .

In this paper, we will focus on **functionals** of A of the form

$$\left(\int_{tD} A_x dx\right)_{D \in \mathcal{D}}, \quad \text{as } t \to \infty,$$
 (4.2)

where  $\mathcal{D}$  is a suitable class of compact sets in  $\mathbb{R}^d$ . The well-posedness of (4.2) under the above assumptions is ensured by Proposition 4.2.1.

The asymptotic behavior of (4.2) has been extensively investigated in probability, and plays an important role in many applications. In Statistics, for instance, functionals of the form  $\int_{tD} A_x dx$  are often involved in parameter estimation, and tD plays the role of an increasing observation window, see e.g. [DOV22; MY22].

Moreover, note that (4.2) may represent non-linear functionals of stationary Gaussian fields, that will be discussed in more detail in Section 4.4. The latter corresponds to the choice  $A_x = \varphi(B_x)$ , where  $B = (B_x)_{x \in \mathbb{R}^d}$  is a stationary Gaussian field and  $\varphi : \mathbb{R} \to \mathbb{R}$  is a function which varies depending on the framework. Just to mention two examples (among many),  $\varphi$  could be a polynomial in view of statistical applications, see e.g. [RST12; TV09], or an indicator function if one is interested in the geometry (and in particular the excursion volumes) of B, see e.g. [LRM23; LO14; MN24].

The goal of this paper will be to give the minimal assumptions on K and  $\mathcal{D}$  in order to compute exactly the **asymptotic covariances** of (4.2), i.e. the limit

$$\lim_{t \to \infty} \text{Cov}\left(\frac{\int_{tD} A_x dx}{r_t}, \frac{\int_{tL} A_y dy}{r_t}\right), \qquad D, L \in \mathcal{D},$$
(4.3)

where  $r_t \to \infty$  is chosen so that the limit (4.3) exists finite (and not identically equal to 0 for all  $D, L \in \mathcal{D}$ ) for every  $D, L \in \mathcal{D}$ . The problem (4.3) often arises when studying the fluctuations of (4.2), in particular in the case of non-linear functionals of stationary Gaussian fields (see Section 4.4). Indeed, the first step to prove limit theorems for (4.2), i.e.

$$\frac{\int_{tD} (A_x - \mathbb{E}[A_x]) dx}{r_t} \stackrel{\mathrm{d}}{\to} Z(D), \quad \text{as } t \to \infty,$$

where Z(D) is a limiting random variable, is usually to study the asymptotic variance of  $\int_{tD} A_x dx$ , which corresponds to (4.3) when D=L. Moreover, to extend such results to a multi-dimensional setting, i.e.

$$\left(\frac{\int_{tD_1}(A_x-\mathbb{E}[A_x])dx}{r_t},\ldots,\frac{\int_{tD_n}(A_x-\mathbb{E}[A_x])dx}{r_t}\right)\overset{\mathrm{d}}{\to} (Z(D_1),\ldots,Z(D_n)),\qquad\text{as }t\to\infty,$$

it is often necessary to compute (4.3) for  $D=D_i$  and  $L=D_j$ , for all  $i,j=1,\ldots,n$ .

A number of different methods have been used in the literature to compute (4.3), such as:

Spectral representations and Fejer-type kernels or approximate identity for

4.1. Introduction 83

convolutions, see e.g. [ALS15].

- Spectral representations and Abelian-Tauberian theorems, see e.g. [LO13].
- When *K* is radial, the method of geometric probabilities, see e.g. [LRM23].
- A direct approach. By Fubini's theorem one has <sup>1</sup>

$$\operatorname{Cov}\left(\int_{tD} A_x dx, \int_{tL} A_y dy\right) = \int_{tD} \int_{tL} K(x - y) dx dy. \tag{4.4}$$

In this paper, we will focus on this latter direct approach, combining it with regularity conditions for cross covariograms developed in [Gal11] and standard properties of regularly varying functions. This method will allow to compute (4.3) under assumptions that encompass the classical ones found in the literature.

#### 4.1.1 Classical assumptions, existing literature, motivating examples

To highlight the main novelty introduced by this paper, let us look at the classical assumptions considered in the literature.

The **first classical assumption**, probably the most popular in the literature (see e.g. [BH02; BM83; CNN20; KS18; LO14; NPP11]), is

$$K \in L^1(\mathbb{R}^d), \tag{4.5}$$

see also Section 4.4.1. In this case, it is well known that for every  $D, L \subseteq \mathbb{R}^d$  compact, choosing  $r_t = t^{d/2}$ , we have

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2}}, \frac{\int_{tL} A_y dy}{t^{d/2}}\right) = \operatorname{Vol}(D \cap L) \int_{\mathbb{R}^d} K(z) dz, \tag{4.6}$$

where the latter follows by (4.4), a change of variable z = x - y and dominated convergence. Then, we have two possibilities:

- If  $\int_{\mathbb{R}^d} K(z) dz = 0$ , then  $r_t = t^{d/2}$  is not a correct choice for computing the asymptotic covariances, since the limit is identically 0 for every D, L compact (see also Example 4.4.1).
- If  $\int_{\mathbb{R}^d} K(z)dz \neq 0$ , then  $r_t = t^{d/2}$  is a correct choice for computing the asymptotic covariances (4.3), as well as  $r_t = t^{d/2} w_t^{1/2}$  (recall the definition (4.1) of w and note that  $K \in L^1(\mathbb{R}^d)$ ), and we have

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_y dy}{t^{d/2} w_t^{1/2}}\right) = \operatorname{Vol}(D \cap L). \tag{4.7}$$

The **second classical assumption**, which covers some of the cases where  $K \notin L^1(\mathbb{R}^d)$ , is K radial and regularly varying with index  $-\beta \in (-d,0)$  (see e.g. [ALO15;

<sup>&</sup>lt;sup>1</sup>Here the equality is ensured by Proposition 4.2.1.

DM79; LO14; Ros60; Taq79]), namely

$$K(z) = k(|z|) = \frac{\ell(|z|)}{|z|^{\beta}}, \qquad \beta \in (0, d),$$
 (4.8)

for some  $k:\mathbb{R}_+\to\mathbb{R}$ , where  $|\cdot|$  is the Euclidean norm in  $\mathbb{R}^d$  and  $\ell:\mathbb{R}_+\to\mathbb{R}$  is a slowly varying function, (i.e.  $\ell$  is definitively positive and  $\ell(rs)/\ell(r)\to 1$  for s>0, as  $r\to\infty$ , see the seminal book [BGT89] or Section 4.2.1 for more details). In this case, it is a standard fact that for every  $D,L\subseteq\mathbb{R}^d$  compact, choosing  $r_t=t^dk(t)^{1/2}=t^{d-\frac{\beta}{2}}\ell(t)^{1/2}$ , we have

$$\lim_{t \to \infty} \text{Cov}\left(\frac{\int_{tD} A_x dx}{t^d k(t)^{1/2}}, \frac{\int_{tL} A_y dy}{t^d k(t)^{1/2}}\right) = \lim_{t \to \infty} \int_D \int_L \frac{k(t|x-y|)}{k(t)} dx dy = \int_D \int_L |x-y|^{-\beta} dx dy,$$
(4.9)

where the latter follows by (4.4), Theorem 4.2.5 and dominated convergence (see also Remark 4.2.1). This means that  $r_t = t^d k(t)^{1/2}$  is a correct choice to compute the asymptotic covariances (4.9). Moreover, by Theorem 4.2.4 we have <sup>2</sup>

$$w_t = \omega_{d-1} \int_0^t k(r) r^{d-1} dr \sim \frac{\omega_{d-1}}{(d-\beta)} t^d k(t) = \frac{\omega_{d-1}}{(d-\beta)} t^{d-\beta} \ell(t), \qquad \text{as } t \to \infty,$$

which implies that  $r_t = t^{d/2} w_t^{1/2}$  is also a correct choice, and

$$\lim_{t \to \infty} \text{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_y dy}{t^{d/2} w_t^{1/2}}\right) = \frac{(d-\beta)}{\omega_{d-1}} \int_D \int_L |x-y|^{-\beta} dx dy. \tag{4.10}$$

Summarizing, and excluding the case  $K \in L^1(\mathbb{R}^d)$ ,  $\int_{\mathbb{R}^d} K(z) dz = 0$ , we have:

- 1. Under both assumptions,  $r_t = t^{d/2} w_t^{1/2}$  is a correct choice for computing the asymptotic covariances.
- 2. Under both assumptions  $w_t$  is a regularly varying function, with index:  $\alpha = 0$  if  $K \in L^1(\mathbb{R}^d)$ ,  $\int_{\mathbb{R}^d} K(z) dz \neq 0$ ;  $\alpha = d \beta \in (0, d)$  if K is radial and regularly varying with index  $-\beta \in (-d, 0)$ .

The intuition we can develop from (i)-(ii) is: if  $w_t$  is regularly varying, then  $r_t = t^{d/2}w_t^{1/2}$  is a correct choice for computing (4.3). This fact will be proved in our main result, Theorem 4.1.1, together with an explicit expression for the asymptotic covariances (see (4.16)), which generalizes (4.7) and (4.10) for D, L in a suitable class  $\mathcal{D}$  of compact domains (see (4.13) and (4.14)).

It is important to note that, excluding the case  $K \in L^1(\mathbb{R}^d)$ ,  $\int_{\mathbb{R}^d} K(z)dz = 0$ , our assumption " $w_t$  regularly varying" (see (4.15)) in Theorem 4.1.1 is **strictly more general than the two forementioned classical assumptions**. This is the main novelty provided by this paper.

In fact, in many cases (see e.g. [GMT24, Lemma 2.6], [LO13, Example 5], or [MN24, Section 6]) the covariance function K is neither integrable nor radial

<sup>&</sup>lt;sup>2</sup>Here  $\omega_0 = 2$  and  $\omega_{d-1}$  is the surface area of the sphere  $S^{d-1}$  in  $\mathbb{R}^d$  when  $d \geq 2$ .

4.1. Introduction 85

and regularly varying, but the integrated covariance function w is regularly varying. Consider, for instance, the situations where K is not absolutely integrable, but  $\lim_{t\to\infty} w_t = \int_{\mathbb{R}^d} K(z) dz \in (0,\infty)$ ; or  $w_t \sim t^{\alpha}$ ,  $\alpha > 0$ , but K is not radial and regularly varying. Some explicit examples will be given in Section 4.4.3.

The case  $K \in L^1(\mathbb{R}^d)$ ,  $\int_{\mathbb{R}^d} K(z) dz = 0$  was not considered in the discussion above, since even if it falls under the first classical assumption, one is not able to compute the asymptotic covariances with (4.6). However, if w is regularly varying, Theorem 4.1.1 allows to derive an expression for the asymptotic covariances, with the rate  $r_t = t^{d/2} w_t^{1/2}$ . An explicit example (see Example 4.4.1) will be given in Section 4.4.1.

As mentioned above, the main novelty provided by Theorem 4.1.1 is the computation of (4.3) in many cases which do not fall under the two classical assumptions.

However, as we will better explain in Section 4.4, Theorem 4.1.1 can also be combined with the Peccati-Tudor theorem [PT05] to prove **new multi-dimensional** central limit theorems for non-linear functionals of Gaussian fields.

An example of this fact, that stands out for its importance in quantum mechanics, is the **Berry's random wave model**, a Gaussian field whose functionals have been extensively studied in the literature (see Section 4.4.3 and the references therein). Recently, it was proved in [MN24] that a large class of functionals of this field (and of many other fields satisfying a specific spectral condition) in the form (4.30) have Gaussian fluctuations. Nevertheless, the authors were not able to extend their result to a multi-dimensional central limit theorem (see also [MN24, Section 1.4]). This paper also aims to partially fill this gap. Indeed, as we will better explain in Section 4.4.3, by means of Theorem 4.1.1 we will prove new multi-dimensional central limit theorems for the Berry's random wave model (see in particular Example 4.4.3 and Example 4.4.5), using the fact that w is regularly varying (i.e. (4.15) holds) even if K does not always satisfy the forementioned classical assumptions.

#### 4.1.2 Main result

In order to state our main result, we need to introduce some notations, quantities and assumptions.

We denote by "Vol" the Lebesgue measure and by "Per" the perimeter in generalized sense in  $\mathbb{R}^d$ , defined for  $D \subseteq \mathbb{R}^d$  measurable as (see [Gal11])

$$\operatorname{Per}(D) = \sup \left\{ \int_{D} \operatorname{div} \varphi(x) \, dx : \varphi \in C_{c}^{1}(\mathbb{R}^{d}, \mathbb{R}^{d}), \|\varphi\|_{\infty} \leq 1 \right\},\,$$

where  $C_c^1(\mathbb{R}^d,\mathbb{R}^d)$  is the set of continuously differentiable functions with compact support. Note that, denoting by  $\mathcal{H}^{d-1}(\partial D)$  the (d-1)-Hausdorff measure of the topological boundary  $\partial D$ , we have  $\operatorname{Per}(D) \leq \mathcal{H}^{d-1}(\partial D)$ , which becomes an equality under additional assumptions on D (see [Gal11] for more details).

Furthermore, for  $f: \mathbb{R}^d \to \mathbb{R}$  and r > 0,

$$\int_{S^{d-1}} f(r\theta) d\theta$$

is the integral of  $\theta\mapsto f(r\theta)$  (when defined) with respect to the uniform measure on  $S^{d-1}$  and  $\omega_{d-1}=\int_{S^{d-1}}d\theta$  is the surface measure of  $S^{d-1}$ . (For d=1 we use the formalism

$$\int_{S^0} f(r\theta)d\theta = f(r) + f(-r), \qquad \omega_0 = 2.$$
 (4.11)

For  $D, L \in \mathcal{D}$ , we consider the **cross covariogram**  $g_{D,L} : \mathbb{R}^d \to \mathbb{R}$ ,

$$g_{D,L}(z) = \operatorname{Vol}(D \cap (L+z)), \tag{4.12}$$

which, by Proposition 4.2.2, is such that all its directional derivatives

$$\frac{d}{dl}g_{D,L}(l\theta), \quad \theta \in S^{d-1},$$

are well-defined in  $L^1(\mathbb{R}_+)$  and bounded by  $Per(D) \wedge Per(L)$ .

Finally, we will always assume to be in one of the following cases:

*Case 1.* K is **radial** and  $\mathcal{D}$  is the class of all compact sets in  $\mathbb{R}^d$  with **finite perimeter**, i.e. for some  $k : \mathbb{R}_+ \to \mathbb{R}$  we have

$$K(z) = k(|z|), \quad \mathcal{D} = \{D \subseteq \mathbb{R}^d : D \text{ compact, } Per(D) < \infty\}.$$
 (4.13)

If d = 1, K is always radial.

Case 2.  $\mathcal{D} = \mathcal{D}_{x_0}$  is the class of the closed balls centered at a fixed point  $x_0 \in \mathbb{R}^d$ , namely

$$\mathcal{D} = \left\{ \left\{ x \in \mathbb{R}^d : |x - x_0| \le r \right\} : r \in \mathbb{R}_+ \right\}. \tag{4.14}$$

We are now ready to state our main result.

**Theorem 4.1.1.** Let  $\mathcal{D}$  be a collection of compact sets in  $\mathbb{R}^d$  and  $A = (A_x)_{x \in \mathbb{R}^d}$  be a measurable, weakly stationary random field with covariance function  $K : \mathbb{R}^d \to \mathbb{R}$ . Furthermore, let K and  $\mathcal{D}$  satisfy the assumption (4.13) or the assumption (4.14), and assume that w in (4.1) is **regularly varying** with index  $\alpha \in (-1, d]$ , that is

$$w_t = \ell(t)t^{\alpha} \quad \forall t \in \mathbb{R}_+ \tag{4.15}$$

where  $\ell : \mathbb{R}_+ \to \mathbb{R}$  is slowly varying (see Definition 4.2.1 and Lemma 4.2.3). Then, for all  $D, L \in \mathcal{D}$ , we have as  $t \to \infty$ 

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_*^{1/2}}, \frac{\int_{tL} A_x dx}{t^{d/2} w_*^{1/2}}\right) \to \frac{1}{\omega_{d-1}} \int_{S^{d-1}} d\theta \int_0^\infty dl \left(-\frac{d}{dl} g_{D,L}(l\theta)\right) l^\alpha. \tag{4.16}$$

**Remark 4.1.1.** (Restriction of the parameter  $\alpha$ ) Note that in Theorem 4.1.1 we consider  $\alpha \in (-1, d]$  instead of  $\alpha \in \mathbb{R}$ . This happens for different reasons:

•  $\alpha > -1$  is assumed for technical reasons. For example, if  $D = L = \{|z| \le 1\} \subseteq$ 

4.1. Introduction 87

 $\mathbb{R}^d$ , we have for all  $\theta \in S^{d-1}$  and some constant  $c_d > 0$ 

$$\begin{split} g_{D,D}(l\theta) &= c_d \int_{\frac{l}{2}}^1 \left(1 - r^2\right)^{\frac{d-1}{2}} dr \ \mathbf{1}_{[0,2]}(l), \\ &- \frac{d}{dl} g_{D,D}(l\theta) = \frac{c_d}{2} \left(1 - \left(\frac{l}{2}\right)^2\right)^{\frac{d-1}{2}} \mathbf{1}_{[0,2]}(l), \end{split}$$

implying that the RHS of (4.16) is not a finite integral if  $\alpha \le -1$ . Anyway, if (4.15) holds with  $\alpha \le -1$ , another scaling could yield a different limit.

• Since  $|K(z)| \le K(0) < \infty$  for every  $z \in \mathbb{R}^d$ , we have

$$|w_t| \le \int_{\{|z| \le t\}} |K(z)| dz \le \operatorname{const} K(0) t^d,$$

implying that  $w_t$  can not be regularly varying with index  $\alpha > d$ .

**Remark 4.1.2.** (Comparison to known covariance structures) Before going on, it is worth discussing analogies and differences with some known covariance structures, highlighting when (4.16) can be seen as their generalization. First of all, note that when d = 1, D = [0, s] and L = [0, r] with r > s > 0, then for l > 0 we have

$$\begin{array}{rcl} g_{D,L}(l) & = & (s-l)\mathbf{1}_{[0,s]}(l) \\ g_{D,L}(-l) & = & s\mathbf{1}_{[0,r-s]}(l) + (r-l)\mathbf{1}_{[r-s,r]}(l) \\ -\frac{d}{dl}(g_{D,L}(l) + g_{D,L}(-l)) & = & \mathbf{1}_{[0,s]}(l) + \mathbf{1}_{[r-s,r]}(l) \end{array}$$

and the RHS of (4.16) becomes

$$\frac{1}{2} \left( \int_0^s l^{\alpha} dl + \int_{r-s}^r l^{\alpha} dl \right) = \frac{1}{2(\alpha+1)} \left( s^{\alpha+1} + r^{\alpha+1} - |r-s|^{\alpha+1} \right). \tag{4.17}$$

In (4.17) we observe the covariance structure of a fractional Brownian motion  $(B_s^H)_{s>0}$  with Hurst index  $H=\frac{\alpha+1}{2}$ . Note that this is also the covariance structure of other (non-Gaussian) stochastic processes, like the Hermite process (see e.g. [Tud13, Section 3]).

For  $\alpha = 0$ , the RHS of (4.16) has the special form

$$Vol(D \cap L), \tag{4.18}$$

which is the covariance structure of a Gaussian noise (a set-indexed generalization of the Brownian motion, see e.g. [AT07, Section 1.4.3]). Note that if D is fixed and u, v > 0, (4.16) becomes

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tDu^{1/d}} A_x dx}{t^{d/2}}, \frac{\int_{tDv^{1/d}} A_y dy}{t^{d/2}}\right) = \operatorname{Vol}(D) \min\{u, v\},$$

which is, up to a scaling factor, the covariance structure of a Brownian motion (see also (4.37)). This rescaling is used in the literature to prove multidimensional central limit theorems (to a Brownian motion) for functionals of Gaussian fields, see e.g.

[IL89, Theorem 2.3.1] and the discussion in Section 4.4.1.

For  $\alpha > 0$ , the RHS of (4.16) coincides with

$$\begin{split} \frac{1}{\omega_{d-1}} \int_{S^{d-1}} d\theta \int_0^\infty dl \left( -\frac{d}{dl} g_{D,L}(l\theta) \right) l^\alpha &= \frac{\alpha}{\omega_{d-1}} \int_{S^{d-1}} d\theta \int_0^\infty dl g_{D,L}(l\theta) l^{\alpha-1} \end{aligned} \tag{4.19}$$

$$= \frac{\alpha}{\omega_{d-1}} \int_{\mathbb{R}^d} g_{D,L}(z) |z|^{\alpha-d} dz, \tag{4.20}$$

where the first equality follows integrating by parts and the second passing from polar to standard coordinates. Note that

$$\int_{\mathbb{R}^d} g_{D,L}(z)|z|^{\alpha-d}dz = \int_{\mathbb{R}^d} \left( \int_{\mathbb{R}^d} \mathbf{1}_D(x) \mathbf{1}_L(x-z) dx \right) |z|^{\alpha-d}dz = \int_D \int_L |x-y|^{\alpha-d}dx dy.$$

Thus, (4.19) is exactly the RHS of (4.10) with  $\beta = d - \alpha$ , which is perfectly consistent with the discussion in Section 4.1.1.

Finally, choosing  $d \geq 2$ ,  $D = [0, \underline{s}]$  and  $L = [0, \underline{r}]$ , with  $\underline{r}, \underline{s} \in \mathbb{R}^d$ , we have:

- If  $\alpha=0$ , then (4.18) is the covariance structure of a d-dimensional Brownian sheet.
- If  $\alpha \neq 0$ , then the RHS of (4.16) is not the covariance structure of a *d*-dimensional fractional Brownian sheet (see e.g. [Wan07] for the definition).

#### 4.1.3 Plan of the paper

The rest of the paper is organized as follows. In Section 4.2 we ensure the well-posedness of (4.2), prove (4.4) and give some preliminaries about cross covariograms and regularly varying functions. In Section 4.3 we prove Theorem 4.1.1. In Section 4.4 we apply Theorem 4.1.1 to prove multi-dimensional central limit theorems for non-linear functionals of stationary Gaussian fields, giving several examples. Finally, in Section 4.5 we show that (4.3) for A with continuous covariance function K can be reduced to the same problem for a radial, continuous covariance function  $K_{\rm iso}$ , providing a class of non-Gaussian, non-stationary, weakly stationary random fields where this reduction principle can be applied.

## 4.2 Preliminaries for the proof of Theorem 4.1.1

The goal of this section is to ensure the well-posedness of (4.2), prove (4.4) and give preliminary results on cross covariograms and regularly varying functions.

#### 4.2.1 Well-posedness of (4.2) and proof of (4.4)

The following proposition explains why (4.2) is well posed and (4.4) holds.

**Proposition 4.2.1.** Fix  $d \geq 1$  and let  $A = (A_x)_{x \in \mathbb{R}^d}$  be a measurable (in the sense that  $(\omega, x) \mapsto A_x(\omega)$  is measurable), weakly stationary random field, with covariance function  $K : \mathbb{R}^d \to \mathbb{R}$ . Then:

1. We have

$$\mathbb{P}\left(\int_{D}|A_{x}|dx<\infty,\quad\forall\ D\subseteq\mathbb{R}^{d}\ \textit{compact}\right)=1$$

which implies that (4.2) is well defined.

2. For every  $D, L \subseteq \mathbb{R}^d$  compact we have

$$\operatorname{Cov}\left(\int_{D} A_{x} dx, \int_{L} A_{y} dy\right) = \int_{D} \int_{L} K(x-y) dx dy,$$

which implies (4.4).

*Proof.* For simplicity, since A is weakly stationary, let us use the notation  $m:=\mathbb{E}[A_x]$ ,  $\forall \ x\in\mathbb{R}^d$ . Note that the function  $(\omega,x,y)\mapsto |A_x(\omega)-m||A_y(\omega)-m|$  is measurable because we are assuming that  $(\omega,x)\mapsto A_x(\omega)$  is measurable. Moreover, we have

$$|A_x - m||A_y - m| \le \frac{1}{2} (|A_x - m|^2 + |A_y - m|^2),$$

which implies

$$\mathbb{E}\left[\int_{D} \int_{L} |A_{x} - m| |A_{y} - m| dx dy\right] \leq \frac{1}{2} \left(\operatorname{Var}(A_{x}) + \operatorname{Var}(A_{y})\right) \operatorname{Vol}(D) \operatorname{Vol}(L) \quad (4.21)$$

$$= K(0) \operatorname{Vol}(D) \operatorname{Vol}(L) < \infty. \quad (4.22)$$

In particular, for  $D \subseteq \mathbb{R}^d$  compact we have

$$\mathbb{E}\left[\int_{D} |A_x - m| dx\right] \le \mathbb{E}\left[\left(\int_{D} |A_x - m| dx\right)^2\right]^{1/2}$$
$$= \mathbb{E}\left[\int_{D} \int_{D} |A_x - m| |A_y - m| dx dy\right]^{1/2} < \infty,$$

implying that for almost every  $\omega \in \Omega$  the function  $x \mapsto A_x(\omega)$  is integrable on the compact D. To prove that for almost every  $\omega \in \Omega$  the function  $x \mapsto A_x(\omega)$  is integrable on every compact set of  $\mathbb{R}^d$  (i.e. the almost sure local integrability stated in (i)), note that as  $n \to \infty$ 

$$\begin{split} \Omega_n := \left\{ \omega \in \Omega : \int_{\{|x| \leq n\}} |A_x(\omega)| dx < \infty \right\} \\ \searrow \Omega_\infty := \left\{ \omega \in \Omega : \int_D |A_x(\omega)| dx < \infty, \forall D \subseteq \mathbb{R}^d \text{ compact} \right\}. \end{split}$$

Thus, since  $\mathbb{P}(\Omega_n) = 1$  for every  $n \in \mathbb{N}$ , we have  $\mathbb{P}(\Omega_\infty) = 1$  and (i) is proved.

Since (4.2) is now well defined by (i) and (4.21) holds, by Fubini-Tonelli we obtain

$$\operatorname{Cov}\left(\int_{D} A_{x} dx, \int_{L} A_{y} dy\right) = \mathbb{E}\left[\int_{D} \int_{L} (A_{x} - m)(A_{y} - m) dx dy\right]$$
$$= \int_{D} \int_{L} \operatorname{Cov}(A_{x}, A_{y}) dx dy = \int_{D} \int_{L} K(x - y) dx dy,$$

for every D, L compact domains, which concludes the proof of (ii).

#### 4.2.2 Cross covariograms

First of all, let us recall the definition (4.12) of the cross covariogram  $g_{D,L}$  of two compact sets D, L in  $\mathbb{R}^d$ . In particular, when D = L, denote by  $g_D$  the **covariogram** of D

$$g_D(z) := g_{D,D}(z) = \operatorname{Vol}(D \cap (D+z)), \qquad z \in \mathbb{R}^d.$$

Moreover, for a bounded set  $E \subseteq \mathbb{R}^d$ , recall the definition of diameter

$$\operatorname{diam}(E) := \sup_{x, y \in E} |x - y| < \infty. \tag{4.23}$$

Let us now list some properties related to cross covariograms, which are easily derived from the results in [Gal11] and will be needed in the sequel.

**Proposition 4.2.2.** Consider D, L compact sets in  $\mathbb{R}^d$ . Then we have:

- 1.  $g_{D,L}(z) = g_{L,D}(-z)$  and in particular  $g_D$  is symmetric.
- 2. For every  $z, h \in \mathbb{R}^d$

$$|g_{D,L}(z+h) - g_{D,L}(z)| \le 2\min\{g_D(0) - g_D(h), g_L(0) - g_L(h)\}.$$

- 3.  $g_{D,L}(z) = 0$  if  $|z| > \text{diam}(D \cup L)$ . In particular, since  $\text{diam}(D \cup L) < \infty$ ,  $g_{D,L}$  has compact support.
- 4.  $Per(D) < \infty$  if and only if  $g_D$  is Lipschitz continuous (with Lipschitz constant Per(D)/2).
- 5. If  $\operatorname{Per}(D) < \infty$  or  $\operatorname{Per}(L) < \infty$ , then  $g_{D,L}$  is Lipschitz with Lipschitz constant  $\operatorname{Per}(D) \wedge \operatorname{Per}(L)$ . Moreover, all the directional derivatives of  $g_{D,L}$  exist almost everywhere and are bounded by  $\operatorname{Per}(D) \wedge \operatorname{Per}(L)$ . In addition, for every  $\theta \in S^{d-1}$  the function  $r \to g_{D,L}(r\theta)$  is absolutely continuous and we have

$$g_{D,L}(r\theta) = \int_{r}^{\operatorname{diam}(D \cup L)} \left( -\frac{d}{dl} g_{D,L}(l\theta) \right) dl, \qquad \forall r > 0.$$
 (4.24)

*Proof.* (i) follows from  $Vol(D \cap (L+z)) = Vol((D-z) \cap L)$  and (ii) from

$$|\operatorname{Vol}(D \cap (L+z+h)) - \operatorname{Vol}(D \cap (L+z))| \le \int_{\mathbb{R}^d} |\mathbf{1}_{L+z}(x) - \mathbf{1}_{L+z+h}(x)| dx$$

$$= \int_{\mathbb{R}^d} |\mathbf{1}_{L+z}(x) - \mathbf{1}_{L+z+h}(x)|^2 dx$$

$$= \int_{\mathbb{R}^d} \mathbf{1}_{L+z}(x) + \mathbf{1}_{L+z+h}(x) - 2\mathbf{1}_{L+z}(x)\mathbf{1}_{L+z+h}(x) dx = 2(g_L(0) - g_L(h))$$

and using (i). (iii) simply follows by definition, since  $g_{D,L}(z) \neq 0$  is possible only if  $z \in D - L \subseteq (D \cup L) - (D \cup L)$ , and (iv) is proved in [Gal11, Theorem 14]. Regarding (v), the fact that  $g_{D,L}$  is Lipschitz with Lipschitz constant  $\operatorname{Per}(D) \wedge \operatorname{Per}(L)$  easily follows from (iv) and (ii). As a consequence, for every fixed  $\theta \in S^{d-1}$  the function

$$l \mapsto g_{D,L}(l\theta)$$

is Lipschitz, its derivative exists almost everywhere and is bounded by  $Per(D) \wedge Per(L)$ . Moreover, the fact that every Lipschitz function is absolutely continuous, together with (iii), implies (4.24).

#### 4.2.3 Regularly varying functions

**Definition 4.2.1.** A measurable function  $h : \mathbb{R}_+ \to \mathbb{R}$  is said **regularly varying** if h is positive on  $[a, \infty)$  for some a > 0 and if we have, for all l > 0

$$\frac{h(tl)}{h(t)} \to g(l) \qquad \text{as } t \to \infty.$$

In particular, if  $g \equiv 1$  then h is said slowly varying.

Definition 4.2.1 is one of the equivalent definitions used for regularly varying functions, as explained in Lemma 4.2.3 (see [BGT89, Section 1.4]).

**Lemma 4.2.3.** Consider a measurable function  $h : \mathbb{R}_+ \to \mathbb{R}$ . Then the following statements are equivalent:

- 1. h is regularly varying.
- 2. h is regularly varying with limit  $g(l) = l^{\alpha}$ , for some  $\alpha \in \mathbb{R}$ .
- 3.  $h(l) = \ell(l)l^{\alpha}$  for some  $\alpha \in \mathbb{R}$  and  $\ell : \mathbb{R}_{+} \to \mathbb{R}$  slowly varying.

If (one of) the three statements hold, we say that h is **regularly varying with index**  $\alpha$ .

The following proposition says that the condition (4.15) in Theorem 4.1.1 is more general than the classical assumption "K(z) = k(|z|) with k regularly varying with index  $\beta \in (-d,0)$ ", discussed in Section 4.1.1.

**Proposition 4.2.4.** [BGT89, Proposition 1.5.11] If  $k : \mathbb{R}_+ \to \mathbb{R}$  is regularly varying with index  $-\beta \in (-d, 0)$  and locally bounded on  $\mathbb{R}_+$ , then

$$\frac{t^d k(t)}{\int_0^t k(r) r^{d-1} dr} \to (d-\beta) \qquad \text{as } t \to \infty$$

and  $w_t := \omega_{d-1} \int_0^t k(r) r^{d-1} dr$  is regularly varying with index  $\alpha = d - \beta$ .

A fundamental tool for the proof of Theorem 4.1.1 will be Potter's theorem (see e.g. [BGT89, Theorem 1.5.6]).

**Theorem 4.2.5.** (Potter's Theorem) If  $h : \mathbb{R}_+ \to \mathbb{R}$  is regularly varying with index  $\alpha \in \mathbb{R}$ , then for every A > 1,  $\delta > 0$ ,  $\exists X = X(h, A, \delta) > 0$  such that

$$|h(lt)/h(t)| \le A \max\{l^{\alpha+\delta}, l^{\alpha-\delta}\} \qquad \forall l > \frac{X}{t}, \quad \forall t > X.$$

Remark 4.2.1. (Justifying equation (4.9)) Note that if  $k : \mathbb{R}_+ \to \mathbb{R}$  is measurable, bounded and regularly varying with index  $-\beta \in (-d,0)$ , by Theorem 4.2.5 we have for  $D, L \subseteq \mathbb{R}^d$  compact sets

$$\mathbf{1}_{\{|x-y| \ge X/t\}} \mathbf{1}_{D}(x) \mathbf{1}_{L}(y) \frac{k(t|x-y|)}{k(t)} \le c \, \mathbf{1}_{D}(x) \mathbf{1}_{L}(y) \max \left\{ |x-y|^{-\beta-\delta}, |x-y|^{-\beta+\delta} \right\}$$

where X,c>0 are suitable constants. Moreover, since  $\beta\in(0,d)$ , choosing  $\delta>0$  small enough, the function on the RHS is integrable in  $\mathbb{R}^{2d}$ . Therefore, by dominated convergence theorem, we have as  $t\to\infty$ 

$$\int_{D} \int_{L} \mathbf{1}_{\{|x-y| \ge X/t\}} \frac{k(t|x-y|)}{k(t)} dx dy \to \int_{D} \int_{L} |x-y|^{-\beta} dx dy.$$

This last fact, together with

$$\int_D \int_L \mathbf{1}_{\{|x-y| \leq X/t\}} \frac{k(t|x-y|)}{k(t)} dx dy \leq c \, \frac{1}{k(t)t^d} = \frac{1}{\ell(t)t^{d-\beta}} \to 0, \qquad \text{as } t \to \infty,$$

concludes the proof of (4.9).

**Corollary 4.2.6.** Consider  $K : \mathbb{R}^d \to \mathbb{R}$  measurable, bounded function and  $w : \mathbb{R}_+ \to \mathbb{R}$  as defined in (4.1). Assume that w is regularly varying with index  $\alpha > -1$ . Then for every  $U \in (0, \infty)$ 

$$\int_0^U |w_{lt}/w_t| dl o \int_0^U l^{lpha} dl \qquad \text{as } t o \infty.$$

*Proof.* First of all, observe that by Potter's Theorem 4.2.5, choosing A=2 and  $\delta=\frac{\alpha+1}{2}>0, \exists \ X=X(w,A,\delta)=X(K,\alpha)>0$  such that

$$|w_{lt}/w_t| \le 2\max\left\{l^{\frac{3\alpha+1}{2}}, l^{\frac{\alpha-1}{2}}\right\} \qquad \forall l > \frac{X}{t}, \quad \forall t > X, \tag{4.25}$$

with  $l \mapsto \max\{l^{\frac{3\alpha+1}{2}}, l^{\frac{\alpha-1}{2}}\}$  integrable on [0, U] because  $\alpha > -1$ . Moreover, since w is of the form (4.1), for 0 < l < X/t we have

$$|w_{lt}/w_t| \le \frac{\max_{x \in \mathbb{R}^d} |K(x)|}{|w_t|} \omega_{d-1} X^d \quad \forall \, 0 < l < X/t.$$

Putting all together, we obtain

$$\int_{0}^{U} |w_{lt}/w_{t}| dl = \int_{0}^{X/t} |w_{lt}/w_{t}| dl + \int_{X/t}^{U} |w_{lt}/w_{t}| dl.$$

Note that  $w_t = \ell(t)t^{\alpha}$  with  $\alpha > -1$ , so  $\ell(t)t^{1+\alpha} \to \infty$  and

$$\int_0^{X/t} |w_{lt}/w_t| dt \le \frac{\max_{x \in \mathbb{R}^d} |K(x)|}{t|w_t|} \omega_{d-1} X^{d+1} = \frac{\max_{x \in \mathbb{R}^d} |K(x)|}{|\ell(t)t^{1+\alpha}|} \omega_{d-1} X^{d+1} \to 0.$$

Moreover, by (4.25) and dominated convergence theorem we have

$$\int_{X/t}^{U} |w_{lt}/w_{t}| dl \to \int_{0}^{U} l^{\alpha} dl,$$

which concludes the proof.

#### 4.3 Proof of Theorem 4.1.1

*Proof.* (of Theorem 4.1.1) By Proposition 4.2.1 and the change of variable x = x, y = x - z, we have

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_y dy}{t^{d/2} w_t^{1/2}}\right)$$

$$= \int_{tD} \int_{tL} K(x - y) \frac{dx dy}{t^d w_t}$$

$$= \int_{\mathbb{R}^d} K(z) g_{tD, tL}(z) \frac{dz}{t^d w_t}$$

$$= \int_{\{|z| \leq \operatorname{diam}(D \cup L)t\}} K(z) g_{D, L}\left(\frac{z}{t}\right) \frac{dz}{w_t}$$

$$(4.26)$$

where  $g_{D,L}$  is defined in (4.12),  $\operatorname{diam}(D \cup L)$  is defined in (4.23) and the last equality follows from (iii) of Proposition 4.2.2 and

$$g_{tD,tL}(z) = \operatorname{Vol}(tD \cap (tL+z)) = t^d \operatorname{Vol}\left(D \cap \left(L + \frac{z}{t}\right)\right) = t^d g_{D,L}\left(\frac{z}{t}\right).$$

Passing to polar coordinates (if d = 1, recall the notation (4.11)), we have

$$\begin{split} & \int_{\{|z| \leq \operatorname{diam}(D \cup L)t\}} K(z) g_{D,L} \left(\frac{z}{t}\right) dz \\ & = \int_{S^{d-1}} d\theta \int_0^{\operatorname{diam}(D \cup L)t} dr \, r^{d-1} K(r\theta) g_{D,L} \left(\frac{r}{t}\theta\right), \end{split}$$

and since  $D, L \in \mathcal{D}$  have finite perimeter in both *Case 1.* (4.13) and *Case 2.* (4.14), by (v) in Proposition 4.2.2 the covariance (4.26) becomes

$$\begin{split} &\operatorname{Cov}\left(\frac{\int_{tD}A_{x}dx}{t^{d/2}w_{t}^{1/2}},\frac{\int_{tL}A_{y}dy}{t^{d/2}w_{t}^{1/2}}\right)\\ &= w_{t}^{-1}\int_{S^{d-1}}d\theta\int_{0}^{\operatorname{diam}(D\cup L)}dl\left(-\frac{d}{dl}(g_{D,L}(l\theta))\right)\left(\int_{0}^{tl}r^{d-1}K(r\theta)dr\right). \end{split}$$

Now we distinguish the two cases in the statement of Theorem 4.1.1.

*Case 1.* If K(z) = k(|z|) for some  $k : \mathbb{R}_+ \to \mathbb{R}$ , then  $K(r\theta) = k(r)$  and we have (recall the definition (4.1) of w)

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_y dy}{t^{d/2} w_t^{1/2}}\right)$$

$$= \frac{1}{\omega_{d-1}} \int_{S^{d-1}} d\theta \int_0^{\operatorname{diam}(D \cup L)} dl \left(-\frac{d}{dl}(g_{D,L}(l\theta))\right) \frac{w_{lt}}{w_t}$$

Case 2. If D, L are balls centered in the same point  $x_0 \in \mathbb{R}^d$ , then  $g_{D,L}(r\theta) = g_{D,L}(r\theta')$  for every  $\theta, \theta' \in S^{d-1}$  and by a Fubini-Tonelli argument (and (v) in Proposition 4.2.2) we have, for all  $\theta' \in S^{d-1}$ 

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_y dy}{t^{d/2} w_t^{1/2}}\right)$$

$$= \int_0^{\operatorname{diam}(D \cup L)} \left(-\frac{d}{dl} (g_{D,L}(l\theta'))\right) \frac{w_{lt}}{w_t} dl$$

$$= \frac{1}{\omega_{d-1}} \int_{S^{d-1}} d\theta \int_0^{\operatorname{diam}(D \cup L)} dl \left(-\frac{d}{dl} (g_{D,L}(l\theta))\right) \frac{w_{lt}}{w_t}$$

$$(4.27)$$

which is equal to what we obtained in Case 1.

Since the expression for the covariance is the same in both cases, to prove (4.16) (and conclude the proof of Theorem 4.1.1) we only need to show

$$\int_{S^{d-1}} d\theta \int_{0}^{\operatorname{diam}(D \cup L)} dl \left( -\frac{d}{dl} (g_{D,L}(l\theta)) \right) \frac{w_{lt}}{w_{t}}$$

$$\rightarrow \int_{S^{d-1}} d\theta \int_{0}^{\operatorname{diam}(D \cup L)} dl \left( -\frac{d}{dl} (g_{D,L}(l\theta)) \right) l^{\alpha} \quad \text{as } t \to \infty, \qquad (4.28)$$

where  $\alpha$  is the index of regular variation of  $w_t$ . First of all, note that  $w_{lt}/w_t \to l^{\alpha}$  for every l > 0, because w is regularly varying with index  $\alpha$ , implying the point-wise convergences (for almost every l > 0,  $\theta \in S^{d-1}$ )

$$F_t(l,\theta) := \left( -\frac{d}{dl} (g_{D,L}(l\theta)) \right) \frac{w_{lt}}{w_t} \to F(l,\theta) := \left( -\frac{d}{dl} (g_{D,L}(l\theta)) \right) l^{\alpha}, \quad \text{as } t \to \infty$$

and

$$M_t(l) := \operatorname{Per}(D) \wedge \operatorname{Per}(L) |w_{lt}/w_t| \to M(l) := \operatorname{Per}(D) \wedge \operatorname{Per}(L) l^{\alpha}, \quad \text{as } t \to \infty.$$

Moreover, by Proposition 4.2.2 the inequality

$$|F_t(l,\theta)| = \left| \left( -\frac{d}{dl} (g_{D,L}(l\theta)) \right) \frac{w_{tl}}{w_t} \right| \le M_t(l) := \operatorname{Per}(D) \wedge \operatorname{Per}(L) |w_{lt}/w_t|$$

holds for almost every  $l>0, \theta\in S^{d-1}$  , and by Corollary 4.2.6 we have

$$\int_{S^{d-1}} d\theta \int_0^{\operatorname{diam}(D \cup L)} dl \, M_t(l) \to \int_{S^{d-1}} d\theta \int_0^{\operatorname{diam}(D \cup L)} dl M(l).$$

Putting all together, by the generalized dominated convergence theorem we obtain

$$\int_{S^{d-1}} d\theta \int_0^{\operatorname{diam}(D \cup L)} dl \, F_t(l,\theta) dl \to \int_{S^{d-1}} d\theta \int_0^{\operatorname{diam}(D \cup L)} dl \, F(l,\theta), \qquad \text{as } t \to \infty,$$

which is exactly (4.28).

### 4.4 Non-linear functionals of stationary Gaussian fields

In this section, we show how Theorem 4.1.1 can be applied in the setting of non-linear functionals of stationary Gaussian fields to obtain multi-dimensional limit theorems. Through several examples, we will also compare the results obtained to the ones found in the existing literature.

Let  $B=(B_x)_{x\in\mathbb{R}^d}$  be a continuous, centered, stationary Gaussian field with  $B_x\sim N(0,1)\ \forall\ x\in\mathbb{R}^d$  and (continuous) covariance function

$$C: \mathbb{R}^d \to \mathbb{R}, \quad \text{Cov}(B_x, B_y) = C(x - y), \quad |C(z)| \le C(0) = 1.$$
 (4.29)

Denote again by  $\mathcal{D}$  a collection of compact sets in  $\mathbb{R}^d$  and consider the non-linear functional<sup>3</sup> of B

$$\int_{tD} \varphi(B_x) dx, \qquad D \in \mathcal{D}, \quad t > 0, \tag{4.30}$$

where  $tD:=\{tx:x\in D\}$ ,  $\varphi:\mathbb{R}\to\mathbb{R}$  is not constant (to avoid trivialities),  $\varphi\in L^2(\mathbb{R},\gamma(dx)), \int_{\mathbb{R}}\varphi(x)\gamma(dx)=0 \text{ and } \gamma \text{ is the standard Gaussian measure on } \mathbb{R}.$  Then, we consider the  $L^2$ -decomposition of  $\varphi$  (see e.g. [NP12, Section 1.4])

$$\varphi = \sum_{q=1}^{\infty} a_q H_q,$$

where  $H_q$  is the q-th Hermite polynomial and

$$\|\varphi\|_{L^2(\mathbb{R},\gamma(dx))}^2 = \sum_{q=1}^{\infty} q! a_q^2 < \infty.$$
 (4.31)

Therefore,  $A_x := \varphi(B_x)$  can be expressed as

$$A_x := \varphi(B_x) = \sum_{q=R}^{\infty} a_q H_q(B_x), \qquad a_R \neq 0,$$

where  $R = \inf\{q \ge 1 : a_q \ne 0\} < \infty$  denotes the **Hermite rank** of  $\varphi$  and the equality holds in  $L^2(\Omega)$  sense. Moreover, we have the isometry property (see e.g. [NP12, Section 1.4])

$$\mathbb{E}[H_q(B_x)H_r(B_y)] = q!C^q(x-y)\delta_{qr}$$

where  $\delta_{qr}$  is the Kronecker delta. As a consequence, the covariance function  $K(x-y) = \text{Cov}(A_x, A_y)$  of  $A = (A_x)_{x \in \mathbb{R}^d}$  is

$$K(z) = \sum_{q=R}^{\infty} q! a_q^2 C^q(z).$$
 (4.32)

Note that A is obviously weakly stationary and K is continuous, because uniform limit of continuous functions (thanks to (4.31) and (4.29)). Moreover, w in (4.1) in

 $<sup>^{3}</sup>$ (4.30) is always well-posed in the  $L^{2}$ -sense, see e.g. [MN24, Proposition 3].

this case is

$$w_t = \int_{\{|z| \le t\}} K(z)dz = \sum_{q=R}^{\infty} a_q^2 w_{q,t}, \qquad t > 0,$$
 (4.33)

where we additionally introduced the notation

$$w_{q,t} = q! \int_{\{|z| \le t\}} C^q(z) dz.$$

Now we can finally understand what new results we obtain in this setting applying Theorem 4.1.1, dividing the study in different cases.

#### 4.4.1 The Breuer-Major case.

If  $C \in L^R(\mathbb{R}^d)$ , we are in the Breuer-Major case. Since  $C \in L^R(\mathbb{R}^d)$  implies  $K \in L^1(\mathbb{R}^d)$  (see (4.29) and (4.32)), reasoning as in (4.6) we get

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} \varphi(B_x) dx}{t^{d/2}}, \frac{\int_{tL} \varphi(B_y) dy}{t^{d/2}}\right) = \operatorname{Vol}(D \cap L) \int_{\mathbb{R}^d} K(z) dz. \tag{4.34}$$

Note that if D is fixed and  $u, v \in [0, 1]$ , we have

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tDu^{1/d}} \varphi(B_x) dx}{t^{d/2}}, \frac{\int_{tDv^{1/d}} \varphi(B_y) dy}{t^{d/2}}\right) = \operatorname{Vol}(D) \min\{u, v\} \int_{\mathbb{R}^d} K(z) dz,$$

which is, up to a scaling factor, the covariance function of a Brownian motion (see also (4.37)). Moreover, we have the following fundamental result, proved for the first time in the discrete setting by Breuer and Major in their seminal paper [BM83], and extended to different settings by several authors, see e.g. [BH02], [CNN20], [IL89], [LO14], [NP09], [NPP11], [NZ20].

**Theorem 4.4.1.** (Breuer-Major) If  $C \in L^R(\mathbb{R}^d)$ , then

$$t^{-d/2} \int_{tD} \varphi(B_x) dx \stackrel{\text{law}}{\to} N(0, \sigma^2 \text{Vol}(D))$$
 (4.35)

where

$$\sigma^2 = \int_{\mathbb{R}^d} K(z) dz = \sum_{q=R}^{\infty} q! a_q^2 \int_{\mathbb{R}^d} C(z)^q dz \ge 0.$$

Note that also multi-dimensional (see e.g. [NZ20, Theorem 1.2]) and stronger type (see e.g. [CNN20, Theorem 1.1]) central limit theorems of the form

$$\left(t^{-d/2} \int_{tD} \varphi(B_x) dx\right)_{D \in \mathcal{D}'} \stackrel{\text{f.d.d.}}{\to} \sigma G = (\sigma G(D))_{D \in \mathcal{D}'}$$
(4.36)

have been proved in the literature, where G is a Gaussian noise, with

$$Cov(G(D), G(L)) = Vol(D \cap L),$$

and  $\mathcal{D}'$  is a suitable class of compact sets. In particular, considering  $\mathcal{D}' = \{Du^{1/d}, u \in \mathcal{D}'\}$ 

[0, 1]} with fixed  $D = \{x \in \mathbb{R}^d : |x| \le 1\}$ , we have (see [IL89, Theorem 2.3.1])

$$\left(t^{-d/2} \int_{tDu^{1/d}} \varphi(B_x) dx\right)_{u \in [0,1]} \stackrel{\text{f.d.d.}}{\to} \sigma \text{Vol}(D)W = (\sigma \text{Vol}(D)W_u)_{u \in [0,1]}, \quad (4.37)$$

where W is a standard Brownian motion.

If  $\sigma^2 > 0$ , then (4.15) holds with  $\alpha = 0$  and Theorem 4.1.1 implies

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} \varphi(B_x) dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} \varphi(B_y) dy}{t^{d/2} w_t^{1/2}}\right) = \operatorname{Vol}(D \cap L), \tag{4.38}$$

which is (4.34). Moreover, Theorem 4.4.1 (and its multi-dimensional generalizations) ensure Gaussian fluctuations, which are not guaranteed by Theorem 4.1.1. Nevertheless, our Theorem 4.1.1 allows to obtain the asymptotic covariances (4.38) in many other situations (see e.g. Example 4.4.3 and 4.4.4), whenever (4.15) holds with  $\alpha = 0$ , i.e.  $w_t$  is slowly varying.

Conversely, if  $\sigma^2=0$ , (4.34) and (4.35) only imply that  $t^{d/2}$  is not the correct normalization if we hope in a non-degenerate limit in distribution (see also the discussion in Section 4.1.1). However,  $w_t$  could be regularly varying, allowing us to apply Theorem 4.1.1. This is exactly what happens in the following example.

**Example 4.4.1. (Breuer-Major case,**  $\sigma^2 = 0$ .) Fix d = 1, consider a fractional Brownian motion  $(W_x^H)_{x \in \mathbb{R}}$  with Hurst index  $H \in (0, 1/2)$  and recall its covariance structure

$$Cov(W_x^H, W_y^H) = \frac{1}{2} (|x|^{2H} + |y|^{2H} - |x - y|^{2H}).$$

Furthermore, consider the associated continuous, stationary Gaussian process  $B = (B_x)_{x \in \mathbb{R}^d}$ , defined by

$$B_x := W_{x+1}^H - W_x^H.$$

Note that we consider the 1-increment process B instead of  $W^H$  because B is stationary (hence weakly stationary, and we can apply Theorem 4.1.1), while  $W^H$  is not even weakly stationary (it only has stationary increments). A standard computation yields that the covariance function  $C: \mathbb{R} \to \mathbb{R}$  of B is given by

$$C(z) = \frac{1}{2} \left( |1 + z|^{2H} + |1 - z|^{2H} - 2|z|^{2H} \right)$$

and  $C \in L^1(\mathbb{R}^d)$ , because as  $|z| \to \infty$  we have (by Taylor expansion)

$$C(z) \sim 2H(2H-1)|z|^{2H-2}, \qquad 2H-2 \in (-2,-1).$$

If we choose  $\varphi(x)=x$ , then  $\int_{tD}\varphi(B_x)dx=\int_{tD}B_xdx$  is Gaussian for every t>0. The important fact to note in this example is that

$$\sigma^2 = \int_{\mathbb{R}^d} K(z)dz = \int_{\mathbb{R}^d} C(z)dz = 0,$$

implying that (4.34) and the Breuer-Major theorem only say that  $t^{d/2}=t^{1/2}$  is not

the correct normalization. Indeed, as  $t \to \infty$  we have

$$w_t = w_{1,t} = 2\int_0^t C(z)dz = \int_0^t |1+z|^{2H} + |1-z|^{2H} - 2|z|^{2H}dz$$
$$= \frac{1}{2H+1} \left( (t+1)^{2H+1} + (t-1)^{2H+1} - 2t^{2H+1} \right) \sim 2Ht^{2H-1}.$$

which means that  $w_t$  is regularly varying with index  $\alpha=2H-1\in(-1,0)$ . Therefore, applying Theorem 4.1.1 we obtain that the correct normalization is  $t^{1/2}w_t^{1/2}\sim\sqrt{2H}t^H$  and  $\forall D,L\in\mathcal{D}$ 

$$\operatorname{Cov}\left(\frac{\int_{tD} B_x dx}{t^{1/2} w_t^{1/2}}, \frac{\int_{tL} B_y dy}{t^{1/2} w_t^{1/2}}\right) \to \frac{1}{2} \int_0^\infty -\frac{d}{dl} (g_{D,L}(l) + g_{D,L}(-l)) l^{2H-1} dl$$

which, if D = [0, r] and L = [0, s], is the covariance function of a fractional Brownian motion with Hurst index  $H \in (0, 1/2)$  (see Remark 4.1.2).

## 4.4.2 The long memory case.

Another frequent assumption is  $C(z) = \rho(|z|)$  radial with  $\rho$  regularly varying with index  $-\beta \in (-d/R, 0)$  (see e.g. [LO13] for some examples), that is:

$$\rho(r) = \frac{\ell(r)}{r^{\beta}}, \qquad r > 0,$$

with  $\ell: \mathbb{R}_+ \to \mathbb{R}$  slowly varying (see Section 4.2.3). In this case, we say that we are in the long-memory case. As  $|z| \to \infty$ , we have

$$K(z) = k(|z|) = \sum_{q=R}^{\infty} q! a_q^2 \rho^q(|z|) \sim R! a_R^2 \rho^R(|z|) = R! a_R^2 \frac{\ell^R(|z|)}{|z|^{R\beta}},$$

implying that  $k : \mathbb{R}_+ \to \mathbb{R}$  is regularly varying with index  $-R\beta \in (-d, 0)$ . Therefore, reasoning as in (4.9) we obtain

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} \varphi(B_x) dx}{t^d k(t)^{1/2}}, \frac{\int_{tL} \varphi(B_y) dy}{t^d k(t)^{1/2}}\right) = \int_{\mathbb{R}^d} g_{D,L}(z) |z|^{-R\beta} dz$$
$$= \int_{D} \int_{L} |x - y|^{-R\beta} dx dy, \quad (4.39)$$

and a correct normalization turns out to be

$$t^d k(t)^{1/2} \sim \sqrt{R! \, a_R^2} \, t^{d-\frac{R\beta}{2}} \ell^{R/2}(t), \qquad \text{as } t \to \infty,$$

i.e. the one usually observed in the long memory context (see e.g. [LO14]). Note that  $C \notin L^R(\mathbb{R}^d)$ , so we are not in the Breuer-Major case, and

$$w_t = \sum_{q=R}^{\infty} a_q^2 w_{q,t} \sim a_R^2 w_{R,t} \quad \text{as } t \to \infty,$$
 (4.40)

which by Proposition 4.2.4 is regularly varying with index  $\alpha = d - R\beta \in (0, d)$ . For this reason, we can apply Theorem 4.1.1, obtaining

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} \varphi(B_x) dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} \varphi(B_y) dy}{t^{d/2} w_t^{1/2}}\right) = \frac{\alpha}{\omega_{d-1}} \int_D \int_L |x - y|^{\alpha - d} dx dy,$$

which is exactly (4.39) (indeed,  $w_t \sim \frac{\omega_{d-1}}{\alpha} t^d k(t)$  by Proposition 4.2.4).

The first limit theorems in the long memory context were proved in the seminal works [DM79], [Ros60], [Taq79] and then extended in many directions. What happens in this case is that (4.30) (suitably normalized by  $t^dk(t)$ ) has asymptotically an Hermite distribution of order R, which is not Gaussian if  $R \geq 2$ . Moreover, the result can be extended to multi-dimensional (and stronger type) central limit theorems, see e.g. [LO14].

Summarizing, if B has a radial covariance function  $C(z) = \rho(|z|)$  with  $\rho$  regularly varying with index  $-\beta \in (-d/R,0)$ , then we can compute the asymptotic covariances of B using (4.39) or Theorem 4.1.1 (since  $w_t$  is regularly varying with index  $\alpha = d - R\beta$ ). Moreover, the forementioned limit theorems apply, yielding an Hermite limiting distribution. However, Theorem 4.1.1 allows to obtain the asymptotic covariances in many other situations in which (4.39) does not hold, whenever  $w_t$  is regularly varying with index  $\alpha > 0$  and k is not regularly varying (see Example 4.4.5).

# 4.4.3 The Berry's case.

The Berry's random wave model is a Gaussian field which arises as the local scaling limit of a variety of random fields on two-dimensional manifolds, see [Die+23] for the details. It was first introduced by Berry in [Ber77], and is used in quantum mechanics to model the Laplace eigenfunctions of classically chaotic billiards with large eigenvalue (see also [NPR19] and the references therein). It is defined as a smooth, stationary, centered Gaussian field B on the plane (d=2) with radial covariance function

$$C(z) = \rho(|z|) = J_0(|z|),$$

where  $J_0$  is the Bessel function of the first kind of order 0 (see e.g. [Kra14]), with series expansion

$$J_0(r) = \sum_{j=0}^{\infty} \frac{(-1)^j}{(j!)^2} \left(\frac{r}{2}\right)^{2j}$$

and asymptotic behavior

$$J_0(r) = \sqrt{\frac{2}{\pi}} r^{-\frac{1}{2}} \cos\left(r - \frac{\pi}{4}\right) + O\left(r^{-\frac{3}{2}}\right) \quad \text{as } r \to \infty.$$
 (4.41)

Note that  $\rho^q$  is not regularly varying. This means that we can not use Proposition 4.2.4 to prove that  $w_{q,t}$  is regularly varying, as we did in Section 4.4.2. Nevertheless, this latter fact is given by the following lemma (for  $q \neq 1$ ) and is a consequence of the results in [GMT24].

**Remark 4.4.1.** For q = 1,  $w_{1,t}$  is not regularly varying. To prove this fact, consider

 $J_1$ , the Bessel function of the first kind of order 1 (see e.g. [Kra14]), with series expansion

$$J_1(r) = \left(\frac{r}{2}\right) \sum_{j=0}^{\infty} \frac{(-1)^j}{j!(j+1)!} \left(\frac{r}{2}\right)^{2j}$$

and asymptotic behavior

$$J_1(r) = \sqrt{\frac{2}{\pi}} r^{-\frac{1}{2}} \cos\left(r - \frac{3}{4}\pi\right) + O\left(r^{-\frac{3}{2}}\right) \quad \text{as } r \to \infty.$$

Note that using the series expansions of  $J_0$  and  $J_1$  we have

$$\frac{d}{dr}\left(J_1(r)r\right) = J_0(r)r.$$

Therefore, since

$$w_{1,t} = 2\pi \int_0^t J_0(r) r \, dr = J_1(t) t = 2\pi \sqrt{\frac{2}{\pi}} \, t^{\frac{1}{2}} \, \cos \left( t - \frac{3}{4} \pi \right) + O\left( t^{-\frac{1}{2}} \right) \quad \text{as } t \to \infty,$$

 $w_{1,t}$  is not a regularly varying function, and Theorem 4.1.1 can not be applied to compute the asymptotic covariances of  $\int_{tD} H_1(B_x) dx = \int_{tD} B_x dx$  and  $\int_{tL} H_1(B_x) dx = \int_{tL} B_x dx$  (recall that  $H_1(x) = x$  is the first Hermite polynomial).

Nevertheless, with a different argument, something can be said about the variance of  $\int_{tD} B_x dx = \int_{tD} H_1(B_x) dx$ . Indeed, we have the formula (see e.g. [Sch38])

$$C(z) = J_0(|z|) = \frac{1}{\omega_{d-1}} \int_{S^{d-1}} e^{i\langle x, \theta \rangle} d\theta.$$

and after applying Fubini-Tonelli theorem we obtain

$$\operatorname{Var}\left(\int_{tD} B_x dx\right) = \int_{tD} \int_{tD} J_0(|x-y|) dx dy = \frac{1}{\omega_{d-1}} \int_{S^{d-1}} |\mathcal{F}[\mathbf{1}_{tD}]|^2(\theta) d\theta = \frac{t^{2d}}{\omega_{d-1}} \int_{S^{d-1}} |\mathcal{F}[\mathbf{1}_{D}]|^2(t\theta) d\theta,$$

where  $\mathcal{F}$  denotes the Fourier transform. Therefore, if  $|\mathcal{F}[\mathbf{1}_D](x)| = o(|x|^{-d/2})$ , we have

$$\operatorname{Var}\left(\int_{tD} B_x dx\right) = o(t^d), \quad \text{as } t \to \infty.$$

This fact will be needed in Example 4.4.6.

**Lemma 4.4.2.** Let  $(B_x)_{x \in \mathbb{R}^2}$  be the Berry's random wave model, and recall the notation introduced in the first part of Section 4.4. Then for  $q \geq 2$ , as  $t \to \infty$ , we have

$$w_{q,t} \sim \begin{cases} ct & \text{if } q = 2\\ c\log(t) & \text{if } q = 4\\ c & \text{if } q = 3 \text{ or } q \ge 5 \end{cases},$$

where  $c \in (0, \infty)$  depends on q, but does not depend on t.

*Proof.* In the sequel, let c be a positive constant which may change value. For q=2

even, by (4.41) and the trigonometric identity  $2\cos^2(x) = 1 + \cos(2x)$  we have

$$w_{2,t} = 2\pi \int_0^t J_0^2(r) r \, dr = c \left( \int_1^t \cos^2(r - \pi/4) dr + O\left( \int_0^t \frac{dr}{r} \right) \right)$$
  
=  $c \left( t - 1 + \int_1^t \cos(2r - \pi/2) dr + O\left( \int_0^t \frac{dr}{r} \right) \right)$   
=  $c \left( t + O(\log(t)) \right) \sim ct$ .

For q = 4, by (4.41) and the trigonometric identity  $8\cos^4(x) = 2(1 + \cos(2x))^2 = 2 + 4\cos(2x) + 2\cos^2(2x) = 2 + 4\cos(2x) + 1 + \cos(4x)$ , we have

$$w_{4,t} = 2\pi \int_0^t J_0^4(r)r \, dr = c \left( \int_1^t \frac{\cos^4(r - \pi/4)}{r} dr + O\left( \int_0^t \frac{dr}{r^2} \right) \right)$$

$$= c \left( 3\log(t) + 4 \int_1^t \cos(2r - \pi/2) dr + \int_1^t \cos(4r - \pi) dr + O\left( \int_0^t \frac{dr}{r^2} \right) \right)$$

$$= c \left( \log(t) + O(1) \right) \sim c \log(t).$$

If  $q \ge 6$  is even,  $w_{q,t} > 0$ ,  $w_{q,t}$  increasing in t, and  $w_{q,t} = O(1)$  (see Remark 4.4.2). Therefore, we have  $w_{q,t} \sim c > 0$  for some positive constant c > 0.

If  $q \geq 3$  is odd, the fact that  $w_{q,t}$  converges to a positive constant is more difficult to see, and is proved in [GMT24] by means of Pearson's random walks.

**Remark 4.4.2.** Note that since  $|C(z)| = |J_0(|z|)| \le 1$  and (4.41) holds, we have

$$\left|J_0^5(r)\right| \leq c\,\left(1\,\wedge\,\frac{1}{r^{5/2}}\right), \quad r>0,$$

where c is a positive constant. Moreover, since  $|C(z)| = |J_0(|z|)| \le 1$  we have

$$\int_{\mathbb{R}^d} |C^q(z)| \, dz \le \int_{\mathbb{R}^d} \left| C^5(z) \right| \, dz = 2\pi \int_0^\infty \left| J_0^5(r) \right| r \, dr \le c \int_1^\infty \frac{dr}{r^{3/2}} < \infty, \qquad q \ge 5.$$

Therefore, we obtain the uniform bound

$$\sum_{q=5}^{\infty} a_q^2 w_{q,t} \le \|\varphi\|_{L^2(\mathbb{R}_+,\gamma(dx))}^2 \int_{\mathbb{R}^d} \left| C^5(x) \right| dx < \infty.$$
 (4.42)

Combining Lemma 4.4.2 with Theorem 4.1.1, we are now able to compute the asymptotic covariances

$$\lim_{t\to\infty} \operatorname{Cov}\left(\frac{\int_{tD} \varphi(B_x) dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} \varphi(B_y) dy}{t^{d/2} w_t^{1/2}}\right),$$

where  $B=(B_x)_{x\in\mathbb{R}^2}$  is the Berry's random wave model, for a large class of functions  $\varphi$  and compact domains D,L. If we further apply [MN24, Theorem 2] and the Peccati-Tudor multi-dimensional fourth moment theorem (see [PT05]), we are able to prove multi-dimensional central limit theorems for functionals of the Berry's random wave model. All these facts are explained in the upcoming Examples 4.4.2-4.4.6. Before we move on to the latter, to ease the exposition, let us state a simplified version of

the Peccati-Tudor theorem in our context.

**Theorem 4.4.3.** [PT05, Proposition 1] Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a continuous, centered, stationary Gaussian field with covariance function C. Let  $\mathcal{D}'$  be a class of compact sets in  $\mathbb{R}^d$ . Assume that the following three conditions hold:

- 1. Var  $(\int_{tD} H_q(B_x) dx) > 0$ , for every  $D \in \mathcal{D}'$  and t large enough.
- 2. For every  $D \in \mathcal{D}'$

$$\frac{\int_{tD} H_q(B_x) dx}{\sqrt{\operatorname{Var}\left(\int_{tD} H_q(B_x) dx\right)}} \stackrel{law}{\to} N,$$

where  $N \sim N(0,1)$  has a standard Gaussian distribution.

3. For every  $D, L \in \mathcal{D}'$ , we have

$$\operatorname{Cov}\left(\frac{\int_{tD} H_q(B_x) dx}{\sqrt{\operatorname{Var}\left(\int_{tD} H_q(B_x) dx\right)}}, \frac{\int_{tL} H_q(B_y) dy}{\sqrt{\operatorname{Var}\left(\int_{tL} H_q(B_y) dy\right)}}\right) \to b(D, L),$$

where b(D, L) is a constant depending only on D and L.

Then, as  $t \to \infty$  we have the multi-dimensional central limit theorem

$$\left(\frac{\int_{tD} H_q(B_x) dx}{\sqrt{\operatorname{Var}\left(\int_{tD} H_q(B_x) dx\right)}}\right) \xrightarrow{f.d.d.} G = (G(D))_{D \in \mathcal{D}'}$$

where G is a centered, set-indexed Gaussian field with covariances

$$Cov(G(D), G(L)) = b(D, L).$$

**Example 4.4.2.** (Berry with  $R \ge 5$ .) If  $R \ge 5$ , then  $C \in L^R(\mathbb{R}^2)$  and we are in the Breuer-Major case of Section 4.4.1. In particular, by Lemma 4.4.2 we have that  $\sigma^2 > 0$  and  $w_t$  is slowly varying (see (4.42)). Therefore, the asymptotic covariances are given by (4.34) or (applying Theorem 4.1.1) by (4.38), and Gaussian fluctuations follow by multi-dimensional Breuer-Major theorems of the form (4.36)-(4.37).

**Example 4.4.3.** (Berry with R=4.) When R=4, note that K(z)=k(|z|) is radial, non-integrable and k is not regularly varying, implying that we are not in the classical cases discussed in Sections 4.4.1 and 4.4.2. Despite this,  $w_t$  is slowly varying (see Lemma 4.4.2, (4.33) and (4.42)) and in particular

$$w_t \sim a_4^2 w_{4,t} \sim c \log(t)$$
 as  $t \to \infty$ ,

where c is a positive constant. Then, Theorem 4.1.1 can be applied to obtain the asymptotic covariances

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} \varphi(B_x) dx}{t^{d/2} w_{\star}^{1/2}}, \frac{\int_{tL} \varphi(B_y) dy}{t^{d/2} w_{\star}^{1/2}}\right) = \operatorname{Vol}(D \cap L), \tag{4.43}$$

for every  $D, L \in \mathcal{D}$ , where  $\mathcal{D}$  is the class of compact sets with finite perimeter introduced in (4.13). Moreover, by using reduction techniques (see e.g. the proof of

[MN24, Proposition 4]) in (4.44) and the spectral CLT [MN24, Theorem 2] in (4.45), we have for  $D \subseteq \mathbb{R}^2$  compact with Vol(D) > 0:

$$\mathbb{E}\left[\left(\frac{\int_{tD}\varphi(B_x)dx}{\sqrt{\operatorname{Var}\left(\int_{tD}\varphi(B_x)dx\right)}} - sgn(a_4)\frac{\int_{tD}H_4(B_x)dx}{\sqrt{\operatorname{Var}\left(\int_{tD}H_4(B_x)dx\right)}}\right)^2\right] \to 0 \tag{4.44}$$

and

$$\frac{\int_{tD} H_4(B_x) dx}{\sqrt{\operatorname{Var}\left(\int_{tD} H_4(B_x) dx\right)}} \stackrel{law}{\to} N(0,1). \tag{4.45}$$

Combining (4.43), (4.44), (4.45) and Theorem 4.4.3 with  $\mathcal{D}' = \mathcal{D}$ , we obtain

$$\left(\frac{\int_{tD} \varphi(B_x) dx}{t^{d/2} w_t^{1/2}}\right)_{D \in \mathcal{D}} \stackrel{f.d.d.}{\to} (G(D))_{D \in \mathcal{D}},\tag{4.46}$$

where  $G = (G(D))_{D \in \mathcal{D}}$  is a Gaussian noise with covariances given by (4.43). Note that we have proved (4.46) on  $\mathcal{D}$ , excluding the compact sets with infinite perimeter, since in the latter cases Theorem 4.1.1 can not be applied as in (4.43) to verify condition (iii) of Theorem 4.4.3. Note also that (4.46) could have been obtained (for a less general class of domains  $\mathcal{A}$  defined in [PV20]) combining the results proved in [PV20] for the nodal lengths of the Berry's random wave model and the reduction principle in [Vid21].

**Example 4.4.4.** (Berry with R=3.) If  $a_4 \neq 0$ , then  $w_t \sim a_4^2 w_{4,t}$  and we can reason exactly as in Example 4.4.3. So we can assume that R=3 and  $a_4=0$ . If this is the case, by Lemma 4.4.2, (4.33) and (4.42) we have that  $w_t$  converges to a positive constant, so that it is slowly varying. Then, by Theorem 4.1.1 we have for every D, L in the class  $\mathcal{D}$  of compact sets with finite perimeter (see (4.13))

$$\lim_{t\to\infty}\operatorname{Cov}\left(\frac{\int_{tD}\varphi(B_x)dx}{t^{d/2}w_{\scriptscriptstyle t}^{1/2}},\frac{\int_{tL}\varphi(B_y)dy}{t^{d/2}w_{\scriptscriptstyle t}^{1/2}}\right)=\operatorname{Vol}(D\cap L).$$

Note that in this case  $C \notin L^3(\mathbb{R}^d)$ , so we are not in the Breuer-Major case, where the covariances can be obtained for D,L compact domains as in (4.34). In fact, in this case  $\int_{\mathbb{R}^d} C^3(z) dz$  is only conditionally convergent, but not absolutely convergent. Nevertheless, excluding compact domains with infinite perimeter, we can compute the asymptotic covariances using Theorem 4.1.1.

To the best of our knowledge, limit theorems in the case R=3 and  $a_4=0$  have never been proved.

**Example 4.4.5. (Berry with** R = 2.) When R = 2, we have (by Lemma 4.4.2, (4.33) and (4.42)), c being a positive constant,

$$w_t \sim a_2^2 w_{2,t} \sim c t$$
 as  $t \to \infty$ .

Again, we are not in the classical cases discussed in Section 4.4.1 or 4.4.2, but  $w_t$  is regularly varying with index  $\alpha = 1$ . Then, by Theorem 4.1.1 we have for every D, L

in the class  $\mathcal{D}$  of compact sets with finite perimeter (see (4.13))

$$\lim_{t \to \infty} \operatorname{Cov}\left(\frac{\int_{tD} \varphi(B_x) dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} \varphi(B_y) dy}{t^{d/2} w_t^{1/2}}\right) = \frac{1}{2\pi} \int_{S^1} d\theta \int_0^\infty dl \left(-\frac{d}{dl} g_{D,L}(l\theta)\right) l \quad (4.47)$$

$$= \frac{1}{2\pi} \int_D \int_L \frac{dx dy}{|x - y|}. \quad (4.48)$$

For the last equality, see Remark 4.1.2. Note that the integral on the RHS of (4.47) is finite, since  $g_{D,L}$  has bounded derivative and compact support if D, L are compact sets with finite perimeter (see Proposition 4.2.2).

Let us define the following class of compact sets:

$$\mathcal{D}_O := \left\{ D \subseteq \mathbb{R}^d \text{ compact } : \operatorname{Vol}(D) > 0, |\mathcal{F}[\mathbf{1}_D](x)| = O(|x|^{-d/2}) \text{ as } |x| \to \infty \right\},$$

where  $\mathcal{F}$  denotes the Fourier transform. For example,  $D \in \mathcal{D}_O$  when D is compact,  $D = \mathring{D}$  and  $\partial D$  is smooth with non-vanishing Gaussian curvature, see [BHI03]. By using reduction techniques (see e.g. the proof of [MN24, Proposition 4]) in (4.49) and the spectral CLT [MN24, Theorem 2], we obtain for every  $D \in \mathcal{D}_O$ :

$$\mathbb{E}\left[\left(\frac{\int_{tD}\varphi(B_x)dx}{\sqrt{\operatorname{Var}\left(\int_{tD}\varphi(B_x)dx\right)}} - sgn(a_2)\frac{\int_{tD}H_2(B_x)dx}{\sqrt{\operatorname{Var}\left(\int_{tD}H_2(B_x)dx\right)}}\right)^2\right] \to 0 \tag{4.49}$$

and

$$\frac{\int_{tD} H_2(B_x) dx}{\sqrt{\operatorname{Var}\left(\int_{tD} H_2(B_x) dx\right)}} \stackrel{law}{\to} N(0, 1). \tag{4.50}$$

Moreover, combining (4.47), (4.49), (4.50) and Theorem 4.4.3 with  $\mathcal{D}' = \mathcal{D} \cap \mathcal{D}_O$ , we obtain the multi-dimensional central limit theorem

$$\left(\frac{\int_{tD}\varphi(B_x)dx}{t^{d/2}w_t^{1/2}}\right)_{D\in\mathcal{D}\cap\mathcal{D}_O} \stackrel{f.d.d.}{\to} (G'(D))_{D\in\mathcal{D}\cap\mathcal{D}_O}, \tag{4.51}$$

where the set-indexed Gaussian field  $G' = (G'(D))_{D \in \mathcal{D} \cap \mathcal{D}_O}$  has covariances given by (4.47). Note that the restriction on  $\mathcal{D}$  in (4.51) is needed to exclude the compact sets with infinite perimeter, since in the latter cases Theorem 4.1.1 can not be applied as in (4.47) to verify condition (iii) of Theorem 4.4.3. Moreover, the restriction on  $\mathcal{D}_O$  in (4.51) is needed to conclude (4.50) by means of Theorem [MN24, Theorem 2], verifying the condition (ii) in Theorem 4.4.3.

**Example 4.4.6.** (Berry with R=1.) If  $\varphi$  is linear, then  $\varphi(x)=a_1x=a_1H_1(x)$ , where  $H_1(x)=x$  is the first Hermite polynomial and  $a_1\neq 0$  by definition of Hermite rank R. In this case,  $w_t=a_1^2w_{1,t}$  is not regularly varying, see Remark 4.4.1. Therefore, we can not apply Theorem 4.1.1 to compute the asymptotic covariances.

Let us now assume that  $\varphi(x)$  is **not linear**, splitting the functional as follows

$$\int_{tD} \varphi(B_x) dx = a_1 \int_{tD} H_1(B_x) dx + \int_{tD} \hat{\varphi}(B_x) dx,$$

where  $\hat{\varphi} := \varphi - a_1 H_1 \neq 0$  because  $\varphi$  is not linear (i.e.  $\exists q \geq 2$  with  $a_q \neq 0$ ). In

addition, recall the notation  $\mathcal{D}$  for the class of compact sets with finite perimeter, and consider

$$\mathcal{D}_o := \left\{ D \subseteq \mathbb{R}^d \text{ compact } : \operatorname{Vol}(D) > 0, |\mathcal{F}[\mathbf{1}_D](x)| = o(|x|^{-d/2}) \text{ as } |x| \to \infty \right\}.$$

For example,  $D \in \mathcal{D}_o$  when D is compact,  $D = \overset{\circ}{D}$  and  $\partial D$  is smooth with non-vanishing Gaussian curvature, see [BHI03].

Then, we can study the asymptotic covariances and multi-dimensional fluctuations of  $\int_{tD} \varphi(B_x) dx$ , as we did in the previous examples, following two steps:

1. We observe that as  $t \to \infty$  (note that the equality holds by definition of  $\hat{\varphi}$ )

$$\mathbb{E}\left[\left(\frac{\int_{tD}\varphi(B_{x})dx}{t^{d/2}\left(\sum_{q=2}^{\infty}a_{q}^{2}w_{q,t}\right)^{1/2}} - \frac{\int_{tD}\hat{\varphi}(B_{x})dx}{t^{d/2}\left(\sum_{q=2}^{\infty}a_{q}^{2}w_{q,t}\right)^{1/2}}\right)^{2}\right]$$
(4.52)

$$= \frac{\operatorname{Var}\left(\int_{tD} a_1 H_1(B_x) dx\right)}{t^d \sum_{o=2}^{\infty} a_o^2 w_{q,t}} \to 0, \qquad \forall D \in \mathcal{D}_o.$$

$$(4.53)$$

In fact, by Remark 4.4.1 we have (note that  $t^d = O\left(t^d \sum_{q=2}^{\infty} a_q^2 w_{q,t}\right)$  by Lemma 4.4.2, since  $\exists q \geq 2$  with  $a_q \neq 0$ )

$$\operatorname{Var}\left(a_1 \int_{tD} H_1(B_x) dx\right) = o\left(t^d \sum_{q=2}^{\infty} a_q^2 w_{q,t}\right), \quad \forall D \in \mathcal{D}_o.$$

2. Since the asymptotic  $L^2$ -equivalence (4.52) holds, the problem of studying the asymptotic covariances and the multi-dimensional Gaussian fluctuations of  $\int_{tD} \varphi(B_x) dx$  is reduced to the same problem for  $\int_{tD} \hat{\varphi}(B_x) dx$ . Hence, since  $\hat{\varphi}$  has Hermite rank greater or equal than 2, one can solve the problem using the results in the previous examples. For instance, if the Hermite rank of  $\hat{\varphi}$  is 2, then by Example 4.4.5 we have

$$\left(\frac{\int_{tD} \hat{\varphi}(B_x) dx}{t^{d/2} \left(\sum_{q=2}^{\infty} a_q^2 w_{q,t}\right)^{1/2}}\right)_{D \in \mathcal{D} \cap \mathcal{D}_O} \stackrel{f.d.d.}{\to} (G'(D))_{D \in \mathcal{D} \cap \mathcal{D}_O},$$

and combining the latter with (4.52) we obtain (note that  $\mathcal{D}_o \subseteq \mathcal{D}_O$ , so we add an additional restriction on the class of domains in order to use (4.52)).

$$\left(\frac{\int_{tD} \varphi(B_x) dx}{t^{d/2} \left(\sum_{q=2}^{\infty} a_q^2 w_{q,t}\right)^{1/2}}\right)_{D \in \mathcal{D} \cap \mathcal{D}_o} \stackrel{f.d.d.}{\to} (G'(D))_{D \in \mathcal{D} \cap \mathcal{D}_o},$$

where the set-indexed Gaussian field  $G' = (G'(D))_{D \in \mathcal{D} \cap \mathcal{D}_o}$  has covariances given by (4.47).

**Remark 4.4.3.** Note that all the arguments above could be generalized to prove analogous multi-dimensional central limit theorems for the d-dimensional Berry's random wave model (and partially extended to a larger class of fields, see [MN24, Section 6]). We only focused on the case d = 2 for the sake of brevity.

## 4.5 Non-radial covariance functions.

When  $d \geq 2$ , the covariance function K of the field A could be non-radial, but Theorem 4.1.1 can still be applied taking  $\mathcal{D}$  as the set of balls centered at a fixed point, see (4.14).

The goal of this section is showing that the problem (4.3) for non-radial, continuous K can be reduced (for D, L balls centered at a fixed point) to the same problem for a radial covariance function  $K_{iso}$ , which will be defined in the sequel. This reduction principle will be proved in Proposition 4.5.1.

In order to state Proposition 4.5.1, we need to introduce some additional quantities.

Let  $A=(A_x)_{x\in\mathbb{R}^d}$  be a measurable, weakly stationary random field with **continuous** covariance function  $K:\mathbb{R}^d\to\mathbb{R}$ , K(0)=1. By Bochner's theorem, there exists a unique, symmetric, Borel probability measure G on  $\mathbb{R}^d$  such that

$$K(z) = \int_{\mathbb{R}^d} e^{i\langle x, \lambda \rangle} G(d\lambda).$$

G is called the **spectral measure** of A (or associated to K). In other words, K is the characteristic function of  $\Lambda$ , where  $\Lambda \sim G$  is a random variable with values in  $\mathbb{R}^d$ .

Note that we can always write

$$K(z) = \mathbb{E}\left[e^{i\langle z,\Lambda\rangle}\right] = \mathbb{E}\left[e^{i|\Lambda|\langle z,\hat{\Lambda}\rangle}\right],$$

where  $|\Lambda| \sim \mu$  and  $\hat{\Lambda} := \mathbf{1}_{\Lambda \neq 0} \Lambda / |\Lambda| \sim \sigma$  are respectively the random norm and the random direction of  $\Lambda$ . We will refer respectively to the probability measures  $\mu$  on  $\mathbb{R}_+$  and  $\sigma$  on  $S^{d-1}$  as the **isotropic spectral measure** and **spherical spectral measure** of A (or associated to K or G).

**Remark 4.5.1.** Let us denote by  $\nu$  the uniform probability measure on  $S^{d-1}$ . Note that K is radial if and only if  $\sigma = \nu$ . Indeed, K is radial, i.e. K(z) = K(Pz) for every P orthogonal matrix, if and only if  $P\Lambda \sim \Lambda$  for every P orthogonal matrix, that is  $\hat{\Lambda}$  is uniformly distributed on  $S^{d-1}$  (i.e.  $\sigma = \nu$ ). In particular, given a probability measure  $\mu$  on  $\mathbb{R}_+$ , there is a unique continuous, radial covariance function with isotropic spectral measure  $\mu$  (and spherical spectral measure  $\nu$ ).

Thanks to Remark 4.5.1, we can now give the crucial definition of this section.

**Definition 4.5.1.** Let  $A=(A_x)_{x\in\mathbb{R}^d}$  be a weakly stationary random field with continuous covariance function  $K:\mathbb{R}^d\to\mathbb{R}$ , isotropic spectral measure  $\mu$  and spherical spectral measure  $\sigma$ , that is

$$K(z) = \int_{0}^{\infty} \mu(dr) \int_{S^{d-1}} \sigma(d\theta) \, e^{i\langle r\theta, z\rangle}.$$

The **isotropic covariance function** of A is the only continuous, radial covariance function  $K_{iso}: \mathbb{R}^d \to \mathbb{R}$  with isotropic spectral measure  $\mu$ , namely (see Remark 4.5.1)

$$K_{\mathrm{iso}}(z) := \int_0^\infty \mu(dr) \int_{S^{d-1}} \nu(d\theta) \, e^{i\langle r\theta, z \rangle},$$

where  $\nu(d\theta) = d\theta/\omega_{d-1}$  is the uniform probability measure on  $S^{d-1}$ .

The main result of this section is the following.

**Proposition 4.5.1.** Let  $A = (A_x)_{x \in \mathbb{R}^d}$  be a measurable, weakly stationary random field with continuous covariance function  $K : \mathbb{R}^d \to \mathbb{R}$  and isotropic covariance function  $K_{iso}$ . Then we have (recall the definition (4.1) of w)

$$w_t^{iso} := \int_{\{|z| \le t\}} K_{iso}(z) dz = \int_{\{|z| \le t\}} K(z) dz = w_t, \qquad t > 0, \tag{4.54}$$

and for D, L balls centered at the same point

$$\int_{D} \int_{L} K(x - y) dx dy = \int_{D} \int_{L} K_{iso}(x - y) dx dy.$$
 (4.55)

In particular, if  $w_t = w_t^{\text{iso}}$  is regularly varying with index  $\alpha \in (-1, d]$  (see (4.15)), by Theorem 4.1.1 we have, for D, L balls centered at the same point (see (4.14)), as  $t \to \infty$ 

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} w_t^{1/2}}, \frac{\int_{tL} A_x dx}{t^{d/2} w_t^{1/2}}\right) = \frac{\int_{tD} \int_{tL} K(x - y) dx dy}{t^d w_t}$$

$$= \frac{\int_{tD} \int_{tL} K_{iso}(x - y) dx dy}{t^d w_t^{iso}}$$

$$\to \frac{1}{\omega_{d-1}} \int_{S^{d-1}} d\theta \int_0^\infty dl \left(-\frac{d}{dl} g_{D,L}(l\theta)\right) l^\alpha. \tag{4.57}$$

*Proof.* By Fubini, we have

$$\begin{split} w_t &= \int_{\{|z| \leq t\}} K(z) dz = \int_{S^{d-1}} \sigma(d\theta) \int_0^\infty \mu(dr) \int_{\{|z| \leq t\}} e^{i\langle z, r\theta \rangle} \\ &= \int_{S^{d-1}} \sigma(d\theta) \int_0^\infty \mu(dr) \mathcal{F}[\mathbf{1}_{\{|z| \leq t\}}](r\theta), \end{split}$$

where  $\mathcal{F}$  denotes the Fourier transform. Since  $\mathcal{F}[\mathbf{1}_{\{|z|\leq t\}}](r\theta)$  is a radial function (i.e. does not depend on  $\theta$ ) and  $\sigma, \nu$  are probability measures on  $S^{d-1}$ , we have

$$\int_{S^{d-1}} \sigma(d\theta) \int_0^\infty \mu(dr) \mathcal{F}[\mathbf{1}_{\{|z| \le t\}}](r\theta) = \int_{S^{d-1}} \nu(d\theta) \int_0^\infty \mu(dr) \mathcal{F}[\mathbf{1}_{\{|z| \le t\}}](r\theta)$$
$$= \int_{\{|z| \le t\}} K_{iso}(z) dz = w_t^{iso},$$

where the last equality follows again by Fubini. Therefore, (4.54) is proved. As a consequence of (4.54) and (4.27), we also obtain (4.55). Moreover, (4.56) follows by (4.54), (4.55) and a direct application of Theorem 4.1.1.

Proposition 4.5.1 allows to reduce the problem (4.3) for K non-radial (and D, L balls centered at the same point) to the easiest problem (4.3) for  $K_{iso}$  radial.

Note that we can also make the argument in the opposite direction: if  $K_{\rm iso}$  is a radial covariance function and we can check (4.15) for  $K_{\rm iso}$ , then by Proposition 4.5.1 we can apply Theorem 4.1.1 not only for random fields with covariance function  $K_{\rm iso}$ , but for every random field with isotropic covariance function  $K_{\rm iso}$ .

To be more concrete about the possible applications of Proposition 4.5.1, we conclude with the following example.

**Example 4.5.1.** Fix  $d \geq 2$  and consider the **non-Gaussian** random field  $A = (A_x)_{x \in \mathbb{R}^d}$  given by the rescaled sum of N independent random waves with random phases  $\varphi_i$ , random directions  $\theta_i$  and random wavenumbers  $k_i$ . Namely, we have

$$A_x := \sqrt{\frac{2}{N}} \sum_{i=1}^{N} \cos(k_i \langle x, \theta_i \rangle + \varphi_i), \qquad (4.58)$$

where  $\varphi_1, \theta_1, k_1, \ldots, \varphi_N, \theta_N, k_N$  are all independent and for all  $i = 1, \ldots, N$  we have:  $\varphi_i$  is uniformly distributed on  $[0, 2\pi]$ ;  $\theta_i \sim \sigma$  symmetric probability measure on  $S^{d-1}$ ;  $k_i \sim \mu$  probability measure on  $(0, \infty)$ .

Note that when  $\sigma$  is the uniform distribution on  $S^{d-1}$  and  $\mu = \delta_k$  for some constant k > 0, random fields of the form (4.58) are called random superposition of independent waves, and have been extensively studied as good local models for wavefunctions (i.e. eigenfunctions of the Laplacian), see e.g. [Ber77; NPR19; PV20; Vid21].

Moreover, note that A is **not stationary**, since  $A_x$  and  $A_y$  have not the same distribution in general. For instance, if N=1, in general  $A_0=\cos(\varphi_1)$  has not the same distribution of  $A_x=\cos(k_1\langle x,\theta_1\rangle+\varphi_1)$  for every  $x\neq 0$ .

However, *A* is **weakly stationary**, implying that Theorem 4.1.1 can be applied to compute the asymptotic covariances of functionals of *A*. Indeed, we have

$$\mathbb{E}[A_x] = \sqrt{2N} \mathbb{E}\left[\cos\left(k_1\langle x, \theta_1\rangle + \varphi_1\right)\right]$$

$$= \sqrt{2N} \left( \mathbb{E}\left[\cos\left(k_1\langle x, \theta_1\rangle\right)\right] \underbrace{\mathbb{E}\left[\cos(\varphi_1)\right]}_{=0} - \mathbb{E}\left[\sin\left(k_1\langle x, \theta_1\rangle\right)\right] \underbrace{\mathbb{E}\left[\sin(\varphi_1)\right]}_{=0} \right) = 0$$

and

$$\operatorname{Cov}(A_{x}, A_{y}) = \mathbb{E}\left[A_{x}A_{y}\right] = 2\mathbb{E}\left[\cos\left(k_{1}\langle x, \theta_{1}\rangle + \varphi_{1}\right)\cos\left(k_{1}\langle y, \theta_{1}\rangle + \varphi_{1}\right)\right]$$

$$= 2\mathbb{E}\left[\cos\left(k_{1}\langle x, \theta_{1}\rangle\right)\cos\left(k_{1}\langle y, \theta_{1}\rangle\right)\right] \mathbb{E}\left[\cos^{2}(\varphi_{1})\right]$$

$$= 1/2$$

$$+ 2\mathbb{E}\left[\sin\left(k_{1}\langle x, \theta_{1}\rangle\right)\sin\left(k_{1}\langle y, \theta_{1}\rangle\right)\right] \mathbb{E}\left[\sin^{2}(\varphi_{1})\right]$$

$$= 1/2$$

$$- 2\mathbb{E}\left[\cos\left(k_{1}\langle x, \theta_{1}\rangle\right)\sin\left(k_{1}\langle y, \theta_{1}\rangle\right)\right] \mathbb{E}\left[\cos(\varphi_{1})\sin(\varphi_{1})\right]$$

$$= 0$$

$$- 2\mathbb{E}\left[\sin\left(k_{1}\langle x, \theta_{1}\rangle\right)\cos\left(k_{1}\langle y, \theta_{1}\rangle\right)\right] \mathbb{E}\left[\sin(\varphi_{1})\cos(\varphi_{1})\right]$$

$$= 0$$

$$= \mathbb{E}\left[\cos\left(k_{1}\langle x - y, \theta_{1}\rangle\right)\right].$$

Therefore, denoting by K the covariance function of A, by the symmetry of  $\theta_1$  we have

$$K(z) = \mathbb{E}\left[\cos\left(k_1\langle z, \theta_1\rangle\right)\right] = \mathbb{E}\left[e^{ik_1\langle z, \theta_1\rangle}\right] = \int_{\mathbb{R}_+ \times S^{d-1}} e^{i\langle z, r\theta\rangle} (\mu \times \sigma)(dr, d\theta),$$

implying that the isotropic (resp. spherical) spectral measure of A is  $\mu$  (resp.  $\sigma$ ). By Proposition 4.5.1 the integral covariance function w defined in (4.1) does not depend on the spherical spectral measure. In other words, whatever is the distribution of the random directions  $\theta_1, \ldots, \theta_N$  fixed in the definition (4.58) of A, one can apply Theorem 4.1.1 with the isotropic covariance function of A, i.e.

$$K_{\rm iso}(z) = \int_{\mathbb{R}_+ \times S^{d-1}} e^{i\langle z, r\theta \rangle} (\mu \times \nu) (dr, d\theta),$$

where  $\nu$  is the uniform probability measure on  $S^{d-1}$ . More precisely, whatever is the covariance function K, whatever is the spherical spectral measure  $\sigma$ , if  $K_{\rm iso}$  satisfies

$$w_t^{\mathrm{iso}} = \int_{\{|z| < t\}} K_{\mathrm{iso}}(z) dz = \ell(t) t^{\alpha}, \qquad \alpha \in (-1, d],$$

with  $\ell : \mathbb{R}_+ \to \mathbb{R}$  slowly varying, we have for D, L balls centered at the same point (see (4.14)), as  $t \to \infty$ 

$$\operatorname{Cov}\left(\frac{\int_{tD} A_x dx}{t^{d/2} (w_t^{\mathsf{iso}})^{1/2}}, \frac{\int_{tL} A_x dx}{t^{d/2} (w_t^{\mathsf{iso}})^{1/2}}\right) \to \frac{1}{\omega_{d-1}} \int_{S^{d-1}} d\theta \int_0^\infty dl \left(-\frac{d}{dl} g_{D,L}(l\theta)\right) l^\alpha.$$

# Limit theorems for p-domain functionals of stationary Gaussian fields

This chapter contains the preprint [Leo+24], "Limit theorems for p-domain functionals of stationary Gaussian fields", written in collaboration with Nikolai Leonenko, Ivan Nourdin and Francesca Pistolato. arXiv (2024+): 2402.16701.

## 5.1 Introduction

Gaussian fields are widely used to model random quantities arising in different applications, see e.g. [Chr92; DR07; Gne02; LRM23]. These random quantities may depend for instance on space  $x_1 \in \mathbb{R}^{d_1}$  and time  $x_2 \in \mathbb{R}_+$  (e.g. [Gne02]) or more generally on several variables  $x_1, x_2, x_3, \ldots$  belonging to (possibly non-Euclidean) spaces of different dimensions (e.g. [BPP22; MRV21; MRV24]).

Throughout this paper, we consider a continuous, stationary, real-valued, centered Gaussian field  $B=(B_x)_{x\in\mathbb{R}^d}$  with unit variance, where  $d\geq 2$ . We denote by  $C:\mathbb{R}^d\to\mathbb{R}$  the **covariance function** of B, defined as

$$Cov(B_x, B_y) = C(x - y), \quad x, y \in \mathbb{R}^d.$$

We also consider a non constant measurable function  $\varphi: \mathbb{R} \to \mathbb{R}$  satisfying  $\mathbb{E}[\varphi^2(N)] < \infty$  for  $N \sim N(0,1)$ . As is well-known,  $\varphi$  admits an Hermite decomposi-

tion (see, e.g. [NP12, Section 1.4]) of the form

$$\varphi = \mathbb{E}[\varphi(N)] + \sum_{q=R}^{\infty} a_q H_q, \quad \text{with } R \ge 1 \text{ such that } a_R \ne 0,$$
 (5.1)

where  $H_q$  is the qth Hermite polynomial and  $a_q = a_q(\varphi) = \frac{1}{q!} \mathbb{E}\left[\varphi(N) H_q(N)\right]$ . The integer  $R \geq 1$  is called the **Hermite rank** of  $\varphi$ .

Finally, we consider a family of compact subsets  $D_i \subseteq \mathbb{R}^{d_i}$ ,  $1 \le i \le p$ , satisfying  $\operatorname{Vol}(D_i) > 0$  for each i. The number  $d = d_1 + \cdots + d_p$  is called the total dimension.

The main object of interest of our paper is the additive functional

$$Y(t_1, \dots, t_p) := \int_{t_1 D_1 \times \dots \times t_p D_p} \varphi(B_x) dx \quad \text{for } t_1, \dots, t_p > 0.$$
 (5.2)

We remark that the integral in (5.2) is well defined thanks to the continuity assumption made on B as well as the square integrability of  $\varphi$  with respect to the standard Gaussian measure (see e.g. [MN24, Proposition 3]).

Under sufficient conditions ensuring that  $\mathrm{Var}(Y(t_1,\ldots,t_p))>0$  for every  $t_1,\ldots,t_p>0$  large enough, we study the limit in distribution as  $t_1,\ldots,t_p\to\infty$  of the normalized version of the functional  $Y(t_1,\ldots,t_p)$  defined as

$$\widetilde{Y}(t_1, \dots, t_p) := \frac{Y(t_1, \dots, t_p) - \mathbb{E}[Y(t_1, \dots, t_p)]}{\sqrt{\operatorname{Var}(Y(t_1, \dots, t_p))}}.$$
(5.3)

While the case p=1 has been extensively studied since the eighties (see e.g. the seminal works [BM83], [DM79], [Ros60], [Taq79]), not much literature can be found when the integration domain of (5.2) does not grow uniformly with respect to all its directions (that is, when  $p \geq 2$  and when  $t_1, \ldots, t_p$  go to infinity at possibly different rates). In fact, the extended class of functionals (5.2) has aroused interest only in the past decade (see e.g. [RST12], [PR16], [Bre11], [LRM23], [AO18], [PS17]), because of the more and more important role that **spatio-temporal** functionals of random fields or random fields with **separable covariance function** (e.g. the fractional Brownian sheet) play now in applications.

In our work, the number p of growing domains and their dimensions  $d_i$  can be arbitrary. This is why we refer to (5.2) as a p-domain (rather than a spatio-temporal) functional. To give at least one explicit motivation for studying the asymptotic behavior of this type of functionals, let us consider the case where  $\varphi = \mathbf{1}_{[a,\infty)}$ . It corresponds to the excursion volume of B at level  $a \in \mathbb{R}$  in the observation window  $t_1D_1 \times \cdots \times t_pD_p$ , namely

$$Y(t_1, ..., t_p) = \text{Vol}\left(\{(x_1, ..., x_p) \in t_1 D_1 \times ... \times t_p D_p : B_{(x_1, ..., x_p)} \ge a\}\right),$$

which is a (random) geometrical object that has been extensively studied in the literature, see e.g [AT07]. Our approach offers more flexibility than the usual one (corresponding to p=1), as it gives the possibility to study the excursion sets of B when the parameters are in domains  $t_1D_1, \ldots, t_pD_p$  that can grow at different rates.

In a broader setting, the aim of the present work is to enhance our understanding of the asymptotic behaviour of p-domain functionals associated with stationary

5.1. Introduction 113

Gaussian fields, when the covariance function has a specific form. In particular, we will show how, in some cases, this asymptotic behaviour can be simply obtained from that of 1-domain functionals, explaining in a new light and in a more systematic way some results from the recent literature (e.g. those contained in [RST12]).

# 5.1.1 Separable case

In this section, we investigate the asymptotic behavior of (5.3) when we assume that the covariance function of B is separable.

A covariance function  $C: \mathbb{R}^d \to \mathbb{R}$  is said **separable** when it can be written as  $C = C_1 \otimes \ldots \otimes C_p$ , that is, when

$$C(x_1, \dots, x_p) = \prod_{i=1}^p C_i(x_i), \quad x_i \in \mathbb{R}^{d_i}, \quad 1 \le i \le p,$$
 (5.4)

for some functions  $C_i:\mathbb{R}^{d_i}\to\mathbb{R}$  satisfying  $C_i(0)=1$  for each i. It is easy to check that C is non-negative definite if and only if  $C_i$  is non-negative definite for every i. Also, by stationarity of B on  $\mathbb{R}^d$ , we have that  $C_i$  is the covariance function of the field  $B^{(i)}:=(B_{(x_1,\ldots,x_i,\ldots,x_p)})_{x_i\in\mathbb{R}^{d_i}}$  for any fixed values of  $x_j, j\neq i$ . Since we are only interested in distributions, we can define the **marginal functionals** 

$$Y_i(t_i) := \int_{t_i D_i} \varphi(B_{x_i}^{(i)}) dx_i, \quad i = 1, \dots p,$$
 (5.5)

and their normalized versions

$$\widetilde{Y}_i(t_i) := \frac{Y_i(t_i) - \mathbb{E}[Y_i(t_i)]}{\sqrt{\operatorname{Var}(Y_i(t_i))}}, \quad i = 1, \dots p,$$
(5.6)

with the convention that, for any given i, the values of  $x_j$ ,  $j \neq i$ , in (5.5) are arbitrary but fixed.

Let us denote by  $\stackrel{d}{\rightarrow}$  the convergence in distribution. The main result of this section is the following.

**Theorem 5.1.1.** Let  $\varphi: \mathbb{R} \to \mathbb{R}$  be a measurable function satisfying  $\mathbb{E}[\varphi^2(N)] < \infty$  for  $N \sim N(0,1)$ , with Hermite rank  $R \geq 1$ . Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $C: \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B, assume it is separable in the sense of (5.4), and also that it satisfies, for each i:

$$C^R \geq 0$$
 and  $C_i \in \bigcup_{M=1}^{\infty} L^M(\mathbb{R}^{d_i}).$ 

Let us consider  $\widetilde{Y}$  given by (5.3) and  $\widetilde{Y}_i$  given in (5.6). Then, the following two assertions are equivalent:

(a) 
$$\widetilde{Y}_i(t_i) \stackrel{d}{\to} N(0,1)$$
 as  $t_i \to \infty$  for at least one  $i \in \{1,\ldots,p\}$ ;

(b) 
$$\widetilde{Y}(t_1,\ldots,t_p) \stackrel{d}{\to} N(0,1)$$
 as  $t_1,\ldots,t_p \to \infty$ .

**Remark 5.1.1.** The integrability assumptions on  $C_i$  for Theorem 5.1.1 to hold may be removed when  $\varphi = H_q$ ; moreover, in this case we also have **quantitative results** for the convergence in distribution (see Subsection 5.3.1).

**Remark 5.1.2.** Given a *discrete* stationary centered Gaussian field  $B = (B_k)_{k \in \mathbb{Z}^d}$  with unit variance, the definition of separable covariance function  $C : \mathbb{Z}^d \to \mathbb{R}$  is similar, i.e.  $C(z) = \prod_{i=1}^p C_i(z_i)$  with  $C_i : \mathbb{Z}^{d_i} \to \mathbb{R}$  for  $i = 1, \ldots, p$ , whereas the functionals (5.2) and (5.5) are defined as follows:

$$Y(n_1, \dots, n_p) = \sum_{k \in (\prod_{i=1}^p [0, n_i]^{d_i}) \cap \mathbb{Z}^d} \varphi(B_k), \qquad Y_i(n_i) = \sum_{k_i \in [0, n_i]^{d_i} \cap \mathbb{Z}^{d_i}} \varphi(B_{k_1, \dots, k_p}),$$

where  $n_i \in \mathbb{N}$  for every  $i = 1, \dots, p$ . In this setting, analogous results to Theorem 5.1.1 could be obtained, see e.g. Remark 5.3.1 where we highlight that Proposition 5.3.3 directly translates with straightforward modifications for a discrete Gaussian field.

**Remark 5.1.3.** The previous result provides a large class of fields B with long-range dependence, i.e. with covariance function  $C \notin L^R$ , such that the functional  $Y_t := Y(t, \ldots, t)$  exhibits Gaussian fluctuations around its mean. See e.g. Example 5.4.3. We refer to [MN24] and the references therein for a deeper analysis and further examples of this phenomenon.

Since central limit theorems for 1-domain functionals have been extensively explored in the literature, it is not difficult to imagine how useful and powerful the implication  $(a)\Rightarrow (b)$  in Theorem 5.1.1 can be. It is noteworthy that a specific instance of this implication had previously been observed in the papers [RST12] and [PR16]; however, it was restricted to a very specific context – rectangular increments of a fractional Brownian sheet – and was not part of a comprehensive systematic investigation, as we undertake in this work.

Since Theorem 5.1.1 establishes that (5.2) displays Gaussian fluctuations (if and only) if at least one of its marginal functionals does, a natural question arises: what happens when none of the  $Y_i(t_i)$ 's exhibits Gaussian fluctuations? We investigate this problem in the classical framework of regularly varying covariance functions.

Given two functions f and g, we write  $f(x) \sim g(x)$  to indicate that  $\lim_{\|x\| \to 0} f(x)g(x)^{-1} = 1$ . Recalling that R denotes the Hermite rank of  $\varphi$  (see (5.1)), we consider the following conditions:

•  $C_i$  is regularly varying with parameter  $-\beta_i \in (-d_i/R, 0)$ , that is

$$C_i(z_i) = \rho_i(\|z_i\|) = L_i(\|z_i\|)\|z_i\|^{-\beta_i} \qquad \beta_i \in (0, d_i/R),$$
 (5.7)

where  $L_i:(0,\infty)\to\mathbb{R}$  is slowly varying, i.e.  $L_i(rs)/L_i(s)\to 1$  as  $s\to\infty$ ,  $\forall r>0$ .

•  $C_i$  admits an absolutely continuous spectral measure  $G_i(d\lambda_i) = g_i(\lambda_i)d\lambda_i$  on  $\mathbb{R}^{d_i}$ , and for some constant  $c_i > 0$  we have

$$g_i(\lambda_i) \sim c_i L_i(1/\|\lambda_i\|) \|\lambda_i\|^{\beta_i - d_i}$$
 as  $\|\lambda_i\| \to 0$ . (5.8)

If (5.7)-(5.8) hold and  $R \ge 2$ , then it is known that  $\widetilde{Y}_i(t_i)$  converges in distribution to a non-Gaussian random variable (see Theorem 5.2.4). In the following Theorem 5.1.2, we give conditions so that Y has non-Gaussian fluctuations either.

5.1. Introduction 115

**Theorem 5.1.2.** Let  $\varphi: \mathbb{R} \to \mathbb{R}$  be a measurable function satisfying  $\mathbb{E}[\varphi^2(N)] < \infty$  for  $N \sim N(0,1)$ , with Hermite rank  $R \geq 1$  (in particular, we have  $a_R \neq 0$ , see (5.1)). Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $C: \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B, assume it is separable in the sense of (5.4), and satisfies

$$C^R \ge 0$$
 and (5.7)-(5.8) hold  $\forall i \in \{1, \dots, p\}.$ 

Let us consider  $\widetilde{Y}$  given by (5.3). Then, as  $t_1, \ldots, t_p \to \infty$ , we have

$$\widetilde{Y}(t_1, \dots, t_p) \stackrel{d}{\to} \frac{\operatorname{sgn}(\mathbf{a}_{\mathbf{R}}) I_{\nu_1 \times \dots \times \nu_p, R} \left( f_{R, t_1 D_1 \times \dots \times t_p D_p} \right)}{\sqrt{\operatorname{Var} \left( I_{\nu_1 \times \dots \times \nu_p, R} \left( f_{R, t_1 D_1 \times \dots \times t_p D_p} \right) \right)}},$$
(5.9)

where  $f_{R,t_1D_1 \times ... \times t_pD_p} : (\mathbb{R}^d)^R \to \mathbb{C}$  is defined by

$$f_{R,t_1D_1 \times \dots \times t_pD_p}(\lambda_1, \dots, \lambda_R) := \int_{t_1D_1 \times \dots \times t_pD_p} e^{i\langle x, \lambda_1 + \dots + \lambda_R \rangle} dx,$$
 (5.10)

the product measure  $\nu_1 \times \ldots \times \nu_p$  on  $\mathbb{R}^d$  is defined by  $\nu_i(dx_i) = \|x_i\|^{\beta_i - d_i} dx_i$  and  $I_{\nu_1 \times \ldots \times \nu_p, R}$  denotes the Rth Wiener-Itô integral with respect to a  $\sigma$ -finite measure (see Section 5.2). In particular, the limit (5.9) is not Gaussian as soon as  $R \geq 2$ .

Theorem 5.1.2 represents a generalization from 1-domain to p-domain of the celebrated Dobrushin-Major-Taqqu (see Theorem 5.2.4). We remark that depending on the choice of the domains, the function f can be computed explicitly (see e.g. Remark 5.2.1). In the particular case of the rectangular increments of a fractional Brownian sheet, we note that Theorem 5.1.2 for p=2 is already proved in [RST12] and [PR16].

**Remark 5.1.4.** Analogous results (both central and non-central) hold if we fix some of the domains, i.e. considering  $t_1D_1 \times \cdots \times t_{p-1}D_{p-1} \times D_p$  and  $Y(t_1, \dots, t_{p-1}, 1)$ . In this setting, it is possible to prove the following.

- Under the same assumptions of Theorem 5.1.1, the following two assertions are equivalent:
  - (a)  $\widetilde{Y}_i(t_i) \stackrel{d}{\to} N(0,1)$  as  $t_i \to \infty$  for at least one  $i \in \{1, \dots, p-1\}$ ;
  - (b)  $\widetilde{Y}(t_1, ..., t_{p-1}, 1) \stackrel{d}{\to} N(0, 1)$  as  $t_1, ..., t_{p-1} \to \infty$ .
- Analogously to Theorem 5.1.2, if (5.7)-(5.8) hold for  $i=1,\ldots,p-1$  and  $C^R \geq 0$ , if  $G_p$  is the spectral measure of  $C_p$ , then the convergence of Theorem 5.1.2 still holds replacing  $\nu_p$  with  $G_p$ , namely

$$\widetilde{Y}(t_1,\ldots,t_{p-1},1) \stackrel{d}{\to} \frac{\operatorname{sgn}(\mathbf{a}_{\mathbf{R}})I_{\nu_1 \times \ldots \times \nu_{p-1} \times G_p,R}\left(f_{R,D_1 \times \ldots \times D_p}\right)}{\sqrt{\operatorname{Var}\left(I_{\nu_1 \times \ldots \times \nu_{p-1} \times G_p,R}\left(f_{R,D_1 \times \ldots \times D_p}\right)\right)}}, t_1,\ldots,t_{p-1} \to \infty.$$

The proofs of these facts follow from similar arguments to the ones of Theorem 5.1.1 and Theorem 5.1.2 (see Section 5.3).

#### 5.1.2 Non separable case

It is easy to construct examples illustrating that the normal convergence of functionals  $\tilde{Y}_i(t_i)$  is in general not enough to determine the behavior of  $\tilde{Y}(t_1, \dots, t_p)$  when the separability of the covariance function (5.4) is dropped. See Example 5.5.1 for such a situation.

This is why we examine in Section 5.5 what can happen when we go beyond the separable case, by investigating two different classes: Gneiting covariance functions (Section 5.5.1) and additively separable covariance functions (Section 5.5.2).

Although not separable, the covariance functions belonging to the Gneiting class are wedged between two separable functions. It is therefore not surprising that a comparable phenomenon can still be proven in this context, see Theorem 5.5.1.

In contrast, the situation in the additively separable case is much more complicated. We still manage to prove a kind of "reduction" theorem, see Theorem 5.5.2, with however a big difference: the marginal functionals to be considered in the additively separable case are really different from the  $Y_i$ 's of Theorem 5.1.1.

We refer to Section 5 for details.

# 5.1.3 Plan of the paper

The paper is organized as follows. Section 5.2 contain some needed preliminaries. In Section 5.3 we prove Theorem 5.1.1 and Theorem 5.1.2. In Section 5.4 we provide several examples where our results apply, and we compare them with the existing literature. In Section 5.5 we go beyond the separability assumption by investigating two other frameworks. Finally, in the Appendix we prove some auxiliary results.

### 5.2 Preliminaries

In this section we briefly present selected results on Malliavin-Stein method and classical results for 1-domain functionals.

#### 5.2.1 Elements of Malliavin-Stein method

Theorem 5.1.1 is proved using the Fourth Moment Theorem by Nualart and Peccati (see [NP05]) and its quantitative version by Nourdin and Peccati (see [NP12, Theorem 5.2.6]). For all the missing details on Malliavin calculus we refer to [Nua06] or [NP12].

Let us fix a probability space  $(\Omega, \mathcal{F}, \mathbb{P})$ . Consider a continuous, stationary, centered Gaussian field  $B = (B_x)_{x \in \mathbb{R}^d}$  with unit variance and covariance function  $C : \mathbb{R}^d \to \mathbb{R}$ . We assume that  $\mathcal{F}$  is the  $\sigma$ -field generated by B. By continuity and stationarity of B, the covariance function C is continuous. As a result, Bochner's theorem yields the existence of a unique real, symmetric<sup>1</sup>, finite measure G on  $\mathbb{R}^d$  endowed with the Borel  $\sigma$ -algebra  $\mathcal{B}(\mathbb{R}^d)$ , called the spectral measure of B, satisfying

$$C(x) = \int_{\mathbb{R}^d} e^{i\langle x, \lambda \rangle} G(d\lambda), \quad x \in \mathbb{R}^d.$$
 (5.11)

<sup>&</sup>lt;sup>1</sup>In the sense that G(A) = G(-A) for every  $A \in \mathcal{B}(\mathbb{R}^d)$ .

5.2. Preliminaries 117

We define the real separable Hilbert space

$$\mathcal{H} := L^2(G) = \left\{ h : \mathbb{R}^d \to \mathbb{C} : \int_{\mathbb{R}^d} |h(\lambda)|^2 G(d\lambda) < \infty, \ \overline{h(\lambda)} = h(-\lambda) \right\},$$
 (5.12)

where  $|\cdot|$  denotes the complex norm, endowed with the inner product<sup>2</sup>

$$\langle h, g \rangle_{\mathcal{H}} = \int_{\mathbb{R}^d} h(\lambda) \overline{g(\lambda)} G(d\lambda) = \int_{\mathbb{R}^d} h(\lambda) g(-\lambda) G(d\lambda).$$
 (5.13)

Thanks to [NP12, Proposition 2.1.1] we may consider an isonormal Gaussian process X on  $\mathcal{H}$ , with covariance kernel

$$\mathbb{E}\left[X_G(h)X_G(g)\right] = \langle h, g \rangle_{\mathcal{H}}.\tag{5.14}$$

Moreover, by (5.11), the field  $(X_G(e_x))_{x\in\mathbb{R}^d}$ , where  $e_x:=\mathrm{e}^{i\langle x,\cdot\rangle}$ , is stationary and Gaussian, with covariance function C and spectral measure G. Hence, the two fields share the same distribution, that is

$$(X_G(e_x))_{x \in \mathbb{R}^d} \stackrel{d}{=} (B_x)_{x \in \mathbb{R}^d}.$$
 (5.15)

Since we study limit theorems in distribution, we assume from now on that  $B_x = X_G(e_x)$  for any  $x \in \mathbb{R}^d$ . For  $q \ge 1$ , we also define the qth Wiener chaos as the linear subspace of  $L^2(\Omega)$  generated by  $\{H_q(X_G(h)) : h \in \mathcal{H}\}$ , and for every  $h \in \mathcal{H}$  such that  $\|h\|_{\mathcal{H}} = 1$ , we define the qth Wiener-Itô integral as

$$I_q(h^{\otimes q}) = H_q(X_G(h)), \tag{5.16}$$

where  $h^{\otimes q}:(\mathbb{R}^d)^q\to\mathbb{R}$  is such that

$$h^{\otimes q}(x_1, \dots, x_q) = \prod_{l=1}^q h(x_l).$$
 (5.17)

By density the definition of  $I_q$  can be linearly extended to every function in the space  $\mathcal{H}^{\odot q}=L^2_s((\mathbb{R}^d)^q,G^{\otimes q})$  of the *symmetric* functions in  $\mathcal{H}^{\otimes q}=L^2((\mathbb{R}^d)^q,G^{\otimes q})$ . This allows one to write

$$\int_{D} H_q(B_x) dx = I_q(f), \tag{5.18}$$

where  $f:(\mathbb{R}^d)^q\to\mathbb{R}$  is given by

$$f(\lambda_1, \dots, \lambda_q) = \int_D e^{i\langle x, \lambda_1 + \dots + \lambda_q \rangle} dx.$$
 (5.19)

In the following, we write  $I_{G,q}$  when we need to explicitly refer to the qth Wiener-Itô integral acting on  $\mathcal{H}^{\odot q}$  with respect to the measure G.

For  $q \in \mathbb{N}$ , r = 1, ..., q - 1 and h, g symmetric functions with unit norm in  $\mathcal{H}$ , we can define the rth contraction of  $h^{\otimes q}$  and  $g^{\otimes q}$  as the (generally non-symmetric)

<sup>&</sup>lt;sup>2</sup>Note that  $\langle h, g \rangle_{\mathcal{H}}$  is real for every  $h, g \in \mathcal{H}$ , because G is symmetric and h, g are even.

element of  $\mathcal{H}^{\otimes 2q-2r}$  given by

$$h^{\otimes q} \otimes_r g^{\otimes q} = \langle h, g \rangle_{\mathcal{H}}^r h^{\otimes q - r} \otimes g^{\otimes q - r}. \tag{5.20}$$

We then extend the definition of contraction to every pair of elements in  $\mathcal{H}^{\odot q}$ . We will denote the norm in this space by  $\|\cdot\|_q$ .

We are finally ready to state the celebrated Fourth Moment Theorem.

**Theorem 5.2.1** (Fourth Moment Theorem, [NP05]). Fix  $q \geq 2$ , consider  $(h_t)_{t>0} \subset \mathcal{H}^{\odot q}$  and assume that  $\mathbb{E}[I_q(h_t)^2] \to 1$  as  $t \to \infty$ . Then, the following three assertions are equivalent:

- $I_q(h_t)$  converges in distribution to a standard Gaussian random variable  $N \sim N(0,1)$ ;
- $\mathbb{E}[I_q(h_t)^4] \to 3 = \mathbb{E}[N^4]$ , where  $N \sim N(0, 1)$ ;
- $||h_t \otimes_r h_t||_{2q-2r} \to 0$  as  $t \to \infty$ , for all  $r = 1, \ldots, q-1$ .

We will also need a quantitative version of the Fourth Moment Theorem, which can be stated as follows (see [NP12, Theorem 5.2.6 and (5.2.6)]):

**Theorem 5.2.2.** Fix  $q \geq 2$  and consider  $h \in \mathcal{H}^{\odot q}$ . Then

$$\mathbb{E}[I_{q}(h)^{4}] - 3\mathbb{E}[I_{q}(h)^{2}]^{2} = \frac{3}{q} \sum_{r=1}^{q-1} rr!^{2} \binom{q}{r}^{4} (2q - 2r)! \|h\widetilde{\otimes}_{r}h\|_{2q-2r}^{2} \\
= \sum_{r=1}^{q-1} q!^{2} \binom{q}{r}^{2} \left\{ \|h\otimes_{r}h\|_{2q-2r}^{2} + \binom{2q-2r}{q-r} \|h\widetilde{\otimes}_{r}h\|_{2q}^{2} (5_{27})^{2} \right\}$$

and, with  $N \sim N(0,1)$ ,

$$d_{TV}(I_q(h), N) \le \frac{2}{\sqrt{3}} \sqrt{\mathbb{E}[I_q(h)^4] - 3\mathbb{E}[I_q(h)^2]^2}.$$
 (5.23)

#### **5.2.2** Classical results for 1-domain functionals

In this section, we state the two most popular results in the framework of limit theorems for 1-domain functionals. The first result, the celebrated Breuer-Major theorem, was first proved in [BM83] in a discrete version, and then extended to several settings. Here we state the continuous version of the result. Its proof can be found e.g. in [NZ20].

**Theorem 5.2.3** (Breuer-Major). Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered Gaussian field on  $\mathbb{R}^d$ , assumed to be stationary and to have unit-variance. Let C denotes its covariance function. Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be a measurable function satisfying  $\mathbb{E}[\varphi^2(N)] < \infty$ ,  $N \sim N(0,1)$ , with Hermite rank R. Let us consider

$$Y(t) := \int_{tD} \varphi(B_x) dx, \tag{5.24}$$

5.2. Preliminaries 119

where  $D \subset \mathbb{R}^d$  is a compact set, and  $t \geq 0$ . If  $C \in L^R(\mathbb{R}^d)$ , then, as  $t \to \infty$ ,

$$\frac{Y(t) - \mathbb{E}[Y(t)]}{t^{d/2}} \stackrel{d}{\longrightarrow} N(0, \sigma^2), \tag{5.25}$$

where  $\sigma^2 = \operatorname{Vol}(D) \sum_{q=R}^{\infty} q! a_q^2 \int_{\mathbb{R}^d} C(x)^q dx \geq 0$ . In particular, if  $\sigma^2 > 0$ , then  $\operatorname{Var}(Y(t)) \sim \sigma^2 t^d$  and a central limit theorem holds for  $(Y(t) - \mathbb{E}[Y(t)]) / \sqrt{\operatorname{Var}(Y(t))}$ .

The idea behind Breuer-Major theorem is that if the fields is not "too correlated" at infinity (precisely, if  $\int_{\mathbb{R}^d} |C(x)|^R dx < \infty$ ), then the fluctuations of the functional Y are Gaussian. Conversely, if this is not the case, then one can have non-Gaussian fluctuations (this does not mean that we necessarily have non-Gaussian fluctuations, see e.g. [MN24]). The following Theorem 5.1.2 provides a non-central limit theorem for functionals of Gaussian fields having a regularly varying covariance function, see (5.7), satisfying (5.8). In the discrete case, a first proof for functionals with Hermite rank R=1,2, was given by Taqqu in [Taq75], and then generalized to any R by Dobrushin and Major in [DM79]. The following sticks to the continuous case and a proof can be found e.g. in [LO14].

**Theorem 5.2.4** (Dobrushin-Major-Taqqu). Let  $B=(B_x)_{x\in\mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $\varphi:\mathbb{R}\to\mathbb{R}$  be a measurable function satisfying  $\mathbb{E}[\varphi^2(N)]<\infty$ ,  $N\sim N(0,1)$ , with Hermite rank R and Rth coefficient  $a_R\neq 0$ , see (5.1). Let us consider

$$Y(t) := \int_{tD} \varphi(B_x) dx,$$

where D is a compact set with Vol(D) > 0 and  $t \ge 0$ . Let  $C : \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B and suppose that (5.7)-(5.8) hold for C with parameter  $-\beta \in (-d/R, 0)$ . Then, as  $t \to \infty$ , we have

$$\widetilde{Y}(t) \xrightarrow{d} \frac{\operatorname{sgn}(a_R)I_{\nu,R}(f_{R,D})}{\sqrt{\operatorname{Var}(I_{\nu,R}(f_{R,D}))}},$$
(5.26)

where  $\nu$  is the measure on  $\mathbb{R}^d$  defined as  $\nu(dx) := |x|^{\beta-d} dx$ , and  $f_{R,D}$  is defined as

$$f_{R,D}(\lambda_1, \dots, \lambda_R) := \int_D e^{i\langle x, \lambda_1 + \dots + \lambda_R \rangle} dx.$$
 (5.27)

In particular, the limit  $I_{\nu,R}(f_{R,D})$  is not Gaussian as soon as  $R \geq 2$ .

Remark 5.2.1. The previous theorem shows that the limit  $I_{\nu,R}(f_{R,D})$  depends on R, d,  $\beta$  and the domain D. In fact, the integrand is  $f_{R,D}(\lambda_1,\ldots,\lambda_R)=\mathcal{F}[\mathbf{1}_D](\lambda_1+\cdots+\lambda_R)$ , where  $\mathcal{F}[\mathbf{1}_D]$  is the Fourier transform of the indicator function of the compact set D. The most common choices for D are the rectangles  $D=\textstyle \times_{i=1}^d [0,u_i]$ ,  $u_i\in\mathbb{R}_+$ , with Fourier transform

$$\mathcal{F}[\mathbf{1}_D](\lambda) = \prod_{j=1}^d \int_0^{u_j} e^{i\lambda_j x_j} dx_j = \prod_{j=1}^d \frac{e^{i\lambda_j u_j} - 1}{i\lambda_j},$$

and domains  $D=\{x\in\mathbb{R}^d:\|x\|\leq u\},\,u\in\mathbb{R}_+,$  with Fourier transform

$$\mathcal{F}[\mathbf{1}_D](\lambda) = c_d J_{d/2}(u||\lambda||) \left(\frac{u}{||\lambda||}\right)^{d/2},$$

where  $J_{d/2}$  is a Bessel function of the first kind and order d/2, and  $c_d$  is a positive constant, see e.g. [MN24, Section 2.2].

# **5.3** Proof of Theorems **5.1.1** and **5.1.2**

We split the proofs of Theorems 5.1.1 and 5.1.2 in four subsections. In Subsection 5.3.1, we prove Theorem 5.1.1 when  $\varphi = H_q$ , under weaker assumptions, providing also a quantitative result. In Subsection 5.3.2, we extend Theorem 5.2.3 to p-domain functionals. Finally, in Subsection 5.3.3 and 5.3.4 we prove Theorem 5.1.1 and Theorem 5.1.2, respectively.

## **5.3.1** Proof of Theorem **5.1.1** when $\varphi = H_q$

When  $\varphi = H_q$ , the functional (5.2) is given by

$$Y(t_1, \dots, t_p) = Y(t_1, \dots, t_p)[q] := \int_{t_1 D_1 \times \dots \times t_p D_p} H_q(B_x) dx.$$
 (5.28)

Therefore, see Section 5.2, we may express it as follows

$$\int_{t_1 D_1 \times ... \times t_n D_n} H_q(B_x) dx = I_q(f(t_1, ..., t_p)),$$
(5.29)

where  $I_q: \mathcal{H}^{\odot q} \to L^2(\Omega)$  stands for the qth Wiener-Ito integral (5.16), and  $f(t_1, \ldots, t_p) \in \mathcal{H}^{\odot q}$  is given by

$$f(t_1, \dots, t_p) := \int_{t_1 D_1 \times \dots \times t_p D_p} e_x^{\otimes q} dx.$$
 (5.30)

Let us recall the definition of the marginal functionals (5.5), and, accordingly, set

$$Y_i(t_i)[q] = I_q\left(\int_{t_i D_i} (e_{x_i}^{(i)})^{\otimes q} dx_i\right) = I_q(f_i(t_i)), \quad i = 1, \dots, p.$$
 (5.31)

**Proposition 5.3.1.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $C : \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B and assume it is separable in the sense of (5.4). Then

$$Var(Y(t_1, ..., t_p)[q]) = (q!)^{1-p} \prod_{i=1}^{p} Var(Y_i(t_i)[q]).$$
 (5.32)

Moreover, the following holds: if there exists  $i \in \{1, ..., p\}$  such that

$$\widetilde{Y}_i(t_i)[q] \stackrel{d}{\to} N(0,1)$$
 as  $t_i \to \infty$ , (5.33)

then

$$\widetilde{Y}(t_1,\ldots,t_p)[q] \stackrel{d}{\to} N(0,1)$$
 as  $t_1,\ldots,t_p \to \infty$ . (5.34)

*Proof.* Note that  $Var(Y_i(t_i)[q]) = q! ||f_i(t_i)||_q^2$ . As a result, (5.32) follows from the separability of C:

$$\operatorname{Var}(Y(t_{1}, \dots, t_{p})[q]) = q! \int_{(t_{1}D_{1} \times \dots \times t_{p}D_{p})^{2}} C(x_{1} - y_{1}, \dots, x_{p} - y_{p})^{q} dx dy \qquad (5.35)$$

$$= \prod_{i=1}^{p} \int_{(t_{i}D_{i})^{2}} C_{i}(x_{i} - y_{i})^{q} dx_{i} dy_{i} = (q!)^{1-p} \prod_{i=1}^{p} \operatorname{Var}(Y_{i}(t_{i})[q]). \qquad (5.36)$$

If q=1 everything is Gaussian and there is nothing more to show. Thus, let us assume that  $q\geq 2$  and let us compute the norm of the rth contraction of  $f(t_1,\ldots,t_p)$  with itself, for  $r=1,\ldots,q-1$ :

$$||f(t_1, \dots, t_p) \otimes_r f(t_1, \dots, t_p)||_{2q-2r}^2 =$$

$$\int ||f(t_1, \dots, t_p) \otimes_r f(t_1, \dots, t_p)||_{2q-2r}^2 =$$
(5.37)

$$= \int_{(t_1 D_1 \times \dots \times t_p D_p)^4} C(x-z)^r C(y-u)^r C(x-y)^{q-r} C(z-u)^{q-r} dx dy du dz \quad (5.38)$$

$$= \prod_{i=1}^{p} \int_{(t_i D_i)^4} C_i (x_i - z_i)^r C_i (y_i - u_i)^r C_i (x_i - y_i)^{q-r} C_i (z_i - u_i)^{q-r} dx_i dy_i du_i dz_i,$$
(5.39)

where  $x_i$  denotes the projection of x onto the ith block of  $\mathbb{R}^d = \prod_{i=1}^p \mathbb{R}^{d_i}$ . Let us define  $\tilde{f} := \frac{f}{\|f\|}$ . By linearity, it immediately follows that  $\widetilde{Y}(t_1, \dots, t_p)[q] = I_q(\widetilde{f}(t_1, \dots, t_p))$  and  $\widetilde{Y}_i(t_i)[q] = I_q(\widetilde{f}_i(t_i))$ . Then, combining (5.37) and (5.32), we obtain

$$\|\widetilde{f}(t_1,\ldots,t_p)\otimes_r \widetilde{f}(t_1,\ldots,t_p)\|_{2q-2r}^2 = \prod_{i=1}^p \|\widetilde{f}_i(t_i)\otimes_r \widetilde{f}_i(t_i)\|_{2q-2r}^2.$$
 (5.40)

Since for every  $f \in \mathcal{H}^{\otimes q}$  and  $r = 1, \dots, q - 1$ , we have

$$||f \otimes_r f||_{2q-2r}^2 \le ||f||_q^4, \tag{5.41}$$

we obtain

$$\|\widetilde{f}(t_1,\ldots,t_p)\otimes_r \widetilde{f}(t_1,\ldots,t_p)\|_{2q-2r}^2 \le \min\{\|\widetilde{f}_i(t_i)\otimes_r \widetilde{f}_i(t_i)\|_{2q-2r}^2 : i=1,\ldots,p\}.$$
(5.42)

It remains to apply Theorem 5.2.1 to conclude that, if  $\widetilde{Y}_i(t_i) \to N(0,1)$  as  $t_i \to \infty$ , then  $\|\widetilde{f}_i(t_i) \otimes_r \widetilde{f}_i(t_i)\|_{2q-2r} \to 0$  for all  $r \in \{1,\ldots,q-1\}$ , implying that  $\|\widetilde{f}(t_1,\ldots,t_p) \otimes_r \widetilde{f}(t_1,\ldots,t_p)\|_{2q-2r} \to 0$  for all  $r \in \{1,\ldots,q-1\}$ , implying finally that  $\widetilde{Y}(t_1,\ldots,t_p) \overset{d}{\to} N(0,1)$ .

The following result provides the converse implication of Proposition 5.3.1, under additional assumptions.

**Proposition 5.3.2.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $C : \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B and

assume it is separable in the sense of (5.4). Finally, assume either that  $C^q \ge 0$  or q = 3. Then, the following holds: if

$$\widetilde{Y}(t_1,\ldots,t_p)[q] \stackrel{d}{\to} N(0,1)$$
 as  $t_1,\ldots,t_p \to \infty$ , (5.43)

then there exists at least one  $i \in \{1, ..., p\}$  such that

$$\widetilde{Y}_i(t_i)[q] \stackrel{d}{\to} N(0,1)$$
 as  $t_i \to \infty$ . (5.44)

*Proof.* When q=1, we have that  $\widetilde{Y}$  and the  $\widetilde{Y}_i$ 's are Gaussian, meaning that the statement is correct but empty.

So, let us assume from now on that  $q \geq 2$  and that  $\widetilde{Y}(t_1,\ldots,t_p)[q] \stackrel{d}{\to} N(0,1)$ . By the Fourth Moment Theorem 5.2.1 one has that  $\|\widetilde{f}(t_1,\ldots,t_p)\otimes_r\widetilde{f}(t_1,\ldots,t_p)\|_{2q-2r} \to 0$  as  $t_1,\ldots,t_p\to\infty$  for any  $r\in\{1,\ldots,q-1\}$ .

When q=2, there is only one contraction to consider. Looking at (5.40), we deduce that at least one of the factors in (5.40) must go to 0 which, by Theorem 5.2.1, implies that at least one among the  $\tilde{Y}_i(t_i)$ 's must have a Gaussian limit.

Consider now the case where q = 3. By Fubini, we have for every  $i = 1, \dots, p$ 

$$\|\widetilde{f}_{i}(t_{i}) \otimes_{1} \widetilde{f}_{i}(t_{i})\|_{4}^{2} = \|\widetilde{f}_{i}(t_{i}) \otimes_{2} \widetilde{f}_{i}(t_{i})\|_{2}^{2},$$
 (5.45)

and (5.40) allows again to conclude.

Finally, let us suppose that  $q \ge 4$  and  $C^q \ge 0$ , that is either q even, or q odd and  $C \ge 0$ . Since the second contraction satisfies

$$\|\widetilde{f}(t_1,\ldots,t_p)\otimes_2\widetilde{f}(t_1,\ldots,t_p)\|_{2q-4}\longrightarrow 0, \quad \text{as } t_1,\ldots,t_p\to\infty,$$
 (5.46)

we deduce from (5.40) that there exists  $i \in \{1, ..., p\}$  such that

$$\|\widetilde{f}_i(t_i) \otimes_2 \widetilde{f}_i(t_i)\|_{2q-4}^2 \to 0 \text{ as } t_i \to \infty.$$
 (5.47)

Let us show that this is sufficient to conclude about the asymptotic normality of  $\widetilde{Y}_i(t_i)[q]$  when  $C^q \geq 0$ . For this, let us consider a positive sequence  $(a_{t_i})_{t_i>0}$  with  $a_{t_i} \to \infty$  as  $t_i \to \infty$  (the exact expression of  $a_{t_i}$  will be made precise later) and define the subset  $A_{t_i}$  of  $\mathbb{R}^{4d_i}$  by

$$A_{t_i} = \left\{ (x_i, y_i, u_i, z_i) \in \mathbb{R}^{4d_i} : |C_i(x_i - z_i)C_i(y_i - u_i)| \le a_{t_i}|C_i(x_i - y_i)C_i(z_i - u_i)| \right\}.$$

We can then write, for every  $3 \le r \le q - 1$ :

$$||f_{i}(t_{i}) \otimes_{r} f_{i}(t_{i})||_{2q-2r}^{2} =$$

$$= \int_{(t_{i}D_{i})^{4}} C_{i}(x_{i} - z_{i})^{r} C_{i}(y_{i} - u_{i})^{r} C_{i}(x_{i} - y_{i})^{q-r} C_{i}(z_{i} - u_{i})^{q-r} dx_{i} dy_{i} du_{i} dz_{i}$$

$$\leq \int_{(t_{i}D_{i})^{4} \cap A_{t,i}} |C_{i}(x_{i} - z_{i})^{r} C_{i}(y_{i} - u_{i})^{r} C_{i}(x_{i} - y_{i})^{q-r} C_{i}(z_{i} - u_{i})^{q-r} |dx_{i} dy_{i} du_{i} dz_{i}$$

$$\leq \int_{(t_{i}D_{i})^{4} \cap A_{t,i}} |C_{i}(x_{i} - z_{i})^{r} C_{i}(y_{i} - u_{i})^{r} C_{i}(x_{i} - y_{i})^{q-r} C_{i}(z_{i} - u_{i})^{q-r} |dx_{i} dy_{i} du_{i} dz_{i}$$

$$(5.50)$$

$$+ \int_{(t_{i}D_{i})^{4} \cap (\mathbb{R}^{4d_{i}} \setminus A_{t_{i}})} |C_{i}(x_{i} - z_{i})^{r} C_{i}(y_{i} - u_{i})^{r} C_{i}(x_{i} - y_{i})^{q-r} C_{i}(z_{i} - u_{i})^{q-r} |dx_{i}dy_{i}du_{i}dz_{i}|$$

$$(5.51)$$

$$\leq a_{t_i}^{r-2} \int_{(t_i D_i)^4} C_i(x_i - z_i)^2 C_i(y_i - u_i)^2 |C_i(x_i - y_i)|^{q-2} |C_i(z_i - u_i)|^{q-2} dx_i dy_i du_i dz_i$$
(5.52)

$$+ a_{t_i}^{-(q-r)} \int_{(t_i,D_i)^4} |C_i(x_i - z_i)|^q |C_i(y_i - u_i)|^q dx_i dy_i du_i dz_i$$
 (5.53)

$$= a_{t_i}^{r-2} \int_{(t_i D_i)^4} C_i(x_i - z_i)^2 C_i(y_i - u_i)^2 C_i(x_i - y_i)^{q-2} C_i(z_i - u_i)^{q-2} dx_i dy_i du_i dz_i$$
(5.54)

$$+ a_{t_i}^{-(q-r)} \int_{(t_i,D_i)^4} C_i(x_i - z_i)^q C_i(y_i - u_i)^q dx_i dy_i du_i dz_i$$
 (5.55)

$$= a_{t_i}^{r-2} \|f_i(t_i) \otimes_2 f_i(t_i)\|_{2q-4}^2 + a_{t_i}^{-(q-r)} \|f_i(t_i)\|_q^4.$$
(5.56)

Summarizing, we have, for every  $r \in \{3, \dots, q-1\}$ ,

$$\|\widetilde{f}_{i}(t_{i}) \otimes_{r} \widetilde{f}_{i}(t_{i})\|_{2q-2r}^{2} \leq a_{t_{i}}^{r-2} \|\widetilde{f}_{i}(t_{i}) \otimes_{2} \widetilde{f}_{i}(t_{i})\|_{2q-4}^{2} + a_{t_{i}}^{-(q-r)}.$$

$$(5.57)$$

Now, let us choose  $a_{t_i} = (\|\widetilde{f}_i(t_i) \otimes_2 \widetilde{f}_i(t_i)\|_{2q-4})^{\frac{2}{2-q}}$  and observe that  $a_{t_i} \to \infty$  as  $t_i \to \infty$ . Plugging into (5.57) allows to obtain, for every  $r \in \{3, \dots, q-1\}$ ,

$$\|\widetilde{f}_{i}(t_{i}) \otimes_{r} \widetilde{f}_{i}(t_{i})\|_{2q-2r}^{2} \leq 2(\|\widetilde{f}_{i}(t_{i}) \otimes_{2} \widetilde{f}_{i}(t_{i})\|_{2q-4})^{\frac{2(q-r)}{q-2}} \to 0 \quad \text{as } t_{i} \to \infty,$$

But this convergence of the rth contraction to zero also holds for r=1, since the norms of the 1-contraction and the (q-1)-contraction are equal. Finally, the desired conclusion follows from the Fourth Moment Theorem 5.2.1.

**Proposition 5.3.3.** Let the same notations and assumptions of Proposition 5.3.1 prevail. In particular  $N \sim N(0,1)$ . Then, the following estimate holds:

$$d_{TV}(\tilde{Y}(t_1, \dots, t_p)[q], N) \le c_q \prod_{i=1}^p \sqrt{\mathbb{E}[\tilde{Y}_i[q]^4] - 3},$$
 (5.58)

where 
$$c_q = \sqrt{\frac{4}{q} \sum_{r=1}^{q-1} rr!^2 \binom{q}{r}^4 (2q-2r)!}$$
.

Proof of Proposition 5.3.3. Recall the result of Theorem 5.2.2, namely

$$d_{TV}(\widetilde{Y}(t_1,\ldots,t_p)[q],N) \le \frac{2}{\sqrt{3}} \sqrt{\mathbb{E}\left[(\widetilde{Y}(t_1,\ldots,t_p)[q])^4\right] - 3},\tag{5.59}$$

and

$$\mathbb{E}\left[\left(\widetilde{Y}(t_1,\ldots,t_p)[q]\right)^4\right] - 3 \tag{5.60}$$

$$= \frac{3}{q} \sum_{r=1}^{q-1} rr!^2 \binom{q}{r}^4 (2q - 2r)! \|\widetilde{f}(t_1 \dots, t_p) \widetilde{\otimes}_r \widetilde{f}(t_1 \dots, t_p)\|_{2q-2r}^2$$
 (5.61)

$$= \sum_{r=1}^{q-1} q!^2 \binom{q}{r}^2 \{ \|\widetilde{f}(t_1, \dots, t_p) \otimes_r \widetilde{f}(t_1, \dots, t_p) \|_{2q-2r}^2$$
 (5.62)

$$+ \binom{2q-2r}{q-r} \|\widetilde{f}(t_1, \dots, t_p)\widetilde{\otimes}_r \widetilde{f}(t_1, \dots, t_p)\|_{2q-2r}^2 \}.$$
 (5.63)

We deduce

$$d_{TV}(\widetilde{Y}(t_1, \dots, t_p)[q], N) \le \frac{2}{\sqrt{3}} \sqrt{\mathbb{E}[(\widetilde{Y}(t_1, \dots, t_p)[q])^4] - 3}$$
 (5.64)

$$\leq c_q \max_{r=1,\dots,q-1} \|\widetilde{f}(t_1\dots,t_p) \otimes_r \widetilde{f}(t_1\dots,t_p)\|_{2q-2r}^2.$$
 (5.65)

But, thanks to (5.40) and recalling (5.41), we have

$$\max_{r=1,\dots,q-1} \|\widetilde{f}(t_1\dots,t_p) \otimes_r \widetilde{f}(t_1\dots,t_p)\|_{2q-2r}^2 = \max_{r=1,\dots,q-1} \prod_{i=1}^p \|\widetilde{f}_i(t_i) \otimes_r \widetilde{f}_i(t_i)\|_{2q-2r}^2$$

$$\leq \prod_{i=1}^p \max_{r=1,\dots,q-1} \|\widetilde{f}_i(t_i) \otimes_r \widetilde{f}_i(t_i)\|_{2q-2r}^2.$$
(5.67)

Also, as a consequence of the second equality in (5.60) (with  $\widetilde{Y}_i$  instead of  $\widetilde{Y}$ ), we have

$$\|\widetilde{f}_i(t_i) \otimes_r \widetilde{f}_i(t_i)\|_{2q-2r}^2 \le \mathbb{E}\left[\widetilde{Y}_i[q]^4\right] - 3,$$

and the desired conclusions now easily follows.

**Remark 5.3.1.** In Example 5.4.1 below we will use the previous result to improve the bound for the central convergence of the rescaled qth Hermite variation of the rectangular increments of the fractional Brownian sheet obtained in [RST12].

## 5.3.2 A Breuer-Major theorem for p-domain functionals

In this subsection, we provide an extension of the celebrated Breuer-Major theorem from the classical setting of 1-domain functionals (see Theorem 5.2.3) to the setting of p-domain functionals.

**Theorem 5.3.4.** Let  $B=(B_x)_{x\in\mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $\varphi:\mathbb{R}\to\mathbb{R}$  be a measurable function satisfying  $\mathbb{E}[\varphi^2(N)]<\infty$ ,  $N\sim N(0,1)$ , with Hermite rank  $R\geq 1$ . Let us consider  $\widetilde{Y}$  as in (5.3). Let  $C:\mathbb{R}^d\to\mathbb{R}$  be the covariance function of B and assume it is separable in the sense

of (5.4). If  $C_i \in L^R(\mathbb{R}^{d_i})$  for any  $i \in \{1, \dots, p\}$ , then, as  $t_1, \dots, t_p \to \infty$ ,

$$\frac{Y(t_1, \dots, t_p) - \mathbb{E}[Y(t_1, \dots, t_p)]}{t_1^{d_1/2} \dots t_p^{d_p/2}} \stackrel{d}{\to} N(0, \sigma^2), \tag{5.68}$$

where

$$\sigma^{2} = \sum_{q > R} a_{q}^{2} q! \prod_{i=1}^{p} \text{Vol}(D_{i}) \int_{\mathbb{R}^{d_{i}}} C_{i}^{q}(z_{i}) dz_{i}.$$
 (5.69)

*Moreover, if*  $\sigma^2 > 0$  *then* 

$$Var(Y(t_1, ..., t_p)) \sim \sigma^2 t_1^{d_1} ... t_p^{d_p}$$
 (5.70)

and we have a central limit theorem for (5.3), that is

$$\widetilde{Y}(t_1,\ldots,t_p)\stackrel{d}{ o} N(0,1)$$
 as  $t_1,\ldots,t_p o\infty$ .

**Remark 5.3.2.** The previous theorem may be seen as a Breuer-Major theorem for p-domains. Its proof follows from similar arguments to the ones in [NP12, Theorem 7.2.4]. However, we observe that the separable assumption allows us to recover the rate of the variance as a function of the (possibly distinct) growth rates of the domains.

*Proof of Theorem 5.3.4.* We proceed as in the proof of [NP12, Theorem 7.2.4], using [NP12, Theorem 6.3.1]. First of all, we have

$$\frac{Y(t_1,\dots,t_p) - \mathbb{E}[Y(t_1,\dots,t_p)]}{t_1^{d_1/2}\dots t_p^{d_p/2}} = \frac{\sum_{q=R}^{\infty} a_q \int_{t_1 D_1 \times \dots \times t_p D_p} H_q(B_x) dx}{t_1^{d_1/2}\dots t_p^{d_p/2}}$$
(5.71)

$$= \sum_{q=R}^{\infty} I_q(f_q(t_1, \dots, t_p)),$$
 (5.72)

where

$$f_q(t_1, \dots, t_p) := \frac{a_q \int_{t_1 D_1 \times \dots \times t_p D_p} e_x^{\otimes q} dx}{t_1^{d_1/2} \dots t_p^{d_p/2}}.$$

To conclude the proof we only need to check the conditions (a)-(d) of [NP12, Theorem 6.3.1].

Condition (a): We need to check that

$$\sigma^{2}[q] := \lim_{t_{1},\dots,t_{p}\to\infty} q! ||f_{q}(t_{1},\dots,t_{p})||^{2}$$

exists in  $[0,\infty)$  for each  $q \geq R$ . Since  $C_i \in L^R(\mathbb{R}^{d_i})$  (implying  $C_i \in L^q(\mathbb{R}^{d_i})$ ) for every  $i \in \{1,\ldots,p\}$ , the change of variable  $z_i = x_i - y_i$  yields

$$\int_{(t_i D_i)^2} C_i^q(x_i - y_i) dx_i dy_i = t_i^{d_i} \int_{\mathbb{R}^{d_i}} C_i^q(z_i) \operatorname{Vol}(D_i \cap (D_i + z_i/t_i)) dz_i$$
 (5.73)

$$\sim t_i^{d_i} \operatorname{Vol}(D_i) \int_{\mathbb{R}^{d_i}} C_i^q(z_i) dz_i, \quad \text{as } t_i \to \infty.$$
 (5.74)

Combining the previous equivalent with the separability of C, we obtain

$$q! \|f_q(t_1, \dots, t_p)\|^2 = q! a_q^2 \frac{\prod_{i=1}^p \int_{(t_i D_i)^2} C_i^q(x_i - y_i) dx_i dy_i}{t_1^{d_1/2} \dots t_p^{d_p/2}} \to \sigma^2[q],$$
 (5.75)

as  $t_1, \ldots, t_p \to \infty$ , where

$$\sigma^{2}[q] := q! a_{q}^{2} \prod_{i=1}^{p} \operatorname{Vol}(D_{i}) \int_{\mathbb{R}^{d_{i}}} C_{i}^{q}(z_{i}) dz_{i}.$$

Condition (b): We need to check that

$$\sum_{q=R}^{\infty} \sigma^2[q] < \infty.$$

Since  $||C_i||_{\infty} \le 1$  for every i = 1, ..., p, we have

$$\sum_{q=R}^{\infty} \sigma^2[q] \le \left(\sum_{q=R}^{\infty} q! a_q^2\right) \prod_{i=1}^p \left(\operatorname{Vol}(D_i) \int_{\mathbb{R}^{d_i}} |C_i(z_i)|^R dz\right).$$

Thanks to  $\operatorname{Var}(\varphi(N)) = \sum_{q=R}^{\infty} q! a_q^2 < \infty$ , the claim follows.

Condition (c): We need to check that for every  $q \geq R$  and every  $r = 1, \dots, q-1$ 

$$||f_q(t_1,\ldots,t_p)\otimes_r f_q(t_1,\ldots,t_p)||^2 \to 0$$
 as  $t_1,\ldots,t_p \to \infty$ .

Since C is separable, we deduce from (5.37) that

$$||f_q(t_1,\ldots,t_p)\otimes_r f_q(t_1,\ldots,t_p)||^2 = \prod_{i=1}^p ||f_{i,q}(t_i)\otimes_r f_{i,q}(t_i)||^2,$$

where  $f_{i,q}(t_i)$  is given by

$$f_{i,q}(t_i) := \int_{t_i D_i} (e_{x_i}^{(i)})^{\otimes q} dx_i.$$

Therefore, it is enough to prove that

$$||f_{i,q}(t_i) \otimes_r f_{i,q}(t_i)||^2 \to 0$$
 (5.76)

for at least one  $i \in \{1, ..., p\}$  as  $t_i \to \infty$ . But (5.76) is actually true for any  $i \in \{1, ..., p\}$ , see indeed point (c) in the proof of [NP12, Theorem 7.2.4] (which uses that  $C_i \in L^R(\mathbb{R}^{d_i})$ ).

Condition (d): We need to check that

$$\lim_{N \to \infty} \sup_{t_1, \dots, t_p \ge 1} \sum_{q=N+1}^{\infty} q! \|f_q(t_1, \dots, t_p)\|^2 = 0.$$

By (5.73) and since  $||C_i||_{\infty} \le 1$  for every i = 1, ..., p, we have

$$\begin{split} \sup_{t_1,\dots,t_p \geq 1} \sum_{q=N+1}^{\infty} q! \|f_q(t_1,\dots,t_p)\|^2 \\ &= \sup_{t_1,\dots,t_p \geq 1} \sum_{q=N+1}^{\infty} q! a_q^2 \frac{\prod_{i=1}^p \int_{(t_i D_i)^2} C_i^q(x_i - y_i) dx_i dy_i}{t_1^{d_1/2} \dots t_p^{d_p/2}} \\ &\leq \left(\sum_{q=N+1}^{\infty} q! a_q^2\right) \prod_{i=1}^p \left( \operatorname{Vol}(D_i) \int_{\mathbb{R}^{d_i}} |C_i(z_i)|^R dz_i \right) \to 0 \qquad \text{as } N \to \infty, \end{split}$$

where  $\left(\sum_{q=N+1}^{\infty} q! a_q^2\right) \to 0$  being the tail of a convergent series.

## **5.3.3** Reduction to *R*th chaos and proof of Theorem **5.1.1**

The chaotic decomposition of (5.2) is

$$Y(t_1, \dots, t_p) = \mathbb{E}[Y(t_1, \dots, t_p)] + \sum_{q \ge R} a_q Y(t_1, \dots, t_p)[q],$$
 (5.77)

see (5.28)-(5.29). In particular, this decomposition gives

$$Var(Y(t_1, ..., t_p)) = \sum_{q=R}^{\infty} a_q^2 Var(Y(t_1, ..., t_p)[q]).$$
 (5.78)

In order to prove Theorem 5.1.1, we reduce the study of  $\widetilde{Y}(t_1,\ldots,t_p)$  to that of  $\widetilde{Y}(t_1,\ldots,t_p)[R]$ , the normalization of  $Y(t_1,\ldots,t_p)[R]$ , thanks to the following extension of [MN24, Proposition 4].

**Proposition 5.3.5.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $\varphi : \mathbb{R} \to \mathbb{R}$  be a measurable function satisfying  $\mathbb{E}[\varphi^2(N)] < \infty$ ,  $N \sim N(0,1)$ , with Hermite rank R. Let us consider  $\widetilde{Y}$  as in (5.3), and  $\widetilde{Y}[R]$  as above. Let  $C : \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B, and assume that it is separable in the sense of (5.4) and that it satisfies the following two hypotheses:

- 1.  $C^R \geq 0$ , that is,  $C_i^R \geq 0$  for every  $i = 0, \dots, p$ ;
- 2. for some  $j \in \{1, ..., p\}$ , we have  $C_j \in \bigcup_{M=R+1}^{\infty} L^M(\mathbb{R}^{d_j}) \setminus L^R(\mathbb{R}^{d_j})$ .

Then, with  $sgn(a_R)$  denoting the sign of the Rth Hermite coefficient in (5.1),

$$\mathbb{E}\left[\left(\operatorname{sgn}(a_R)\widetilde{Y}(t_1,\ldots,t_p)[R]-\widetilde{Y}(t_1,\ldots,t_p)\right)^2\right]\to 0 \quad \text{as } t_1,\ldots,t_p\to\infty. \quad (5.79)$$

*Proof.* We divide the proof in three steps.

Step 1: upper and lower bounds for the variance. By assumption, their exist j and  $M \geq R+1$  such that  $C_j \notin L^R(\mathbb{R}^{d_j})$ , but  $C_j \in L^M(\mathbb{R}^{d_j})$ . Since  $C_j^R \geq 0$  by assumption, by doubling conditions for non-negative definite functions (see [GT19]) and properties of covariograms (see [Gal11]), reasoning exactly as in the first step of

the proof of [MN24, Proposition 9], there exist two positive constants  $c_1 > c_2 > 0$  such that

$$c_2 \int_{\{\|x_j\| \le t_j\}} C_j^R(z_j) dz_j \le t_j^{-d_j} \operatorname{Var}(Y_j(t_j)[R]) \le c_1 \int_{\{\|x_j\| \le t_j\}} C_j^R(z_j) dz_j.$$

In particular, this implies

$$t_j^{-d_j} \operatorname{Var}(Y_j(t_j)[R]) \to \infty \quad \text{as } t_j \to \infty.$$
 (5.80)

Step 2: we prove that  $\operatorname{Var}(Y(t_1,\ldots,t_p)) \sim a_R^2 \operatorname{Var}(Y(t_1,\ldots,t_p)[R])$  as  $t_1,\ldots,t_p \to \infty$ . Since  $|C| \le 1$  and by assumption  $C_i^R \ge 0$  for  $i \in \{1,\ldots,p\}$ , we have that, for any q > R and any  $j \in \{1,\ldots,p\}$ ,

$$\frac{\operatorname{Var}(Y(t_1, \dots, t_p)[q])}{\operatorname{Var}(Y(t_1, \dots, t_p)[R])} = \frac{q!}{R!} \prod_{i=1}^p \frac{\int_{(t_i D_i)^2} C_i^q(x_i - y_i) dx_i dy_i}{\int_{(t_i D_i)^2} C_i^R(x_i - y_i) dx_i dy_i}$$
(5.81)

$$\leq \frac{q!}{R!} \frac{\int_{(t_j D_j)^2} C_j^q(x_j - y_j) dx_j dy_j}{\int_{(t_j D_j)^2} C_j^R(x_j - y_j) dx_j dy_j} = \frac{\text{Var}(Y(t_j)[q])}{\text{Var}(Y(t_j)[R])}. \quad (5.82)$$

Now, by applying Cauchy-Schwarz n times, we obtain

$$\begin{split} \frac{\int_{(t_{j}D_{j})^{2}} C_{j}^{q}(x_{j} - y_{j}) dx_{j} dy_{j}}{\int_{(t_{j}D_{j})^{2}} C_{j}^{R}(x_{j} - y_{j}) dx_{j} dy_{j}} &\leq \left( \frac{\int_{(t_{j}D_{j})^{2}} |C_{j}(x_{j} - y_{j})|^{2q - R} dx_{j} dy_{j}}{\int_{(t_{j}D_{j})^{2}} |C_{j}(x_{j} - y_{j})|^{4q - 3R} dx_{j} dy_{j}} \right)^{1/2} \\ &\leq \left( \frac{\int_{(t_{j}D_{j})^{2}} |C_{j}(x_{j} - y_{j})|^{4q - 3R} dx_{j} dy_{j}}{\int_{(t_{j}D_{j})^{2}} C_{j}^{R}(x_{j} - y_{j}) dx_{j} dy_{j}} \right)^{1/4} \\ &\leq \cdots \\ &\leq \left( \frac{\int_{(t_{j}D_{j})^{2}} |C_{j}(x_{j} - y_{j})|^{R + 2^{n}(q - R)} dx_{j} dy_{j}}{\int_{(t_{j}D_{j})^{2}} C_{j}^{R}(x_{j} - y_{j}) dx_{j} dy_{j}} \right)^{1/2^{n}} \\ &\leq \left( \operatorname{Vol}(D_{j}) t_{j}^{d_{j}} \frac{\int_{\{\|x_{j}\| \leq \operatorname{diam}(D_{j})t_{j}\}} |C_{j}(z_{j})|^{R + 2^{n}} dz_{j}}{\int_{(t_{j}D_{j})^{2}} C_{j}^{R}(x_{j} - y_{j}) dx_{j} dy_{j}} \right)^{1/2^{n}} \\ &= \left( \operatorname{Vol}(D_{j}) t_{j}^{d_{j}} R! \frac{\int_{\{\|x_{j}\| \leq \operatorname{diam}(D_{j})t_{j}\}} |C_{j}(z_{j})|^{R + 2^{n}} dz_{j}}{\operatorname{Var}(Y_{j}(t_{j})[R])} \right)^{1/2^{n}} \end{split}$$

where the last inequality follows by a change of variable  $x_j - y_j = z_j$ , and the fact that  $q - R \ge 1$ . Since  $C_j \in L^{R+2^n}(\mathbb{R}^{d_j})$  for n sufficiently large, we deduce from (5.80) for every q > R

$$\frac{\operatorname{Var}(Y(t_1,\ldots,t_p)[q])}{\operatorname{Var}(Y(t_1,\ldots,t_p)[R])} \le cq! \left(t_j^{d_j} / \operatorname{Var}(Y_j(t_j)[R])\right)^{1/2^n},$$

where c > 0 depends on R and n but <u>not</u> on q. Combining this with (5.78), we get

$$\left| \frac{\operatorname{Var}(Y(t_1, \dots, t_p))}{\operatorname{Var}(Y(t_1, \dots, t_p)[R])} - a_R^2 \right| \le c \left( \sum_{q=R+1}^{\infty} a_q^2 q! \right) \left( t_j^{d_j} / \operatorname{Var}(Y_j(t_j)[R]) \right)^{1/2^n}.$$
 (5.83)

Using (5.80), this implies  $Var(Y(t_1,\ldots,t_p)) \sim a_R^2 Var(Y(t_1,\ldots,t_p)[R])$  as  $t_1,\ldots,t_p \to a_R^2 Var(Y(t_1,\ldots,t_p)[R])$ 

Step 3:  $\mathbb{E}\left[\left(\operatorname{sgn}(a_R)\widetilde{Y}(t_1,\ldots,t_p)[R]-\widetilde{Y}(t_1,\ldots,t_p)\right)^2\right]\to 0$ . To prove this last step, considering the decomposition

$$\widetilde{Y}(t_1, \dots, t_p) - \operatorname{sgn}(a_R) \widetilde{Y}(t_1, \dots, t_p)[R]$$
(5.84)

$$= \frac{\operatorname{sgn}(a_R)(Y(t_1, \dots, t_p) - \mathbb{E}[Y(t_1, \dots, t_p)] - a_R Y(t_1, \dots, t_p)[R])}{a_R \sqrt{\operatorname{Var}(Y(t_1, \dots, t_p)[R])}}$$
(5.85)

$$+ \frac{Y(t_{1}, \dots, t_{p}) - \mathbb{E}[Y(t_{1}, \dots, t_{p})]}{\sqrt{\operatorname{Var}(Y(t_{1}, \dots, t_{p}))}} \left\{ 1 - \frac{1}{|a_{R}|} \sqrt{\frac{\operatorname{Var}(Y(t_{1}, \dots, t_{p}))}{\operatorname{Var}(Y(t_{1}, \dots, t_{p})[R])}} \right\},$$
(5.86)

we get that

$$\mathbb{E}\left[\left(\widetilde{Y}(t_{1},\ldots,t_{p})-\operatorname{sgn}(a_{R})\widetilde{Y}(t_{1},\ldots,t_{p})[R]\right)^{2}\right] \\
\leq 2\frac{\mathbb{E}\left[\left(Y(t_{1},\ldots,t_{p})-\mathbb{E}\left[Y(t_{1},\ldots,t_{p})\right]-a_{R}Y(t_{1},\ldots,t_{p})[R]\right)\right)^{2}}{a_{R}^{2}\operatorname{Var}(Y(t_{1},\ldots,t_{p})[R])} \\
+2\left(1-\frac{1}{|a_{R}|}\sqrt{\frac{\operatorname{Var}(Y(t_{1},\ldots,t_{p}))}{\operatorname{Var}(Y(t_{1},\ldots,t_{p})[R])}}\right)^{2}.$$
(5.87)

$$+2\left(1-\frac{1}{|a_R|}\sqrt{\frac{\text{Var}(Y(t_1,\ldots,t_p))}{\text{Var}(Y(t_1,\ldots,t_p)[R])}}\right)^2.$$
 (5.88)

By the previous step, the second addend converges to 0. Regarding the first addend, since by (5.78) we have

$$\mathbb{E}\Big[(Y(t_1,\ldots,t_p)-\mathbb{E}[Y(t_1,\ldots,t_p)]-a_RY(t_1,\ldots,t_p)[R])^2\Big] = \sum_{q=R+1}^{\infty} a_q^2 \operatorname{Var}(Y(t_1,\ldots,t_p)[q]),$$

we deduce from the previous step that

$$\frac{\mathbb{E}[(Y(t_1,\dots,t_p) - \mathbb{E}[Y(t_1,\dots,t_p)] - a_R Y(t_1,\dots,t_p)[R]))^2]}{a_R^2 \text{Var}(Y(t_1,\dots,t_p)[R])} \to 0.$$
 (5.89)

This concludes the proof of Step 3 and Proposition 5.3.5.

We are now in a position to prove Theorem 5.1.1.

*Proof of Theorem 5.1.1.* Let us recall that  $B = (B_x)_{x \in \mathbb{R}^d}$  is a stationary and continuous Gaussian random field, and that its covariance function is

$$C(x) = C_1(x_1) \dots C_p(x_p), \quad x_i \in \mathbb{R}^{d_i}, i = 1, \dots, p,$$
 (5.90)

where  $C_i : \mathbb{R}^{d_i} \to \mathbb{R}$  is a non-negative definite function such that  $C_i(0) = 1$  for every  $i = 1, \dots, p$ .

The functionals  $\widetilde{Y}(t_1,\ldots,t_p)$  and  $\widetilde{Y}_i(t_i)$  are well-defined when  $C^R$  is positive. Indeed, by the decomposition (5.78) and the formula (5.36), we get that their variances are strictly positive.

Let us now distinguish two cases:

- (i) The first one is when  $C_i \in L^R(\mathbb{R}^{d_i})$  for every i. In this case, each functional  $Y_i(t_i)$  satisfies Theorem 5.2.3, hence  $\widetilde{Y}_i(t_i)$  has a Gaussian limit. Moreover, Theorem 5.3.4 applies, too, and we can conclude the same for  $\widetilde{Y}(t_1, \ldots, t_p)$ .
- (ii) Secondly, let us suppose that for at least one i one has  $C_i \in L^M(\mathbb{R}^{d_i}) \setminus L^R(\mathbb{R}^{d_i})$  with M > R. Then, by Proposition 5.3.5, it is equivalent to study the limit in distribution of  $\widetilde{Y}(t_1,\ldots,t_p)[R]$  only. Being in a fixed chaos and because  $C_j^R \geq 0$ , both Proposition 5.3.1 and Proposition 5.3.2 apply. Hence, if there exist j (possibly equal or distinct from i) such that  $\widetilde{Y}_j(t_j)[R]$  converges in distribution to a standard Gaussian, we can deduce the same for the functional  $\widetilde{Y}(t_1,\ldots,t_p)[R]$ , and vice-versa. To conclude, it remains to show that  $\widetilde{Y}_j(t_j)$  converges in distribution to a standard Gaussian random variable if and only if  $\widetilde{Y}_j(t_j)[R]$  does. When  $C_j \in L^R(\mathbb{R}^{d_j})$ , it is a consequence of the usual Breuer-Major theorem (Theorem 5.2.3), since in this case both  $\widetilde{Y}_j(t_j)$  and  $\widetilde{Y}_j(t_j)[R]$  are converging to N(0,1). When, on the contrary, we have  $C_j \in L^M(\mathbb{R}^{d_j}) \setminus L^R(\mathbb{R}^{d_j})$ , it is a consequence of the reduction theorem proved in [MN24, Proposition 4].

#### **5.3.4 Proof of Theorem 5.1.2**

We state two results on Wiener-Itô integrals, which are needed for our proof of Theorem 5.1.2. We note that they should not be considered as new, but rather as a generalization of well-known results contained in, e.g., [Maj81]. However, for completeness we provide their detailed proofs in the Appendix.

**Lemma 5.3.6** (Change of variable formula). Let  $\nu, \nu'$  be real,  $\sigma$ -finite measures on  $\mathbb{R}^d$  endowed with the Borel  $\sigma$ -algebra  $\mathcal{B}(\mathbb{R}^d)$  and let us define the real separable<sup>3</sup> Hilbert spaces  $\mathcal{H}_{\nu} := L^2(\nu), \mathcal{H}_{\nu'} := L^2(\nu')$  as in (5.12). Let us suppose that

$$\nu(dx) = |a(x)|^2 \nu'(dx) \tag{5.91}$$

where a is a complex valued, even function (i.e.  $a(-x) = \overline{a(x)}$ ). Then, for every  $h \in \mathcal{H}_{\nu}{}^{\odot q}$  we have

$$I_{\nu,q}(h) \stackrel{\text{law}}{=} I_{\nu',q}(a^{\otimes q}h),$$

where  $a^{\otimes q}$  is as in (5.17).

**Lemma 5.3.7.** Let us denote by  $I_q$  the Wiener-Itô integral acting on  $\mathcal{H}^{\odot q}$ , with respect to the Lebesgue measure dx on  $\mathbb{R}^d$ . Then, for every  $h \in \mathcal{H}^{\odot q} = L_s^2((\mathbb{R}^d)^q, dx)$  and  $s_1, \ldots, s_d > 0$ , we have

$$I_q(h) \stackrel{\text{law}}{=} s_1^{-q/2} \dots s_d^{-q/2} I_q\left(\tilde{h}(s_1,\dots,s_d)\right),$$

<sup>&</sup>lt;sup>3</sup>Note that if  $\nu$  is real,  $\sigma$ -finite and symmetric, then  $\mathcal{H}_{\nu}$  is a real separable Hilbert space, and we can consider an isonormal Gaussian process on it, as well as every notion and result introduced in Section 5.2.1 for  $L^2(G)$ .

where  $\tilde{h}(s_1, \ldots, s_d) \in \mathcal{H}^{\odot q}$  is such that

$$\tilde{h}(s_1, \dots, s_d)(x_1, \dots, x_q) := h(x_{11}/s_1, \dots, x_{1d}/s_d, \dots, x_{q1}/s_1, \dots, x_{qd}/s_d).$$
 (5.92)

*Proof of Theorem 5.1.2.* We divide the proof in four steps. In the sequel, c will denote a positive constant which may vary depending on the instance.

Step 1: Reduction to the Rth chaos. Since  $C^R \geq 0$  and  $C_i \in \bigcup_{M=R+1}^{\infty} L^M(\mathbb{R}^{d_i}) \setminus L^R(\mathbb{R}^{d_i})$  for every i, by Proposition 5.3.5 we are left to study the limit in distribution of  $\widetilde{Y}(t_1,\ldots,t_p)[R]$ .

*Step 2: Variance analysis.* It is a standard fact (see e.g. [LO13], [LO14], [Leo99] or [Mai24]) that if (5.7) holds for i, then

$$\operatorname{Var}(Y_i(t_i)[R]) \sim c L_i(t_i)^R t_i^{2d_i - R\beta_i} \quad \text{as } t_i \to \infty.$$

We conclude that

$$Var(Y(t_1, \dots, t_p)[R]) = \frac{1}{(R!)^{p-1}} \prod_{i=1}^p Var(Y_i(t_i)[R]) \sim c \prod_{i=1}^p L_i(t_i)^R t_i^{2d_i - R\beta_i}.$$

Step 3: A suitable expression (in law) for  $Y(t_1, ..., t_p)[R]$ . Recall the definition (5.10). By assumption (5.8), we have  $G_i(d\lambda_i) = g_i(\lambda_i)d\lambda_i$  for every i. Let us define  $g: \mathbb{R}^d \to \mathbb{R}_+$  as

$$g(x_1,\ldots,x_p)=\prod_{i=1}^p g_i(x_i).$$

We get from Lemma 5.3.6 that

$$Y(t_1,\ldots,t_p)[R] = I_{G,R}(f_{R,t_1D_1\times\cdots\times t_pD_p}) \stackrel{\text{law}}{=} I_R((\sqrt{g})^{\otimes R} f_{R,t_1D_1\times\cdots\times t_pD_p}),$$

where  $I_R$  is the Rth Wiener-Itô integral with respect to the Lebesgue measure. By applying Lemma 5.3.7, we also obtain

$$I_{R}((\sqrt{g})^{\otimes R} f_{R,t_{1}D_{1} \times \dots \times t_{p}D_{p}}) = \left(\prod_{i=1}^{p} t_{i}^{d_{i} - \frac{Rd_{i}}{2}}\right) I_{R}\left(\left(\sqrt{g(t_{1}, \dots, t_{p})}\right)^{\otimes R} f_{R,D_{1} \times \dots \times D_{p}}\right)$$

$$(5.93)$$

where  $g(t_1,\ldots,t_p):\mathbb{R}^d\to\mathbb{R}_+$  is given by

$$g(t_1, \dots, t_p)(x_1, \dots, x_p) = \prod_{i=1}^p g_i(x_i/t_i).$$

Then, we have obtained the following expression in distribution for  $\widetilde{Y}(t_1,\ldots,t_p)[R]$ :

$$\widetilde{Y}(t_1, \dots, t_p)[R] \stackrel{\text{law}}{=} \frac{I_R\left(\left(\sqrt{g(t_1, \dots, t_p)}\right)^{\otimes R} f_{R, D_1 \times \dots \times D_p}\right)}{\sqrt{\operatorname{Var}\left(I_R\left(\left(\sqrt{g(t_1, \dots, t_p)}\right)^{\otimes R} f_{R, D_1 \times \dots \times D_p}\right)\right)}}.$$
 (5.94)

Step 4: Proving the  $L^2$  convergence. The last step of the proof consists in showing the following  $L^2$  convergence:

$$\frac{I_{R}\left(\left(\sqrt{g(t_{1},\ldots,t_{p})}\right)^{\otimes R}f_{R,D_{1}\times\cdots\times D_{p}}\right)}{\sqrt{\operatorname{Var}\left(I_{R}\left(\left(\sqrt{g(t_{1},\ldots,t_{p})}\right)^{\otimes R}f_{R,D_{1}\times\cdots\times D_{p}}\right)\right)}} \xrightarrow{L^{2}(\Omega)} \frac{I_{R}((\sqrt{g'})^{\otimes R}f_{R,D_{1}\times\cdots\times D_{p}})}{\sqrt{\operatorname{Var}\left(I_{R}((\sqrt{g'})^{\otimes R}f_{R,D_{1}\times\cdots\times D_{p}})\right)}} \\
\stackrel{\underline{\operatorname{law}}}{=} \frac{I_{\nu_{1}\times\cdots\times\nu_{p},R}\left(f_{R,D_{1}\times\cdots\times D_{p}}\right)}{\sqrt{\operatorname{Var}\left(I_{\nu_{1}\times\cdots\times\nu_{p},R}\left(f_{R,D_{1}\times\cdots\times D_{p}}\right)\right)}}, \tag{5.95}$$

where the equality in distribution follows from Lemma 5.3.6 and where  $g'(x) := \prod_{i=1}^p \|x_i\|^{\beta_i - d_i}$ . By the previous steps, it is enough to prove that for some positive constant c we have

$$c \prod_{i=1}^{p} t_i^{\frac{R(\beta_i - d_i)}{2}} L_i(t_i)^{-R/2} I_R \left( \left( \sqrt{g(t_1, \dots, t_p)} \right)^{\otimes R} f_{R, D_1 \times \dots \times D_p} \right)$$
 (5.96)

$$\stackrel{L^{2}(\Omega)}{\to} I_{R}\left((\sqrt{g'})^{\otimes R} f_{R,D_{1} \times \dots \times D_{p}}\right). \tag{5.97}$$

It is equivalent to show that

$$\left(\prod_{i=1}^{p} Q_{t_i}(\lambda_{1i}, \dots, \lambda_{qi})\right) f_{R, D_1 \times \dots \times D_p}(\lambda_1, \dots, \lambda_R) \prod_{j=1}^{R} \prod_{i=1}^{p} \|\lambda_{ji}\|^{\frac{\beta_i - d_i}{2}}$$
(5.98)

$$\stackrel{L^2((\mathbb{R}^d)^R)}{\longrightarrow} f_{R,D_1 \times \dots \times D_p}(\lambda_1, \dots, \lambda_R) \prod_{j=1}^R \prod_{i=1}^p \|\lambda_{ji}\|^{\frac{\beta_i - d_i}{2}}, \tag{5.99}$$

where the  $Q_{t_i}:(\mathbb{R}^{d_i})^R\to\mathbb{C}$  are defined as

$$Q_{t_i}(\lambda_{1i}, \dots, \lambda_{Ri}) := \sqrt{\prod_{i=1}^{R} \|\lambda_{ji}/t_i\|^{d_i-\beta_i} c_i^{-1} L_i^{-1}(t_i) g_i(\lambda_{ji}/t_i)},$$

and  $c_i$  are some positive constant. In particular, we will prove (5.98) with the constant  $c_i$  given by assumption (5.8). To prove (5.98), we proceed by induction on p. If p = 1, then (5.98) reduces to

$$\int_{(\mathbb{R}^{d_1})^R} |Q_{t_1}(\lambda_{11}, \dots, \lambda_{R1}) - 1|^2 |f_{R, D_1}(\lambda_{11}, \dots, \lambda_{R1})|^2 \frac{d\lambda_{11} \dots d\lambda_{R1}}{\prod_{j=1}^R \|\lambda_{j1}\|^{d_1 - \beta_1}} \to 0,$$

and this convergence is shown in the proof of [LO14, Theorem 5]. Now, let us assume that (5.98) holds for p-1, and let us prove the result for p. For this, note

5.4. Examples 133

that xy - 1 = (x - 1)(y - 1) + (x - 1) + (y - 1), implying

$$\left(\prod_{i=1}^{p} Q_{t_i}(\lambda_{1i}, \dots, \lambda_{Ri})\right) - 1 = \left(\left(\prod_{i=1}^{p-1} Q_{t_i}(\lambda_{1i}, \dots, \lambda_{Ri})\right) - 1\right) \left(Q_{t_p}(\lambda_{1p}, \dots, \lambda_{Rp}) - 1\right)$$

$$(5.100)$$

$$+\left(\left(\prod_{i=1}^{p-1}Q_{t_i}(\lambda_{1i},\ldots,\lambda_{Ri})\right)-1\right) \tag{5.101}$$

$$+\left(Q_{t_p}(\lambda_{1p},\ldots,\lambda_{Rp})-1\right). \tag{5.102}$$

Then, by applying the triangular inequality in  $L^2((\mathbb{R}^d)^R)$ , we get that (5.98) holds by inductive hypothesis.

# 5.4 Examples

In this Section we collect some examples.

**Example 5.4.1** (Hermite variations of a fractional Brownian sheet (fBs)). Let us consider a fBs

$$W^{\alpha,\beta} = (W^{\alpha,\beta})_{(x_1,x_2) \in \mathbb{R}^2_+} \tag{5.103}$$

with parameter  $(\alpha, \beta) \in \mathbb{R}^2_+$  and its rectangular increments:

$$R_{x_1,x_2} := W_{x_1+1,x_2+1}^{\alpha,\beta} - W_{x_1,x_2+1}^{\alpha,\beta} - W_{x_1+1,x_2}^{\alpha,\beta} + W_{x_1,x_2}^{\alpha,\beta}. \tag{5.104}$$

The field R defined as above is a stationary centered Gaussian field, with covariance function given by

$$Cov(R_{x_1,x_2}, R_{y_1,y_2}) = r_{\alpha}(x_1 - y_1)r_{\beta}(x_2 - y_2)$$
(5.105)

where

$$r_H(u) = \frac{1}{2} \left( |u+1|^{2H} + |u-1|^{2H} - 2|u|^{2H} \right), \qquad u \in \mathbb{R}, \ H \in (0,1).$$
 (5.106)

Notice that the covariance function of R is separable in the sense of (5.4). Moreover,  $r_{\alpha}$  (resp.  $r_{\beta}$ ) is the covariance function of the process defined as the (one-dimensional) increments of a fractional Brownian motion (fBm) with Hurst index  $\alpha$  (resp.  $\beta$ ). We briefly recall some results about this process, see e.g. [NP12]. A fractional Brownian motion  $W^H = (W^H_t)_{t \in \mathbb{R}}$  of Hurst index H, is a centered Gaussian process such that

$$\mathbb{E}[W_{x_1}^H W_{y_1}^H] = \frac{1}{2} \left( |x_1|^{2H} + |y_1|^{2H} - |x_1 - y_1|^{2H} \right), \quad x_1, y_1 \in \mathbb{R}.$$
 (5.107)

Its increment process

$$X_u = W_{u+1}^H - W_k^H, \quad u \in \mathbb{R}_+$$
 (5.108)

is known as fractional Gaussian noise. It is a centered, stationary Gaussian process

with covariance function as in (5.106). Note that (5.106) is regularly varying with parameter 2H - 2 (see [NP12, (7.4.3)], or [Mai24, Example 1]), i.e. it behaves asymptotically as

$$r_H(u) = H(2H-1)|u|^{2H-2} + o(|u|^{2H-2}), \text{ as } |u| \to \infty.$$
 (5.109)

Breuer-Major theorem (see Theorem 5.2.3) and [NP12, Theorem 7.4] allow us to deduce that the sequence

$$Y_N = \sum_{k=0}^{N-1} H_q(X_k), \quad N \ge 1,$$
(5.110)

under a proper renormalization, converges to a Gaussian distribution if  $H \le 1 - 1/2q$ . Moreover, in [NP09] and [BN08], the authors quantify the convergence in total variation distance.

**Theorem 5.4.1** (Theorem 1.1, 1.2 in [BN08]). If  $H \in (0, 1 - 1/2q)$  and  $V_N = \frac{Y_N}{\sqrt{\text{Var}(Y_N)}}$ , then

$$d_{TV}(V_N, N(0, 1)) \le c_{H,q} \begin{cases} N^{-1/2} & \text{if } H \in (0, \frac{1}{2}) \\ N^{H-1} & \text{if } H \in \left[\frac{1}{2}, \frac{2q-3}{2q-2}\right] \\ N^{2Hq-2q+1/2} & \text{if } H \in \left(\frac{2q-3}{2q-2}, 1 - \frac{1}{2q}\right) \end{cases} =: c_{H,q} \cdot g(q, H, N).$$

$$(\log N)^{-1/2} & \text{if } H = 1 - \frac{1}{2q}$$

$$(5.111)$$

In the previous,  $c_{H,q}$  a positive constant, that may vary, only dependent on q and H.

If H > 1 - 1/2q, we have convergence towards a non Gaussian distribution (see [DM79]).

In [RST12], the authors show that a proper renormalization of

$$V_{N,M} = \sum_{x_1=0}^{N-1} \sum_{x_2=0}^{M-1} H_q(R_{x_1,x_2}), \quad N, M \in \mathbb{N}$$
 (5.112)

converges to a Gaussian distribution as  $N, M \to \infty$  when  $\alpha \le 1 - 1/2q$  or  $\beta \le 1 - 1/2q^4$ . Moreover, they bound the total variation distance between  $V_{N,M}$  and its Gaussian limit.

**Theorem 5.4.2** (Theorem 3.1 in [RST12]). Let us denote by  $c_{\alpha,\beta}$  a generic positive constant which depends on  $\alpha$ ,  $\beta$  and q, but which is independent of N and M. We have

(1) If both  $0 < \alpha, \beta < 1 - 1/2q$ , then  $\widetilde{V}_{N,M}$  converges in law to N(0,1) with normalization  $\varphi(\alpha,\beta,N,M) = \sqrt{\frac{q!}{s_{\alpha}s_{\beta}}}N^{\alpha q - 1/2}M^{\beta q - 1/2}$ . In addition

$$d_{TV}(\widetilde{V}_{N,M}, N(0,1)) \le c_{\alpha,\beta} \sqrt{N^{-1} + N^{2\alpha - 2} + N^{2\alpha q - 2q + 1} + M^{-1} + M^{2\beta - 2} + M^{2\beta q - 2q + 1}}.$$
(5.113)

<sup>&</sup>lt;sup>4</sup>More precisely, they showed it for the Hermite variations of the field  $N^{\alpha}M^{\beta}R_{i/N,j/M}^{\alpha,\beta}$ . However, by self-similarity of the process R (see [RST12, Definition 2.3]), the two share the same law.

5.4. Examples 135

(2) If  $0 < \alpha < 1 - 1/2q$  and  $\beta = 1 - 1/2q$ , then  $\widetilde{V}_{N,M}$  converges in law to N(0,1) with normalization  $\varphi(\alpha,\beta,N,M) = \sqrt{\frac{q!}{s_\alpha \iota_\beta}} N^{\alpha q - 1/2} M^{q-1} (\log M)^{-1/2}$ . In addition

$$d_{TV}(\widetilde{V}_{N,M}, N(0,1)) \le c_{\alpha,\beta} \sqrt{N^{-1} + N^{2\alpha - 2} + N^{2\alpha q - 2q + 1} + (\log M)^{-1}}.$$
(5.114)

If  $0 < \beta < 1 - 1/2q$  and  $\alpha = 1 - 1/2q$ , then we get an analogous estimate as the previous one.

(3) If both  $\alpha = \beta = 1 - 1/2q$ , then  $\widetilde{V}_{N,M}$  converges in law to N(0,1) with normalization  $\varphi(\alpha,\beta,N,M) = \sqrt{\frac{q!}{\iota_\alpha\iota_\beta}}N^{q-1}(\log N)^{-1/2}M^{q-1}(\log M)^{-1/2}$ . In addition

$$d_{TV}(\widetilde{V}_{N,M}, N(0,1)) \le c_{\alpha,\beta} \sqrt{(\log N)^{-1} + (\log M)^{-1}}.$$
 (5.115)

(4) If  $\alpha < 1 - 1/2q$  and  $\beta > 1 - 1/2q$ , then  $\widetilde{V}_{N,M}$  converges in law to N(0,1) with normalization  $\varphi(\alpha,\beta,N,M) = \sqrt{\frac{q!}{s_{\alpha}\kappa_{\beta}}}N^{\alpha q - 1/2}M^{q-1}$ . In addition<sup>5</sup>

$$d_{TV}(\widetilde{V}_{N,M}, N(0,1)) \le c_{\alpha,\beta} \sqrt{N^{-1} + N^{2\alpha - 2} + N^{2\alpha q - 2q + 1} + M^{-(2\beta q - 2q + 1)}}.$$
(5.116)

(5) If  $\alpha=1-1/2q$  and  $\beta>1-1/2q$ , then  $\widetilde{V}_{N,M}$  converges in law to N(0,1) with normalization  $\varphi(\alpha,\beta,N,M)=\sqrt{\frac{q!}{\iota_{\alpha}\kappa_{\beta}}}N^{q-1}(\log N)^{-1/2}M^{q-1}$ . In addition

$$d_{TV}(\tilde{V}_{N,M}, N(0,1)) \le c_{\alpha,\beta} \sqrt{(\log N)^{-1} + M^{-(2\beta q - 2q + 1)}}$$
(5.117)

Our Theorem 5.1.1 translates in the discrete setting, too, and returns the same qualitative phenomenon described above. Moreover, by using the bounds obtained in the proof of [NP09, Theorem 4.1], Proposition 5.3.3 allows us to improve the previous rates.

**Corollary 5.4.3.** *Recall the definition of* g *in* (5.111).

(1) If both  $0 < \alpha, \beta \le 1 - 1/2q$ , then

$$d_{TV}(\tilde{V}_{N,M}, N(0,1)) \le c_{\alpha,\beta,q} \ g(q,\alpha,N) \cdot g(q,\beta,M). \tag{5.118}$$

(2) If  $\alpha \le 1 - 1/2q$  and  $\beta > 1 - 1/2q$ , then

$$d_{TV}(\tilde{V}_{N,M}, N(0,1)) \le c_{\alpha,\beta,q} g(q,\alpha,N),$$
 (5.119)

where  $c_{\alpha,\beta,q}$  are constants depending on  $\alpha,\beta,q$  that may differ from the ones in [RST12, Theorem 3.1], and the normalization terms are as before.

**Example 5.4.2** (Tensor product of regularly varying covariance functions). Let us define a centered Gaussian field  $B = (B_x)_{x \in \mathbb{R}^d}$  with separable covariance function as

<sup>&</sup>lt;sup>5</sup>The exponent in red is the correction of a typo in [RST12, Theorem 3.1].

in (5.4), with  $p \geq 2$  and  $C_i : \mathbb{R}^{d_i} \to \mathbb{R}$  defined as

$$C_i(x_i) := \frac{1}{(1 + ||x_i||^2)^{\beta_i/2}}, \quad \beta_i > 0, \quad i = 1, \dots, p.$$
 (5.120)

Let us consider the quadratic variation of this field, that is, the functional  $Y(t_1,\ldots,t_p)[2]$  as in (5.29) as well as its rescaled version  $\widetilde{Y}$ . The parameters  $\beta_i$  determine the behavior of the functional. Indeed, combining Theorem 5.2.3 and Proposition 5.3.1, it is enough to have  $\beta_i > d_i/2$  for at least one i for  $\widetilde{Y}(t_1,\ldots,t_p)[2]$  to have a Gaussian limit. Moreover, by the application of Proposition 5.3.3, we may also deduce an upper bound for the rate of convergence in total variation. For a fixed i satisfying  $\beta_i > d_i/2$ , an upper bound for the rate of convergence in total variation distance of  $\widetilde{Y}_i(t_i)[2]$  (recall (5.6)) towards a Gaussian distribution is given by the rate of convergence to 0 of the norm of the contraction:

$$\|\widetilde{f}_{i}(t_{i}) \otimes_{1} \widetilde{f}_{i}(t_{i})\|_{2}^{2} \lesssim \frac{\left(\int_{\|x_{i}\| \leq t_{i}} C_{i}(x_{i}) dx_{i}\right)^{2}}{t_{i}^{d_{i}}} \approx \begin{cases} t_{i}^{-d_{i}} & \text{if } \beta_{i} > d_{i} \\ \log(t_{i})^{2} t_{i}^{-d_{i}} & \text{if } \beta_{i} = d_{i} \\ t_{i}^{d_{i} - 2\beta_{i}} & \text{if } \beta_{i} \in (d_{i}/2, d_{i}) \end{cases} =: g(\beta_{i}, t_{i})^{2},$$

$$(5.121)$$

where we denote by  $\lesssim$  the inequality up to a positive constant. From the above equivalence and (5.58), we obtain that, if  $\beta_i > d_i/2$ ,

$$d_{TV}(\widetilde{Y}_i(t_i)[2], N(0,1)) \lesssim g(\beta_i, t_i).$$
 (5.122)

Denoting  $J = \{i \in \{1, ..., p\} : \beta_i > d_i/2\}$ , we can deduce that

$$d_{TV}(\widetilde{Y}(t_1, \dots, t_p)[2], N(0, 1)) \lesssim \prod_{i \in J} g(\beta_i, t_i).$$
 (5.123)

**Example 5.4.3** (Gaussian fluctuations in a long-range dependence setting). Let us define a centered Gaussian field  $B = (B_x)_{\mathbb{R}^2}$  with separable covariance function

$$C(x_1, x_2) = \frac{1}{(1 + |x_1|^2)^{1/4}} \cdot \frac{1}{(1 + |x_2|^2)^{3/2}}.$$
 (5.124)

Let us consider

$$Y(t_1, t_2) = \int_{t_1 D_1 \times t_2 D_2} H_2(B_x) dx, \qquad t_1, t_2 > 0,$$
 (5.125)

where  $H_2$  is the second Hermite polynomial and  $D_i \subset R$  are compact sets. Then, R=2. We have that  $C \notin L^2(\mathbb{R}^2)$ , hence we are in the long-range dependence case. However, the marginal functional

$$\widetilde{Y}_{2}(t_{2}) = \frac{\int_{t_{2}D_{2}} H_{2}(B_{x}^{(2)}) dx}{\sqrt{\operatorname{Var}\left(\int_{t_{2}D_{2}} H_{2}(B_{x}^{(2)}) dx\right)}},$$
(5.126)

see also (5.5), where  $(B_x^{(2)})_{\mathbb{R}}$  is centered Gaussian field with covariance function  $C_2(x_2)=\frac{1}{(1+|x_2|^2)^{3/2}}$ , exhibits Gaussian fluctuations as  $t_2\to\infty$  by Theorem 5.2.3.

Hence, by Theorem 5.1.1 also  $\widetilde{Y}(t_1, t_2)$  does.

### 5.5 Going beyond the separability assumption

While the separability assumption holds true in numerous applications (see, e.g., [Chr92, Chapter 5] for examples in hydrology and fluid dynamics, or think of the many frameworks where the fractional Brownian sheet arises, see, e.g., [ØZ01; SW17]), there also many instances where it does not.

In this section, we aim to examine what can happen when we go beyond the separable case. First, let us give a counterexample that shows that we cannot always expect a result as simple as Theorem 5.1.1 in the non separable context.

**Example 5.5.1** (Separability matters!). Let us consider p=2,  $d_1=d_2=1$ , and thus d=2. Fix  $R\geq 2$  and consider  $\alpha\in (1/R,2/R)$ . Let  $B=(B_x)_{\mathbb{R}^d}$  be a centered stationary Gaussian field with unit-variance and covariance function  $C:\mathbb{R}^d\to\mathbb{R}$  as

$$C(x) = \frac{L(\|x\|)}{\|x\|^{\alpha}}, \quad x \in \mathbb{R}^d \setminus \{0\},$$
 (5.127)

with L a slowly varying function, such that C is continuous in 0. Assuming that C satisfies (5.8), since  $\alpha < \frac{2}{R}$ , Theorem 5.2.4 applies to the functional

$$Y(t,t)[R] = \int_{t(D_1 \times D_2)} H_R(B_x) dx$$
 (5.128)

and we obtain that  $\widetilde{Y}(t,t)[R]$  is asymptotically not Gaussian. However, both  $\widetilde{Y}_1(t)[R]$  and  $\widetilde{Y}_2(t)[R]$  constructed as in (5.6) have Gaussian fluctuations as  $t\to\infty$ . Indeed, they both have covariance functions as follows:

$$C_i(x_i) = \frac{L(|x_i|)}{|x_i|^{\alpha}}, \quad i = 1, 2;$$
 (5.129)

and since  $\alpha > 1/R$ , they both satisfy Theorem 5.2.3. This shows how the convergence of both functionals  $Y_i(t_i)$  is in general not enough to determine the behavior of  $Y(t_1, \ldots, t_p)$ , when the separability of the covariance function (5.4) is not satisfied.

In what follows, we drop the separability assumption by examining two different classes: Gneiting covariance functions (Section 5.5.1) and additively separable covariance functions (Section 5.5.2). We will explore whether, akin to the separable case, it is possible to simplify the asymptotic study of  $Y(t_1, \ldots, t_p)$  given by (5.2) by reducing it to that of simpler functionals  $Y_i(t_i)$ .

#### 5.5.1 Gneiting covariance functions

The class of Gneiting covariance functions was first introduced by Gneiting in [Gne02]. These functions are very popular in many applications, including geostatistics, environmental science, climatology and meteorology. A **Gneiting covariance function** 

 $C: \mathbb{R}^d \to \mathbb{R}$  has the form

$$C(x_1, x_2) = \psi\left(\|x_2\|^2\right)^{-d_1/2} \varphi\left(\frac{\|x_1\|^2}{\psi\left(\|x_2\|^2\right)}\right), \qquad x_1 \in \mathbb{R}^{d_1}, x_2 \in \mathbb{R}^{d_2}, \tag{5.130}$$

where  $\varphi:[0,\infty)\to(0,\infty)$  is a **completely monotone function**, that is

$$(-1)^n \varphi^{(n)}(t) \ge 0 \quad \forall n \in \mathbb{N}, \ t \ge 0,$$

and  $\psi:[0,\infty)\to\mathbb{R}_+$  has completely monotone derivative. The fact that (5.130) defines a non-negative definite function on  $\mathbb{R}^{d_1+d_2}$  if  $d_2=1$  was proved in [Gne02], and can be extended to  $d_2\geq 1$  using similar arguments. Assuming that B is a centered Gaussian random field with unit-variance and covariance function as in (5.130), we may infer that  $C(0,0)=\psi(0)^{-d_1/2}\varphi(0)=1$ , and suppose, without loss of generality, that  $\varphi(0)=\psi(0)=1$ . As a consequence, the Gaussian fields  $(B^{(1)}_{x_1})_{x_1\in\mathbb{R}^{d_1}}:=(B_{x_1,0})_{x_1\in\mathbb{R}^{d_1}}$  and  $(B^{(2)}_{x_2})_{x_2\in\mathbb{R}^{d_2}}:=(B_{0,x_2})_{x_2\in\mathbb{R}^{d_2}}$  have the following covariance functions

$$C_1(x_1) := C(x_1, 0) = \varphi(\|x_1\|^2);$$
 (5.131)

$$C_2(x_2) := C(0, x_2) = \psi(\|x_2\|^2)^{-d_1/2}$$
 (5.132)

For this reason, we rewrite (5.130) as

$$C(x_1, x_2) = C_2(x_2)C_1\left(x_1C_2(x_2)^{2/d_1}\right), \qquad x_1 \in \mathbb{R}^{d_1}, x_2 \in \mathbb{R}^{d_2}.$$
 (5.133)

From the expression (5.133), since  $C_1(x_1)$ ,  $C_2(x_2)$  are positive and non-increasing in the norms  $||x_1||$ ,  $||x_2||$ , we deduce that

$$C_2(x_2)C_1(x_1) \le C(x_1, x_2) \le C_2(x_2)C_1\left(x_1\underline{C_2}(D_2)^{2/d_1}\right),$$
 (5.134)

where  $D_2$  denotes the domain of the variable  $x_2$ , and

$$\underline{C_2}(D_2) = \inf_{x_2 \in (D_2 - D_2)} C_2(x_2) = \psi \left( \operatorname{diam}(D_2)^2 \right)^{-d_1/2}.$$
 (5.135)

Therefore,  $C(x_1,x_2)$  is wedged between two separable covariance functions,  $C_1(x_1)C_2(x_2)$  and  $C_1(x_1)C_2(x_2)$ . This suggests that the Gneiting case may be studied combining the bounds (5.134) and Theorem 5.1.1. We will partially formalize this intuition in Theorem 5.5.1, and we conjecture that the latter could be extended to more general functionals, growing domains and classes of non-separable covariance functions satisfying properties analogous to (5.134). These extensions are left for future research.

Analogously to the version of Theorem 5.1.1 explained in Remark 5.1.4, Theorem 5.5.1 allows to reduce the study of

$$Y(t_1, t_2)[q] = \int_{t_1 D_1 \times t_2 D_2} H_q(B_x) dx, \quad \text{as } t_i \to \infty,$$

for only one index  $i \in \{1, 2\}$  (i.e. only the *i*-th domain is growing), to that of the

respective marginal functional

$$Y_i(t_i)[q] = \int_{t_i D_i} H_q(B_{x_i}^{(i)}) dx_i, \quad \text{as } t_i \to \infty,$$

where  $D_i \subseteq \mathbb{R}^{d_i}$  are compact sets with  $\operatorname{Vol}(D_i) > 0$  and  $(B_{x_i}^{(i)})_{x_i \in \mathbb{R}^{d_i}}$  are Gaussian fields with covariance functions  $C_i$ , as defined in (5.131).

**Theorem 5.5.1.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $C : \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B, and assume that C is a Gneiting covariance function, see (5.130)-(5.133). Let  $Y(t_1, t_2)[q]$ ,  $Y_i(t_i)[q]$  be defined as above, and  $\widetilde{Y}$ ,  $\widetilde{Y}_i$  their normalized versions (see e.g. (5.3)). Fix  $j \in \{1, 2\}$ . Then, the following holds:

$$\widetilde{Y}_j(t_j)[q] \stackrel{d}{\to} N(0,1), \quad \text{as } t_j \to \infty$$

implies

$$\widetilde{Y}(t_1, t_2)[q] \stackrel{d}{\to} N(0, 1), \qquad \text{as } t_j \to \infty$$

where  $t_k$  is fixed, for  $k \neq j$ .

*Proof of Theorem 5.5.1.* The case q=1 is trivial because everything is Gaussian, so let us focus on  $q \ge 2$ . We will show the convergence by means of the Fourth Moment Theorem 5.2.1. First, recall that

$$Var(Y(t_1, t_2)[q]) = q! \int_{(t_1 D_1 \times t_2 D_2)^2} C^q(x - y) dx dy.$$

Thus, by (5.134) and (5.32), we have

$$\operatorname{Var}(Y_1(t_1)[q])\operatorname{Var}(Y_2(t_2)[q]) \le q! \operatorname{Var}(Y(t_1, t_2)[q]). \tag{5.136}$$

For the sake of brevity, let use denote  $h(x_2) := C_2(x_2)^{2/d_1}$ . We have

$$||f(t_1, t_2) \otimes_r f(t_1, t_2)||^2 = \int_{(t_1 D_1 \times t_2 D_2)^4} C_2(x_2 - y_2)^r C_2(z_2 - u_2)^r C_2(x_2 - z_2)^{q-r} C_2(y_2 - u_2)^{q-r}$$

$$\times C_1((x_1 - y_1)h(x_2 - y_2))^r C_1((z_1 - u_1)h(z_2 - u_2))^r$$

$$\times C_1((x_1 - z_1)h(x_2 - z_2))^{q-r} C_1((y_1 - u_1)h(y_2 - u_2))^{q-r} dx dy dz du.$$
(5.137)

If we fix  $t_1$  and let  $t_2 \to \infty$ , we get from  $0 \le C_1 \le 1$  that

$$||f(t_1, t_2) \otimes_r f(t_1, t_2)||^2 \le ||f_2(t_2) \otimes_r f_2(t_2)||^2 \operatorname{Vol}(t_1 D_1)^4.$$
(5.138)

If we fix  $t_2$  and let  $t_1 \to \infty$ , setting  $A := \underline{C_2}(t_2D_2)^{2/d_1} < \infty$ , by (5.134) we obtain

$$||f(t_{1},t_{2})\otimes_{r}f(t_{1},t_{2})||^{2} \leq \int_{(t_{1}D_{1}\times t_{2}D_{2})^{4}} C_{2}(x_{2}-y_{2})^{r}C_{2}(z_{2}-u_{2})^{r}C_{2}(x_{2}-z_{2})^{q-r}C_{2}(y_{2}-u_{2})^{q-r}$$

$$(5.139)$$

$$\times C_{1}((x_{1}-y_{1})A)^{r}C_{1}((z_{1}-u_{1})A)^{r}C_{1}((x_{1}-z_{1})A)^{q-r}C_{1}((y_{1}-u_{1})A)^{q-r}$$

$$(5.140)$$

$$dx dy dz du$$

$$(5.141)$$

$$= A^{-4d_{1}} \int_{(At_{1}D_{1}\times t_{2}D_{2})^{4}} C_{2}(x_{2}-y_{2})^{r}C_{2}(z_{2}-u_{2})^{r}C_{2}(x_{2}-z_{2})^{q-r}C_{2}(y_{2}-u_{2})^{q-r}$$

$$(5.142)$$

$$\times C_{1}(x_{1}-y_{1})^{r}C_{1}(z_{1}-u_{1})^{r}C_{1}(x_{1}-z_{1})^{q-r}C_{1}(y_{1}-u_{1})^{q-r}dx dy dz du$$

$$(5.143)$$

$$\leq A^{-4d_{1}} \operatorname{Vol}(t_{2}D_{2})^{4}||f_{1}(t_{1})\otimes_{r}f_{1}(t_{1})||^{2},$$

$$(5.144)$$

where in the last inequality we used that  $A \le 1$  and the positivity of the integrand. Applying the Fourth-Moment Theorem 5.2.1 leads to the desired conclusion.

#### 5.5.2 Additively separable covariance functions

In this subsection we consider **additively separable covariance functions** of the form

$$C(x_1, x_2) = K_1(x_1) + K_2(x_2), \qquad x_1 \in \mathbb{R}^{d_1}, \ x_2 \in \mathbb{R}^{d_2},$$
 (5.145)

where  $K_1$  and  $K_2$  are the covariance functions of two stationary, continuous, centered Gaussian fields  $(B^{(1)})_{x_1 \in \mathbb{R}^{d_1}}$  and  $(B^{(2)})_{x_2 \in \mathbb{R}^{d_2}}$ , with  $K_1(0), K_2(0) > 0$ . From the point of view of applications, the study of Gaussian fields with additively separable covariance function (5.145) is motivated by the fact that they can model the sum of two independent Gaussian fields. Unlike the Gneiting class considered in the previous subsection, which is comparable to the separable case (see (5.134)), here reduction theorems have to be developed in a different way. In this case, we define the marginal functionals as

$$A_i(t_i)[q] = \int_{t_i D_i} H_q\left(\frac{B_{x_i}^{(i)}}{K_i(0)}\right) dx_i.$$
 (5.146)

We introduce new quantities to describe the interplay between the growth rates of the volumes and the variances, that are the quotients

$$\gamma_{t_i}^i[q] := \frac{\int_{(t_i D_i)^2} K_i^q(x_i - y_i) dx_i dy_i}{\text{Vol}(t_i D_i)^2}.$$
 (5.147)

**Theorem 5.5.2.** Let  $B = (B_x)_{x \in \mathbb{R}^d}$  be a real-valued, continuous, centered, stationary Gaussian field with unit-variance. Let  $C : \mathbb{R}^d \to \mathbb{R}$  be the covariance function of B, and assume it is additively separable in the sense of (5.145), with  $K_1, K_2 \geq 0$ . Let  $\widetilde{Y}$  be as

in (5.3). For i=1,2, let  $A_i(t_i)[q]$  be as in (5.146),  $\widetilde{A}_i$  its normalized versions (see e.g. (5.3)), and recall the definition (5.147) of  $\gamma_{t_i}^i[q]$ . Finally, assume that

$$rac{\gamma_{t_1}^1[q]}{\gamma_{t_2}^2[q]} 
ightarrow 0 \quad \textit{as } t_1, t_2 
ightarrow \infty.$$

Then, the following holds:

$$\widetilde{A}_2(t_2)[q] \stackrel{d}{\to} N(0,1), \quad \text{as } t_2 \to \infty,$$

if and only if

$$\widetilde{Y}_1(t_1, t_2)[q] \stackrel{d}{\to} N(0, 1), \quad \text{as } t_1, t_2 \to \infty.$$

**Remark 5.5.1.** Note that, by symmetry, the roles of  $K_1$  and  $K_2$  can be exchanged in the statement of Theorem 5.5.2, obtaining an equivalence between  $\widetilde{A}_1(t_1)[q] \stackrel{d}{\to} N(0,1)$  and  $\widetilde{Y}(t_1,t_2)[q] \stackrel{d}{\to} N(0,1)$  as  $t_1,t_2 \to \infty$ , if  $\frac{\gamma_{t_2}^2[q]}{\gamma_{t_1}^1[q]} \to 0$ .

**Remark 5.5.2.** To ease the computation of the quotients  $\gamma_{t_i}^i[q]$ , i=1,2, as noted in Equation (5.188), assuming that  $K_i$  is both non-negative and non-negative definite, one can observe that

$$\gamma_{t_i}^i[q] \approx \frac{\int_{\{\|x\| \le t_i\}} K_i^q(x_i) dx_i}{t_i^{d_i}}.$$
 (5.148)

Just like Theorem 5.1.1 and Theorem 5.5.1, Theorem 5.5.2 should be interpreted as a reduction theorem, since it allows to reduce the asymptotic problem of a 2-domain functional  $Y(t_1,t_2)[q]$  to that of a 1-domain functional. The difference here is that the marginal functionals to be considered in the additively separable case are not the same considered for the separable and Gneiting classes. Moreover, unlike the latter cases, here the growth rates of the integration domains come into play by means of the quotients  $\gamma_{t_i}^i[q]$  in (5.147) (see also Example 5.5.2). In order to prove Theorem 5.5.2, we state and prove the following lemma.

**Lemma 5.5.3.** Under the notations and assumptions of Theorem 5.5.2, we have that

$$\operatorname{Var}(Y(t_1, t_2)[q]) = \operatorname{Vol}(t_1 D_1)^2 \operatorname{Var}(A_2(t_2)[q]) + \operatorname{Var}(A_1(t_1)[q]) \operatorname{Vol}(t_2 D_2)^2 + \sum_{k=1}^{q-1} {q \choose k}^2 \operatorname{Var}(A_1(t_1)[k]) \operatorname{Var}(A_2(t_2)[q-k]).$$
 (5.149)

In addition, assuming  $q \geq 2$ , we have that

$$\frac{\operatorname{Var}(Y(t_1, t_2)[q])}{\operatorname{Vol}(t_1 D_1)^2 \operatorname{Var}(A_2(t_2)[q])} \longrightarrow 1 \qquad \text{as } t_1, t_2 \to \infty.$$
 (5.150)

*Proof.* Recall  $q \ge 2$ . Applying Newton's binomial formula, we may write

$$Var(Y(t_1, t_2)[q]) = q! \int_{(t_1 D_1 \times t_2 D_2)^2} C(x - y)^q dx dy$$
(5.151)

$$= q! \int_{(t_1 D_1 \times t_2 D_2)^2} (K_1(x_1 - y_1) + K_2(x_2 - y_2))^q dx_1 dx_2 dy_1 dy_2$$
 (5.152)

$$= \sum_{k=0}^{q} q! \binom{q}{k} \int_{(t_1 D_1)^2} K_1(x_1 - y_1)^k dx_1 dy_1 \cdot \int_{(t_2 D_2)^2} K_2(x_2 - y_2)^{q-k} dx_2 dy_2,$$
(5.153)

which is (5.149). Now assume that  $\gamma_{t_1}^1[q]/\gamma_{t_2}^2[q] \to 0$ . Then, as  $t_1, t_2 \to \infty$ 

$$\frac{\operatorname{Var}(Y(t_1,t_2)[q])}{\operatorname{Vol}(t_1D_1)^2\operatorname{Var}(A_2(t_2)[q])} \longrightarrow 1.$$

Indeed,

$$\frac{\operatorname{Var}(A_1(t_1)[q])\operatorname{Vol}(t_2D_2)^2}{\operatorname{Vol}(t_1D_1)^2\operatorname{Var}(A_2(t_2)[q])} = \frac{\gamma_{t_1}^1[q]}{\gamma_{t_2}^2[q]} \longrightarrow 0$$

by assumption. By Jensen inequality in  $t_i$  and recalling that  $K_i \geq 0$ , we have that

$$\operatorname{Var}(A_{i}(t_{i})[k]) = k! \int_{(t_{i}D_{i})^{2}} K_{i}^{k}(x_{i} - y_{i}) dx_{i} dy_{i} \lesssim \operatorname{Vol}(t_{i}D_{i})^{2(q-k)/q} \operatorname{Var}(A_{i}(t_{i})[q])^{k/q}.$$
(5.154)

Therefore, the proof is concluded by observing that, for k > 0, we have

$$\frac{\operatorname{Var}(A_{1}(t_{1})[k])\operatorname{Var}(A_{2}(t_{2})[q-k])}{\operatorname{Vol}(t_{1}D_{1})^{2}\operatorname{Var}(A_{2}(t_{2})[q])} \lesssim \frac{\operatorname{Var}(A_{1}(t_{1})[q])^{k/q}\operatorname{Vol}(t_{2}D_{2})^{2k/q}}{\operatorname{Vol}(t_{1}D_{1})^{2k/q}\operatorname{Var}(A_{2}(t_{2})[q])^{k/q}} = \frac{\gamma_{t_{1}}^{1}[q]^{k/q}}{\gamma_{t_{2}}^{2}[q]^{k/q}} \longrightarrow 0.$$
(5.155)

*Proof of Theorem 5.5.2.* By assumption,  $\gamma_{t_1}^1[q]/\gamma_{t_2}^2[q] \to 0$ . By Lemma 5.5.3, as  $t_1, t_2 \to \infty$  we have

$$Var(Y(t_1, t_2)[q]) \sim Vol(t_1D_1)^2 Var(A_2(t_2)[q]).$$

(5.166)

Regarding contractions, by using the Newton binomial formula, we have

$$\begin{aligned} &\|f(t_1,t_2)\otimes_r f(t_1,t_2)\|^2 & (5.156) \\ &= \int_{(t_1D_1\times t_2D_2)^4} C(x-y)^r C_2(z-u)^r C(x-z)^{q-r} C_2(y-u)^{q-r} dx \ dy \ dz \ du & (5.157) \\ &= \int_{(t_1D_1\times t_2D_2)^4} dx \ dy \ dz \ du & (5.158) \\ & (K_1(x_1-y_1)+K_2(x_2-y_2))^r (K_1(z_1-u_1)+K_2(z_2-u_2))^r & (5.159) \\ & \times (K_1(x_1-z_1)+K_2(x_2-z_2))^{q-r} (K_1(y_1-u_1)+K_2(y_2-u_2))^{q-r} \\ & (5.160) \end{aligned}$$

$$&= \sum_{k_1,k_2=0}^r \sum_{k_3,k_4=0}^{q-r} \binom{r}{k_1} \binom{r}{k_2} \binom{q-r}{k_3} \binom{q-r}{k_4} & (5.161) \\ \int_{(t_1D_1)^4} K_1(x_1-y_1)^{k_1} K_1(z_1-u_1)^{k_2} K_1(x_1-z_1)^{k_3} K_1(y_1-u_1)^{k_4} \ dx_1 \ dy_1 \ dz_1 \ du_1 \\ & (5.162) \\ \int_{(t_2D_2)^4} K_2(x_2-y_2)^{r-k_1} K_2(z_2-u_2)^{r-k_2} K_2(x_2-z_2)^{q-r-k_3} K_2(y_2-u_2)^{q-r-k_4} \ dx_2 \ dy_2 \ dz_2 \ du_2 \\ & (5.163) \\ &= \sum_{k_1,k_2=0}^r \sum_{k_3,k_4=0}^{q-r} \mathbf{1}_{\{(k_1,k_2,k_3,k_4)\neq(0,0,0,0)\}} \binom{r}{k_1} \binom{r}{k_2} \binom{q-r}{k_3} \binom{q-r}{k_4} & (5.164) \\ \int_{(t_1D_1)^4} K_1(x_1-y_1)^{k_1} K_1(z_1-u_1)^{k_2} K_1(x_1-z_1)^{k_3} K_1(y_1-u_1)^{k_4} & (5.165) \\ \int_{(t_2D_2)^4} K_2(x_2-y_2)^{r-k_1} K_2(z_2-u_2)^{r-k_2} K_2(x_2-z_2)^{q-r-k_3} K_2(y_2-u_2)^{q-r-k_4} \end{aligned}$$

+ 
$$\operatorname{Vol}(t_1 D_1)^4 || f_2(t_2) \otimes_r f_2(t_2) ||^2$$
. (5.167)

Moreover, we have

$$\sum_{k_{1},k_{2}=0}^{r} \sum_{k_{3},k_{4}=0}^{q-r} \mathbf{1}_{\{(k_{1},k_{2},k_{3},k_{4})\neq(0,0,0,0)\}} \binom{r}{k_{1}} \binom{r}{k_{2}} \binom{q-r}{k_{3}} \binom{q-r}{k_{4}}$$

$$\int_{(t_{1}D_{1})^{4}} K_{1}(x_{1}-y_{1})^{k_{1}} K_{1}(z_{1}-u_{1})^{k_{2}} K_{1}(x_{1}-z_{1})^{k_{3}} K_{1}(y_{1}-u_{1})^{k_{4}} dx_{1} dy_{1} dz_{1} du_{1}$$

$$(5.169)$$

$$\int_{(t_{2}D_{2})^{4}} K_{2}(x_{2}-y_{2})^{r-k_{1}} K_{2}(z_{2}-u_{2})^{r-k_{2}} K_{2}(x_{2}-z_{2})^{q-r-k_{3}} K_{2}(y_{2}-u_{2})^{q-r-k_{4}}$$

$$(5.170)$$

$$\lesssim \sum_{k_{1},k_{2}=0}^{r} \sum_{k_{3},k_{4}=0}^{q-r} \mathbf{1}_{\{(k_{1},k_{2},k_{3},k_{4})\neq(0,0,0,0)\}} \binom{r}{k_{1}} \binom{r}{k_{2}} \binom{q-r}{k_{3}} \binom{q-r}{k_{4}} \times$$

$$\times \int_{(t_{1}D_{1})^{2}} K_{1}(x_{1}-y_{1})^{k_{1}+k_{3}} dx_{1} dy_{1} \int_{(t_{2}D_{2})^{2}} K_{2}(x_{2}-y_{2})^{q-k_{1}-k_{3}} dx_{2} dy_{2}$$

$$\times \int_{(t_{1}D_{1})^{2}} K_{1}(z_{1}-u_{1})^{k_{2}+k_{4}} dz_{1} du_{1} \int_{(t_{2}D_{2})^{2}} K_{2}(z_{2}-u_{2})^{q-k_{2}-k_{4}} dz_{2} du_{2},$$

$$(5.173)$$

where the last inequality follows from the positivity of  $K_1, K_2$  and Lemma 5.5.4 applied to every term of the sum. Then, using Jensen as in the proof of Lemma 5.5.3, we obtain that, as  $t_1, t_2 \to \infty$ ,

$$\|\widetilde{f}(t_{1}, t_{2}) \otimes_{r} \widetilde{f}(t_{1}, t_{2})\|_{2q-2r}^{2} = (q!)^{2} \frac{\|f(t_{1}, t_{2}) \otimes_{r} f(t_{1}, t_{2})\|_{2q-2r}^{2}}{\operatorname{Var}(Y(t_{1}, t_{2}))^{2}} \sim \frac{\|f(t_{1}, t_{2}) \otimes_{r} f(t_{1}, t_{2})\|^{2}}{\operatorname{Vol}(t_{1}D_{1})^{4} \operatorname{Var}(A_{2}(t_{2}))^{2}}$$

$$= \frac{\|f_{2}(t_{2}) \otimes_{r} f_{2}(t_{2})\|^{2}}{\operatorname{Var}(A_{2}(t_{2}))^{2}}$$

$$+ O\left(\sum_{k_{1}, k_{2}=0}^{r} \sum_{k_{3}, k_{4}=0}^{q-r} \mathbf{1}_{\{(k_{1}, k_{2}, k_{3}, k_{4}) \neq (0, 0, 0, 0)\}} \left(\frac{\gamma_{t_{1}}^{1}[q]}{\gamma_{t_{2}}^{2}[q]}\right)^{(k_{1}+k_{3})/q} \left(\frac{\gamma_{t_{1}}^{1}[q]}{\gamma_{t_{2}}^{2}[q]}\right)^{(k_{2}+k_{4})/q}\right)$$

$$(5.176)$$

$$\sim (q!)^{2} \frac{\|f_{2}(t_{2}) \otimes_{r} f_{2}(t_{2})\|_{2q-2r}^{2}}{\operatorname{Var}(A_{2}(t_{2}))^{2}} = \|\widetilde{f}_{2}(t_{2}) \otimes_{r} \widetilde{f}_{2}(t_{2})\|_{2q-2r}^{2}.$$

$$(5.177)$$

Therefore, the proof is again concluded by means of the Fourth Moment theorem 5.2.1.

We conclude the section with an example.

**Example 5.5.2.** Let us consider a Gaussian random field  $B = (B_x)_{x \in \mathbb{R}^{d_1 + d_2}}$  with covariance function of the form (5.145), choosing

$$K_1(x_1) = \frac{1}{(1 + \|x_1\|^2)^{\beta_1/2}}, \qquad K_2(x_2) = \frac{1}{(1 + \|x_2\|^2)^{\beta_2/2}}.$$
 (5.178)

and consider its quadratic variation, that is

$$Y(t_1, t_2)[2] = \int_{t_1 D_1 \times t_2 D_2} H_2(B_x) dx.$$
 (5.179)

Suppose that  $\beta_1 \in (0, d_1/2)$  and  $\beta_2 > d_2/2$ . Note that  $K_1$  satisfies both (5.7) and (5.8) (see [LO13, Example 3]). Then, by Theorem 5.2.4 the functional  $\widetilde{A}_1(t_1)[2]$ , see (5.146) properly normalized, is not asymptotically Gaussian; on the other hand,  $\widetilde{A}_2(t_2)[2]$  is asymptotically Gaussian, thanks to Theorem 5.2.3. Note that, up to constants, recalling Remark 5.5.2,

$$\frac{\gamma_{t_1}^1[2]}{\gamma_{t_2}^2[2]} \sim \frac{t_1^{-2\beta_1}}{t_2^{-d_2}}.$$
 (5.180)

Therefore, we may observe two different behaviors depending on the choice of the rate for  $t_1$  and  $t_2$ : applying Theorem 5.5.2, whenever  $t_1^{-2\beta_1} = o(t_2^{-d_2})$ , we have that  $\widetilde{Y}(t_1,t_2)$  is not asymptotically Gaussian; conversely, when  $t_2^{-d_2} = o(t_1^{-2\beta_1})$  we have a Gaussian limiting behavior.

#### **Funding**

Nikolai Leonenko was partially supported under the ARC Discovery Grant DP220101680 (Australia), LMS grant 42997 (UK) and grant FAPESP 22/09201-8 (Brazil). Leonardo Maini was supported by the Luxembourg National Research Fund PRIDE17/1224660/GPS. Ivan Nourdin and Francesca Pistolato were partially supported by the Luxembourg National Research Fund (Grant: O22/17372844/FraMStA).

## **Appendix**

*Proof of Lemma 5.3.6.* The following proof is merely a reformulation of [Maj81, Theorem 4.5] with our notation. Since  $h \in \mathcal{H}_{\nu}^{\odot q}$ , we have the equality

$$h = \sum_{i_1, \dots, i_q = 1}^{\infty} c_{i_1, \dots, i_q} \operatorname{Sym}(e_{i_1} \otimes \dots \otimes e_{i_q}),$$

where  $\{e_i\}_{i=1}^{\infty}$  is an orthonormal basis of  $\mathcal{H}_{\nu}$  and Sym is the symmetrization operator. If we define  $e_i'(x) := e_i(x) \, a(x) \in L^2_E(\nu')$ , then  $\{e_i'\}_{i=1}^{\infty}$  is an orthonormal basis of  $\mathcal{H}_{\nu'}$  and we obtain

$$a^{\otimes q}h = \sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} \operatorname{Sym}(e'_{i_1} \otimes \dots \otimes e'_{i_q}).$$

Applying the operators  $I_{\nu,q}$  and  $I_{\nu',q}$  we obtain

$$I_{\nu,q}(h) = \sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} I_{\nu,q} \left( \operatorname{Sym}(e_{i_1} \otimes \dots \otimes e_{i_q}) \right)$$

and

$$I_{\nu',q}(a^{\otimes q}h) = \sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} I_{\nu',q} \left( \operatorname{Sym}(e'_{i_1} \otimes \dots \otimes e'_{i_q}) \right).$$

Therefore, we conclude by means of product formula (see e.g. [NP12]), observing that

$$I_{\nu,q}(h) = \sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} X_{\nu}(e_{i_1}) \dots X_{\nu}(e_{i_q}) \stackrel{d}{=} \sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} N_{i_1} \dots N_{i_q}$$

and

$$I_{\nu',q}(a^{\otimes q}h) = \sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} X_{\nu}(e'_{i_1}) \dots X_{\nu}(e'_{i_q}) \stackrel{d}{=} \sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} N_{i_1} \dots N_{i_q},$$

where  $\{N_i\}_{i=1}^{\infty}$  is a sequence of i.i.d. standard Gaussian random variables.

*Proof of Lemma 5.3.7.* Acting as in the proof of Lemma 5.3.6, we have

$$I_q(h(\cdot)) = \sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} X(e_{i_1}(\cdot)) \dots X(e_{i_q}(\cdot))),$$

where  $\{e_i\}_{i=1}^{\infty}$  is an orthonormal basis of  $\mathcal{H}_{\lambda}$ . Note that using the multi-index notation  $\underline{s}^{\alpha} = \prod_i s_i^{\alpha}$  for  $s \in \mathbb{N}^d$  and  $\alpha \in \mathbb{R}$ , also the family  $\{\underline{s}^{-1/2}e_i(\cdot/s_1,\ldots,\cdot/s_d)\}_{i=1}^{\infty}$  is an orthonormal basis of  $\mathcal{H}_{\lambda}$ . Therefore, the two Gaussian families  $(X(e_i(\cdot)))_{i=1}^{\infty}$  and  $(X(\underline{s}^{-1/2}e_i(\cdot/s_1,\ldots,\cdot/s_d)))_{i=1}^{\infty}$  share the same law, and, by extension, the following random variables are equally distributed:

$$I_{q}(h(\cdot)) \stackrel{\text{law}}{=} \sum_{i_{1},\dots,i_{q}=1}^{\infty} c_{i_{1},\dots,i_{q}} X(\underline{s}^{-1/2}e_{i_{1}}(\cdot/s_{1},\dots,\cdot/s_{d})) \dots X(\underline{s}^{-1/2}e_{i_{q}}(\cdot/s_{1},\dots,\cdot/s_{d})).$$

$$(5.181)$$

Moreover, by product formula (see e.g. [NP12]), we have that

$$\sum_{i_1,\dots,i_q=1}^{\infty} c_{i_1,\dots,i_q} X(\underline{s}^{-1/2} e_{i_1}(\cdot/s_1,\dots,\cdot/s_d)) \dots X(\underline{s}^{-1/2} e_{i_q}(\cdot/s_1,\dots,\cdot/s_d)) \stackrel{d}{=} I_q(\underline{s}^{-\frac{q}{2}} \tilde{h}(s_1,\dots,s_d)).$$

$$(5.182)$$

By linearity,  $I_q(\underline{s}^{-\frac{q}{2}}\tilde{h}(s_1,\ldots,s_d))=\underline{s}^{-\frac{q}{2}}I_q(\tilde{h}(s_1,\ldots,s_d))$ , which concludes the proof.

**Lemma 5.5.4.** Let  $K : \mathbb{R}^d \to \mathbb{R}$  be a non-negative definite function such that  $K \geq 0$ . Suppose that  $D \subseteq \mathbb{R}^d$  compact with  $\operatorname{Vol}(D) > 0$ . Then, for every  $k_1, k_2, k_3, k_4 \in \mathbb{N}$ , there exists a positive constant c, depending on K, D and the exponents  $k_i$ , i = 1, 2, 3, 4, satisfying for all t > 0 the following inequality:

$$\int_{(tD)^4} K(x-y)^{k_1} K(z-u)^{k_2} K(x-z)^{k_3} K(y-u)^{k_4} dx dy dz du$$

$$\leq c \int_{(tD)^2} K(x-y)^{k_1+k_3} dx dy \int_{(tD)^2} K(z-u)^{k_2+k_4} dz du. \quad (5.183)$$

*Proof.* First, by using the inequality  $x^a y^b \leq x^{a+b} + y^{a+b}$ , the LHS of the inequality

(5.183) can be bounded by

$$2\int_{(tD)^4} K(x-y)^{k_1+k_3} K(z-u)^{k_2+k_4} dx dy dz du + 2\int_{(tD)^4} K(x-y)^{k_1+k_3} K(y-u)^{k_2+k_4} dx dy dz du.$$
(5.184)

Then, by the change of variable a = x - y and b = y - u and compactness of the domain D, we may also bound the second term in the previous equation, up to a positive constant, by

$$\operatorname{Vol}(tD) \int_{(tD-tD)^2} K(a)^{k_1+k_3} K(b)^{k_2+k_4} \operatorname{Vol}(tD \cap (tD-a) \cap (tD-b)) dadb$$

$$\leq \operatorname{Vol}(tD)^2 \int_{tD-tD} K(a)^{k_1+k_3} da \int_{tD-tD} K(b)^{k_2+k_4} db. \quad (5.185)$$

To conclude, we need to show that

$$Vol(tD) \int_{tD-tD} K(a)^{k_1+k_3} da \lesssim \int_{(tD)^2} K(x-y)^{k_1+k_3} dx dy.$$
 (5.186)

By positivity of the integrand and the doubling conditions for non-negative definite functions that are also non-negative proved in [GT19], we deduce that

$$\int_{tD-tD} K(a)^{k_1+k_3} da \le \int_{\{\|x\| \le t \cdot \operatorname{diam}(D)\}} K(a)^{k_1+k_3} da \lesssim \int_{\{\|x\| \le t\}} K(a)^{k_1+k_3} da.$$
(5.187)

Reasoning as in [MN24, Proof of Proposition 9, Step 1], we have that

$$t^{d} \int_{\{\|x\| \le t\}} K(a)^{k_1 + k_3} da \approx \int_{(tD)^2} K(x - y)^{k_1 + k_3} dx dy, \tag{5.188}$$

which is enough to conclude.

- [AT07] R. J. Adler and J. E. Taylor. *Random fields and geometry*. Springer Monographs in Mathematics. Springer, New York, 2007, pp. xviii+448 (cit. on pp. 24, 65, 87, 112).
- [AO18] T. Alodat and A. Olenko. "Weak convergence of weighted additive functionals of long-range dependent fields". In: *Theory Probab. Math. Statist.* 97 (2018), pp. 1–16 (cit. on pp. 27, 112).
- [ALO15] V. Anh, N. Leonenko, and A. Olenko. "On the rate of convergence to Rosenblatt-type distribution". In: *J. Math. Anal. Appl.* 425.1 (2015), pp. 111–132. DOI: 10.1016/j.jmaa.2014.12.016 (cit. on p. 83).
- [ALS15] F. Avram, N. Leonenko, and L. Sakhno. "Limit theorems for additive functionals of stationary fields, under integrability assumptions on the higher order spectral densities". In: *Stochastic Process. Appl.* 125.4 (2015), pp. 1629–1652. DOI: 10.1016/j.spa.2014.11.010 (cit. on p. 83).
- [APY21] E. Azmoodeh, G. Peccati, and X. Yang. "Malliavin-Stein method: a survey of some recent developments". In: *Mod. Stoch. Theory Appl.* 8.2 (2021), pp. 141–177. DOI: 10.15559/21-vmsta184 (cit. on p. 19).
- [Azm+16] E. Azmoodeh et al. "Generalization of the Nualart-Peccati criterion". In: *Ann. Probab.* 44.2 (2016), pp. 924–954. DOI: 10.1214/14-A0P992 (cit. on p. 17).
- [Azo09] T. T. Azomahou. "Memory properties and aggregation of spatial autoregressive models". In: *J. Statist. Plann. Inference* 139.8 (2009), pp. 2581–2597. DOI: 10.1016/j.jspi.2008.11.015 (cit. on p. 23).
- [BPP22] F. Bachoc, A. P. Peron, and E. Porcu. "Multivariate Gaussian Random Fields over Generalized Product Spaces involving the Hypertorus". In: *Theory Probab. Math. Statist.* 107 (2022), pp. 3–14 (cit. on pp. 27, 111).

[BMM22] D. Beliaev, M. McAuley, and S. Muirhead. "Fluctuations of the number of excursion sets of planar Gaussian fields". In: *Probab. Math. Phys.* 3.1 (2022), pp. 105–144. DOI: 10.2140/pmp.2022.3.105 (cit. on p. 3).

- [BH02] S. Ben Hariz. "Limit theorems for the non-linear functional of stationary Gaussian processes". In: *J. Multivariate Anal.* 80.2 (2002), pp. 191–216. DOI: 10.1006/jmva.2001.1986 (cit. on pp. 24, 83, 96).
- [Ber77] M. V. Berry. "Regular and irregular semiclassical wavefunctions". In: *J. Phys. A* 10.12 (1977), pp. 2083–2091 (cit. on pp. 7, 40, 67, 68, 99, 108).
- [Ber02] M. V. Berry. "Statistics of nodal lines and points in chaotic quantum billiards: perimeter corrections, fluctuations, curvature". In: *J. Phys. A* 35.13 (2002), pp. 3025–3038. DOI: 10.1088/0305-4470/35/13/301 (cit. on p. 69).
- [BGT89] N. H. Bingham, C. M. Goldie, and J. L. Teugels. *Regular variation*. Vol. 27. Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, 1989, pp. xx+494 (cit. on pp. 84, 91).
- [BSV16] J. M. Borwein, A. Straub, and C. Vignat. "Densities of short uniform random walks in higher dimensions". In: *J. Math. Anal. Appl.* 437.1 (2016), pp. 668–707. DOI: 10.1016/j.jmaa.2016.01.017 (cit. on pp. 72, 73, 75).
- [BSW13] J. M. Borwein, A. Straub, and J. Wan. "Three-step and four-step random walk integrals". In: *Exp. Math.* 22.1 (2013), pp. 1–14. DOI: 10.1080/10586458.2013.748379 (cit. on p. 72).
- [Bor+11] J. M. Borwein et al. "Some arithmetic properties of short random walk integrals". In: *Ramanujan J.* 26.1 (2011), pp. 109–132. DOI: 10.1007/s11139-011-9325-y (cit. on p. 72).
- [Bor+12] J. M. Borwein et al. "Densities of short uniform random walks". In: *Canad. J. Math.* 64.5 (2012). With an appendix by Don Zagier, pp. 961–990. DOI: 10.4153/CJM-2011-079-2 (cit. on pp. 72, 75).
- [BHI03] L. Brandolini, S. Hofmann, and A. Iosevich. "Sharp rate of average decay of the Fourier transform of a bounded set". In: *Geom. Funct. Anal.* 13.4 (2003), pp. 671–680. DOI: 10.1007/s00039-003-0426-7 (cit. on pp. 43, 55, 59, 63, 104, 105).
- [Bre11] J. C. Breton. "On the rate of convergence in non-central asymptotics of the Hermite variations of fractional Brownian sheet". In: *Probab. Math. Statist.* 31.2 (2011), pp. 301–311 (cit. on pp. 27, 112).
- [BN08] J. C. Breton and I. Nourdin. "Error bounds on the non-normal approximation of Hermite power variations of fractional Brownian motion". In: *Electron. Commun. Probab.* 13 (2008), no. 46, 482–493. DOI: 10.1214/ECP.v13-1415 (cit. on p. 134).

[BM83] P. Breuer and P. Major. "Central limit theorems for nonlinear functionals of Gaussian fields". In: *J. Multivariate Anal.* 13.3 (1983), pp. 425–441. DOI: 10.1016/0047-259X(83)90019-2 (cit. on pp. 24, 28, 38, 39, 70, 83, 96, 112, 118).

- [Cam19] V. Cammarota. "Nodal area distribution for arithmetic random waves". In: *Trans. Amer. Math. Soc.* 372.5 (2019), pp. 3539–3564. DOI: 10. 1090/tran/7779 (cit. on p. 3).
- [CMR23] V. Cammarota, D. Marinucci, and M. Rossi. "Lipschitz-Killing curvatures for arithmetic random waves". In: *Ann. Sc. Norm. Super. Pisa Cl. Sci. (5)* 24.2 (2023), pp. 1095–1147. DOI: 10.2422/2036–2145.202010\\_065 (cit. on p. 3).
- [CNN20] S. Campese, I. Nourdin, and D. Nualart. "Continuous Breuer-Major theorem: tightness and nonstationarity". In: *Ann. Probab.* 48.1 (2020), pp. 147–177. DOI: 10.1214/19-A0P1357 (cit. on pp. 24, 83, 96).
- [Cam+16] S. Campese et al. "Multivariate Gaussian approximations on Markov chaoses". In: *Electron. Commun. Probab.* 21 (2016), Paper No. 48, 9. DOI: 10.1214/16-ECP4615 (cit. on p. 17).
- [CH15] Y. Canzani and B. Hanin. "Scaling limit for the kernel of the spectral projector and remainder estimates in the pointwise Weyl law". In: *Anal. PDE* 8.7 (2015), pp. 1707–1731. DOI: 10.2140/apde.2015.8.1707 (cit. on p. 3).
- [CH18] Y. Canzani and B. Hanin. " $C^{\infty}$  scaling asymptotics for the spectral projector of the Laplacian". In: *J. Geom. Anal.* 28.1 (2018), pp. 111–122. DOI: 10.1007/s12220-017-9812-5 (cit. on p. 3).
- [CH20] Y. Canzani and B. Hanin. "Local universality for zeros and critical points of monochromatic random waves". In: *Comm. Math. Phys.* 378.3 (2020), pp. 1677–1712. DOI: 10.1007/s00220-020-03826-w (cit. on p. 3).
- [Chr92] G. Christakos. *Random Field Models in Earth Sciences*. Academic Press, 1992 (cit. on pp. 27, 111, 137).
- [CL12] S. Cohen and M. A. Lifshits. "Stationary Gaussian random fields on hyperbolic spaces and on Euclidean spheres". In: *ESAIM Probab. Stat.* 16 (2012), pp. 165–221. DOI: 10.1051/ps/2011105 (cit. on p. 67).
- [DEL21] F. Dalmao, A. Estrade, and J. R. León. "On 3-dimensional Berry's model". In: *ALEA Lat. Am. J. Probab. Math. Stat.* 18.1 (2021), pp. 379–399. DOI: 10.30757/alea.v18-17 (cit. on p. 3).
- [DER20] E. Di Bernardino, A. Estrade, and M. Rossi. "On the excursion area of perturbed Gaussian fields". In: *ESAIM Probab. Stat.* 24 (2020), pp. 252–274. DOI: 10.1051/ps/2020002 (cit. on p. 3).
- [Die+23] G. Dierickx et al. "Small scale CLTs for the nodal length of monochromatic waves". In: *Comm. Math. Phys.* 397.1 (2023), pp. 1–36. DOI: 10.1007/s00220-022-04422-w (cit. on pp. 3, 40, 99).

[DR07] P. J. Diggle and P. J. Ribeiro. "Gaussian models for geostatistical data". In: *Model-based Geostatistics*. New York, NY: Springer New York, 2007, pp. 46–78. DOI: 10.1007/978-0-387-48536-2\_3 (cit. on p. 111).

- [DP17] C. Döbler and G. Peccati. "Quantitative de Jong theorems in any dimension". In: *Electronic Journal of Probability* 22.none (2017), pp. 1 –35. DOI: 10.1214/16-EJP19 (cit. on p. 17).
- [DP18] C. Döbler and G. Peccati. "The fourth moment theorem on the Poisson space". In: *Ann. Probab.* 46.4 (2018), pp. 1878–1916. DOI: 10.1214/17–A0P1215 (cit. on p. 17).
- [DM79] R. L. Dobrushin and P. Major. "Non-central limit theorems for nonlinear functionals of Gaussian fields". In: *Z. Wahrsch. Verw. Gebiete* 50.1 (1979), pp. 27–52. DOI: 10.1007/BF00535673 (cit. on pp. 25, 38–40, 43, 84, 99, 112, 119, 134).
- [DOV22] I. Donhauzer, A. Olenko, and A. Volodin. "Strong law of large numbers for functionals of random fields with unboundedly increasing covariances". In: *Comm. Statist. Theory Methods* 51.20 (2022), pp. 6947–6962. DOI: 10.1080/03610926.2020.1868515 (cit. on p. 82).
- [EF20] A. Estrade and J. Fournier. "Anisotropic Gaussian wave models". In: *ALEA Lat. Am. J. Probab. Math. Stat.* 17.1 (2020), pp. 329–353. DOI: 10.30757/alea.v17-13 (cit. on p. 5).
- [Flo97] K. Floret. "Natural norms on symmetric tensor products of normed spaces". In: *Proceedings of the Second International Workshop on Functional Analysis (Trier, 1997)*. Vol. 17. 1997, pp. 153–188 (cit. on pp. 14, 50).
- [Gal11] B. Galerne. "Computation of the perimeter of measurable sets via their covariogram. Applications to random sets". In: *Image Anal. Stereol.* 30.1 (2011), pp. 39–51. DOI: 10.5566/ias.v30.p39-51 (cit. on pp. 20, 53, 83, 85, 90, 127).
- [Gas23] L. Gass. "Almost-sure asymptotics for Riemannian random waves". In: Bernoulli 29.1 (2023), pp. 625–651. DOI: 10.3150/22-bej1471 (cit. on p. 3).
- [GS23] L. Gass and M. Stecconi. *The number of critical points of a Gaussian field: finiteness of moments*. 2023. arXiv: 2305.17586 [math.PR] (cit. on p. 3).
- [Gne02] T. Gneiting. "Nonseparable, stationary covariance functions for spacetime data". In: *J. Amer. Statist. Assoc.* 97.458 (2002), pp. 590–600. DOI: 10.1198/016214502760047113 (cit. on pp. 27, 111, 137, 138).
- [GMO18] L. Golinskii, M. Malamud, and L. Oridoroga. "Radial positive definite functions and Schoenberg matrices with negative eigenvalues". In: *Trans. Amer. Math. Soc.* 370.1 (2018), pp. 1–25. DOI: 10.1090/tran/6876 (cit. on p. 65).

[GT19] D. Gorbachev and S. Tikhonov. "Doubling condition at the origin for non-negative positive definite functions". In: *Proc. Amer. Math. Soc.* 147.2 (2019), pp. 609–618. DOI: 10.1090/proc/14191 (cit. on pp. 53, 54, 58, 59, 61, 127, 147).

- [GMT24] F. Grotto, L. Maini, and A. P. Todino. "Fluctuations of polyspectra in spherical and Euclidean random wave models". In: *Electronic Communications in Probability* 29.none (2024), pp. 1 –12. DOI: 10.1214/24–ECP578 (cit. on pp. 27, 67, 84, 99, 101).
- [GP23] F. Grotto and G. Peccati. "Nonlinear Functionals of Hyperbolic Random Waves: the Wiener Chaos Approach". In: *arXiv e-prints*, arXiv:2301.08315 (2023), arXiv:2301.08315 (cit. on pp. 3, 67).
- [HMP24] R. Herry, D. Malicet, and G. Poly. "Superconvergence phenomenon in Wiener chaoses". In: *Ann. Probab., to appear*, arXiv:2303.02628 (2024), arXiv:2303.02628. arXiv: 2303.02628 [math.PR] (cit. on p. 17).
- [HLX23] M. Hong, H. Liu, and F. Xu. *Limit theorems for additive functionals of some self-similar Gaussian processes*. 2023. arXiv: 2305.13146 [math.PR] (cit. on p. 5).
- [IL71] I. A. Ibragimov and Y. V. Linnik. *Independent and stationary sequences of random variables*. Ed. by J. F. C. Kingman. With a supplementary chapter by I. A. Ibragimov and V. V. Petrov. Wolters-Noordhoff Publishing, Groningen, 1971, p. 443 (cit. on pp. 2, 22).
- [IL89] A. V. Ivanov and N. N. Leonenko. *Statistical analysis of random fields*. Vol. 28. Mathematics and its Applications (Soviet Series). With a preface by A. V. Skorokhod, Translated from the Russian by A. I. Kochubinskii. Kluwer Academic Publishers Group, Dordrecht, 1989, pp. x+244. DOI: 10.1007/978-94-009-1183-3 (cit. on pp. 88, 96, 97).
- [Jar+23] A. Jaramillo et al. "Limit theorems for additive functionals of the fractional Brownian motion". In: *Ann. Probab.* 51.3 (2023), pp. 1066–1111. DOI: 10.1214/22-aop1612 (cit. on p. 5).
- [KS18] J. Kampf and E. Spodarev. "A functional central limit theorem for integrals of stationary mixing random fields". In: *Teor. Veroyatn. Primen*. 63.1 (2018), pp. 167–185. DOI: 10.4213/tvp5161 (cit. on pp. 3, 83).
- [Klu06] J. C. Kluyver. "Some formulae concerning the integers less than n and prime to n". In: *Koninklijke Nederlandse Akademie van Wetenschappen Proceedings Series B Physical Sciences* 9 (Jan. 1906), pp. 408–414 (cit. on p. 72).
- [Kra14] I. Krasikov. "Approximations for the Bessel and Airy functions with an explicit error term". In: *LMS J. Comput. Math.* 17.1 (2014), pp. 209–225. DOI: 10.1112/S1461157013000351 (cit. on pp. 29, 40, 46, 66, 74, 99, 100).

[LPS16] G. Last, G. Peccati, and M. Schulte. "Normal approximation on Poisson spaces: Mehler's formula, second order Poincaré inequalities and stabilization". In: *Probability Theory and Related Fields* 165.3 (2016), pp. 667–723. DOI: 10.1007/s00440-015-0643-7 (cit. on p. 19).

- [Lav06] F. Lavancier. "Long memory random fields". In: *Dependence in probability and statistics*. Vol. 187. Lect. Notes Stat. Springer, New York, 2006, pp. 195–220. DOI: 10.1007/0-387-36062-X\\_9 (cit. on p. 23).
- [Led12] M. Ledoux. "Chaos of a Markov operator and the fourth moment condition". In: *Ann. Probab.* 40.6 (2012), pp. 2439–2459. DOI: 10.1214/11–A0P685 (cit. on p. 19).
- [LRM23] N. N. Leonenko and M. D. Ruiz-Medina. "Sojourn functionals for spatiotemporal Gaussian random fields with long memory". In: *J. Appl. Probab.* 60.1 (2023), pp. 148–165. DOI: 10.1017/jpr.2022.30 (cit. on pp. 27, 82, 83, 111, 112).
- [LRM24] N. N. Leonenko and M. D. Ruiz-Medina. *High-level moving excursions for spatiotemporal Gaussian random fields with long range dependence*. 2024. arXiv: 2402.09003 [math.PR] (cit. on p. 27).
- [Leo99] N. Leonenko. *Limit theorems for random fields with singular spectrum*. Vol. 465. Mathematics and its Applications. Kluwer Academic Publishers, Dordrecht, 1999, pp. viii+401. DOI: 10.1007/978-94-011-4607-4 (cit. on pp. 43, 131).
- [LO13] N. Leonenko and A. Olenko. "Tauberian and Abelian theorems for long-range dependent random fields". In: *Methodol. Comput. Appl. Probab.* 15.4 (2013), pp. 715–742. DOI: 10.1007/s11009-012-9276-9 (cit. on pp. 26, 83, 84, 98, 131, 145).
- [LO14] N. Leonenko and A. Olenko. "Sojourn measures of Student and Fisher-Snedecor random fields". In: *Bernoulli* 20.3 (2014), pp. 1454–1483. DOI: 10.3150/13-BEJ529 (cit. on pp. 3, 25, 82–84, 96, 98, 99, 119, 131, 132).
- [Leo+24] N. Leonenko et al. *Limit theorems for p-domain functionals of stationary Gaussian fields*. 2024. arXiv: 2402.16701 [math.PR] (cit. on pp. 28, 111).
- [Mai24] L. Maini. "Asymptotic covariances for functionals of weakly stationary random fields". In: *Stochastic Process. Appl.* 170 (2024), Paper No. 104297, 19. DOI: 10.1016/j.spa.2024.104297 (cit. on pp. 28, 44, 71, 81, 131, 134).
- [MN24] L. Maini and I. Nourdin. "Spectral central limit theorem for additive functionals of isotropic and stationary Gaussian fields". In: *The Annals of Probability* 52.2 (2024), pp. 737 –763. DOI: 10.1214/23-A0P1669 (cit. on pp. 3, 27, 37, 67, 68, 70, 82, 84, 85, 95, 101, 103–105, 112, 114, 119, 120, 127, 128, 130, 147).

[Maj81] P. Major. *Multiple Wiener-Itô integrals*. Vol. 849. Lecture Notes in Mathematics. With applications to limit theorems. Springer, Berlin, 1981, pp. vii+127 (cit. on pp. 130, 145).

- [Mak+21] V. Makogin et al. "Long range dependence for stable random processes". In: *J. Time Series Anal.* 42.2 (2021), pp. 161–185. DOI: 10.1111/jtsa. 12560 (cit. on p. 22).
- [Mal13] A. Malyarenko. *Invariant random fields on spaces with a group action*. Probability and its Applications (New York). With a foreword by Nikolai Leonenko. Springer, Heidelberg, 2013, pp. xviii+261. DOI: 10.1007/978-3-642-33406-1 (cit. on pp. 4, 8).
- [MRV21] D. Marinucci, M. Rossi, and A. Vidotto. "Non-universal fluctuations of the empirical measure for isotropic stationary fields on  $\mathbb{S}^2 \times \mathbb{R}$ ". In: *The Annals of Applied Probability* 31.5 (2021), pp. 2311 –2349. DOI: 10.1214/20-AAP1648 (cit. on pp. 27, 111).
- [MRV24] D. Marinucci, M. Rossi, and A. Vidotto. "Fluctuations of Level Curves for Time-Dependent Spherical Random Fields". In: *Annales Henri Lebesgue* (in press) (2024) (cit. on p. 111).
- [MP08] D. Marinucci and G. Peccati. "High-frequency asymptotics for subordinated stationary fields on an abelian compact group". In: *Stochastic Process. Appl.* 118.4 (2008), pp. 585–613. DOI: 10.1016/j.spa.2007.05.008 (cit. on p. 68).
- [MP10] D. Marinucci and G. Peccati. "Group representations and high-resolution central limit theorems for subordinated spherical random fields". In: *Bernoulli* 16.3 (2010), pp. 798–824. DOI: 10.3150/09-BEJ230 (cit. on p. 68).
- [MP11] D. Marinucci and G. Peccati. *Random fields on the sphere*. Vol. 389. London Mathematical Society Lecture Note Series. Representation, limit theorems and cosmological applications. Cambridge University Press, Cambridge, 2011, pp. xii+341. DOI: 10.1017/CB09780511751677 (cit. on pp. 4, 67).
- [MR15] D. Marinucci and M. Rossi. "Stein-Malliavin approximations for non-linear functionals of random eigenfunctions on S<sup>d</sup>". In: *J. Funct. Anal.* 268.8 (2015), pp. 2379–2420. DOI: 10.1016/j.jfa.2015.02.004 (cit. on pp. 68, 71, 78).
- [MW11] D. Marinucci and I. Wigman. "The defect variance of random spherical harmonics". In: *Journal of Physics A: Mathematical and Theoretical* 44.35 (2011), p. 355206. DOI: 10.1088/1751-8113/44/35/355206 (cit. on pp. 71, 78).
- [MW14] D. Marinucci and I. Wigman. "On nonlinear functionals of random spherical eigenfunctions". In: *Comm. Math. Phys.* 327.3 (2014), pp. 849–872. DOI: 10.1007/s00220-014-1939-7 (cit. on pp. 3, 28, 30, 41, 68, 71).

[MPU19] F. Merlevède, M. Peligrad, and S. Utev. *Functional Gaussian approximation for dependent structures*. Vol. 6. Oxford Studies in Probability. Oxford University Press, Oxford, 2019, pp. xv+478. DOI: 10.1093/oso/9780198826941.001.0001 (cit. on p. 2).

- [MY22] Y. Mishura and N. Yoshidae. "Divergence of an integral of a process with small ball estimate". In: *Stochastic Process. Appl.* 148 (2022), pp. 1–24. DOI: 10.1016/j.spa.2022.02.006 (cit. on p. 82).
- [NS16] F. Nazarov and M. Sodin. "Asymptotic laws for the spatial distribution and the number of connected components of zero sets of Gaussian random functions". In: *J. Math. Phys. Anal. Geom.* 12.3 (2016), pp. 205–278. DOI: 10.15407/mag12.03.205 (cit. on p. 3).
- [NS20] F. Nazarov and M. Sodin. "Fluctuations in the number of nodal domains". In: *J. Math. Phys.* 61.12 (2020), pp. 123302, 39. DOI: 10.1063/5.0018588 (cit. on p. 3).
- [Not21] M. Notarnicola. "Probabilisitic limit theorems and the geometry of random fields". PhD thesis. Unilu - University of Luxembourg, Esch sur Alzette, Luxembourg, 2021 (cit. on pp. 3, 40, 71).
- [NPV23] M. Notarnicola, G. Peccati, and A. Vidotto. "Functional convergence of Berry's nodal lengths: approximate tightness and total disorder". In: *J. Stat. Phys.* 190.5 (2023), Paper No. 97, 41. DOI: 10.1007/s10955-023-03111-9 (cit. on pp. 3, 44).
- [Nou11] I. Nourdin. "Yet another proof of the Nualart-Peccati criterion". In: *Electron. Commun. Probab.* 16 (2011), pp. 467–481. DOI: 10.1214/ECP. v16–1642 (cit. on p. 17).
- [NN20] I. Nourdin and D. Nualart. "The functional Breuer-Major theorem". In: *Probab. Theory Related Fields* 176.1-2 (2020), pp. 203–218. DOI: 10.1007/s00440-019-00917-1 (cit. on pp. 28, 39).
- [NNP21] I. Nourdin, D. Nualart, and G. Peccati. "The Breuer-Major theorem in total variation: improved rates under minimal regularity". In: *Stochastic Process. Appl.* 131 (2021), pp. 1–20. DOI: 10.1016/j.spa.2020.08.007 (cit. on p. 24).
- [NNT10] I. Nourdin, D. Nualart, and C. A. Tudor. "Central and non-central limit theorems for weighted power variations of fractional Brownian motion".
   In: Ann. Inst. Henri Poincaré Probab. Stat. 46.4 (2010), pp. 1055–1079.
   DOI: 10.1214/09-AIHP342 (cit. on pp. 28, 39).
- [NP09] I. Nourdin and G. Peccati. "Stein's method on Wiener chaos". In: *Probab. Theory Related Fields* 145.1-2 (2009), pp. 75–118. DOI: 10.1007/s00440-008-0162-x (cit. on pp. 17, 18, 96, 134, 135).
- [NP12] I. Nourdin and G. Peccati. *Normal approximations with Malliavin calculus*. Vol. 192. Cambridge Tracts in Mathematics. From Stein's method to universality. Cambridge University Press, Cambridge, 2012, pp. xiv+239 (cit. on pp. 10, 12, 13, 15, 16, 18, 19, 38, 42, 47, 49, 61, 70, 95, 112, 116–118, 125, 126, 133, 134, 146).

[NPP11] I. Nourdin, G. Peccati, and M. Podolskij. "Quantitative Breuer-Major theorems". In: *Stochastic Process. Appl.* 121.4 (2011), pp. 793–812. DOI: 10.1016/j.spa.2010.12.006 (cit. on pp. 47, 83, 96).

- [NPR19] I. Nourdin, G. Peccati, and M. Rossi. "Nodal statistics of planar random waves". In: *Comm. Math. Phys.* 369.1 (2019), pp. 99–151. DOI: 10. 1007/s00220-019-03432-5 (cit. on pp. 3, 5, 7, 29, 40, 53, 68–71, 99, 108).
- [NPY19] I. Nourdin, G. Peccati, and X. Yang. "Berry-Esseen bounds in the Breuer-Major CLT and Gebelein's inequality". In: *Electron. Commun. Probab.* 24 (2019), Paper No. 34, 12. DOI: 10.1214/19-ECP241 (cit. on p. 24).
- [Nou+16] I. Nourdin et al. "Multidimensional limit theorems for homogeneous sums: a survey and a general transfer principle". In: *ESAIM Probab. Stat.* 20 (2016), pp. 293–308. DOI: 10.1051/ps/2016014 (cit. on p. 17).
- [Nua06] D. Nualart. *The Malliavin calculus and related topics*. Second. Probability and its Applications (New York). Springer-Verlag, Berlin, 2006, pp. xiv+382 (cit. on pp. 16, 49, 116).
- [NP05] D. Nualart and G. Peccati. "Central limit theorems for sequences of multiple stochastic integrals". In: *Ann. Probab.* 33.1 (2005), pp. 177–193. DOI: 10.1214/009117904000000621 (cit. on pp. 17, 50, 116, 118).
- [NZ20] D. Nualart and G. Zheng. "Oscillatory Breuer-Major theorem with application to the random corrector problem". In: *Asymptot. Anal.* 119.3-4 (2020), pp. 281–300 (cit. on pp. 24, 96, 118).
- [ØZ01] B. Øksendal and T. Zhang. "Multiparameter Fractional Brownian Motion And QuasiLinear Stochastic Partial Differential Equations". In: Stochastics An International Journal of Probability and Stochastic Processes 71 (Feb. 2001), pp. 141–163. DOI: 10.1080/17442500108834263 (cit. on p. 137).
- [PR16] M. S. Pakkanen and A. Réveillac. "Functional limit theorems for generalized variations of the fractional Brownian sheet". In: *Bernoulli* 22.3 (2016), pp. 1671–1708. DOI: 10.3150/15-BEJ707 (cit. on pp. 27, 35, 112, 114, 115).
- [Pea05] K. Pearson. "The Problem of the Random Walk". In: *Nature* 72.1865 (1905), pp. 294–294. DOI: 10.1038/072294b0 (cit. on p. 72).
- [PT05] G. Peccati and C. A. Tudor. "Gaussian Limits for Vector-valued Multiple Stochastic Integrals." In: *Séminaire de Probabilités XXXVIII. Lecture Notes in Mathematics, vol.1857, Springer, Berlin* (2005) (cit. on pp. 17, 18, 85, 101, 102).
- [PV20] G. Peccati and A. Vidotto. "Gaussian random measures generated by Berry's nodal sets". In: *J. Stat. Phys.* 178.4 (2020), pp. 996–1027. DOI: 10.1007/s10955-019-02477-z (cit. on pp. 3, 44, 103, 108).
- [Pel90] M. Peligrad. "On Ibragimov-Iosifescu conjecture for  $\varphi$ -mixing sequences". In: *Stochastic Process. Appl.* 35.2 (1990), pp. 293–308. DOI: 10.1016/0304-4149(90)90008-G (cit. on pp. 2, 22).

[PS17] V. Pilipauskaitė and D. Surgailis. "Scaling transition for nonlinear random fields with long-range dependence". In: *Stochastic Process. Appl.* 127.8 (2017), pp. 2751–2779. DOI: 10.1016/j.spa.2016.12.011 (cit. on p. 112).

- [RST12] A. Réveillac, M. Stauch, and C. A. Tudor. "Hermite variations of the fractional Brownian sheet". In: *Stoch. Dyn.* 12.3 (2012), pp. 1150021, 21. DOI: 10.1142/S0219493711500213 (cit. on pp. 27, 35, 82, 112–115, 124, 134, 135).
- [Ros60] M. Rosenblatt. "Independence and dependence". In: Proc. 4th Berkeley Sympos. Math. Statist. and Prob., Vol. II. Univ. California Press, Berkeley-Los Angeles, Calif., 1960, pp. 431–443 (cit. on pp. 38, 84, 99, 112).
- [Ros19] M. Rossi. "The defect of random hyperspherical harmonics". In: *J. Theoret. Probab.* 32.4 (2019), pp. 2135–2165. DOI: 10.1007/s10959-018-0849-6 (cit. on p. 71).
- [Sch38] I. J. Schoenberg. "Metric spaces and completely monotone functions". In: *Ann. of Math.* (2) 39.4 (1938), pp. 811–841. DOI: 10.2307/1968466 (cit. on pp. 42, 100).
- [SW17] X. Song and Y. Wang. "Quasi-Monte Carlo simulation of Brownian sheet with application to option pricing". In: *Statistical Theory and Related Fields* 1.1 (2017), pp. 82–91. DOI: 10.1080/24754269.2017.1332965 (cit. on p. 137).
- [Ste93] E. M. Stein. *Harmonic analysis: real-variable methods, orthogonality, and oscillatory integrals*. Vol. 43. Princeton Mathematical Series. With the assistance of Timothy S. Murphy, Monographs in Harmonic Analysis, III. Princeton University Press, Princeton, NJ, 1993, pp. xiv+695 (cit. on p. 9).
- [Swi02] R. J. Swift. "Applications of harmonizable isotropic random fields". In: *Teor. Ĭmovīr. Mat. Stat.* 66 (2002), pp. 137–146 (cit. on p. 9).
- [Sze39] G. Szegö. *Orthogonal Polynomials*. Vol. Vol. 23. American Mathematical Society Colloquium Publications. American Mathematical Society, New York, 1939, pp. ix+401 (cit. on pp. 31, 69, 77).
- [Taq75] M. S. Taqqu. "Weak convergence to fractional brownian motion and to the rosenblatt process". In: *Zeitschrift für Wahrscheinlichkeitstheorie und Verwandte Gebiete* 31.4 (1975), pp. 287–302. DOI: 10.1007/BF00532868 (cit. on pp. 25, 119).
- [Taq79] M. S. Taqqu. "Convergence of integrated processes of arbitrary Hermite rank". In: *Z. Wahrsch. Verw. Gebiete* 50.1 (1979), pp. 53–83. DOI: 10. 1007/BF00535674 (cit. on pp. 38, 84, 99, 112).
- [Tod19] A. P. Todino. "A quantitative central limit theorem for the excursion area of random spherical harmonics over subdomains of  $\mathbb{S}^2$ ". In: *J. Math. Phys.* 60.2 (2019), pp. 023505, 33. DOI: 10.1063/1.5048976 (cit. on pp. 3, 28, 70, 71).

[Tud13] C. A. Tudor. *Analysis of variations for self-similar processes*. Probability and its Applications (New York). A stochastic calculus approach. Springer, Cham, 2013, pp. xii+268. DOI: 10.1007/978-3-319-00936-0 (cit. on pp. 39, 87).

- [TV09] C. A. Tudor and F. G. Viens. "Variations and estimators for self-similarity parameters via Malliavin calculus". In: *Ann. Probab.* 37.6 (2009), pp. 2093–2134. DOI: 10.1214/09-A0P459 (cit. on p. 82).
- [Vid21] A. Vidotto. "A note on the reduction principle for the nodal length of planar random waves". In: *Statist. Probab. Lett.* 174 (2021), Paper No. 109090, 5. DOI: 10.1016/j.spl.2021.109090 (cit. on pp. 3, 103, 108).
- [WC10] L. Wang and H. Cai. "Asymptotic properties of nonparametric regression for long memory random fields". In: *Journal of Statistical Planning and Inference* 140.3 (2010), pp. 837–850. DOI: https://doi.org/10.1016/j.jspi.2009.09.012 (cit. on p. 23).
- [Wan07] W. Wang. "Almost-sure path properties of fractional Brownian sheet". In: *Ann. Inst. H. Poincaré Probab. Statist.* 43.5 (2007), pp. 619–631. DOI: 10.1016/j.anihpb.2006.09.005 (cit. on p. 88).
- [Wat95] G. N. Watson. *A treatise on the theory of Bessel functions*. Cambridge Mathematical Library. Reprint of the second (1944) edition. Cambridge University Press, Cambridge, 1995, pp. viii+804 (cit. on p. 74).
- [Wig10] I. Wigman. "Fluctuations of the nodal length of random spherical harmonics". In: *Comm. Math. Phys.* 298.3 (2010), pp. 787–831. DOI: 10.1007/s00220-010-1078-8 (cit. on p. 78).
- [Wig23] I. Wigman. "On the nodal structures of random fields: a decade of results". In: *Journal of Applied and Computational Topology* (2023), pp. 1–43 (cit. on p. 67).
- [Xia09] Y. Xiao. "Sample path properties of anisotropic Gaussian random fields". In: *A minicourse on stochastic partial differential equations*. Vol. 1962. Lecture Notes in Math. Springer, Berlin, 2009, pp. 145–212. DOI: 10. 1007/978-3-540-85994-9\\_5 (cit. on p. 5).
- [Yag60] A. M. Yaglom. "Second-order homogeneous random fields". In: *Proc.* 4th Berkeley Sympos. Math. Statist. and Prob., Vol. II. Univ. California Press, Berkeley-Los Angeles, Calif., 1960, pp. 593–622 (cit. on p. 8).
- [Zel09] S. Zelditch. "Real and complex zeros of Riemannian random waves". In: *Spectral analysis in geometry and number theory*. Vol. 484. Contemp. Math. Amer. Math. Soc., Providence, RI, 2009, pp. 321–342. DOI: 10. 1090/conm/484/09482 (cit. on pp. 3, 67).
- [Zhe19] G. Zheng. "A Peccati-Tudor type theorem for Rademacher chaoses". In: *ESAIM Probab. Stat.* 23 (2019), pp. 874–892. DOI: 10.1051/ps/2019013 (cit. on p. 17).